

Nonlocal models in cell migration

Vom Fachbereich Mathematik der Rheinland-Pfälzischen Technischen Universität Kaiserslautern-Landau zur Verleihung des akademischen Grades Doktor der Naturwissenschaften (Doctor rerum naturalium, Dr. rer. nat.) genehmigte Dissertation

von

Maria Eckardt

Gutachter:

Prof. Dr. Christina Surulescu, Rheinland-Pfälzische Technische Universität Kaiserslautern-Landau Prof. Dr. Christian Stinner, Technische Universität Darmstadt

Datum der Disputation: 10. April 2025

DE-386

Acknowledgements

First and foremost, I would like to express my deepest gratitude to my supervisor Prof. Dr. Christina Surulescu for the opportunity to work and write my dissertation in her research group. Her constant support and encouragement as well as her advice in mathematical and biological questions were essential for this work. Moreover, I thank her for the space she gave me to grow professionally and personally.

My greatest thank you also goes to Dr. Anna Zhigun for her enthusiasm and determination during the collaboration on our joint papers and for her invaluable input on further mathematical problems. In addition, I would like to thank Prof. Dr. Christian Stinner for his review. Further, I thank Dr. Nataliya Kraynyukova for the helpful discussion.

I would like to thank my current and former colleagues for the friendly working environment and my friends for the great time we have together.

Last but not least, I sincerely thank my family without whose support the completion of this dissertation would not have been possible: my husband Marcel for being a loving husband and a caring father, my parents and parents-in-law for being dedicated grandparents, my brother Mark for being a funny uncle and, finally, my children Felix and Emily for always giving me a reason to smile.

Abstract

Experimental evidence suggests that cells can perceive signals not only at their actual location but also within a large neighborhood compared to the cell size. These biochemical and biophysical cues influence the migration, proliferation, and differentiation of cells. In this work, we examine four nonlocal models describing the movement of cell populations. These models are represented by reaction-diffusion(-advection) equations containing nonlocal spatial integral terms that describe the influence of the surroundings on the development of the cell population. Our focus is on the mathematical analysis of these models. Numerical simulations are performed to illustrate the solution behavior.

First, we consider two models, in which the gradient in the advection term of the respective local model is replaced by a nonlocal integral. For the first adhesion or nonlocal chemotaxis model, we show convergence of the weak solution to the weak solution of the corresponding local haptotaxis or chemotaxis model, respectively, as the sensing radius decreases. Then, we show the existence of a very weak solution for the second cell-cell-adhesion model with degenerated myopic diffusion.

Furthermore, we consider two models with a nonlocality in the reaction term. Specifically, for a model for cancer invasion with myopic diffusion, repellent pH-taxis, and nonlocal intraspecific interaction, we show the global existence of a bounded unique weak solution and visualize its behavior with numerical simulations. Additionally, we perform a 1D pattern analysis. Finally, we show the global existence of a bounded weak solution for a model with two nonlocal interaction terms and perform numerical simulations.

Zusammenfassung

Experimente haben nachgewiesen, dass Zellen Signale nicht nur an ihrer Position empfangen können, sondern innerhalb eines im Vergleich zur Zellgröße großen Wahrnehmungsradius. Diese biochemischen und biophysischen Signale beeinflussen die Bewegung, Proliferation und Differenzierung von Zellen. In dieser Arbeit betrachten wir vier nichtlokale Modelle, die die Bewegung von Zellpopulationen beschreiben. Die Nichtlokalität wird mittels eines nichtlokalen Raumintegrals modelliert, das in verschiedenen Termen der betrachteten Reaktions-Diffusions-(Advektions-) Gleichungen enthalten ist. Der Fokus liegt dabei auf der mathematischen Analyse dieser Modelle. Auch numerische Simulationen werden durchgeführt, um das Verhalten der Lösung zu veranschaulichen.

Wir betrachten zwei Modelle, in denen der Gradient im Advektionsterm durch ein nichtlokales Integral ersetzt wird. Zuerst zeigen wir die Konvergenz der schwachen Lösung eines Modells, das Adhesion oder nichtlokale Chemotaxis beschreibt, gegen die schwache Lösung des jeweils entsprechenden lokalen Haptoxis- oder Chemotaxismodells für einen verschwindenden Wahrnehmungsradius. Anschließend zeigen wir die Existenz einer sehr schwachen Lösung eines Modells für Zell-Zell-Adhesion mit degenerierter myopischer Diffusion.

Darüber hinaus betrachten wir zwei Modelle mit Nichtlokalität im Reaktionsterm. Wir zeigen die globale Existenz einer eindeutigen beschränkten Lösung eines Modells für Krebsinvasion mit myopischer Diffusion, abstoßender pH-Taxis und einem nichtlokalen innerartlichen Interaktionsterm und eines Modells mit zwei nichtlokalen Interaktionstermen. Das Verhalten der jeweiligen Lösung wird mithilfe von numerischen Simulationen veranschaulicht. Darüber hinaus analysieren wir für eines der Modelle das Auftreten von Mustern in 1D.

Contents

1	Intr	roduction	1
2	Pre	liminaries	7
	2.1	Matrices	8
	2.2	Function spaces	9
	2.3	Convolutions on bounded domains	13
Ι	No	onlocal models with nonlocality in the advection term	15
3	Nor	nlocal and local models for taxis in cell migration: a rigorous limit pro-	
	\mathbf{ced}	ure	17
	3.1	Motivation	17
	3.2	Operators \mathcal{A}_r and $\mathring{\nabla}_r$ and averages of ∇	20
		3.2.1 Properties of the averaging operators \mathcal{T}_r and \mathcal{S}_r	23
	3.3	Well-posedness for a class of evolution equations involving \mathcal{T}_r or \mathcal{S}_r	28
	3.4	Nonlocal models involving averaging operators \mathcal{T}_r and \mathcal{S}_r	31
		3.4.1 Problem setting and main result of the section	32
		3.4.2 Global existence of solutions to (3.4.1): the case of f_c Lipschitz	35
		3.4.3 Global existence of solutions to (3.4.1): the case of f_c dissipative	46
		3.4.4 Limiting behaviour of the nonlocal model (3.4.1) as $r \to 0 \ldots \ldots \ldots$	49
	3.5	Numerical simulations in 1D	51
		3.5.1 Comparison of nonlocal operator representations	51
		3.5.2 Comparison between nonlocal and local formulation	54
	3.6	Discussion	57
4	Glo	bal existence of solutions to a nonlocal equation with degenerate anisotropi	с
	diff	usion	61
	4.1	Motivation	61
	4.2	Preliminaries	62
		4.2.1 The adhesion operator \ldots	62
		4.2.2 A lemma about sets of "sufficiently small" dimension	63
	4.3	Problem setting and main result	65
	4.4	Approximate problems	67
		4.4.1 Construction of a regular matrix family $(\mathbb{D}_{\varepsilon})$	67
		4.4.2 Existence of a global classical solution to the approximate problem	70
	4.5	Existence of a very weak solution to the original problem $\hfill \hfill \ldots \hfill \hf$	74
	4.6	Smooth very weak solutions are classical	79

Π	Ν	onlocal models with nonlocality in the reaction term	83		
5	On a mathematical model for cancer invasion with repellent pH-taxis and				
	non	local intraspecific interaction	85		
	5.1	Motivation	85		
	5.2	Modeling	86		
	5.3	Mathematical analysis	92		
		5.3.1 Local existence in an approximate problem	93		
		5.3.2 Global existence and boundedness of u in the approximate problem \ldots	99		
		5.3.3 Global existence and boundedness in the original problem	104		
	5.4	Long time behavior	106		
	5.5	Pattern formation: a 1D study	110		
		5.5.1 Stability in the local model without diffusion and taxis	110		
		5.5.2 Stability in the local model with diffusion and taxis	111		
		5.5.3 Stability in the nonlocal model	111		
	5.6	Numerical simulations	114		
	5.7	Discussion	117		
6	On	a PDE-ODE-PDE model for two interacting cell populations under the			
	infl	uence of an acidic environment and with nonlocal intra- and interspecific			
	gro	wth limitation	119		
	6.1	Motivation	119		
	6.2	Model	120		
	6.3	Global existence of a classical solution to an approximate problem $\ldots \ldots \ldots$	122		
	6.4	Existence of a weak solution to the original problem $\ldots \ldots \ldots \ldots \ldots \ldots$	133		
	6.5	1D Simulations	138		
	6.6	Discussion	144		
7	Sun	nmary and outlook	147		
A	Ado	litional Theorems	153		
Bi	bliog	graphy	168		
A	Academic curriculum vitae				
A	Akademischer Lebenslauf 1				

List of Figures

3.1	Comparison between nonlocal formulations $(3.1.1)$ and $(3.4.1)$.	52
3.2	Comparison between nonlocal formulations $(3.1.1)$ and $(3.4.1)$ near boundaries	
	and between the two forms of nonlocal operator. \ldots \ldots \ldots \ldots \ldots \ldots	53
3.3	Convergence between nonlocal and local/classical formulations under negligible	
	cell-cell adhesion.	55
3.4	Time restricted convergence under moderate cell-cell adhesion.	56
3.5	Convergence between nonlocal and local/classical formulations under a set of min-	
	imalistic linear functional forms.	57
5.1	Simulation results for (5.5.1) with $\beta = \mu = 1$	115
5.2	Simulation results for (5.5.1) with $\beta = 20$ and $\mu = 100$	116
5.3	Simulation results for (5.5.1) with local source term $\mu(h)u^{\alpha}(1-u^{\beta})$ replacing the	
	one in the equation for u	116
6.1	Initial conditions $u_0, w_0, h_0, \ldots, \ldots, \ldots, \ldots, \ldots, \ldots, \ldots$	139
6.2	Simulation results of model (6.2.1) with J_1 and J_2 logistic kernels	140
6.3	Simulation results of model (6.2.1) with J_1 and J_2 uniform kernels $\ldots \ldots \ldots$	141
6.4	Simulations of models (6.2.1) and (6.5.3) with $\alpha = 2, \mu_1 = 1, \beta = 1$ and $\gamma = 1000$	
	or $\gamma = 1, \beta = 10, 100, 1000.$	142
6.5	Simulations of model $(6.5.3)$ without w	143
6.6	Simulations of model (6.2.1) with <i>h</i> -dependent kernels J_1 and J_2	144

List of Tables

6.1	Minimal value α^* for which the solution ceases to exist for $\beta = \gamma = \mu_1 = 1$	
	depending on the kernels J_1 and J_2	141
6.2	Minimal value α^* for which the solution ceases to exist, depending on parameters	
	β, γ, μ_1 for logistic kernels J_1 and $J_2, \ldots, \ldots, \ldots, \ldots, \ldots, \ldots$	141

CHAPTER 1

Introduction

In this dissertation, we investigate nonlocal models in cell migration. In recent decades these models have attracted increasing interest, see the review [28]. They reflect the ability of cells to receive environmental signals within a sensing region surrounding their current position. These signals can impact multiple processes, including cell migration. The nonlocal approach is suggested by experimental evidence and appears more realistic, as cells are surrounded by other cells and tissue. In certain contexts, the reduced consideration of these effects by local models could be oversimplifying and can lead to analytical and modeling problems. While there are many studies of local models, nonlocal models have been considered far less often, especially from an analytical point of view. However, the nonlocal models provide a biologically plausible description of certain processes involved in cell migration and, as we discuss later, can avoid in certain situations analytical problems of local models, such as finite-time blow-up of the solution.

The orientation of directional migration of cells is largely influenced by the extracellular environment and primarily determined by protrusions (e.g., filopodia, lamellipodia) which are outward extensions of the cell membrane. Cells respond to external diffusible and non-diffusible signals by extending protrusions in the direction of movement [86]. The main functions of protrusions include sampling the cell's environment and establishing initial dynamic adhesions to the extracellular matrix (ECM) or other cells within a sensing region that can be large compared to the cell size [1, 71, 90, 130]. Furthermore, protrusions are involved in the communication of cells over long distances, thereby transmitting signals [17, 57, 87]. The information obtained via protrusions influences the subsequent behavior of the cell, e.g., the choice of the following direction of migration [1, 122]. In areas with hard borders, such as bones, cartilage, or the walls of a Petri dish, cells receive hardly any information from outside, as the ability of cells to stretch their protrusions outwards is limited there [28].

After sampling their environment cells form cell-matrix or cell-cell adhesions to move [35, 86, 152]. The adhesion of cells to the ECM is facilitated by the attachment of specific cell receptors (e.g., integrins) to tissue fibers [38, 152]. Besides, cells adhere to other cells by binding specific cell adhesion molecules (cadherins) on the cell surface [96], which also enables the formation of cell clusters. This process is essential in organizing cells into tissue, organs, and organisms [5, 66]. The strength and number of bindings depend on chemical signals [68]. Both adhesion structures are often dynamic to allow cells to react to changes in environmental cues [34]. In addition to their role in cell movement, they are essential for embryonic development, homeostasis, immune responses, wound healing, and cell sorting [5, 34, 152]. Hence, nonlocal models align with many

biological observations and empirically collected data [97].

Apart from this, nonlocal models can solve mathematical inconsistencies present in some local models, such as a finite-time blow-up that is unrealistic from a biological point of view. For example, in [79], the authors show that the considered nonlocal model has a bounded global solution, while a finite-time blow-up occurs in the corresponding local model. Thereby, the solutions of the nonlocal models have increasingly larger peaks in numerical simulations for diminishing sensing regions. It is assumed that the behavior of nonlocal models with a parameter whose reduction leads to a vanishing sensing region can be approximated by the corresponding local model [68, 79]. Related local and nonlocal models include chemotaxis, which refers to migration in response to differences in the concentration of a soluble signal, and nonlocal chemotaxis, which takes into account that migrating cells can detect this signal within a sensing region. Besides, haptotaxis referring to the migration of cells in response to different concentrations of a bound signal is the local counterpart to cell-matrix (and cell-cell) adhesion. Nonlocal and local models can also exhibit different behavior in other contexts, e.g., in the occurrence of Turing patterns [111]. Continuous local models are often incompatible with biological effects such as sorting, while nonlocal models can replicate this effect [5].

Typically, nonlocal models include a spatial integral that increases the regularity of the equation; however, nonlocalities can also be introduced with respect to other variables (e.g., time, speed). Nevertheless, these models are mathematically challenging, as comparison principles to show the biologically important boundedness of solutions do not hold for this type of equation. Further challenges arise if the nonlocal equation is coupled with other differential equations, especially if the involved equations have a different type. A numerical simulation of the integral terms is numerically costly and requires efficient numerical methods to deal with them. Additionally, a unifying analytical framework that could deal with different kinds of nonlocal terms could be advantageous. Up to our knowledge, no such framework exists, as the analysis strongly depends on the specific form of the nonlocality.

Moreover, spatially nonlocal (and local) models describing the migration of a cell population u are usually of reaction-diffusion-advection (RDA) type, i.e., of the form

$$u_t = \underbrace{\nabla \cdot (D\nabla u)}_{\text{diffusion}} - \underbrace{\nabla \cdot (vu)}_{\text{advection}} + \underbrace{f(u)}_{\text{reaction}},$$

possibly coupled with further dynamics. Here, D denotes the diffusion coefficient, v the advection velocity, and f(u) a reaction function. Most nonlocal models consider RDA-equations with nonlocality in the advection term describing cell-cell or cell-matrix adhesion or nonlocal chemotaxis. Examples of such adhesion models can be found in [5, 21, 23, 68, 80, 156] and for nonlocal chemotaxis in [21, 79]. The solvability of models with nonlocal advection term was studied in [45, 46, 56, 79, 80, 82, 128]. Moreover, there are few studies on the existence and long-time behavior of solutions to local models including potentially degenerating myopic diffusion and taxis. Most studies feature haptotaxis and consider only the one-dimensional case [77, 149–151].

Less studied models contain nonlocalities in the source term. This term impacts cell movement indirectly since the evolving cell density leads to modified density-dependent coefficients. Possible applications are competition for resources, differentiation, proliferation, and growth; see [28, 99] and references therein. Examples of the modeling and analysis of this kind of problems can be found in [12–14, 99, 103, 104, 113, 136] but even for comparatively easy settings there are no complete results involving existence, boundedness, pattern formation, numerical simulations, etc. of solutions. We refer to [28, 51, 87] for more detailed reviews on these and other types of nonlocal models.

In the present work, we examine four nonlocal models with no-flux boundary conditions in bounded domains and describing the movement of cell populations. The nonlocality is modeled via a density-dependent spatial integral included in the advection (*Chapters 3* and 4) or reaction term (*Chapters 5* and 6) of a reaction-diffusion(-advection) equation. Our focus is on the mathematical analysis of the models. Additionally we perform a 1D pattern analysis in *Chapter 5* and numerical simulations in *Chapters 3*, 5, and 6 to visualize the results.

The convergence of the nonlocal operators to the local gradient presented in *Chapter 3* was shown heuristically via Taylor expansion in [68, 81]. In [79], the question of convergence of nonlocal models to its local counterpart was raised. However, up to our knowledge, a rigorous proof of convergence has not been established before. Moreover, the combination of myopic diffusion and adhesion has not been analyzed so far. Therefore, our existence proof for a solution to the degenerated PDE in *Chapter 4* significantly contributes to this new field. Notably, our assumptions regarding the degeneracy set seem new in the context of degenerated diffusion. Furthermore, the existence proof and the analysis of the long-time behavior and pattern formation of solutions to the model in *Chapter 5* contribute to the rarely studied field of models that feature nonlocality in the source term and models involving myopic diffusion and advection. There, we add to the existing literature the analysis of a PDE-PDE-system, one of the PDEs combining myopic diffusion and advection with a nonlocality in the source term. Also, considering a PDE with (two) nonlocalities in the source term coupled with an ODE from *Chapter 6* is novel.

Outline

The first part of this thesis deals with nonlocal models with nonlocality in the advection term describing cell-cell or cell-matrix adhesion or nonlocal chemotaxis. Thereby, we consider models with adhesion velocity of the form

$$\mathcal{A}_r u(x) = \frac{1}{r} \int_{B_r} u(x+\xi) \frac{\xi}{|\xi|} F_r(|\xi|) d\xi$$

where u denotes some interaction function taking into account cell-cell and cell-matrix interactions and depends on the cell and tissue density. The magnitude of the interaction force F_r depends on the distance within the sensing region B_r , where r is called sensing radius. In the case of nonlocal chemotaxis, we consider a similar integral (over a sphere), where the interaction function depends on the concentration of some dissolved chemical signal.

This part consists of two chapters:

• Chapter 3 considers a PDE-ODE-system describing adhesion and a PDE-PDE-system for nonlocal chemotaxis including the aforementioned operators. The adhesion model is related to its local counterpart characterizing haptotaxis by replacing the gradient of the cell-cell and cell-matrix interaction function u by $\mathcal{A}_r u$. Similarly, we relate nonlocal chemotaxis to chemotaxis. We show the existence of a global weak-strong solution to each of the nonlocal models and link it via a limit procedure for a diminishing sensing radius to the weak solution of the corresponding local model. Our proof relies on a reformulation of the involved nonlocal operator as an integral operator that is applied directly to the gradient of the interaction function. Both types of models are treated in a unified framework. Numerical simulations in 1D are cited for completeness reasons. This chapter is largely based on [47].

• Chapter 4 shows the existence of a global very weak solution to a nonlocal reaction-diffusionadvection equation including degenerated myopic diffusion, cell-cell adhesion, and a generalized logistic-type growth term in dimensions $n \ge 3$. Thereby, the degeneracy set is sufficiently low-dimensional (in terms of upper box fractal dimension) and has a positive distance to the boundary of the domain. We deal with the nonlocal operator upon rewriting it to a convolution with a bounded function. The corresponding equation without growth term was derived in [156]. Besides its biological foundation, we included the growth term to deal with analytical challenges arising especially from the degeneracy of the diffusion tensor. This chapter is largely based on [50].

The second part of this dissertation deals with models involving nonlocality in the reaction term. The nonlocal terms are of the form

$$u^{\alpha}(1 - J_1 * u^{\beta} - J_2 * w^{\gamma})(x) = u^{\alpha}(x) \left(1 - \int_{\Omega} J_1(x - y) u^{\beta}(y) \, \mathrm{d}y - \int_{\Omega} J_2(x - y) w^{\gamma}(y) \, \mathrm{d}y \right)$$

with $J_1 > 0$ and $J_2 \ge 0$, where u and w denote the density of two cell populations. Such terms describe intra- and interspecific competition between cells for available resources in their surrounding, e.g., to prevent overcrowding. The assumption of strict positivity of J_1 and the integration over the whole domain indicate that the sensing region of a cell corresponds to the whole domain independent from its position.

This part consists of two chapters:

- Chapter 5 shows the existence of a unique global bounded weak solution to a PDE-PDEmodel for tumor cell migration with myopic diffusion, repellent pH-taxis, and a nonlocal source term of the above form. Moreover, we analyze the asymptotic behavior of the solution. In 1D we perform a pattern analysis for constant diffusion and numerical simulations to illustrate the behavior of the solution. The model deduction based on a mesoscopic description of cell migration with a kinetic transport equation is included for completeness. Our results extend [99], where a Fischer-KPP-equation with nonlocal intraspecific competition (but in an unbounded domain) was examined. This chapter is largely based on [49].
- In *Chapter 6* we prove the existence of a global bounded weak solution to a PDE-ODE-PDE-system describing the dynamics of active and inactive cells and a repellent signal. It has two nonlocalities in an equation, one of them depending on another cell population. Moreover, both nonlocalities depend on the signal produced by both cell populations. Also numerical simulations are performed. This chapter is largely based on [48].

The dissertation is structured as follows. *Chapter 2* contains the mathematical preliminaries of this work and introduces the relevant function spaces, convolutions in bounded domains, and notation. The notation may differ slightly from chapter to chapter, as this work consists of

four independently considered models. Chapters 3-6 begin each with a motivation, in which the biological context and the underlying literature of the concrete model are mentioned. Finally, Chapter 7 provides a brief summary of this work along with some perspectives.

CHAPTER 2

Preliminaries

In this chapter, we introduce the notation and recall mostly without proof some facts concerning matrices, the relevant function spaces and convolutions in bounded domains. More details on these topics can be found e.g., in [20, 58, 84, 94]. Relevant results on different types of differential equations, fixed-point theorems, convergence theorems and functional analysis and the proofs of some lemmas from this chapter can be found in Appendix A.

Notation:

- Throughout this work we consider a bounded domain $\Omega \subset \mathbb{R}^n$, $n \in \mathbb{N}$, with sufficiently smooth boundary $\partial \Omega$ and outer unit normal ν .
- For r > 0 we introduce the subdomain $\Omega_r := \{x \in \Omega : \operatorname{dist}(x, \partial \Omega) > r\}$ of Ω .
- For a function $u: \Omega \to \mathbb{R}$ we assume, by convention, that $u \equiv 0$ on $\mathbb{R}^n \setminus \overline{\Omega}$. This allows for an obvious meaning to be given to the convolution u * v for any $u \in L^1(\Omega)$ and $v \in L^1(\mathbb{R}^n)$. This extends componentwise to any vector-/matrix-valued function u.
- We denote by e_i , $i \in \mathbb{N}$, the *i*th canonical vector in \mathbb{R}^n and by $I_n \in \mathbb{R}^{n \times n}$ the identity matrix.
- · By $|\cdot|$ and $|\cdot|_{\infty}$ we denote the Euclidean and infinity norms in \mathbb{R}^n , respectively, and by |A| the Lebesgue measure of a set A.
- For $x \in \mathbb{R}^n$, $n \ge 2$ we denote $x = (x', x_n)$, where $x' \in \mathbb{R}^{n-1}$ and $x_n \in \mathbb{R}$.
- For $A, B \subset \mathbb{R}^n$, $a \in \mathbb{R}^n$, and s > 0 we use the set notations

$$A + B := \{x + y : x \in A, y \in B\},\$$

$$a + B := B + a := \{a\} + B,\$$

$$O_s(A) := \{x \in \mathbb{R}^n : \operatorname{dist}(x, A) < s\}.$$

• By B_r and S_r , r > 0, we denote the open r-ball and the r-sphere in \mathbb{R}^n , both centred at the origin, and define the mean values of a function u over B_r and S_r , respectively, as

$$\begin{split} & \int_{B_r} u(\xi) \, d\xi := \frac{1}{|B_r|} \int_{B_r} u(\xi) \, d\xi, \\ & \int_{S_r} u(\xi) \, d\xi := \frac{1}{|S_r|} \int_{S_r} u(\xi) \, d\sigma(\xi), \end{split}$$

where $\sigma(\cdot)$ denotes the surface measure corresponding to the Lebesgue measure on \mathbb{R}^n .

• For all indices $i \in \mathbb{N}$, the quantities C_i, K_i, ε_i denote a positive constant or, alternatively, a positive function of its arguments. The constants C_i and ε_i are numbered chapter by chapter, while K_i denotes the constants from Appendix A. Dependencies upon such parameters as the space dimension n, domain Ω , the norms of the initial data, norms and bounds for the coefficient functions and parameters are mostly **not** indicated in an explicit way.

2.1 Matrices

In this section we summarize some definitions and properties of matrices.

Definition 2.1.1. Let $P = (p_{ij})_{i,j=1,...,n}$, $Q = (q_{ij})_{i,j=1,...,n} \in \mathbb{R}^{n \times n}$. We define the following norms and inner product:

- the spectral norm $|P|_2 := |P| := \max_{|x|=1} |Px| = \sqrt{\lambda_{\max}(P^T P)}$, where λ_{\max} denotes the maximal eigenvalue of a matrix,
- · $|P|_{\infty} := \max_{|x|_{\infty}=1} |Px|_{\infty} = \max_{i=1,\dots,n} \sum_{j=1}^{n} |p_{ij}|,$
- the Frobenius inner product

$$P: Q = \sum_{i,j=1}^{n} p_{ij} q_{ij}.$$

If not states otherwise we will use the $|\cdot|_2$ -norm for matrices.

These norms obviously satisfy the following lemma.

Lemma 2.1.2. (i) The norms $|\cdot|_2$ and $|\cdot|_{\infty}$ are equivalent on $\mathbb{R}^{n \times n}$.

(*ii*) Let
$$P = (p_{ij})_{i,j=1,...,n} \in \mathbb{R}^{n \times n}$$
. Then, $|p_{ij}| \leq |P|_2$ for all $i, j = 1,..., n$.

Further, we recall some facts on orthogonal matrices.

Lemma 2.1.3. Consider an orthogonal matrix $O \in \mathbb{R}^{n \times n}$, i.e., satisfying $OO^T = O^T O = I_n$. Then, |Ox| = |x| for all $x \in \mathbb{R}^n$ and |det(O)| = 1.

Moreover, we introduce the following two definitions on matrix functions (see *Definition 2.2.1* for the definition of the involved spaces).

Definition 2.1.4. For a matrix function $\mathbb{D} = (d_{ij})_{i,j=1,...,n} \in C(\overline{\Omega}; \mathbb{R}^{n \times n})$ we write $\mathbb{D} > c$ if $y^T \mathbb{D}(x)y > c$ for all $x \in \overline{\Omega}$ and $y \in \mathbb{R}^n$ (analogously for $\geq, <, \leq$). Further, for $\mathbb{D} \geq 0$ we define the set

$$\{\mathbb{D} \ge 0\} := \{x \in \overline{\Omega} : \exists y \in \mathbb{R}^n \ s.t. \ y^T \mathbb{D}(x)y = 0\}.$$

Definition 2.1.5. For a matrix function $\mathbb{D} = (d_{ij})_{i,j=1,...,n} \in C(\overline{\Omega}; \mathbb{R}^{n \times n})$ and a function $u \in C^2(\overline{\Omega})$ we define

• the divergence

$$\nabla \cdot \mathbb{D}(x) = \sum_{i,j=1}^{n} (d_{ij})_{x_j}(x) e_i$$

• and the myopic diffusion

$$\nabla \nabla : (\mathbb{D}(x)u) = \nabla \cdot (\mathbb{D}(x)\nabla u + \nabla \cdot \mathbb{D}(x)u).$$

2.2 Function spaces

In this section we recall the definitions of certain spaces of continuous functions and Lebesgue and Sobolev spaces in x and t and some of their properties.

First, we define several spaces of continuous and Hölder continuous functions.

Definition 2.2.1. Let $k \in \mathbb{N}$, $j \in \mathbb{N} \cup \{\infty\}$, $l, m \in \mathbb{N}_0 \cup \{\infty\}$, T > 0, $S \subset \mathbb{R}^k$ nonempty and compact, $I \subset \mathbb{R}$ a nonempty interval and X a normed vector space with norm $\|\cdot\|_X$. We define the spaces

- C(S;X) of continuous functions $u: S \to X$,
- $C^{j}(S; X)$ of j-times continuously differentiable functions $u: S \to X$,
- $C^{l,m}(S \times I; X)$ of functions $u: S \times I \to X$ that are *l*-times continuously differentiable in $x \in S$ and m-times continuously differentiable in $t \in I$,
- + $C^{j}_{c}(S)$ of functions from $C^{j}(S;\mathbb{R})$ with compact support,
- $C_b(S)$ of functions from $C(S; \mathbb{R})$ that are bounded,
- $C_w([0,T];X)$ of functions $u:[0,T] \to X$ which are continuous w.r.t. the weak topology of X.

If $X = \mathbb{R}$ we leave out the dependence on \mathbb{R} in the notation of the space. If S is closed and $j < \infty$, we define the norms

$$\|u\|_{C(S;X)} := \max_{x \in S} \|u(x)\|_X,$$
$$\|u\|_{C^j(S;X)} := \sum_{|\alpha| \le j} \|D^{\alpha}u\|_X.$$

Definition 2.2.2. Let $\vartheta \in (0,1]$, $k \in \mathbb{N}_0$ and T > 0. We define the seminorms

$$\begin{split} \langle u \rangle_{\overline{\Omega}}^{\vartheta} &:= \sup_{x,y \in \overline{\Omega}, x \neq y} \frac{|u(x) - u(y)|}{|x - y|^{\vartheta}}, \\ \langle u \rangle_{x,\overline{\Omega} \times [0,T]}^{\vartheta} &:= \sup_{(x,t),(y,t) \in \overline{\Omega} \times [0,T], x \neq y} \frac{|u(x,t) - u(y,t)|}{|x - y|^{\vartheta}}, \\ \langle u \rangle_{t,\overline{\Omega} \times [0,T]}^{\vartheta} &:= \sup_{(x,t),(x,s) \in \overline{\Omega} \times [0,T], t \neq s} \frac{|u(x,t) - u(x,s)|}{|t - s|^{\vartheta}}. \end{split}$$

and the following Hölder spaces

· $C^{k+\vartheta}(\overline{\Omega})$ as the space of functions $u \in C^k(\overline{\Omega})$ with finite norm

$$||u||_{C^{k+\vartheta}(\overline{\Omega})} := \sum_{|\alpha| \leqslant k} ||D^{\alpha}u||_{C(\overline{\Omega})} + \sum_{|\alpha|=k} \langle D^{\alpha}u \rangle_{\overline{\Omega}}^{\vartheta},$$

 $\cdot C^{k+\vartheta}(\Omega)$ as the space

$$C^{k+\vartheta}(\Omega) := \left\{ u : \Omega \to \mathbb{R} : u \big|_{K} \in C^{k+\vartheta}(K) \,\forall K \subset \Omega \, \, compact \right\}.$$

$$(2.2.1)$$

· $C^{\vartheta,\frac{\vartheta}{2}}\left(\overline{\Omega}\times[0,T]\right)$ as the space of functions $u \in C\left(\overline{\Omega}\times[0,T]\right)$ that are ϑ -Hölder continuous in x and $\frac{\vartheta}{2}$ -Hölder continuous in t, i.e., with finite norm

$$||u||_{C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} := ||u||_{C(\overline{\Omega}\times[0,T])} + \langle u \rangle_{x,\overline{\Omega}\times[0,T]}^{\vartheta} + \langle u \rangle_{t,\overline{\Omega}\times[0,T]}^{\frac{\vartheta}{2}},$$

· $C^{1+\vartheta,\frac{1+\vartheta}{2}}(\overline{\Omega}\times[0,T])$ as the space of functions $u \in C^{1,0}(\overline{\Omega}\times[0,T])$ with finite norm

$$\begin{aligned} ||u||_{C^{1+\vartheta,\frac{1+\vartheta}{2}}(\overline{\Omega}\times[0,T])} &:= ||u\|_{C(\overline{\Omega}\times[0,T])} + \langle u \rangle_{t,\overline{\Omega}\times[0,T]}^{\frac{1+\vartheta}{2}} \\ &+ \sum_{i=1}^{n} \left(||u_{x_{i}}\|_{C(\overline{\Omega}\times[0,T])} + \langle u_{x_{i}} \rangle_{x,\overline{\Omega}\times[0,T]}^{\vartheta} + \langle u_{x_{i}} \rangle_{t,\overline{\Omega}\times[0,T]}^{\vartheta} \right), \end{aligned}$$

 $\cdot C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])$ as the space of functions $u \in C^{2,1}(\overline{\Omega}\times[0,T])$ with finite norm

$$\begin{split} ||u||_{C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} &:= ||u\|_{C(\overline{\Omega}\times[0,T])} + ||u_t\|_{C(\overline{\Omega}\times[0,T])} + \langle u_t \rangle_{x,\overline{\Omega}\times[0,T]}^{\vartheta} + \langle u_t \rangle_{t,\overline{\Omega}\times[0,T]}^{\frac{\vartheta}{2}} \\ &+ \sum_{i=1}^n \left(||u_{x_i}\|_{C(\overline{\Omega}\times[0,T])} + \langle u_{x_i} \rangle_{t,\overline{\Omega}\times[0,T]}^{\frac{1+\vartheta}{2}} \right) \\ &+ \sum_{i,j=1}^n \left(||u_{x_ix_j}\|_{C(\overline{\Omega}\times[0,T])} + \langle u_{x_ix_j} \rangle_{x,\overline{\Omega}\times[0,T]}^{\vartheta} + \langle u_{x_ix_j} \rangle_{t,\overline{\Omega}\times[0,T]}^{\vartheta} \right) \end{split}$$

• For k = 0, 1, 2 we define $C^{k+\vartheta, \frac{k+\vartheta}{2}}(\Omega \times (0,T))$ and $C^{k+\vartheta, \frac{k+\vartheta}{2}}(\overline{\Omega} \times (0,T))$ analogously to (2.2.1).

The lemmas below will be proved in the appendix.

Lemma 2.2.3. Let $\vartheta, \kappa \in (0, 1)$, $u \in C^{\vartheta}(\overline{\Omega})$ and $v \in C^{\kappa}(\overline{\Omega})$. Then,

- (i) $uv \in C^{\min\{\vartheta,\kappa\}}(\overline{\Omega}),$
- (ii) if $u \ge 0$ then $u^r \in C^{\vartheta r}(\overline{\Omega})$ if $r \in (0, 1)$,
- (iii) if $u \ge 0$ then $u^r \in C^{\vartheta}(\overline{\Omega})$ if r > 1,
- (iv) $\frac{1}{u} \in C^{\vartheta}(\overline{\Omega})$ if $u \neq 0$ in $\overline{\Omega}$.

Next, we define Lebesgue and Sobolev spaces and introduce dual pairings.

Definition 2.2.4. Let $p \in [1, \infty]$. We define $L^p(\Omega)$ as the space of functions $u : \Omega \to \mathbb{R}$ with finite norm

$$\|u\|_{L^p(\Omega)} := \begin{cases} \left(\int_{\Omega} |u|^p \, dx\right)^{\frac{1}{p}}, & \text{if } p \in [1, \infty), \\ \operatorname{ess\,sup}_{x \in \Omega} |u|, & \text{if } p = \infty. \end{cases}$$

Further, we define $(L^p(\Omega))^n$ and $(L^p(\Omega))^{n \times n}$, respectively, as the space of vector-valued functions $f: \Omega \to \mathbb{R}^n$ or matrix-valued functions $\mathbb{D}: \Omega \to \mathbb{R}^{n \times n}$ with finite norms

$$\begin{split} \|f\|_{(L^{p}(\Omega))^{n}} &:= \||f|\|_{L^{p}(\Omega)}, \\ \|\mathbb{D}\|_{(L^{p}(\Omega))^{n \times n}} &:= \||\mathbb{D}|_{2}\|_{L^{p}(\Omega)}, \end{split}$$

and for T > 0 and a Banach space X with norm $\|\cdot\|_X$ we define the Bochner space $L^p(0,T;X)$ as the space of functions $g:(0,T) \to X$ with finite norm

$$\|g\|_{L^p(0,T;X)} := \begin{cases} \left(\int_0^T \|g(t)\|_X^p \, dt \right)^{\frac{1}{p}}, & \text{if } p \in [1,\infty), \\ \operatorname{ess\,sup}_{t \in (0,T)} \|g(t)\|_X, & \text{if } p = \infty. \end{cases}$$

Definition 2.2.5. Let X a Banach space with dual space X^* . We denote the duality pairing of $x \in X$ and $x^* \in X^*$ as

$$\langle x^*, x \rangle_{X^*, X} := x^*(x) \in \mathbb{R}.$$

A sequence $(x_k)_{k\in\mathbb{N}} \subset X$ converges weakly to $x \in X$ in X if

$$\langle x^*, x_k \rangle_{X^*, X} \xrightarrow[k \to \infty]{} \langle x^*, x \rangle_{X^*, X}$$

for all $x^* \in X^*$. We denote this by $x_k \underset{k \to \infty}{\rightharpoonup} x$.

A sequence $(x_k^*)_{k \in \mathbb{N}} \subset X^*$ converges weakly-* to $x^* \in X^*$ in X^* if

$$\langle x_k^*, x \rangle_{X^*, X} \underset{k \to \infty}{\xrightarrow{}} \langle x^*, x \rangle_{X^*, X}$$

for all $x \in X$. We denote this by $x_k^* \stackrel{*}{\underset{k \to \infty}{\longrightarrow}} x^*$.

Definition 2.2.6. Let $k \in \mathbb{N}$ and $p \in [1, \infty]$. The Sobolev space $W_p^k(\Omega)$ is given by

$$W_p^k(\Omega) := \{ u \in L^p(\Omega) : \text{the weak derivative } D^{\alpha}u \text{ exists for all } \alpha \in \mathbb{N}_0^n \text{ s.t. } |\alpha| \leq k \text{ and } D^{\alpha}u \in L^p(\Omega) \}$$

with norm

$$|u||_{W_p^k(\Omega)} := \sum_{|\alpha| \leq k} ||D^{\alpha}u||_{L^p(\Omega)}.$$

Moreover, we set

$$\overset{\circ}{W_p^k}(\Omega) := \left\{ u \in W_p^k(\Omega) : \exists (u_m)_m \subset C_c^\infty(\Omega) \text{ s.t. } u_m \xrightarrow[m \to \infty]{} u \text{ in } W_p^k(\Omega) \right\}$$

with dual space

$$W_q^{-k}(\Omega) := \left(\overset{\circ}{W_p^k}(\Omega) \right)^*$$

for $q \in [1, \infty]$ s.t. $\frac{1}{p} + \frac{1}{q} = 1$.

For p = 2 we set $H^k(\Omega) := W_2^k(\Omega), \ H_0^k(\Omega) := \overset{\circ}{W_2^k}(\Omega) \ and \ H^{-1} := (H_0^1(\Omega))^*.$

Further, we define the spaces

$$\begin{split} W_p^{1,1}(\Omega \times (0,T)) &:= \{ u \in L^p(\Omega \times (0,T)) : u_{x_i}, u_t \in L^p(\Omega \times (0,T)) \; \forall i \in \{1,\dots,n\} \} \\ W_p^{2,1}(\Omega \times (0,T)) &:= \{ u \in L^p(\Omega \times (0,T)) : u_{x_i}, u_{x_ix_j}, u_t \in L^p(\Omega \times (0,T)) \\ \forall i, j \in \{1,\dots,n\} \} \end{split}$$

with norm

$$\begin{aligned} ||u||_{W_{p}^{1,1}(\Omega\times(0,T))} &:= ||u||_{L^{p}(\Omega\times(0,T))} + ||u_{t}||_{L^{p}(\Omega\times(0,T))} + \sum_{i=1}^{n} ||u_{x_{i}}||_{L^{p}(\Omega\times(0,T))}, \\ ||u||_{W_{p}^{2,1}(\Omega\times(0,T))} &:= ||u||_{L^{p}(\Omega\times(0,T))} + ||u_{t}||_{L^{p}(\Omega\times(0,T))} + \sum_{i=1}^{n} ||u_{x_{i}}||_{L^{p}(\Omega\times(0,T))} \\ &+ \sum_{i,j=1}^{n} ||u_{x_{i}x_{j}}||_{L^{p}(\Omega\times(0,T))}. \end{aligned}$$

We define continuous and compact embeddings and state some embeddings concerning Sobolev spaces.

Definition 2.2.7. Let X, Y be Banach spaces s.t. $X \subset Y$. We say that X is continuously embedded in Y (denoted by $X \hookrightarrow Y$) if there is a constant $C_1 > 0$ s.t. for all $u \in X$ the estimate $||u||_Y \leq C_1 ||u||_X$ holds. A continuous embedding is called compact (denoted by $X \hookrightarrow Y$) if each bounded sequence $(u_m)_{m \in \mathbb{N}} \subset X$ has a subsequence that converges in Y.

Lemma 2.2.8. (i) The space $H^1(\Omega)$ is continuously embedded in $L^p(\Omega)$, where $p \in [1, \infty]$ if $n = 1, p \in [1, \infty)$ if n = 2 and $p \in [1, \frac{2n}{n-2}]$ if n > 2. For such p there is a constant $K_S(p) > 0$ s.t. for all $u \in H^1(\Omega)$ it holds that

$$\|u\|_{L^p(\Omega)} \leq K_S(p) \|u\|_{H^1(\Omega)}.$$

- (ii) If p > n then $W_p^1(\Omega)$ is continuously embedded in $C^{\vartheta}(\overline{\Omega})$ (up to the choice of a continuous version) for $\vartheta = \left| \frac{n}{p} \right| + 1 \frac{n}{p}$.
- (iii) If p > n then $W_p^2(\Omega) \hookrightarrow C^1(\overline{\Omega})$ and u = 0 and $\nabla u = 0$ on $\partial\Omega$ for $u \in W_p^2(\Omega)$.
- (iv) For $p \in (1, \infty)$ the compact embedding $W_n^1(\Omega) \hookrightarrow \hookrightarrow L^p(\Omega)$ holds.

Proof. For (i) and (ii) see Theorem 6 in Section 5.6 in [58] and for (iv) see the remark at the end of Section 5.7 in [58].

If p > n by Theorem 6 in Section 5.6 in [58] the space $W_p^2(\Omega)$ is continuously embedded in $C^1(\overline{\Omega})$. By definition, for $u \in W_p^2(\Omega)$ there is a sequence $(u_m)_m \in C_c^\infty(\Omega)$ that converges to u in the W_p^2 -norm. Due to the continuous embedding $(u_m)_m$ also converges in the C^1 -norm. Hence, u = 0 and $\nabla u = 0$ on $\partial \Omega$.

Lemma 2.2.9. Let ϑ , $\kappa \in (0,1)$, $T \in (0,1)$ and $K_I(\vartheta) > 0$ denote the constant from the continuous embedding of $W^1_{\infty}(\Omega)$ into $C^{\vartheta}(\overline{\Omega})$ from Lemma 2.2.8(ii).

(i) If
$$u \in C^{1+\vartheta, \frac{1+\vartheta}{2}}(\overline{\Omega} \times [0,T])$$
, then it holds that

$$\|u - u(\cdot,0)\|_{C^{\vartheta, \frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])} \leq \max\{1, K_I(\vartheta)\}T^{\frac{\vartheta}{2}}\|u\|_{C^{1+\vartheta, \frac{1+\vartheta}{2}}(\overline{\Omega} \times [0,T])}.$$
(2.2.2)

(ii) If $u \in C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])$, then it holds that

$$\|u - u(\cdot, 0)\|_{C^{1+\vartheta, \frac{1+\vartheta}{2}}(\overline{\Omega} \times [0,T])} \leq 2 \max\{1, K_I(\vartheta)\} T^{\frac{1}{2}\min\{\vartheta, 1-\vartheta\}} \|u\|_{C^{2+\vartheta, 1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])}.$$

$$(2.2.3)$$

(iii) If $u \in C^{2+\kappa,1+\frac{\kappa}{2}}(\overline{\Omega} \times [0,T])$, then it holds that

$$\|u - u(\cdot, 0)\|_{C^{\vartheta, \frac{\vartheta}{2}}(\overline{\Omega} \times [0, T])} \leq 2 \max\{1, K_I(\vartheta)\} T^{\frac{1}{2}\min\{2-\vartheta, 1+\kappa\}} \|u\|_{C^{2+\kappa, 1+\frac{\kappa}{2}}(\overline{\Omega} \times [0, T])}.$$
(2.2.4)

2.3 Convolutions on bounded domains

In this section we will consider convolutions on bounded domains Ω . The definition is obtained from the definition of a convolution over \mathbb{R}^n using our convention that a function defined on a domain Ω is zero outside.

Definition 2.3.1. Let $p, q \in [1, \infty]$ s.t. $\frac{1}{p} + \frac{1}{q} = 1$ and $u \in L^q(\Omega)$. Set $S := \{x - y : x, y \in \Omega\}$ and consider $J \in L^p(S)$. For $x \in \Omega$ we define the convolution over the bounded domain Ω as

$$(J * u) (x) = \int_{\Omega} J(x - y)u(y) \, dy.$$

For a function $h \in L^{\infty}(\Omega)$ and an h-dependent kernel $J(\cdot, h)$ that satisfies $J(\cdot, 0) \in L^p(S)$ and is Lipschitz continuous in the second argument with Lipschitz constant $L \in L^p(S)$ the convolution is defined as

$$(J(\cdot, h) * u)(x) := \int_{\Omega} J(x - y, h(y))u(y) \, dy.$$
(2.3.1)

for $x \in \Omega$. For vector- or matrix-valued kernels J or functions u the convolution is defined componentwise.

The following lemma will be proved in the appendix.

Lemma 2.3.2. Let T > 0, $\vartheta \in (0, 1)$, $p \in (1, \infty)$, $\beta \ge 1$, $S := \{x - y : x, y \in \Omega\}$, J(x, h) a kernel that satisfies $J(\cdot, 0) \in L^p(S)$ and is Lipschitz continuous in the second argument with Lipschitz constant $L \in L^p(S)$.

(i) If $u, h \in L^{\infty}(\Omega \times (0,T))$ then, $J(\cdot,h) * u^{\beta} \in L^{\infty}(\Omega \times (0,T))$ and satisfies

$$\|J(\cdot,h) \ast u^{\beta}\|_{L^{\infty}(\Omega \times (0,T))} \leqslant \|u\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \left(\|L\|_{L^{1}(S)}\|h\|_{L^{\infty}(\Omega \times (0,T))} + \|J(x,0)\|_{L^{1}(S)}\right) \leq \|u\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \left(\|L\|_{L^{1}(S)}\|h\|_{L^{\infty}(\Omega \times (0,T))} + \|J(x,0)\|_{L^{1}(S)}\right) \leq \|u\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \left(\|L\|_{L^{1}(S)}\|h\|_{L^{\infty}(\Omega \times (0,T))} + \|J(x,0)\|_{L^{1}(S)}\right) \leq \|u\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \left(\|L\|_{L^{1}(S)}\|h\|_{L^{\infty}(\Omega \times (0,T))} + \|J(x,0)\|_{L^{1}(S)}\right)$$

(ii) If $u, h \in C^{\vartheta, \frac{\vartheta}{2}}(\overline{\Omega} \times [0, T])$ and $u \ge 0$ then, for $\kappa := \min\{\vartheta, \frac{p-1}{p}\}$ it holds that $J(\cdot, h) * u^{\beta} \in C^{\kappa, \frac{\kappa}{2}}(\overline{\Omega} \times [0, T]).$

If J is independent from h this holds setting J(x,h) := J(x) for $x \in S$ and h = 0 in the above estimate.

Definition 2.3.3. We define the standard mollifier $\varsigma \in C^{\infty}(\mathbb{R}^n)$ as

$$\varsigma(x) := \begin{cases} C_2 e^{\frac{1}{|x|^2 - 1}} & \text{if } |x| < 1, \\ 0 & \text{if } |x| \ge 1, \end{cases}$$

where $C_2 > 0$ is chosen s.t. $\int_{\mathbb{R}^n} \varsigma \, dx = 1$. For $\varepsilon > 0$ we set $\varsigma_{\varepsilon}(x) := \frac{1}{\varepsilon^n} \varsigma\left(\frac{x}{\varepsilon}\right)$ with $supp(\varsigma_{\varepsilon}) \subset \overline{B_{\varepsilon}}$ and $\int_{B_{\varepsilon}} \varsigma_{\varepsilon} \, dx = 1$. **Lemma 2.3.4.** ([58, Appendix C, Theorem 6(i) + (iii)]) Let $u \in L^1_{loc}(\Omega)$. Then, $\varsigma_{\varepsilon} * u \in C^{\infty}(\Omega_{\varepsilon})$. If $u \in C(\Omega)$, then $\varsigma_{\varepsilon} * u \to u$ for $\varepsilon \to 0$ uniformly on compact subsets of Ω .

Part I

Nonlocal models with nonlocality in the advection term

Nonlocal and local models for taxis in cell migration: a rigorous limit procedure

This chapter was first published in Volume 81 of Journal of Mathematical Biology in 2020.¹ The presentation has been adapted for use in this dissertation to clarify the details of the proofs and guarantee consistency of the notation.

3.1 Motivation

Macroscopic equations and systems describing the evolution of populations in response to soluble and insoluble environmental cues have been intensively studied and the palette of such reactiondiffusion-taxis models is continuously expanding. Models of such form are motivated by problems arising in various contexts, a large part related to cell migration and proliferation connected to tumor invasion, embryonal development, wound healing, biofilm formation, insect behavior in response to chemical cues, etc. We refer, e.g., to [9] for a recent review also containing some deduction methods for taxis equations based on kinetic transport equations.

Apart from such purely local PDE systems with taxis, several spatially nonlocal models have been introduced over the last two decades and are attracting ever increasing interest. They involve integro-differential operators in one or several terms of the featured reaction-diffusionadvection equations. Their aim is to characterize interactions between individuals or signal perception happening not only at a specific location, but over a whole set (usually a ball) containing (centered at) that location. In the context of cell populations, for instance, this seems to be a more realistic modeling assumption, as cells are able to extend various protrusions (such as lamellipodia, filopodia, cytonemes, etc.) into their surroundings, which can reach across long distances compared against cell size, see [71, 130] and references therein. Moreover, the cells are able to relay signals they perceive and thus transmit them to cells with which they are not in direct contact, thereby influencing their motility, see e.g., [57, 65]. Cell-cell and cell-tissue adhesion are essential for mutual communication, homeostasis, migration, proliferation, sorting, and many other biological processes. A large variety of models for adhesive behavior at the cellular level have been developed to account for the dynamics of focal contacts, e.g., [6, 7, 146] and to assess their influence on cytoskeleton restructuring and cell migration, e.g., [40, 41, 93,

 $^{^{1}}$ [47] The paper is licensed under a Creative Commons Attribution 4.0 International License.

143]. Continuous, spatially nonlocal models involving adhesion were introduced more recently [5] and are attracting increasing interest from the modeling [16, 21, 23, 43, 68, 69, 109, 118], analytical [26, 45, 46, 80, 132], and numerical [67] viewpoints. Yet more recent models [44, 56] also take into account subcellular level dynamics, thus involving further nonlocalities (besides adhesion), with respect to some structure variable referring to individual cell state. Thereby, multiscale mathematical settings are obtained, which lead to challenging problems for analysis and numerics. Another essential aspect of cell migration is the directional bias in response to a diffusing signal, commonly termed chemotaxis. A model of cell migration with finite sensing radius, thus featuring nonlocal chemotaxis has been introduced in [117] and readdressed in [79] from the perspective of well-posedness, long time behaviour, and patterning. We also refer to [105] for further spatially nonlocal models and their formal deduction.

For adhesion and nonlocal chemotaxis models, a gradient of some nondiffusing or diffusing signal is replaced by a nonlocal integral term. Here we are only interested in this type of model, and refer to [28, 51, 87] for reviews on settings involving other types of nonlocality. Specifically, following [5, 68, 79, 117], we consider the subsequent systems, whose precise mathematical formulations will be specified further below:

1. a prototypical nonlocal model for adhesion

$$\partial_t c_r = \nabla \cdot \left(D_c(c_r, v_r) \nabla c_r - c_r \chi(c_r, v_r) \mathcal{A}_r(g(c_r, v_r)) \right) + f_c(c_r, v_r), \tag{3.1.1a}$$

$$\partial_t v_r = f_v(c_r, v_r), \tag{3.1.1b}$$

where

$$\mathcal{A}_{r}u(x) := \frac{1}{r} \int_{B_{r}} u(x+\xi) \frac{\xi}{|\xi|} F_{r}(|\xi|) \,\mathrm{d}\xi$$
(3.1.2)

is referred to as the adhesion velocity, and the function F_r describes how the magnitude of the interaction force depends on the interaction range $|\xi|$ within the sensing radius r. We require this function to satisfy

Assumptions 3.1.1. (Assumptions on F_r)

- (a) $(r, \rho) \mapsto F_r(\rho)$ is continuous and positive in $[0, r_0]^2$ for some $r_0 > 0$;
- (b) $F_0(0) = n + 1.^2$

The quantity

$$\mathbb{F}(c_r, v_r) = c_r \chi(c_r, v_r) \mathcal{A}_r(g(c_r, v_r))$$

is often referred to as the total adhesion flux, possibly scaled by some constant involving the typical cell size or the sensing radius, see e.g., [5, 21]. Here we also include a coefficient $\chi(c_r, v_r)$ that depends on cell and tissue (extracellular matrix, ECM) densities, which can be seen as characterizing the sensitivity of cells towards their neighbours and the surrounding tissue. It will, moreover, help provide in a rather general framework a unified presentation of this and the subsequent local and nonlocal model classes for adhesion, haptotactic, and chemotactic behavior of moving cells.

 $^{^{2}}$ In Section 3.2 we will see that this is, indeed, the 'right' normalisation. If we assume, as in [5], that this function is a constant involving some viscosity related proportionality, then this choice provides the value of that constant.

System (3.1.1) is a simplification of the integro-differential system (4) in [68]. The main difference between the two settings is that in our case we ignore the so-called matrix-degrading enzymes (MDEs). Instead, we assume cells directly degrade the tissue: this fairly standard simplification (e.g., [118]) effectively assumes that proteolytic enzymes remain localised to the cells, and helps simplify the analysis. On the other hand, (3.1.1) can also be viewed as a nonlocal version of the haptotaxis model with nonlinear diffusion:

$$\partial_t c = \nabla \cdot (D_c(c, v) \nabla c - c\chi(c, v) \nabla g(c, v)) + f_c(c, v), \qquad (3.1.3a)$$

$$\partial_t v = f_v(c, v); \tag{3.1.3b}$$

2. a prototypical nonlocal chemotaxis-growth model

$$\partial_t c_r = \nabla \cdot \left(D_c(c_r, v_r) \nabla c_r - c_r \chi(c_r, v_r) \mathring{\nabla}_r v_r \right) + f_c(c_r, v_r), \tag{3.1.4a}$$

$$\partial_t v_r = D_v \Delta v_r + f_v(c_r, v_r) \tag{3.1.4b}$$

with the nonlocal gradient

$$\overset{\circ}{\nabla}_{r}u(x) := \frac{n}{r^{2}} \int_{S_{r}} u(x+\xi)\xi \,\mathrm{d}\sigma(\xi) = \frac{n}{r} \int_{S_{r}} u(x+\xi)\frac{\xi}{|\xi|} \,\mathrm{d}\sigma(\xi).$$
(3.1.5)

System (3.1.4) can be seen as a nonlocal version of the chemotaxis-growth model

$$\partial_t c = \nabla \cdot (D_c(c, v) \nabla c - c\chi(c, v) \nabla v) + f_c(c, v), \qquad (3.1.6a)$$

$$\partial_t v = D_v \Delta v + f_v(c, v), \tag{3.1.6b}$$

where $\chi(c, v)$ is the chemotactic sensitivity function. As mentioned above, in order to have a unified description of our systems (3.1.3) and (3.1.6) and of their respective nonlocal counterparts (3.1.1) and (3.1.4), we later introduce a more general version of the nonlocal chemotaxis flux, similar to the above adhesion velocity \mathcal{A}_r .

The nonlocal systems (3.1.3) and (3.1.6) are stated for

$$t > 0, \qquad x \in \Omega \subset \mathbb{R}^n.$$

Unless the spatial domain Ω is the whole \mathbb{R}^n , suitable boundary conditions are required. In the latter case, usually periodicity is assumed, which is not biologically realistic in general. Still, this offers the easiest way to properly define the output of the nonlocal operator in the boundary layer where the sensing region is not fully contained in Ω . Very recently various other boundary conditions have been derived and compared in the context of a single equation modeling cell-cell adhesion in 1D [82].

Few previous works focus on solvability for models with nonlocality in a taxis term. Some of them deal with single equations that only involve cell-cell adhesion [45, 46, 82], others study nonlocal systems of the sort considered here for two [79] or more components [56]. The global solvability and boundedness study in [80] is obtained for the case of a nonlocal operator with integration over a set of sampling directions being an open, not necessarily strict subset of \mathbb{R}^n . The systems studied there include settings with a third equation for the dynamics of diffusing MDEs. Conditions which secure uniform boundedness of solutions to such cell-cell and cell-tissue adhesion models in 1D were elaborated in [132]. Some heuristic analysis via local Taylor expansions was performed in [68] and [81] showing that as $r \to 0$ the outputs $\mathcal{A}_r u$ and $\mathring{\nabla}_r u$, respectively, converge pointwise to ∇u for a fixed and sufficiently smooth u. In [79] it was observed that it would be interesting to study rigorously the limiting behaviour of solutions of the nonlocal problems involving $\dot{\nabla}_r u$. The authors ask in which sense, if at all, do these solutions converge to solutions of the corresponding local problem as $r \to 0$. Numerical results appeared to confirm that, in certain cases, the answer is positive. Still, to the best of our knowledge, no rigorous analytical study of this issue has as yet been performed. Clearly, any approach based on representations using Taylor polynomials requires a rather high order regularity of solution components and a suitable control on the approximation errors, and that uniformly in r. This is difficult or even impossible to obtain in most cases, particularly when dealing with weak solutions. In this chapter we propose a different approach based on the representation of the input u in terms of an integral of ∇u over line segments. This leads to a new description of the nonlocal operators \mathcal{A}_r and $\check{\nabla}_r$ in terms of nonlocal operators applied to gradients (see Section 3.2 below). Moreover, it turns out that redefining their outputs inside the vanishing boundary layer in a suitable way allows one to perform a rigorous proof of convergence: Under suitable assumptions on the system coefficients and other parameters, appropriately defined sequences of solutions to nonlocal problems involving the mentioned modified nonlocal operators converge for $r \to 0$ to those of the corresponding local models (3.1.3) and (3.1.6), respectively. Our convergence proof is based on estimates on c_r and v_r which are uniform in r and on a compactness argument. The two models (3.1.1) and (3.1.4) are chosen as illustrations, however our idea can be further applied to other integro-differential systems with similar properties.

The rest of the chapter is organised as follows. In Section 3.2 we introduce the aforementioned adaptations of the nonlocal operators \mathcal{A}_r and $\mathring{\nabla}_r$ and study their limiting properties as r becomes infinitesimally small. This turns out to be useful for our convergence proof later. We also establish in Section 3.3 the well-posedness for a certain class of equations including such operators. In the subsequent Section 3.4 we introduce a couple of nonlocal models that involve the previously considered averaging operators, prove the global existence of solutions of the respective systems, and investigate their limit behaviour as $r \to 0$. Section 3.5 provides some numerical simulations comparing various nonlocal and local models considered in this work in the 1D case. Finally, Section 3.6 contains a discussion of the results and a short outlook on open issues.

3.2 Operators A_r and $\check{\nabla}_r$ and averages of ∇

In this section we study the applications of the nonlocal operators \mathcal{A}_r and $\mathring{\nabla}_r$ to fixed, i.e., independent of r, functions u. Our focus is on the limiting behaviour as $r \to 0$. Formal Taylor expansions performed in [68, 79] anticipate that the limit is the gradient operator in both cases. This we prove here rigorously under rather mild regularity assumptions on u. To be more precise, we replace \mathcal{A}_r and $\mathring{\nabla}_r$ by certain integral operators \mathcal{T}_r and \mathcal{S}_r (see (3.2.2) and (3.2.7) below) applied to ∇u and show that these operators are pointwise approximations of the identity operator in the L^p spaces.

Recall that $\Omega \subset \mathbb{R}^n$ is a bounded domain with smooth enough boundary. Unless explicitly stated, the constants C_i in this chapter **do not** depend upon r.

We start with the operator \mathcal{A}_r . For $r \in (0, r_0]$, $u \in C^1(\Omega)$, and $x \in \Omega_r$ we compute using the mean value and Fubini's theorem that

$$\begin{aligned} \mathcal{A}_{r}u(x) &= \frac{1}{r} \int_{B_{r}} u(x+\xi) \frac{\xi}{|\xi|} F_{r}(|\xi|) \,\mathrm{d}\xi \\ &= \frac{1}{r} \int_{B_{r}} (u(x+\xi) - u(x)) \frac{\xi}{|\xi|} F_{r}(|\xi|) \,\mathrm{d}\xi \\ &= \frac{1}{r} \int_{B_{r}} \int_{0}^{1} (\nabla u(x+s\xi) \cdot \xi) \,\mathrm{d}s \, \frac{\xi}{|\xi|} F_{r}(|\xi|) \,\mathrm{d}\xi \\ &= \frac{1}{r} \int_{0}^{1} \int_{B_{r}} (\nabla u(x+s\xi) \cdot \xi) \frac{\xi}{|\xi|} F_{r}(|\xi|) \,\mathrm{d}\xi \,\mathrm{d}s \\ &= \int_{0}^{1} \int_{B_{1}} (\nabla u(x+rsy) \cdot y) \frac{y}{|y|} F_{r}(r|y|) \,\mathrm{d}y \,\mathrm{d}s. \end{aligned}$$
(3.2.1)

Formula (3.2.1) extends to arbitrary $u \in W_1^1(\Omega)$ by means of a density argument. Motivated by (3.2.1) we introduce the averaging operator

$$\mathcal{T}_{r}w(x) := \int_{0}^{1} \int_{B_{1}} (w(x + rsy) \cdot y) \frac{y}{|y|} F_{r}(r|y|) \,\mathrm{d}y \,\mathrm{d}s.$$
(3.2.2)

In Subsection 3.2.1 we check that $\mathcal{T}_r w(x)$ is well-defined for all $w \in (L^1(\Omega))^n$ and a.e. $x \in \Omega$. In this notation, for all $r \in (0, r_0]$ and $u \in W_1^1(\Omega)$ identity (3.2.1) takes the form

$$\mathcal{A}_r u = \mathcal{T}_r(\nabla u)$$
 a.e. in Ω_r .

In the limiting case r = 0 we have for $x \in \Omega$ that

$$\begin{aligned} \mathcal{T}_{0}w(x) &= \int_{0}^{1} \int_{B_{1}} (w(x) \cdot y) \frac{y}{|y|} F_{0}(0) \, \mathrm{d}y \, \mathrm{d}s, \\ &= F_{0}(0) \sum_{i,j=1}^{n} w_{i}(x) e_{j} \int_{B_{1}} \frac{y_{i}y_{j}}{|y|} \, \mathrm{d}y \\ &= F_{0}(0) \sum_{i,j=1}^{n} w_{i}(x) e_{j} \delta_{ij} \int_{B_{1}} \frac{y_{i}^{2}}{|y|} \, \mathrm{d}y \\ &= F_{0}(0) \sum_{i=1}^{n} w_{i}(x) e_{i} \int_{B_{1}} \frac{y_{i}^{2}}{|y|} \, \mathrm{d}y \\ &= F_{0}(0) \sum_{i=1}^{n} w_{i}(x) e_{i} \frac{1}{n} \int_{B_{1}} \frac{|y|^{2}}{|y|} \, \mathrm{d}y \\ &= F_{0}(0) \frac{1}{n} \int_{B_{1}} |y| \, \mathrm{d}y \, w(x) \\ &= w(x). \end{aligned}$$
(3.2.3)

In the final step we used Assumptions 3.1.1(b) which says that $F_0(0) = n + 1$ (this explains our choice) and the trivial identity

$$\int_{B_1} |y| \, \mathrm{d}y = \frac{n}{n+1}.$$
(3.2.4)

Thus, we have just proved the following lemma:

Lemma 3.2.1. (Adhesion velocity vs. \mathcal{T}_r) Let $u \in W_1^1(\Omega)$. Then for $r \in (0, r_0]$ it holds that

$$\mathcal{A}_r u = \mathcal{T}_r(\nabla u) \qquad a.e. \ in \ \Omega_r. \tag{3.2.5}$$

Moreover, if $F_0(0) = n + 1$, then

$$\nabla u = \mathcal{T}_0(\nabla u) \qquad a.e. \ in \ \Omega. \tag{3.2.6}$$

In a very similar manner one can establish a representation for $\overset{\circ}{\nabla}_r$. For this purpose we define for $r \in (0, r_0]$ the averaging operator

$$\mathcal{S}_r w(x) := n \int_0^1 \oint_{S_1} (w(x + rsy) \cdot y) y \,\mathrm{d}\sigma(y) \,\mathrm{d}s. \tag{3.2.7}$$

The corresponding result then reads:

Lemma 3.2.2. (Nonlocal gradient vs. S_r) Let $u \in W_1^1(\Omega)$. Then for $r \in (0, r_0]$ it holds that

$$\check{\nabla}_r u = \mathcal{S}_r(\nabla u) \qquad a.e. \ in \ \Omega_r, \tag{3.2.8}$$

$$\nabla u = \mathcal{S}_0(\nabla u) \qquad a.e. \ in \ \Omega. \tag{3.2.9}$$

The proof of Lemma 3.2.2 is very similar to that of Lemma 3.2.1 and we omit it here.

Next, we observe that identity (3.2.5) was established for Ω_r . In the boundary layer $\Omega \setminus \Omega_r$ the definition (3.1.2) of the adhesion velocity allows various extensions. For example, one could keep (3.1.2) by assuming (as done here and, e.g., in [56]) that u := 0 in $\mathbb{R}^n \setminus \Omega$. An alternative would be to average over the part of the *r*-ball that lies inside the domain. Let us have a closer look at the first option (the second can be handled similarly). Consider the following example:

Example 3.2.3. $(\mathcal{A}_r \text{ vs. } \mathcal{T}_r(\nabla \cdot) \text{ in 1D})$ Let $\Omega = (-1, 1), r_0 = 1, F_r \equiv 2$, and $u \equiv 1$. In this case, $u' \equiv 0$, hence

$$\mathcal{T}_r(u') \equiv 0 \equiv u'.$$

For \mathcal{A}_r one readily computes by assuming u = 0 in $\mathbb{R} \setminus (-1, 1)$ that for $x \in (-1, 1)$

~

$$\begin{aligned} \mathcal{A}_{r}u(x) &= \frac{2}{r} \frac{1}{2r} \int_{(-1-x,1-x) \cap (-r,r)} \operatorname{sign}(\xi) \,\mathrm{d}\xi \\ &= \begin{cases} \frac{1}{r^{2}}(-1+r-x) & \text{in } [-1,-1+r], \\ 0 & \text{in } (-1+r,1-r) = \Omega_{r}, \\ \frac{1}{r^{2}}(1-r-x) & \text{in } [1-r,1], \end{cases} \end{aligned}$$

so that

$$\begin{aligned} |\mathcal{A}_{r}u||_{L^{1}(-1,1)} &= ||\mathcal{A}_{r}u||_{L^{1}(\Omega\setminus\Omega_{r})} \\ &= \frac{1}{r^{2}} \int_{-1}^{-1+r} |-1+r-x| \, \mathrm{d}x + \frac{1}{r^{2}} \int_{1-r}^{1} |1-r-x| \, \mathrm{d}x \\ &= 1, \end{aligned}$$

although

$$|\Omega \backslash \Omega_r| = 2r \underset{r \to 0}{\to} 0.$$

Thus,

$$\mathcal{A}_r u \underset{r \to 0}{\to} 0 \equiv u'$$

in the measure but not in $L^1(\Omega)$.
Example 3.2.3 supports our idea to average ∇u instead of u itself. The same applies to $\check{\nabla}_r u$ vs. $S_r(\nabla u)$.

Averaging w.r.t. $y \in B_1$ and then also w.r.t. $s \in (0, 1)$ might appear superfluous in the definition of the operator \mathcal{T}_r . The following example compares the effect of \mathcal{T}_r with that of an operator which averages w.r.t. to y only.

Example 3.2.4. Let $\Omega = \mathbb{R}^n$, $n \ge 2$, and r > 0, $F_r \equiv n + 1$. In this case

$$\mathcal{T}_r w(x) := (n+1) \int_0^1 \int_{B_1} (w(x+rsy) \cdot y) \frac{y}{|y|} \,\mathrm{d}y \,\mathrm{d}s.$$

Consider also the operator

$$\widetilde{\mathcal{T}}_r w(x) := (n+1) \int_{B_1} (w(x+ry) \cdot y) \frac{y}{|y|} \,\mathrm{d}y.$$

It is easy to see that both operators are well-defined, linear, continuous, and self-adjoint in the space $(L^2(\mathbb{R}^n))^n$ (see Lemma 3.2.5 below). Moreover, they map the dense subspace $C_c(\mathbb{R}^n;\mathbb{R}^n)$ into itself. This suggests the following natural extension to $(C_c(\mathbb{R}^n;\mathbb{R}^n))^*$:

$$\langle \mathcal{T}_r \mu, \varphi \rangle_{(C_c(\mathbb{R}^n; \mathbb{R}^n))^*, C_c(\mathbb{R}^n; \mathbb{R}^n)} := \langle \mu, \mathcal{T}_r \varphi \rangle_{(C_c(\mathbb{R}^n; \mathbb{R}^n))^*, C_c(\mathbb{R}^n; \mathbb{R}^n)}, \\ \left\langle \tilde{\mathcal{T}}_r \mu, \varphi \right\rangle_{(C_c(\mathbb{R}^n; \mathbb{R}^n))^*, C_c(\mathbb{R}^n; \mathbb{R}^n)} := \left\langle \mu, \tilde{\mathcal{T}}_r \varphi \right\rangle_{(C_c(\mathbb{R}^n; \mathbb{R}^n))^*, C_c(\mathbb{R}^n; \mathbb{R}^n)}.$$

Let, for instance,

$$w := \delta_0 e_1,$$

where δ_0 means the usual Dirac delta. One readily computes that

$$\widetilde{\mathcal{T}}_r(\delta_0 e_1)(x) = \frac{n+1}{|B_r|} \chi_{B_r}(x) \frac{x_1}{r} \frac{x}{|x|},$$

whereas

$$\mathcal{T}_{r}(\delta_{0}e_{1})(x) = \frac{n+1}{|B_{r}|} \int_{0}^{1} s^{-n-1} \chi_{B_{rs}}(x) \,\mathrm{d}s \frac{x_{1}}{r} \frac{x}{|x|}$$
$$= \frac{n+1}{n|B_{r}|} \left(\left(\frac{r}{|x|}\right)^{n} - 1 \right)_{+} \frac{x_{1}}{r} \frac{x}{|x|}.$$

For $n \ge 2$, the operator \mathcal{T}_r retains the singularity at the origin, however making it less concentrated, while $\tilde{\mathcal{T}}_r$ eliminates that singularity entirely and produces instead jump discontinuities all over S_r .

3.2.1 Properties of the averaging operators T_r and S_r

In this section we collect some properties of the averaging operators \mathcal{T}_r and \mathcal{S}_r .

Lemma 3.2.5. (Properties of \mathcal{T}_r) Let F_r satisfy Assumptions 3.1.1 and let $r \in (0, r_0]$. Then:

(i) \mathcal{T}_r is a well-defined continuous linear operator in $(L^p(\Omega))^n$ for all $p \in [1, \infty]$. The corresponding operator norm satisfies

$$\|\mathcal{T}_r\|_{L((L^p(\Omega))^n)} \leqslant C_1(r,p), \tag{3.2.10}$$

where

$$C_1(r,p) := \begin{cases} \left(n \int_0^1 \rho^{n-1+p^*} (F_r(r\rho))^{p^*} d\rho\right)^{\frac{1}{p^*}} & \text{for } p \in (1,\infty], \ p^* = \frac{p}{p-1}, \\ \max_{\rho \in [0,1]} \rho F_r(r\rho) & \text{for } p = 1. \end{cases}$$

(ii) Let $p, p^* \in [1, \infty]$ be such that $p^* = \frac{p}{p-1}$. For all $w_1 \in (L^p(\Omega))^n$ and $w_2 \in (L^{p^*}(\Omega))^n$ it holds:

$$\int_{\Omega} (\mathcal{T}_r w_1(x) \cdot w_2(x)) \, dx = \int_{\Omega} (w_1(x) \cdot \mathcal{T}_r w_2(x)) \, dx. \tag{3.2.11}$$

(iii) Let $p \in [1, \infty)$. For all $w \in (L^p(\Omega))^n$ it holds that

$$\mathcal{T}_r w \xrightarrow[r \to 0]{} \mathcal{T}_0 w = w \qquad in \ (L^p(\Omega))^n.$$
 (3.2.12)

(iv) For p = 2 it holds that

$$\|\mathcal{T}_r\|_{L((L^2(\Omega))^n)} \xrightarrow[r \to 0]{\to} 1.$$
(3.2.13)

Remark 3.2.6. Due to the assumptions on F_r we have in the limit that

$$C_1(r,p) \underset{r \to 0}{\to} C_2(p) := \begin{cases} (n+1) \left(\frac{n}{n+p^*}\right)^{\frac{1}{p^*}} & \text{for } p \in (1,\infty], \ p^* = \frac{p}{p-1}, \\ n+1 & \text{for } p = 1. \end{cases}$$
(3.2.14)

Proof of Lemma 3.2.5. (i) Since w is measurable and $\rho \mapsto F_r(\rho)$, $(x, s, y) \mapsto x + rsy$, $(y, z) \mapsto (z \cdot y) \frac{y}{|y|}$ are continuous, we have that

$$(x, y, s) \mapsto (w(x + rsy) \cdot y) \frac{y}{|y|} F_r(r|y|)$$

is well-defined a.e. in $\Omega \times B_1 \times (0,1)$ and is measurable. Let $p \in (1,\infty)$ and $p^* = \frac{p}{p-1}$. We compute

$$\int_{B_1} \left(|y| F_r(r|y|) \right)^{p^*} dy = \frac{1}{|B_1|} \int_0^1 \rho^{n-1+p^*} (F_r(r\rho))^{p^*} d\rho \cdot 2\pi \prod_{k=2}^{n-1} \int_0^\pi \sin(\phi_{n-k})^{k-1} d\phi_{n-k}$$
$$= n \int_0^1 \rho^{n-1+p^*} (F_r(r\rho))^{p^*} d\rho$$

using spherical coordinates and the properties of the Gamma function. With the help of this equality, Hölder's inequality, Fubini's theorem, and our convention that w vanishes outside Ω , we deduce for all $w \in (L^p(\Omega))^n$ that

$$\begin{split} &\|\mathcal{T}_{r}w\|_{(L^{p}(\Omega))^{n}}^{p} \\ &= \int_{\Omega} \left| \int_{0}^{1} \int_{B_{1}} (w(x+rsy) \cdot y) \frac{y}{|y|} F_{r}(r|y|) \, \mathrm{d}y \, \mathrm{d}s \right|^{p} \, \mathrm{d}x \\ &\leqslant \int_{\Omega} \int_{0}^{1} \int_{B_{1}} |w(x+rsy)|^{p} \, \mathrm{d}y \left(\int_{B_{1}} (|y|F_{r}(r|y|))^{p^{*}} \, \mathrm{d}y \right)^{\frac{p}{p^{*}}} \, \mathrm{d}s \, \mathrm{d}x \\ &= C_{1}^{p}(r,p) \int_{0}^{1} \int_{B_{1}} \int_{\Omega} |w(x+rsy)|^{p} \, \mathrm{d}x \, \mathrm{d}y \, \mathrm{d}s, \end{split}$$

$$\leq C_1^p(r,p) \int_0^1 \oint_{B_1} \int_{\Omega} |w(z)|^p \, \mathrm{d}z \, \mathrm{d}y \, \mathrm{d}s$$
$$= C_1^p(r,p) \|w\|_{(L^p(\Omega))^n}^p.$$

This implies that for all $p \in (1, \infty)$ the operator \mathcal{T}_r is well-defined in $(L^p(\Omega))^n$. It is also clearly linear. Taken together we then have that $\mathcal{T}_r \in L((L^p(\Omega))^n)$ and (3.2.10) holds. The cases p = 1 and $p = \infty$ can be treated similarly.

(ii) Let $w_1 \in (L^p(\Omega))^n$ and $w_2 \in (L^{p^*}(\Omega))^n$. We compute by using Fubini's theorem, the symmetry of B_1 , and simple variable transformations that

$$\begin{split} &\int_{\Omega} \left(\mathcal{T}_{r} w_{1}(x) \cdot w_{2}(x) \right) \mathrm{d}x \\ &= \int_{\Omega} \int_{0}^{1} \int_{B_{1}} \left(w_{1}(x + rsy) \cdot y \right) \frac{y}{|y|} F_{r}(r|y|) \mathrm{d}y \, \mathrm{d}s \cdot w_{2}(x) \, \mathrm{d}x \\ &= \int_{0}^{1} \int_{B_{1}} |y| F_{r}(r|y|) \\ &\quad \cdot \int_{\Omega} \left(w_{1}(x + rsy) \cdot \frac{y}{|y|} \right) \left(w_{2}(x) \cdot \frac{y}{|y|} \right) \, \mathrm{d}x \, \mathrm{d}y \, \mathrm{d}s \\ &= \int_{0}^{1} \int_{B_{1}} |y| F_{r}(r|y|) \\ &\quad \cdot \int_{\Omega \cap (-rsy + \Omega)} \left(w_{1}(x + rsy) \cdot \frac{y}{|y|} \right) \left(w_{2}(x) \cdot \frac{y}{|y|} \right) \, \mathrm{d}x \, \mathrm{d}y \, \mathrm{d}s \end{split}$$
(3.2.15)
$$&= \int_{0}^{1} \int_{B_{1}} |y| F_{r}(r|y|) \\ &\quad \cdot \int_{(rsy + \Omega) \cap \Omega} \left(w_{1}(z) \cdot \frac{y}{|y|} \right) \left(w_{2}(z - rsy) \cdot \frac{y}{|y|} \right) \, \mathrm{d}z \, \mathrm{d}y \, \mathrm{d}s \\ &= \int_{0}^{1} \int_{B_{1}} |y| F_{r}(r|y|) \\ &\quad \cdot \int_{(-rsy + \Omega) \cap \Omega} \left(w_{1}(z) \cdot \frac{y}{|y|} \right) \left(w_{2}(z + rsy) \cdot \frac{y}{|y|} \right) \, \mathrm{d}z \, \mathrm{d}y \, \mathrm{d}s.$$
(3.2.16)

Thereby we used our convention that each function defined in Ω is assumed to be prolonged by zero outside Ω . Comparing (3.2.15) and (3.2.16) we obtain (3.2.11).

(iii) We apply the Banach-Steinhaus theorem. Due to (i) and (3.2.14), $\{\mathcal{T}_r\}_{r\in(0,r_0]}$ is a family of uniformly bounded linear operators in the Banach space $(L^p(\Omega))^n$. Thus, as $C_c(\Omega; \mathbb{R}^n)$ is dense in $(L^p(\Omega))^n$ for $p < \infty$, we only need to check (3.2.12) for $w \in C_c(\Omega; \mathbb{R}^n)$. But for such w we can directly pass to the limit under the integral and thus obtain using (3.2.3) and the dominated convergence theorem that

$$\mathcal{T}_r w \xrightarrow[r \to 0]{} \mathcal{T}_0 w = w$$
 for all $x \in \Omega$ and in $(L^p(\Omega))^n$.

(iv) Here we make use of the Fourier transform, which we denote by the hat symbol. A straightforward calculation using Fubini's theorem and a variable transformation shows that for $w \in (L^p(\Omega))^n$ and $\xi \in \mathbb{R}^n$ it holds that

$$\widehat{\mathcal{T}_r w}(\xi) = \frac{1}{(2\pi)^{\frac{n}{2}}} \int_{\mathbb{R}^n} \mathcal{T}_r w(x) e^{-ix \cdot \xi} \, \mathrm{d}x$$

$$\begin{split} &= \frac{1}{(2\pi)^{\frac{n}{2}}} \int_{\mathbb{R}^n} \int_0^1 \int_{B_1} \left(w(x+rsy) \cdot y \right) \frac{y}{|y|} F_r(r|y|) \, \mathrm{d}y \, \mathrm{d}s \, e^{-ix \cdot \xi} \, \mathrm{d}x \\ &= \frac{1}{(2\pi)^{\frac{n}{2}}} \int_{\mathbb{R}^n} \int_0^1 \int_{B_1} \left(w(z) \cdot y \right) \frac{y}{|y|} F_r(r|y|) \, \mathrm{d}y \, \mathrm{d}s \, e^{-i(z-rsy) \cdot \xi} \, \mathrm{d}z \\ &= \frac{1}{(2\pi)^{\frac{n}{2}}} \int_0^1 \int_{B_1} \left(\int_{\mathbb{R}^n} w(z) e^{-iz \cdot \xi} \, \mathrm{d}z \right) \cdot y \frac{y}{|y|} F_r(r|y|) e^{irsy \cdot \xi} \, \mathrm{d}y \, \mathrm{d}s \\ &= \Phi_r(\xi) \hat{w}(\xi), \end{split}$$

where

$$\Phi_r(\xi) := \int_0^1 \int_{B_1} \frac{yy^T}{|y|} F_r(r|y|) e^{irsy \cdot \xi} \, \mathrm{d}y \, \mathrm{d}s.$$
(3.2.17)

Combining (3.2.17) with the Plancherel theorem and using our convention that w vanishes outside Ω , we can estimate as follows:

$$\begin{aligned} \|\mathcal{T}_{r}\|_{L((L^{2}(\Omega))^{n})} &= \sup_{\|w\|_{(L^{2}(\Omega))^{n}} = 1} \|\mathcal{T}_{r}w\|_{(L^{2}(\Omega))^{n}} \\ &\leq \sup_{\|w\|_{(L^{2}(\Omega))^{n}} = 1} \|\widehat{\mathcal{T}_{r}w}\|_{(L^{2}(\mathbb{R}^{n}))^{n}} \\ &\leq \|\Phi_{r}\|_{(L^{\infty}(\mathbb{R}^{n}))^{n}} \sup_{\|w\|_{(L^{2}(\Omega))^{n}} = 1} \|\widehat{w}\|_{(L^{2}(\mathbb{R}^{n}))^{n}} \\ &= \|\Phi_{r}\|_{(L^{\infty}(\mathbb{R}^{n}))^{n}} \sup_{\|w\|_{(L^{2}(\Omega))^{n}} = 1} \|w\|_{(L^{2}(\Omega))^{n}} \\ &= \|\Phi_{r}\|_{(L^{\infty}(\mathbb{R}^{n}))^{n}}. \end{aligned}$$
(3.2.18)

Further, consider an arbitrary orthogonal matrix $O \in \mathbb{R}^{n \times n}$ and $\xi \in \mathbb{R}^n$. With a variable transformation using the properties of orthogonal matrices from *Lemma 2.1.3* we observe that

$$\begin{split} \Phi_{r}(O\xi) &= \int_{0}^{1} \int_{B_{1}} \frac{yy^{T}}{|y|} F_{r}(r|y|) e^{irsy^{T}O\xi} \, \mathrm{d}y \, \mathrm{d}s \\ &= O \int_{0}^{1} \int_{B_{1}} \frac{O^{T}y(O^{T}y)^{T}}{|O^{T}y|} F_{r}(r|O^{T}y|) e^{irs(O^{T}y)^{T}\xi} \, \mathrm{d}y \, \mathrm{d}s \, O^{T} \\ &= O \Phi_{r}(\xi) O^{T}. \end{split}$$
(3.2.19)

Consequently, we construct an orthogonal matrix O out of an orthonormal basis containing $\frac{\xi}{|\xi|}$ in order for $O\xi = |\xi|e_1$ to hold and obtain that

$$|\Phi_r(\xi)|_2 = |\Phi_r(|\xi|e_1)|_2$$
 for all $\xi \in \mathbb{R}^n$. (3.2.20)

Since

$$\Phi_r(|\xi|e_1) = \int_0^1 \int_{B_1} \frac{yy^T}{|y|} F_r(r|y|) e^{irs|\xi|y_1} \,\mathrm{d}y \,\mathrm{d}s$$
(3.2.21)

is a diagonal matrix, its spectral norm is given by the spectral radius (see *Definition 2.1.1*). Estimating the right-hand side of (3.2.21) we then conclude that

$$|\Phi_r(|\xi|e_1)|_2 \leqslant \int_{B_1} \frac{y_1^2}{|y|} F_r(r|y|) \,\mathrm{d}y = \frac{1}{n} \int_{B_1} |y| F_r(r|y|) \,\mathrm{d}y \xrightarrow[r \to 0]{} 1 \qquad \text{for all } \xi \in \mathbb{R}^n \quad (3.2.22)$$

due to $F_0(0) = n + 1$ and (3.2.4). Combining (3.2.18), (3.2.20), and (3.2.22) we arrive at

$$\limsup_{r \to 0} \|\mathcal{T}_r\|_{L((L^2(\Omega))^n)} \le 1.$$
(3.2.23)

Finally, the pointwise convergence (3.2.12) and the Banach-Steinhaus theorem imply that

$$\liminf_{r \to 0} \|\mathcal{T}_r\|_{L((L^2(\Omega))^n)} \ge 1,$$

concluding the proof.

A similar result holds for S_r :

Lemma 3.2.7. (Operator S_r) Let $r \in (0, r_0]$. Then:

(i) S_r is a well-defined continuous linear operator in $(L^p(\Omega))^n$ for all $p \in [1, \infty]$. The corresponding operator norm satisfies

$$\|\mathcal{S}_r\|_{L((L^p(\Omega))^n)} \leqslant n. \tag{3.2.24}$$

(ii) Let $p, p^* \in [1, \infty]$ be such that $p^* = \frac{p}{p-1}$. For all $w_1 \in (L^p(\Omega))^n$ and $w_2 \in (L^{p^*}(\Omega))^n$ it holds:

$$\int_{\Omega} (\mathcal{S}_r w_1(x) \cdot w_2(x)) \, dx = \int_{\Omega} (w_1(x) \cdot \mathcal{S}_r w_2(x)) \, dx.$$

(iii) Let $p \in [1, \infty)$. For all $w \in (L^p(\Omega))^n$ it holds that

$$\mathcal{S}_r w \xrightarrow[r \to 0]{\to} \mathcal{S}_0 w = w \qquad in \ (L^p(\Omega))^n.$$

(iv) For p = 2 it holds that

$$\|\mathcal{S}_r\|_{L((L^2(\Omega))^n)} \xrightarrow[r \to 0]{} 1.$$

Proof. The proof almost repeats that of Lemma 3.2.5. Therefore, we only check (3.2.24) and omit further details. Let $p \in [1, \infty)$ and $p^* = \frac{p}{p-1}$. Using Hölder's inequality, Fubini's theorem, and our convention that w vanishes outside Ω we deduce for all $w \in (L^p(\Omega))^n$ that

$$\begin{split} \|\mathcal{S}_{r}w\|_{(L^{p}(\Omega))^{n}}^{p} =& n^{p} \int_{\Omega} \left| \int_{0}^{1} \int_{S_{1}} (w(x+rsy) \cdot y)y \,\mathrm{d}\sigma(y) \,\mathrm{d}s \right|^{p} \,\mathrm{d}x \\ \leqslant & n^{p} \int_{\Omega} \int_{0}^{1} \int_{S_{1}} \int_{S_{1}} |w(x+rsy)|^{p} \,\mathrm{d}\sigma(y) \,\mathrm{d}s \,\mathrm{d}x \\ =& n^{p} \int_{0}^{1} \int_{S_{1}} \int_{\Omega} |w(x+rsy)|^{p} \,\mathrm{d}x \,\mathrm{d}\sigma(y) \,\mathrm{d}s, \\ \leqslant & n^{p} \int_{0}^{1} \int_{S_{1}} \int_{\Omega} |w(z)|^{p} \,\mathrm{d}z \,\mathrm{d}\sigma(y) \,\mathrm{d}s \\ =& n^{p} \|w\|_{(L^{p}(\Omega))^{n}}^{p} , \end{split}$$

which means that

$$\|\mathcal{S}_r\|_{L((L^p(\Omega))^n)} \leqslant n. \tag{3.2.25}$$

The proof in the case $p = \infty$ follows the same steps, or, alternatively, one passes to the limit as $p \to \infty$ in (3.2.25).

Remark 3.2.8. The constants in (3.2.10) for any $n \ge 1$ and in (3.2.24) for $n \ge 2$ are not necessarily optimal. For $p \ne 2$ it remains open whether or not

$$\begin{split} & \liminf_{r \to 0} \|\mathcal{T}_r\|_{L((L^p(\Omega))^n)} = 1, \\ & \liminf_{r \to 0} \|\mathcal{S}_r\|_{L((L^p(\Omega))^n)} = 1. \end{split}$$

The answer may depend upon Ω and p.

3.3 Well-posedness for a class of evolution equations involving T_r or S_r

In this section we establish the existence and uniqueness of solutions to a certain class of single evolution equations involving \mathcal{T}_r or \mathcal{S}_r . This result is an important ingredient for our analysis of nonlocal systems in *Section 3.4*. Thus, we consider the following initial boundary value problem:

$$\partial_t c_r = \nabla \cdot (a_1 \nabla c_r - a_2 G_{\varepsilon} (\mathcal{R}_r(a_3 \nabla c_r))) + f \qquad \text{in } \Omega \times (0, T), \tag{3.3.1a}$$

$$(a_1 \nabla c_r - a_2 G_{\varepsilon}(\mathcal{R}_r(a_3 \nabla c_r))) \cdot \nu = 0 \qquad \text{on } \partial\Omega \times (0, T), \qquad (3.3.1b)$$

$$c_r(\cdot, 0) = c_0 \qquad \qquad \text{in } \Omega \qquad (3.3.1c)$$

for $T \in (0, \infty)$. Here

$$\mathcal{R}_r \in \{\mathcal{T}_r, \mathcal{S}_r\},\$$

and for $\varepsilon \ge 0$ we set

$$G_{\varepsilon} : \mathbb{R}^n \to \mathbb{R}^n, \quad x \mapsto \frac{x}{1 + \varepsilon |x|}.$$
 (3.3.2)

The following lemma shows that G_{ε} is globally Lipschitz.

Lemma 3.3.1. The function G_{ε} is globally Lipschitz continuous with Lipschitz constant 1 for $\varepsilon \ge 0$.

Proof. Let $i \in \{1, ..., n\}$. For fixed $x_j \in \mathbb{R}$, $j \in \{1, ..., n\} \setminus \{i\}$ we set $f_i(x_i) := \frac{x_i}{1+\varepsilon|x|} = G_i(x)$. We can estimate the derivative

$$|\partial_{x_i} f_i| = \frac{|1 + \varepsilon |x| - \varepsilon \frac{x_i^2}{|x|}|}{(1 + \varepsilon |x|)^2} \leqslant \frac{1}{1 + \varepsilon |x|} \leqslant 1.$$

The one-dimensional mean value theorem implies that for fixed $x, y \in \mathbb{R}^n$ it holds that

$$|G_i(x) - G_i(y)| = |f_i(x_i) - f_i(y_i)| \le \|\partial_{x_i} f_i\|_{L^{\infty}(\mathbb{R})} |x_i - y_i| \le |x_i - y_i|.$$

Consequently,

$$|G(x) - G(y)|^2 = \sum_{i=1}^n |f_i(x_i) - f_i(y_i)|^2 \leq \sum_{i=1}^n |x_i - y_i|^2 = |x - y|^2.$$

Remark 3.3.2. Observe that for $\varepsilon = 0$ equation (3.3.1a) is linear, whereas for $\varepsilon > 0$ the nonlocal part of the flux is a priori bounded. The latter helps us to construct nonnegative solutions in *Section 3.4.*

We make the following assumptions:

$$a_1, a_2, a_3 \in L^{\infty}(\Omega \times (0, T)),$$
 (3.3.3)

$$a_1 > 0 \text{ and } a_1^{-1} \in L^{\infty}(\Omega \times (0,T)),$$
 (3.3.4)

$$\left\| a_1^{-\frac{1}{2}} a_2 \right\|_{L^{\infty}(\Omega \times (0,T))} \left\| a_1^{-\frac{1}{2}} a_3 \right\|_{L^{\infty}(\Omega \times (0,T))} \left\| \mathcal{R}_r \right\|_{L((L^2(\Omega))^n)} < 1,$$

$$f \in L^2(0, T; (H^1(\Omega))^*)$$

$$(3.3.5)$$

$$f \in L^{2}(0,T;(H^{1}(\Omega))^{*}),$$
(3.3.6)

$$c_0 \in L^2(\Omega). \tag{3.3.7}$$

To shorten the notation, we introduce a pair of constants

$$\alpha_{r} := \|a_{1}^{-1}\|_{L^{\infty}(\Omega \times (0,T))}^{-1} \left(1 - \|a_{1}^{-\frac{1}{2}}a_{2}\|_{L^{\infty}(\Omega \times (0,T))} \|a_{1}^{-\frac{1}{2}}a_{3}\|_{L^{\infty}(\Omega \times (0,T))} \|\mathcal{R}_{r}\|_{L((L^{2}(\Omega))^{n})}\right),$$

$$M_{r} := \|a_{1}\|_{L^{\infty}(\Omega \times (0,T))} + \|a_{2}\|_{L^{\infty}(\Omega \times (0,T))} \|a_{3}\|_{L^{\infty}(\Omega \times (0,T))} \|\mathcal{R}_{r}\|_{L((L^{2}(\Omega))^{n})}.$$

$$(3.3.8)$$

Due to assumptions (3.3.3)–(3.3.5) it is clear that

$$0 < \alpha_r, M_r < \infty.$$

We introduce a family of operators

$$\begin{split} \langle \mathcal{M}(t,u),\varphi\rangle_{(H^{1}(\Omega))^{*},H^{1}(\Omega)} &:= \int_{\Omega} a_{1}(\cdot,t)\nabla u \cdot \nabla \varphi \,\mathrm{d}x - \int_{\Omega} a_{2}(\cdot,t)G_{\varepsilon}(\mathcal{R}_{r}(a_{3}(\cdot,t)\nabla u)) \cdot \nabla \varphi \,\mathrm{d}x, \\ \langle \mathcal{M}(u),\varphi\rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}),L^{2}(0,T;H^{1}(\Omega))} &:= \int_{0}^{T} \langle \mathcal{M}(t,u),\varphi(t)\rangle_{(H^{1}(\Omega))^{*},H^{1}(\Omega)} \,\mathrm{d}t. \end{split}$$

Lemma 3.3.3. Let (3.3.3)-(3.3.5) be satisfied. Then:

(i) For a.e. $t \in [0,T]$ the operator

$$\mathcal{M}(t,\cdot): H^1(\Omega) \to (H^1(\Omega))^*$$

is well-defined, monotone, hemicontinuous, and satisfies the bounds

$$\langle \mathcal{M}(t,u), u \rangle_{(H^1(\Omega))^*, H^1(\Omega)} \ge \alpha_r ||\nabla u||^2_{(L^2(\Omega))^n},$$
(3.3.9)

$$\|\mathcal{M}(t,u)\|_{(H^{1}(\Omega))^{*}} \leq M_{r} \|\nabla u\|_{(L^{2}(\Omega))^{n}}$$
(3.3.10)

for all $u \in H^1(\Omega)$. Moreover, for all $u \in H^1(\Omega)$ the function $\mathcal{M}(\cdot, u) : [0, T] \to (H^1(\Omega))^*$ is measurable.

(ii) The operator

$$\mathcal{M}: L^2(0,T;H^1(\Omega)) \to L^2(0,T;(H^1(\Omega))^*)$$

is well-defined, monotone, hemicontinuous, and satisfies the bounds

$$\begin{split} \langle \mathcal{M}(u), u \rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}), L^{2}(0,T;H^{1}(\Omega))} &\geqslant \alpha_{r} ||\nabla u||_{L^{2}(0,T;(L^{2}(\Omega))^{n})}^{2}, \\ ||\mathcal{M}(u)||_{L^{2}(0,T;(H^{1}(\Omega))^{*})} &\leqslant M_{r} ||\nabla u||_{L^{2}(0,T;(L^{2}(\Omega))^{n})} \end{split}$$

for all $u \in L^2(0,T; H^1(\Omega))$.

Proof. The assumptions on the coefficients a_i together with the Lipschitz continuity of G_{ε} readily imply that for a.e. $t \in [0, T]$ the operator $\mathcal{M}(t, \cdot)$ is well-defined and satisfies (3.3.10). Moreover, due to (3.3.3) and G_{ε} Lipschitz, it is also clear that $\mathcal{M}(\cdot, u) : [0, T] \to (H^1(\Omega))^*$ is measurable on [0, T] for all $u \in H^1(\Omega)$, whereas for a.e. t the mapping $\lambda \mapsto \langle \mathcal{M}(t, u + \lambda v), w \rangle_{(H^1(\Omega))^*, H^1(\Omega)}$ is continuous on \mathbb{R} for all $u, v, w \in H^1(\Omega)$, the latter meaning that $\mathcal{M}(t, \cdot)$ is hemicontinuous. Using Hölder's inequality, the facts that G_{ε} is Lipschitz with Lipschitz constant 1 and $G_{\varepsilon}(0) = 0$, the assumptions on the a_i 's, and the properties of \mathcal{R}_r , we compute that

$$\begin{split} \langle \mathcal{M}(t,u) - \mathcal{M}(t,v), u - v \rangle_{(H^{1}(\Omega))^{*}, H^{1}(\Omega)} \\ &= \int_{\Omega} \nabla(u-v) \cdot a_{1}(\cdot,t) \nabla(u-v) \, dx \\ &- \int_{\Omega} \left(G_{\varepsilon}(\mathcal{R}_{r}(a_{3}(\cdot,t) \nabla u)) - G_{\varepsilon}(\mathcal{R}_{r}(a_{3}(\cdot,t) \nabla v))) \cdot a_{2}(\cdot,t) \nabla(u-v) \, dx \right) \\ &\geq \left\| a_{1}^{\frac{1}{2}} \nabla(u-v) \right\|_{(L^{2}(\Omega))^{n}}^{2} \\ &- \int_{\Omega} \left| \mathcal{R}_{r} \left(a_{1}^{-\frac{1}{2}} a_{3}(\cdot,t) \left(a_{1}^{\frac{1}{2}} \nabla(u-v) \right) \right) \right| \left\| a_{1}^{-\frac{1}{2}} a_{2}(\cdot,t) \left(a_{1}^{\frac{1}{2}} \nabla(u-v) \right) \right| \, dx \\ &\geq \left(1 - \left\| a_{1}^{-\frac{1}{2}} a_{2} \right\|_{L^{\infty}(\Omega \times (0,T))} \left\| a_{1}^{-\frac{1}{2}} a_{3} \right\|_{L^{\infty}(\Omega \times (0,T))} \left\| \mathcal{R}_{r} \right\|_{L((L^{2}(\Omega))^{n})} \right) \left\| a_{1}^{\frac{1}{2}} \nabla(u-v) \right\|_{(L^{2}(\Omega))^{n}}^{2} \\ &\geqslant \alpha_{r} \left\| \nabla(u-v) \right\|_{(L^{2}(\Omega))^{n}}^{2} \geqslant 0 \end{split}$$
(3.3.11)

for $u, v \in H^1(\Omega)$, which proves monotonicity. Further, taking v = 0 in (3.3.11) and using $\mathcal{M}(t,0) = 0$ yields (3.3.9). Part *(i)* is thus proved. A proof of *(ii)* can be done similarly; we omit further details.

Using the properties of the averaging operators proved in *Subsection 3.2.1* we can define weak solutions to (3.3.1) in a manner very similar to that for the classical, purely local case (i.e., when $a_2 \equiv 0$):

Definition 3.3.4. Let (3.3.3)-(3.3.7) hold. We call the function $c_r : \overline{\Omega} \times [0,T] \to \mathbb{R}$ a weak solution of (3.3.1) if:

(i) $c_r \in L^2(0,T; H^1(\Omega)) \cap C([0,T]; L^2(\Omega)), \ \partial_t c_r \in L^2(0,T; (H^1(\Omega))^*);$

(ii) c_r satisfies (3.3.1a)-(3.3.1b) in the following sense: for all $\varphi \in H^1(\Omega)$ and a.e. $t \in (0,T)$

$$\langle \partial_t c_r, \varphi \rangle_{(H^1(\Omega))^*, H^1(\Omega)} = -\int_{\Omega} a_1 \nabla c_r \cdot \nabla \varphi \, dx + \int_{\Omega} a_2 G_{\varepsilon} (\mathcal{R}_r(a_3 \nabla c_r)) \cdot \nabla \varphi \, dx + \langle f, \varphi \rangle_{(H^1(\Omega))^*, H^1(\Omega)}; \quad (3.3.12)$$

(*iii*) $c_r(\cdot, 0) = c_0$ in $L^2(\Omega)$.

Using standard theory one readily proves the following existence result:

Lemma 3.3.5. Let (3.3.3)-(3.3.7) hold. Then there exists a unique weak solution to (3.3.1) in terms of Definition 3.3.4. The solution satisfies the following estimates:

$$\|c_r\|_{C([0,T];L^2(\Omega))}^2 + \alpha_r \|\nabla c_r\|_{L^2(0,T;(L^2(\Omega))^n)}^2 \leq C_3(\alpha_r, T) \left(\|c_0\|_{L^2(\Omega)}^2 + \|f\|_{L^2(0,T;(H^1(\Omega))^*)}^2\right),$$
(3.3.13)

$$\|\partial_t c_r\|_{L^2(0,T;(H^1(\Omega))^*)}^2 \leqslant C_4(\alpha_r, M_r, T) \left(\|c_0\|_{L^2(\Omega)}^2 + \|f\|_{L^2(0,T;(H^1(\Omega))^*)}^2 \right).$$
(3.3.14)

Proof. The existence of a unique weak solution to (3.3.1) is a direct consequence of Lemma 3.3.3(i) and the standard theory of evolution equations with monotone operators, see Theorem A.1.17. It remains to check the bounds (3.3.13) and (3.3.14). Taking $\varphi := c_r$ in the weak formulation (3.3.12) and using Lemma A.3.8, (3.3.9), and the Young inequality, we obtain that

$$\begin{split} \frac{1}{2} \frac{d}{dt} \|c_r\|_{L^2(\Omega)}^2 &\leqslant -\alpha_r \|\nabla c_r\|_{(L^2(\Omega))^n}^2 + \|c_r\|_{H^1(\Omega)} \|f\|_{(H^1(\Omega))} * \\ &= -\alpha_r \|c_r\|_{H^1(\Omega)}^2 + \alpha_r \|c_r\|_{L^2(\Omega)}^2 + \|c_r\|_{H^1(\Omega)} \|f\|_{(H^1(\Omega))} * \\ &\leqslant -\frac{1}{2} \alpha_r \|c_r\|_{H^1(\Omega)}^2 + \alpha_r \|c_r\|_{L^2(\Omega)}^2 + \frac{1}{2} \alpha_r^{-1} \|f\|_{(H^1(\Omega))}^2 *, \end{split}$$

which yields (3.3.13) due to Gronwall's inequality. Finally, using (3.3.10), we obtain from the weak formulation (3.3.12) that

$$\|\partial_t c_r\|_{(H^1(\Omega))^*} \leq M_r \|\nabla c_r\|_{(L^2(\Omega))^n} + \|f\|_{(H^1(\Omega))^*}$$

and consequently,

$$\|\partial_t c_r\|_{L^2(0,T;(H^1(\Omega))^*)}^2 \leq 2M_r^2 \|\nabla c_r\|_{L^2(0,T;(L^2(\Omega))^n)}^2 + 2\|f\|_{L^2(0,T;(H^1(\Omega))^*)}^2$$

Together with (3.3.13) this implies (3.3.14).

3.4 Nonlocal models involving averaging operators \mathcal{T}_r and \mathcal{S}_r

In this section we study the following model IBVP:

$$\partial_t c_r = \nabla \cdot \left(D_c(c_r, v_r) \nabla c_r - c_r \chi(c_r, v_r) \mathcal{R}_r(\nabla g(c_r, v_r)) \right) + f_c(c_r, v_r) \text{ in } \Omega \times (0, \infty), \qquad (3.4.1a)$$

$$\partial_t v_r = D_v \Delta v_r + f_v(c_r, v_r) \qquad \text{in } \Omega \times (0, \infty), \qquad (3.4.1b)$$

$$D_c(c_r, v_r)\partial_{\nu}c_r - c_r\chi(c_r, v_r)\mathcal{R}_r(\nabla g(c_r, v_r)) \cdot \nu = D_v\partial_{\nu}v_r = 0 \qquad \text{on } \partial\Omega \times (0, \infty), \quad (3.4.1c)$$

$$c_r(\cdot, 0) = c_0, \ v_r(\cdot, 0) = v_0$$
 in Ω . (3.4.1d)

Here, as in the previous section, \mathcal{R}_r stands for any of the two averaging operators:

$$\mathcal{R}_r \in \{\mathcal{T}_r, \mathcal{S}_r\}.$$

We assume that the diffusion coefficient D_v is either a positive number, or it is zero.

Equations (3.4.1a)-(3.4.1b) are closely related to (3.1.1) and (3.1.4) in Section 3.1, the difference being that the terms involving the adhesion velocity/nonlocal gradient are now replaced by those including the averaging operators $\mathcal{T}_r/\mathcal{S}_r$ from Section 3.2. Our motivation for introducing this change is twofold. First of all, due to (3.2.5) and (3.2.8) it affects the points in the boundary layer $\Omega \setminus \Omega_r$, at the most. On the other hand, Example 3.2.3 indicates that including, e.g., \mathcal{A}_r can lead to limits with unexpected blow-ups on the boundary of Ω .

System (3.4.1) is a nonlocal version of the hapto-/chemotaxis system

$$\partial_t c = \nabla \cdot (D_c(c, v) \nabla c - c\chi(c, v) \nabla g(c, v)) + f_c(c, v) \qquad \text{in } \Omega \times (0, \infty), \tag{3.4.2a}$$

$$\partial_t v = D_v \Delta v + f_v(c, v) \qquad \text{in } \Omega \times (0, \infty), \qquad (3.4.2b)$$

$$D_c(c,v)\partial_\nu c_r - c\chi(c,v)\partial_\nu g(c,v) = D_v\partial_\nu v = 0 \qquad \text{on } \partial\Omega \times (0,\infty), \qquad (3.4.2c)$$

$$c(\cdot, 0) = c_0, \ v(\cdot, 0) = v_0$$
 in Ω . (3.4.2d)

In this case, the actual diffusion and haptotactic sensitivity coefficients are

$$\begin{split} \tilde{D}_c(c,v) &= D_c(c,v) - c\chi(c,v)\partial_c g(c,v),\\ \tilde{\chi}(c,v) &= \chi(c,v)\partial_v g(c,v), \end{split}$$

so that in the classical formulation (3.4.2a) takes the form

$$\partial_t c = \nabla \cdot \left(\widetilde{D}_c(c, v) \nabla c - c \widetilde{\chi}(c, v) \nabla v \right) + f_c(c, v) \quad \text{in } \Omega \times (0, \infty).$$

The main goal of this section is to establish, under suitable assumptions on the system coefficients which are introduced in *Subsection 3.4.1*, a rigorous convergence as $r \to 0$ of solutions of the nonlocal model family (3.4.1) to those of the local model (3.4.2), see *Theorem 3.4.8*. This is accomplished in the final *Subsection 3.4.4*. Since we are dealing here with a new type of non-local system, we establish for (3.4.1) the existence of nonnegative solutions in *Subsections 3.4.2* and *3.4.3*.

3.4.1 Problem setting and main result of the section

We begin with several general assumptions about the coefficients of system (3.4.1).

Assumptions 3.4.1. Let $D_v \in \mathbb{R}^+_0$, $D_c, \chi \in C_b(\mathbb{R}^+_0 \times \mathbb{R}^+_0)$, and $g, f_c, f_v \in C^1(\mathbb{R}^+_0 \times \mathbb{R}^+_0)$ satisfy

$$C_{5} \leq D_{c} \leq C_{6} \quad in \ \mathbb{R}_{0}^{+} \times \mathbb{R}_{0}^{+} \quad for \ some \ C_{5}, C_{6} > 0,$$

$$\nabla_{(c,v)}g, \ \nabla_{(c,v)}f_{v} \in (L^{\infty}(\mathbb{R}_{0}^{+} \times \mathbb{R}_{0}^{+}))^{2},$$

$$f_{c}(0, \cdot) \equiv 0,$$

$$f_{v}(\cdot, 0) \equiv 0.$$

Assume that the coefficients satisfy the following bounds:

$$C_7 := \sup_{c \ v \ge 0} c |\chi(c, v)| < \infty, \tag{3.4.3}$$

$$C_8 := \sup_{c,v \ge 0} |\partial_c g(c,v)| < \infty.$$

$$(3.4.4)$$

Further, we assume that the initial values satisfy

$$0 \leq c_0 \in L^2(\Omega),$$

$$0 \leq v_0 \in H^1(\Omega).$$
(3.4.5)

Remark 3.4.2. If $D_v > 0$, then assumption (3.4.5) can be replaced by a weaker one, such as

$$v_0 \in L^2(\Omega).$$

We keep (3.4.5) in order to simplify the exposition.

In addition, we will later choose one of the following assumptions on f_c and the nonlocal operator:

Assumptions 3.4.3. (Further assumptions on f_c) One of the following conditions holds:

$$\nabla_{(c,v)} f_c \in \left(L^{\infty} (\mathbb{R}_0^+ \times \mathbb{R}_0^+) \right)^2$$

(b) there exists $s \ge 0$ such that

$$|f_c(c,v)| \leq C_9(1+|c|^s) \qquad in \ \mathbb{R}_0^+ \times \mathbb{R}_0^+ \qquad for \ some \ C_9 \geq 0, \tag{3.4.6}$$

$$cf_c(c,v) \leq C_{10} - C_{11}c^{s+1}$$
 in $\mathbb{R}_0^+ \times \mathbb{R}_0^+$ for some $C_{10} \geq 0, \ C_{11} > 0.$ (3.4.7)

Assumptions 3.4.4 (Assumptions on \mathcal{R}_r). One of the following holds:

(a) for a given fixed $r \in (0, r_0]$

$$C_{12}(\|\mathcal{R}_r\|) := 1 - \frac{C_7 C_8}{C_5} \|\mathcal{R}_r\|_{L((L^2(\Omega))^n)} > 0$$

(b)

$$C_{13} := \frac{C_7 C_8}{C_5} < 1. \tag{3.4.8}$$

Example 3.4.5. Let

$$\begin{split} D_v &= 0, \\ F_r(\rho) &:= (n+1)e^{-r\rho}, \\ g(c,v) &:= \frac{S_{cc}c + S_{cv}v}{1+c+v} & \text{for some constants } S_{cc}, S_{cv} > 0, \\ D_c(c,v) &:= \frac{1+c}{1+c+v}, \\ \chi(c,v) &:= \frac{b}{1+c+v}, & b > 0, \\ f_c(c,v) &:= \mu_c \frac{c}{1+c^2} (K_c - c - \eta_c v) & \text{for some constants } K_c, \eta_c, \mu_c > 0, \\ f_v(c,v) &:= \mu_v v (K_v - v) - \lambda_v v \frac{c}{1+c} & \text{for some constants } K_v, \lambda_v > 0, \ \mu_v \ge 0, \end{split}$$

and assume that

$$0 \leqslant v_0 \leqslant K_v.$$

Then, it holds a priori that

$$0 \leqslant v \leqslant K_v$$

for any v which solves (3.4.1b) due to the form of f_v . Therefore it suffices to consider the coefficient functions in $\mathbb{R}^+_0 \times [0, K_v]$.

For D_c it holds on $\mathbb{R}^+_0 \times [0, K_v]$ that

$$D_c(c,v) \geqslant \frac{1+c}{1+c+K_v} \geqslant \frac{1}{1+K_v} =: C_5$$

and

$$D_c(c,v) \leqslant 1 =: C_6$$

Obviously, $f_c(0, \cdot) = f_v(\cdot, 0) = 0$. Moreover, $\nabla_{(c,v)}g$, $\nabla_{(c,v)}f_v \in (L^{\infty}(\mathbb{R}^+_0 \times \mathbb{R}^+_0))^2$, due to

$$\begin{split} C_8 &= \sup_{c,v \ge 0} |\partial_c g(c,v)| = \max_{0 \le v \le K_v} \max_{c \ge 0} \frac{|S_{cc}(1+v) - S_{cv}v|}{(1+c+v)^2} \\ &= \max\left\{S_{cc}, \left|\frac{S_{cc}}{1+K_v} - \frac{S_{cv}K_v}{(1+K_v)^2}\right|\right\},\\ \sup_{c,v \ge 0} |\partial_v g(c,v)| &= \max_{0 \le v \le K_v} \max_{c \ge 0} \frac{|S_{cv}(1+c) - S_{cc}c|}{(1+c+v)^2} \\ &= \max_{c \ge 0} \frac{|S_{cv}(1+c) - S_{cc}c|}{(1+c)^2} < \infty,\\ \sup_{c,v \ge 0} |\partial_c f_v(c,v)| &= \max_{0 \le v \le K_v} \max_{c \ge 0} \frac{\lambda_v v}{(1+c)^2} = \lambda_v K_v \end{split}$$

and

$$\sup_{c,v \ge 0} \left| \partial_v f_v(c,v) \right| = \max_{0 \le v \le K_v} \max_{c \ge 0} \left| \mu_v(K_v - 2v) - \lambda_v \frac{c}{1+c} \right| < \infty.$$

For $C_9:=\mu_c(K_c+1+\eta_cK_v)$, $C_{10}:=\mu_c(K_c+1)$ and $C_{11}:=\mu_c$ we can estimate on $\mathbb{R}_0^+\times\mathbb{R}_0^+$ that

$$|f_c(c,v)| \leq C_9,$$

$$cf_c(c,v) = \mu_c \frac{c^2}{1+c^2} \left(K_c - c - \eta_c v \right) \leq C_{10} - C_{11}c.$$

Further,

$$C_7 = \sup_{c,v \ge 0} c\chi(c,v) = \sup_{c,v \ge 0} \frac{bc}{1+c+v} = \sup_{c \ge 0} \frac{bc}{1+c} = b$$

holds.

Thus, Assumptions 3.1.1, 3.4.1, 3.4.3(b) and 3.4.4(b) are fulfilled if

$$(1+K_v)b \max\left\{S_{cc}, \left|\frac{S_{cc}}{1+K_v} - \frac{S_{cv}K_v}{(1+K_v)^2}\right|\right\} < 1.$$

This choice of coefficient functions can be used to describe a population of cancer cells which interact among themselves and with the surrounding extracellular matrix (ECM) tissue. Both interaction types are due to adhesion, whether to other cells (cell-cell adhesion) or to the matrix (cell-matrix adhesion). The interaction force $F_r(\rho)$ is taken to diminish with increasing interaction range ρ and/or of the sensing radius r: cells too far apart/out of reach hardly interact in a direct way. Function g(c, v) characterises effective interactions. Here the coefficients S_{cc} and S_{cv} represent cell-cell and cell-matrix adhesion strengths, respectively. Our choice of g accounts for some adhesiveness limitation imposed by high local cell and tissue densities. It is motivated by the fact that overcrowding may preclude further adhesive bonds, e.g., due to saturation of receptors. The diffusion coefficient $D_c(c, v)$ is chosen to be everywhere positive and increase with a growing population density, thus enhancing diffusivity under population pressure, but, further, limited by excessive cell-tissue interaction. The latter also applies to the choice of the sensitivity function χ . Indeed, there is evidence that tight packing of cells and ECM limits diffusivity and the advective effects of haptotaxis [106]. Thereby the constant b > 0 is assumed to be rather small. Finally, f_c and f_v describe growth of cells and tissue limited by concurrence for resources. Next, we introduce weak-strong solutions to our problem. The definition is as follows:

Definition 3.4.6. Let Assumptions 3.4.1 hold. Let $r \in [0, r_0]$. We call a pair of functions $(c_r, v_r) : \overline{\Omega} \times \mathbb{R}^+_0 \to \mathbb{R}^+_0 \times \mathbb{R}^+_0$ a global weak-strong solution of (3.4.1) if for all T > 0:

- $(i) \ c_r \in L^2(0,T; H^1(\Omega)) \cap C_w([0,T]; L^2(\Omega)), \ \partial_t c_r \in L^1(0,T; (W^1_{\infty}(\Omega))^*);$
- (*ii*) $v_r \in W_2^{2,1}(\Omega \times (0,T)) \cap C([0,T]; H^1(\Omega)) \text{ if } D_v > 0 \text{ and } v_r \in L^{\infty}(0,T; H^1(\Omega)) \cap C([0,T]; L^2(\Omega))$ with $\partial_t v_r \in L^2(\Omega \times (0,T)) \text{ if } D_v = 0;$
- (*iii*) $f_c(c_r, v_r) \in L^1(\Omega \times (0, T)), f_v(c_r, v_r) \in L^2(\Omega \times (0, T));$
- (iv) (c_r, v_r) satisfies (3.4.1) in the following weak-strong sense: for all $\varphi \in C^1(\overline{\Omega})$ and a.e. $t \in (0,T)$

$$\langle \partial_t c_r, \varphi \rangle_{(W^1_{\infty}(\Omega))^*, W^1_{\infty}(\Omega)} = -\int_{\Omega} \left(D_c(c_r, v_r) \nabla c_r - c_r \chi(c_r, v_r) \mathcal{R}_r(\nabla g(c_r, v_r)) \right) \cdot \nabla \varphi \, dx$$

$$+ \int_{\Omega} f_c(c_r, v_r) \varphi \, dx,$$
 (3.4.9a)

$$c_r(\cdot, 0) = c_0 \qquad in \ L^2(\Omega),$$
 (3.4.9b)

and

$$\partial_t v_r = D_v \Delta v_r + f_v(c_r, v_r)$$
 a.e. in $\Omega \times (0, T)$, (3.4.9c)

$$D_v \partial_\nu v_r = 0 \qquad a.e. \ on \ \partial\Omega \times (0,T), \qquad (3.4.9d)$$

$$v_r(\cdot, 0) = v_0$$
 in $H^1(\Omega)$. (3.4.9e)

Remark 3.4.7. Observe that for r = 0 we obtain a corresponding solution definition for the local system (3.4.2).

Our main result now reads:

Theorem 3.4.8. Let Assumptions 3.1.1, 3.4.1, 3.4.3(b) and 3.4.4(b) hold. Then, there exists a sequence $r_m \to 0$ as $m \to \infty$ and solutions (c_{r_m}, v_{r_m}) and (c, v) in terms of Definition 3.4.6 corresponding to $r = r_m$ and r = 0, respectively, s.t.

$$\begin{array}{ll} c_{r_m} \underset{m \to \infty}{\to} c & \quad in \; L^2(\Omega \times (0,T)), \\ v_{r_m} \underset{m \to \infty}{\to} v & \quad in \; L^2(\Omega \times (0,T)). \end{array}$$

This Theorem is proved in Subsection 3.4.4.

3.4.2 Global existence of solutions to (3.4.1): the case of f_c Lipschitz

In this subsection we address the existence of solutions to the nonlocal model (3.4.1) for the case when f_c satisfies Assumptions 3.4.3(a). The main result of the Subsection is as follows:

Theorem 3.4.9. Let Assumptions 3.1.1, 3.4.1, and 3.4.3(a) hold and let r satisfy Assumptions 3.4.4(a). Then there exists a global weak-strong solution with $\partial_t c_r \in L^2(0, T; (H^1(\Omega))^*)$ to (3.4.1) in terms of Definition 3.4.6 for $\varphi \in H^1(\Omega)$.

Since we aim at constructing nonnegative solutions, it turns out to be helpful to consider first the following family of approximating problems:

$$\partial_t c_{r\varepsilon} = \nabla \cdot \left(D_c(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon} - c_{r\varepsilon} \chi(c_{r\varepsilon}, v_{r\varepsilon}) \left(G_{\varepsilon}(\mathcal{R}_r(\partial_c g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon})) + G_{\varepsilon}(\mathcal{R}_r(\partial_v g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla v_{r\varepsilon})) \right) \right) + f_c(c_{r\varepsilon}, v_{r\varepsilon}) \quad \text{in } \Omega \times (0, \infty), \quad (3.4.10a)$$
$$\partial_t v_{r\varepsilon} = D_v \Delta v_{r\varepsilon} + f_v(c_{r\varepsilon}, v_{r\varepsilon}) \quad \text{in } \Omega \times (0, \infty), \quad (3.4.10b)$$

$$D_{c}(c_{r\varepsilon}, v_{r\varepsilon})\partial_{\nu}c_{r\varepsilon} - c_{r\varepsilon}\chi(c_{r\varepsilon}, v_{r\varepsilon})\Big(G_{\varepsilon}(\mathcal{R}_{r}(\partial_{c}g(c_{r\varepsilon}, v_{r\varepsilon})\nabla c_{r\varepsilon})) + G_{\varepsilon}(\mathcal{R}_{r}(\partial_{v}g(c_{r\varepsilon}, v_{r\varepsilon})\nabla v_{r\varepsilon}))\Big) \cdot \nu = D_{v}\partial_{\nu}v_{r\varepsilon} = 0 \quad \text{on } \partial\Omega \times (0, \infty), \quad (3.4.10c)$$

$$c_{r\varepsilon}(\cdot,0) = c_0, \ v_{r\varepsilon}(\cdot,0) = v_0 \qquad \text{in } \Omega, \qquad (3.4.10d)$$

where G_{ε} was defined in (3.3.2). In order to obtain existence for the original problem, i.e., for $\varepsilon = 0$, we first prove existence of nonnegative solutions for the cases when $\varepsilon, D_v > 0$. This corresponds to a chemotaxis problem with a nonlocal flux-limited drift. Weak-strong solutions to (3.4.10) are understood as in *Definition 3.4.6*, with the obvious modification of the weak formulation, which now reads:

$$\begin{split} \langle \partial_t c_{r\varepsilon}, \varphi \rangle_{(H^1(\Omega))^*, H^1(\Omega)} &= -\int_{\Omega} D_c(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon} \cdot \nabla \varphi \, \mathrm{d}x \\ &+ \int_{\Omega} c_{r\varepsilon} \chi(c_{r\varepsilon}, v_{r\varepsilon}) G_{\varepsilon}(\mathcal{R}_r(\partial_c g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon})) \cdot \nabla \varphi \, \mathrm{d}x \\ &+ \int_{\Omega} c_{r\varepsilon} \chi(c_{r\varepsilon}, v_{r\varepsilon}) G_{\varepsilon}(\mathcal{R}_r(\partial_v g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla v_{r\varepsilon})) \cdot \nabla \varphi + f_c(c_{r\varepsilon}, v_{r\varepsilon}) \varphi \, \mathrm{d}x. \end{split}$$

$$(3.4.11)$$

Lemma 3.4.10. Let the assumptions of Theorem 3.4.9 be satisfied. Assume further that

 $\varepsilon, D_v > 0.$

Then there exists a global weak-strong solution to (3.4.10) with

$$\partial_t c_{r\varepsilon} \in L^2(0,T;(H^1(\Omega))^*)$$

for all T > 0.

Proof. To begin with, we extend the coefficients: for c < 0 we set

$$(D_c, \chi)(c, v) := (D_c, \chi)(-c, v), \quad f_c(c, v) := -f_c(-c, v), \quad (3.4.12)$$

$$g(c,v) := 2g(0,v) - g(-c,v), \quad f_v(c,v) := 2f_v(0,v) - f_v(-c,v). \tag{3.4.13}$$

These coefficients still satisfy Assumptions 3.4.1, 3.4.3(a), and Assumptions 3.4.4(a) if we consider all suprema over $c \in \mathbb{R}$ instead of $c \in \mathbb{R}_0^+$.

Our approach to proving existence is based on a Schaefer fixed-point argument (see *Theo*rem A.2.3). In order to apply this theorem we first 'freeze' $c_{r\varepsilon}$ in the system coefficients of (3.4.10), replacing it by $\bar{c}_{r\varepsilon}$. Correspondingly, we obtain the following weak formulation in place of (3.4.11): For all $\varphi \in H^1(\Omega)$, T > 0 and a.e. $t \in (0, T)$

$$\langle \partial_t c_{r\varepsilon}, \varphi \rangle_{(H^1(\Omega))^*, H^1(\Omega)} = -\int_{\Omega} D_c(\overline{c}_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon} \cdot \nabla \varphi \, \mathrm{d}x$$

$$+ \int_{\Omega} \overline{c}_{r\varepsilon} \chi(\overline{c}_{r\varepsilon}, v_{r\varepsilon}) G_{\varepsilon} (\mathcal{R}_{r}(\partial_{c}g(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\nabla c_{r\varepsilon})) \cdot \nabla \varphi \, \mathrm{d}x \\ + \int_{\Omega} \overline{c}_{r\varepsilon} \chi(\overline{c}_{r\varepsilon}, v_{r\varepsilon}) G_{\varepsilon} (\mathcal{R}_{r}(\partial_{v}g(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\nabla v_{r\varepsilon})) \cdot \nabla \varphi + f_{c}(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\varphi \, \mathrm{d}x,$$

$$(3.4.14a)$$

$$c_{r\varepsilon}(0,\cdot) = c_0 \qquad \text{in } L^2(\Omega) \tag{3.4.14b}$$

and

$$\partial_t v_{r\varepsilon} = D_v \Delta v_{r\varepsilon} + f_v(\overline{c}_{r\varepsilon}, v_{r\varepsilon}) \qquad \text{a.e. in } \Omega \times (0; T), \tag{3.4.14c}$$

$$\partial_{\nu} v_{r\varepsilon} = 0 \qquad \text{a.e. on } \partial\Omega \times (0,T), \qquad (3.4.14d)$$
$$v_{r\varepsilon}(\cdot,0) = v_0 \qquad \text{in } H^1(\Omega). \qquad (3.4.14e)$$

$$v_{r\varepsilon}(\cdot, 0) = v_0 \qquad \qquad \text{in } H^1(\Omega). \tag{3.4.14e}$$

Let T > 0 and let $\overline{c}_{r\varepsilon} \in L^2(\Omega \times (0,T))$.

Step 1 (Existence of $v_{r\varepsilon}$ satisfying (3.4.14c)-(3.4.14e)). First, we set $f_v(c,v) := -f_v(c,-v)$ for v < 0. We want to perform a Banach fixed-point argument in $C([0,T]; L^2(\Omega))$. Therefore, we fix $\overline{v}_{r\varepsilon} \in C([0,T]; L^2(\Omega))$. Then, Theorem A.1.8 and Lemma A.3.8 imply the existence of a unique solution $v_{r\varepsilon} \in W_2^{2,1}(\Omega \times (0,T)) \cap C([0,T]; H^1(\Omega))$ satisfying

$$\begin{split} \partial_t v_{r\varepsilon} &= D_v \Delta v_{r\varepsilon} + f_v(\overline{c}_{r\varepsilon}, \overline{v}_{r\varepsilon}) & \text{ a.e. in } \Omega \times (0; T), \\ \partial_\nu v_{r\varepsilon} &= 0 & \text{ a.e. on } \partial\Omega \times (0, T), \\ v_{r\varepsilon}(\cdot, 0) &= v_0 & \text{ in } H^1(\Omega). \end{split}$$

Moreover, the map $\Psi: C([0,T]; L^2(\Omega)) \to C([0,T]; L^2(\Omega)), \overline{v}_{r\varepsilon} \mapsto v_{r\varepsilon}$ satisfies

$$\|\Psi(\overline{v}_1) - \Psi(\overline{v}_2)\|_{C([0,T];L^2(\Omega))} \leq C_{14}(T) \|f_v\|_{L^{\infty}(\mathbb{R}^+_0 \times \mathbb{R}^+_0)} \|\overline{v}_1 - \overline{v}_2\|_{C([0,T];L^2(\Omega))}$$

again due to Theorem A.1.8, Lemma A.3.8 and the Lipschitz continuity of f_v . Hence, for small enough T the map Ψ is a contraction and we conclude from Banach's fixed-point theorem (Theorem A.2.1) that the semilinear parabolic initial boundary value problem (3.4.14c)-(3.4.14c)possesses a unique strong solution $0 \leq v_{r\varepsilon} \in W_2^{2,1}(\Omega \times (0,T))$. The solution extends to a global solution as the choice of T only depends on fixed parameters. Multiplying (3.4.14c) by $(v_{r\varepsilon})_{-} := -\min\{v_{r\varepsilon}, 0\}$ and integrating over Ω implies together with the Lipschitz continuity of f_v and $f_v(\cdot, 0) \equiv 0$ that

$$\frac{1}{2}\frac{d}{dt}\|(v_{r\varepsilon})_{-}\|_{L^{2}(\Omega)}^{2} \leq \|\partial_{v}f_{v}\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}_{0}^{+})}\|(v_{r\varepsilon})_{-}\|_{L^{2}(\Omega)}^{2}.$$

Due to $v_0 \ge 0$ and Gronwall's inequality $v_{r\varepsilon} \ge 0$ follows. Analogously, we conclude that

$$\|v_{r\varepsilon}\|_{L^{\infty}(0,T;L^{2}(\Omega))} \leqslant C_{15}(T) \|v_{0}\|_{L^{2}(\Omega)}.$$

Combining this with the Lipschitz continuity of f_v , $f_v(\cdot, 0) \equiv 0$, Theorem A.1.8 and Lemma A.3.8 the estimate

$$\begin{aligned} \|v_{r\varepsilon}\|_{C([0,T];H^{1}(\Omega))}^{2} + \|v_{r\varepsilon}\|_{L^{2}(0,T;H^{2}(\Omega))}^{2} + \|\partial_{t}v_{r\varepsilon}\|_{L^{2}(\Omega\times(0,T))}^{2} \\ \leqslant C_{16}(T) \left(\|v_{0}\|_{H^{1}(\Omega)}^{2} + \|f_{v}(\bar{c}_{r\varepsilon},v_{r\varepsilon})\|_{L^{2}(\Omega\times(0,T))}^{2}\right) \\ \leqslant C_{16}(T) \left(\|v_{0}\|_{H^{1}(\Omega)}^{2} + \|\partial_{v}f_{v}\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}_{0}^{+})}^{2} \|v_{r\varepsilon}\|_{L^{2}(\Omega\times(0,T))}^{2}\right) \\ \leqslant C_{17}(T) \|v_{0}\|_{H^{1}(\Omega)}^{2} + \|\partial_{v}f_{v}\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}_{0}^{+})}^{2} \|v_{r\varepsilon}\|_{L^{2}(\Omega\times(0,T))}^{2} \right) \\ \leqslant C_{16}(T) \left(\|v_{0}\|_{H^{1}(\Omega)}^{2} + \|\partial_{v}f_{v}\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}_{0}^{+})}^{2} \|v_{r\varepsilon}\|_{L^{2}(\Omega\times(0,T))}^{2}\right) \\ \end{cases}$$

follows. Here and further in the proof we omit the dependence of constants upon D_v .

Step 2 (Existence of $c_{r\varepsilon}$ satisfying (3.4.14a) and (3.4.14b)). Set

$$a_{1} := D_{c}(\overline{c}_{r\varepsilon}, v_{r\varepsilon}), \qquad a_{2} := \overline{c}_{r\varepsilon}\chi(\overline{c}_{r\varepsilon}, v_{r\varepsilon}), \qquad a_{3} := \partial_{c}g(\overline{c}_{r\varepsilon}, v_{r\varepsilon}), \langle f, \varphi \rangle_{(H^{1}(\Omega))^{*}, H^{1}(\Omega)} := \int_{\Omega} \overline{c}_{r\varepsilon}\chi(\overline{c}_{r\varepsilon}, v_{r\varepsilon})G_{\varepsilon}(\mathcal{R}_{r}(\partial_{v}g(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\nabla v_{r\varepsilon})) \cdot \nabla \varphi + f_{c}(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\varphi \,\mathrm{d}x.$$

Due to our assumptions about D_c, χ, g , and f_c , these coefficients a_i and f satisfy the requirements of Lemma 3.3.5. Consequently, there exists a unique global weak solution $c_{r\epsilon}$ to problem (3.3.1) with these coefficients. We estimate for the corresponding constants α_r and M_r introduced in (3.3.8):

$$\begin{aligned} \alpha_{r} &= \left\| \frac{1}{D_{c}(\overline{c}_{r\varepsilon}, v_{r\varepsilon})} \right\|_{L^{\infty}(\Omega \times (0,T))}^{-1} \\ &\cdot \left(1 - \left\| \frac{\overline{c}_{r\varepsilon}\chi(\overline{c}_{r\varepsilon}, v_{r\varepsilon})}{D_{c}(\overline{c}_{r\varepsilon}, v_{r\varepsilon})^{\frac{1}{2}}} \right\|_{L^{\infty}(\Omega \times (0,T))} \left\| \frac{\partial_{c}g(\overline{c}_{r\varepsilon}, v_{r\varepsilon})}{D_{c}(\overline{c}_{r\varepsilon}, v_{r\varepsilon})^{\frac{1}{2}}} \right\|_{L^{\infty}(\Omega \times (0,T))} \left\| \mathcal{R}_{r} \right\|_{L((L^{2}(\Omega))^{n})} \right) \\ &\geqslant C_{5}C_{12}(r) =: C_{18}(r), \end{aligned}$$
(3.4.16)

$$M_{r} = \|D_{c}(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\|_{L^{\infty}(\Omega \times (0,T))}$$

+ $\|\overline{c}_{r\varepsilon}\chi(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\|_{L^{\infty}(\Omega \times (0,T))} \|\partial_{c}g(\overline{c}_{r\varepsilon}, v_{r\varepsilon})\|_{L^{\infty}(\Omega \times (0,T))} \|\mathcal{R}_{r}\|_{L((L^{2}(\Omega))^{n})}$
$$\leq C_{6} + C_{7}C_{8} \|\mathcal{R}_{r}\|_{L((L^{2}(\Omega))^{n})} =: C_{19}(r),$$
(3.4.17)

and, due to the Lipschitz continuity of G_{ε} and $G_{\varepsilon}(0) = 0$, the linearity of \mathcal{R}_r , the Lipschitz continuity of g and f_c and (3.4.15),

$$\begin{aligned} \|f\|_{L^{2}(0,T;(H^{1}(\Omega))^{*})} \\ \leqslant \|\bar{c}_{r\varepsilon}\chi(\bar{c}_{r\varepsilon},v_{r\varepsilon})G_{\varepsilon}(\mathcal{R}_{r}(\partial_{v}g(\bar{c}_{r\varepsilon},v_{r\varepsilon})\nabla v_{r\varepsilon}))\|_{L^{2}(0,T;(L^{2}(\Omega))^{n})} + \|f_{c}(\bar{c}_{r\varepsilon},v_{r\varepsilon})\|_{L^{2}(\Omega\times(0,T))} \\ \leqslant C_{7}\|\mathcal{R}_{r}\|_{L(L^{2}(\Omega))^{n})} \|\partial_{v}g\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}^{+}_{0})} \|\nabla v_{r\varepsilon}\|_{L^{2}(0,T;(L^{2}(\Omega))^{n})} + \|\partial_{c}f_{c}\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}^{+}_{0})} \|\bar{c}_{r\varepsilon}\|_{L^{2}(\Omega\times(0,T))} \\ \leqslant C_{20}(r,T) \left(1 + \|\bar{c}_{r\varepsilon}\|_{L^{2}(\Omega\times(0,T))}\right). \end{aligned}$$

$$(3.4.18)$$

Combining (3.3.13)-(3.3.14) and (3.4.16)-(3.4.18), we obtain the following bounds for $c_{r\epsilon}$:

$$\|c_{r\varepsilon}\|_{C([0,T];L^{2}(\Omega))}^{2} + \alpha_{r} \|\nabla c_{r\varepsilon}\|_{L^{2}(0,T;(L^{2}(\Omega))^{n})}^{2} \leqslant C_{21}(r,T) \left(1 + \|\bar{c}_{r\varepsilon}\|_{L^{2}(\Omega\times(0,T))}^{2}\right), \qquad (3.4.19)$$

$$\|\partial_t c_{r\varepsilon}\|_{L^2(0,T;(H^1(\Omega))^*)}^2 \leqslant C_{22}(r,T) \left(1 + \|\bar{c}_{r\varepsilon}\|_{L^2(\Omega \times (0,T))}^2\right).$$
(3.4.20)

Step 3 (Fixed-point argument.). Now consider the mapping

$$\Phi: \overline{c}_{r\varepsilon} \mapsto c_{r\varepsilon}.$$

Thanks to (3.4.19) and (3.4.20), Φ is well-defined in $L^2(\Omega \times (0,T))$ and

$$\Phi: L^2(\Omega \times (0,T)) \to \{ u \in L^2(0,T; H^1(\Omega)) : \partial_t u \in L^2(0,T; (H^1(\Omega))^*) \}$$

maps bounded sets on bounded sets. (3.4.21)

Due to the Lions-Aubin lemma (Lemma A.3.9), (3.4.21) implies that

$$\Phi: L^2(\Omega \times (0,T)) \to L^2(\Omega \times (0,T)) \text{ maps bounded sets on precompact sets.}$$
(3.4.22)

Next, we verify that Φ is closed in $L^2(\Omega \times (0,T))$. Consider a sequence

$$(\overline{c}_{r\varepsilon m})_m \subset L^2(\Omega \times (0,T))$$

s.t.

$$\bar{c}_{r\varepsilon m} \xrightarrow[m \to \infty]{} \bar{c}_{r\varepsilon} \qquad \text{in } L^2(\Omega \times (0,T)), \tag{3.4.23}$$

$$\Phi(\bar{c}_{r\varepsilon m}) =: c_{r\varepsilon m} \xrightarrow[m \to \infty]{} c_{r\varepsilon} \qquad \text{in } L^2(\Omega \times (0,T)).$$
(3.4.24)

We need to check that

$$\Phi(\overline{c}_{r\varepsilon}) = c_{r\varepsilon}.$$

Due to (3.4.23) we have (by switching to a subsequence, if necessary) that

$$\overline{c}_{r\varepsilon m} \xrightarrow[m \to \infty]{} \overline{c}_{r\varepsilon} \qquad \text{a.e. in } \Omega \times (0, T).$$
(3.4.25)

Further, (3.4.21) and (3.4.24) together with the Banach-Alaoglu theorem and Lemma A.3.8 imply that

$$c_{r\varepsilon m} \underset{m \to \infty}{\longrightarrow} c_{r\varepsilon} \qquad \text{in } L^2(0,T;H^1(\Omega)), \qquad (3.4.26)$$

$$\partial_t c_{r\varepsilon m} \xrightarrow[m \to \infty]{} \partial_t c_{r\varepsilon} \qquad \text{in } L^2(0,T;(H^1(\Omega))^*)$$

$$(3.4.27)$$

and $c_{r\varepsilon} \in C([0,T]; L^2(\Omega))$. By the definition of Φ we have that $\overline{c}_{r\varepsilon m}$ and $c_{r\varepsilon m}$ satisfy: for all $\varphi \in H^1(\Omega)$ and a.e. $t \in (0,T)$

$$\begin{split} \langle \partial_t c_{r\varepsilon m}, \varphi \rangle_{(H^1(\Omega))^*, H^1(\Omega)} &= -\int_{\Omega} D_c(\overline{c}_{r\varepsilon m}, v_{r\varepsilon m}) \nabla c_{r\varepsilon m} \cdot \nabla \varphi \, \mathrm{d}x \\ &+ \int_{\Omega} \overline{c}_{r\varepsilon m} \chi(\overline{c}_{r\varepsilon m}, v_{r\varepsilon m}) G_{\varepsilon}(\mathcal{R}_r(\partial_c g(\overline{c}_{r\varepsilon m}, v_{r\varepsilon m}) \nabla c_{r\varepsilon m})) \cdot \nabla \varphi \, \mathrm{d}x \\ &+ \int_{\Omega} \overline{c}_{r\varepsilon m} \chi(\overline{c}_{r\varepsilon m}, v_{r\varepsilon m}) G_{\varepsilon}(\mathcal{R}_r(\partial_v g(\overline{c}_{r\varepsilon m}, v_{r\varepsilon m}) \nabla v_{r\varepsilon m})) \cdot \nabla \varphi \\ &+ f_c(\overline{c}_{r\varepsilon m}, v_{r\varepsilon m}) \varphi \, \mathrm{d}x, \end{split}$$
(3.4.28a)

$$c_{r\varepsilon m}(0,\cdot) = c_0 \qquad \text{in } L^2(\Omega) \tag{3.4.28b}$$

and

$$\partial_t v_{r\varepsilon m} = D_v \Delta v_{r\varepsilon m} + f_v(\bar{c}_{r\varepsilon m}, v_{r\varepsilon m}) \qquad \text{a.e. in } \Omega \times (0, T), \tag{3.4.28c}$$

$$\partial_{\nu} v_{r\varepsilon m} = 0$$
 a.e. on $\partial \Omega \times (0, T)$, (3.4.28d)

$$v_{r\varepsilon m}(\cdot, 0) = v_0 \qquad \qquad \text{in } H^1(\Omega). \tag{3.4.28e}$$

From (3.4.15) and (3.4.23) we conclude that the sequence $(v_{r\varepsilon m})_m$ is uniformly bounded in $L^2(0,T; H^2(\Omega))$ and $(\partial_t v_{r\varepsilon m})_m$ in $L^2(\Omega \times (0,T))$. Hence the Lions-Aubin lemma and the Banach-Alaoglu theorem imply that there exists $v_{r\varepsilon}$ s.t. (after switching to a subsequence, if necessary)

$$v_{r\varepsilon m} \xrightarrow[m \to \infty]{} v_{r\varepsilon} \quad \text{in } L^{2}(0,T;H^{2}(\Omega)),$$

$$\partial_{t}v_{r\varepsilon m} \xrightarrow[m \to \infty]{} \partial_{t}v_{r\varepsilon} \quad \text{in } L^{2}(\Omega \times (0,T)),$$

$$v_{r\varepsilon m} \xrightarrow[m \to \infty]{} v_{r\varepsilon} \quad \text{in } L^{2}(0,T;H^{1}(\Omega)) \text{ and a.e. in } \Omega \times (0,T), \qquad (3.4.29)$$

and due to the Lipschitz continuity of f_v , the fundamental lemma of calculus of variations and the embedding $H^1(\Omega) \hookrightarrow L^2(\partial\Omega)$ this $v_{r\varepsilon}$ satisfies equation (3.4.14c) for $\bar{c}_{r\varepsilon}$ as well as the initial and boundary conditions in the required sense. Moreover, $v_{r\varepsilon} \ge 0$ as pointwise limit of such functions.

Further, we conclude from combining (3.4.26) and (3.4.27) with Lemmas A.3.4 and A.3.8 that

$$c_{r\varepsilon m}(\cdot,t) \underset{m \to \infty}{\longrightarrow} c_{r\varepsilon}(\cdot,t) \quad \text{in } L^2(\Omega)$$

$$(3.4.30)$$

for all $t \in [0, T]$. In particular,

$$c_{r\varepsilon m}(\cdot,0)=c_0,$$

i.e., the initial condition is satisfied.

It remains now to pass to the limit in (3.4.28a). For this purpose we use the Minty-Browder method. To shorten the notation, we introduce for $u \in L^2(0,T; H^1(\Omega))$ and $m \in \mathbb{N} \cup \{\infty\}$

$$\begin{split} &\langle \mathcal{M}_{m}(u),\varphi\rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}),L^{2}(0,T;H^{1}(\Omega))} \\ &:= \int_{0}^{T}\int_{\Omega}D_{c}(\overline{c}_{r\varepsilon m},v_{r\varepsilon m})\nabla u\cdot\nabla\varphi - G_{\varepsilon}(\mathcal{R}_{r}(\partial_{c}g(\overline{c}_{r\varepsilon m},v_{r\varepsilon m})\nabla u))\overline{c}_{r\varepsilon m}\chi(\overline{c}_{r\varepsilon m},v_{r\varepsilon m})\cdot\nabla\varphi\,\mathrm{d}x\,\mathrm{d}t, \\ &\langle f_{m},\varphi\rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}),L^{2}(0,T;H^{1}(\Omega))} \\ &:= \int_{0}^{T}\int_{\Omega}\overline{c}_{r\varepsilon m}\chi(\overline{c}_{r\varepsilon m},v_{r\varepsilon m})G_{\varepsilon}(\mathcal{R}_{r}(\partial_{v}g(\overline{c}_{r\varepsilon m},v_{r\varepsilon m})\nabla v_{r\varepsilon m}))\cdot\nabla\varphi + f_{c}(\overline{c}_{r\varepsilon m},v_{r\varepsilon m})\varphi\,\mathrm{d}x\,\mathrm{d}t, \end{split}$$

where

$$\overline{c}_{r\varepsilon\infty} := \overline{c}_{r\varepsilon}, \qquad \overline{v}_{r\varepsilon\infty} := \overline{v}_{r\varepsilon}.$$

Due to Lemma 3.3.3(ii), (3.4.17) and (3.4.26) each operator \mathcal{M}_m is monotone, hemicontinuous, and satisfies

$$||\mathcal{M}_m(c_{r\varepsilon m})||_{L^2(0,T;(H^1(\Omega))^*)} \leq C_{19}(r) ||c_{r\varepsilon m}||_{L^2(0,T;H^1(\Omega))} \leq C_{23}(r).$$

Consequently, due to weak compactness there is $\eta \in L^2(0,T; (H^1(\Omega))^*)$ s.t.

$$\mathcal{M}_m(c_{r\varepsilon m}) \rightharpoonup \eta \text{ in } L^2(0,T;(H^1(\Omega))^*).$$
 (3.4.31)

Next, from (3.4.25) and (3.4.29), the boundedness and continuity of $(c, v) \mapsto c\chi(c, v)$, ∇g and ∇f_c over $\mathbb{R} \times \mathbb{R}^+_0$, the Lipschitz continuity of G_{ε} , the fact that $\mathcal{R}_r \in L((L^2(\Omega))^n)$ and the dominated convergence theorem we conclude that

$$f_m \xrightarrow[m \to \infty]{} f_\infty \quad \text{in } L^2(0,T;(H^1(\Omega))^*).$$
 (3.4.32)

A similar argument yields

$$\mathcal{M}_m(u) \xrightarrow[m \to \infty]{} \mathcal{M}_\infty(u), \quad \text{in } L^2(0, T; (H^1(\Omega))^*)$$

$$(3.4.33)$$

for all $u \in L^2(0,T; H^1(\Omega))$ so that due to (3.4.26) and compensated compactness (Lemma A.3.2)

$$\langle \mathcal{M}_m(u), c_{r\varepsilon m} \rangle_{L^2(0,T;(H^1(\Omega))^*), L^2(0,T;H^1(\Omega))} \xrightarrow[m \to \infty]{} \langle \mathcal{M}_\infty(u), c_{r\varepsilon} \rangle_{L^2(0,T;(H^1(\Omega))^*), L^2(0,T;H^1(\Omega))}$$

Observe that the weak formulation (3.4.28a) is equivalent to

$$\partial_t c_{r\varepsilon m} = -\mathcal{M}_m(c_{r\varepsilon m}) + f_m \qquad \text{in } (H^1(\Omega))^*$$

$$(3.4.34)$$

for a.e. $t \in (0,T)$. Combining (3.4.27), (3.4.31), and (3.4.32) we can pass to the weak limit in (3.4.34) and obtain

$$\partial_t c_{r\varepsilon} = -\eta + f_{\infty} \qquad \text{in } (H^1(\Omega))^*$$

$$(3.4.35)$$

for a.e. $t \in (0,T)$. For $u \in L^2(0,T; H^1(\Omega))$ and $m \in \mathbb{N}$ we have due to the monotonicity of \mathcal{M}_m that

$$X_m := \langle \mathcal{M}_m(c_{r\varepsilon m}) - \mathcal{M}_m(u), c_{r\varepsilon m} - u \rangle_{L^2(0,T;(H^1(\Omega))^*), L^2(0,T;H^1(\Omega))} \ge 0.$$
(3.4.36)

Moreover, setting $\varphi = c_{r\varepsilon m}$ in (3.4.28), integrating over (0,T) and using Lemma A.3.8 after inserting the obtained term into the definition of X_m , we conclude that

$$X_{m} = - \langle \mathcal{M}_{m}(c_{r\varepsilon m}), u \rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}), L^{2}(0,T;H^{1}(\Omega))} - \langle \mathcal{M}_{m}(u), c_{r\varepsilon m} - u \rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}), L^{2}(0,T;H^{1}(\Omega))} + \frac{1}{2} \|c_{0}\|_{L^{2}(\Omega)}^{2} - \frac{1}{2} \|c_{r\varepsilon m}(T)\|_{L^{2}(\Omega)}^{2} + \langle f_{m}, c_{r\varepsilon m} \rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}), L^{2}(0,T;H^{1}(\Omega))} .$$
(3.4.37)

From (3.4.30) for t = T we conclude $||c_{r\varepsilon}(T)||_{L^2(\Omega)} \leq \liminf_{m \to \infty} ||c_{r\varepsilon m}(T)||_{L^2(\Omega)}$. Combining this with (3.4.26), (3.4.31)–(3.4.33), (3.4.36), and (3.4.37) and compensated compactness, we obtain

$$\begin{split} 0 &\leqslant \limsup_{m \to \infty} X_m \leqslant - \langle \eta, u \rangle_{L^2(0,T;(H^1(\Omega))^*), L^2(0,T;H^1(\Omega))} \\ &- \langle \mathcal{M}_{\infty}(u), c_{r\varepsilon} - u \rangle_{L^2(0,T;(H^1(\Omega))^*), L^2(0,T;H^1(\Omega))} \\ &+ \frac{1}{2} \| c_0 \|_{L^2(\Omega)}^2 - \frac{1}{2} \| c_{r\varepsilon}(T) \|_{L^2(\Omega)}^2 + \langle f_{\infty}, c_{r\varepsilon} \rangle_{L^2(0,T;(H^1(\Omega))^*), L^2(0,T;H^1(\Omega))} \,. \end{split}$$

As $c_{r\varepsilon}$ satisfies (3.4.35), it follows again with Lemma A.3.8 from the last inequality that

$$0 \leq \langle \eta - \mathcal{M}_{\infty}(u), c_{r\varepsilon} - u \rangle_{L^{2}(0,T;(H^{1}(\Omega))^{*}), L^{2}(0,T;H^{1}(\Omega))}$$

holds for all $u \in L^2(0, T; H^1(\Omega))$.

Since \mathcal{M}_{∞} is monotone and hemicontinuous, *Lemma A.1.16* implies that it is maximal monotone. Consequently, $\eta = \mathcal{M}_{\infty}(c_{r\varepsilon})$.

Altogether, we conclude that $(c_{r\varepsilon}, v_{r\varepsilon})$ satisfies (3.4.14) for $\bar{c}_{r\varepsilon}$, meaning that $\Phi(\bar{c}_{r\varepsilon}) = c_{r\varepsilon}$ holds, i.e., Φ is a closed operator. Together with (3.4.22), this implies that

$$\Phi: L^2(0,T;L^2(\Omega)) \to L^2(0,T;L^2(\Omega)) \text{ is a compact operator.}$$
(3.4.38)

Since we aim to apply the Schaefer's fixed-point theorem (*Theorem A.2.3*), it is necessary to consider for $\lambda \in (0,1)$ the system which corresponds to $c_{r\varepsilon} = \lambda \Phi(c_{r\varepsilon})$. The corresponding weak-strong formulation reads:

$$\begin{split} &\langle \partial_t c_{r\varepsilon}, \varphi \rangle_{(H^1(\Omega))^*, H^1(\Omega)} \\ &= -\int_{\Omega} D_c(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon} \cdot \nabla \varphi \, \mathrm{d}x \\ &+ \lambda \int_{\Omega} c_{r\varepsilon} \chi(c_{r\varepsilon}, v_{r\varepsilon}) G_{\varepsilon} (\lambda^{-1} \mathcal{R}_r(\partial_c g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon})) \cdot \nabla \varphi \, \mathrm{d}x \\ &+ \lambda \int_{\Omega} c_{r\varepsilon} \chi(c_{r\varepsilon}, v_{r\varepsilon}) G_{\varepsilon} (\mathcal{R}_r(\partial_v g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla v_{r\varepsilon})) \cdot \nabla \varphi + f_c(c_{r\varepsilon}, v_{r\varepsilon}) \varphi \, \mathrm{d}x, \end{split}$$
(3.4.39a)

$$c_{r\varepsilon}(\cdot, 0) = \lambda c_0 \qquad \text{in } L^2(\Omega) \tag{3.4.39b}$$

and

$$\partial_t v_{r\varepsilon} = D_v \Delta v_{r\varepsilon} + f_v(c_{r\varepsilon}, v_{r\varepsilon}) \qquad \text{a.e. in } \Omega \times (0, T), \tag{3.4.39c}$$

$$\partial_{\nu} v_{r\varepsilon} = 0$$
 a.e. on $\partial \Omega \times (0, T)$, (3.4.39d)

$$v_{r\varepsilon}(\cdot, 0) = v_0 \qquad \text{in } H^1(\Omega). \qquad (3.4.39e)$$

Taking $\varphi := c_{r\varepsilon}$ in (3.4.39) and estimating the right-hand side by using Assumptions 3.4.1 and 3.4.4(a), the Hölder inequality, the fact that $|G_{\varepsilon}(x)| \leq |x|$, Lemmas 3.2.5(i), 3.2.7(i) and A.3.8 we obtain that

$$\begin{aligned} \frac{1}{2} \frac{d}{dt} ||c_{r\varepsilon}||^{2}_{L^{2}(\Omega)} \\ \leqslant &- C_{5} \left\| \nabla c_{r\varepsilon} \right\|^{2}_{(L^{2}(\Omega))^{n}} + \lambda C_{7} \left\| G_{\varepsilon} \left(\lambda^{-1} \mathcal{R}_{r} \left(\partial_{c} g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon} \right) \right) \right\|_{(L^{2}(\Omega))^{n}} \left\| \nabla c_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \\ &+ \lambda C_{7} \left\| G_{\varepsilon} \left(\mathcal{R}_{r} \left(\partial_{v} g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla v_{r\varepsilon} \right) \right) \right\|_{(L^{2}(\Omega))^{n}} \left\| |\nabla c_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \\ &+ \lambda \left\| \partial_{c} f_{c} \right\|_{L^{\infty}(\mathbb{R} \times \mathbb{R}_{0}^{+})} \left\| |c_{r\varepsilon} \right\|_{L^{2}(\Omega)}^{2} \\ \leqslant &- C_{5} \left\| \nabla c_{r\varepsilon} \right\|^{2}_{(L^{2}(\Omega))^{n}} + \lambda C_{7} \left\| \mathcal{R}_{r} \right\|_{L((L^{2}(\Omega))^{n})} \frac{1}{\lambda} C_{8} \left\| \nabla c_{r\varepsilon} \right\|^{2}_{(L^{2}(\Omega))^{n}} \\ &+ C_{7} \left| |\partial_{v} g \right| \right|_{L^{\infty}(\mathbb{R} \times \mathbb{R}_{0}^{+})} \left\| \mathcal{R}_{r} \right\|_{L((L^{2}(\Omega))^{n})} \left\| \nabla v_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \left\| |\nabla c_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \\ &+ \left\| \partial_{c} f_{c} \right\|_{L^{\infty}(\mathbb{R} \times \mathbb{R}_{0}^{+})} \left\| \nabla c_{r\varepsilon} \right\|^{2}_{(L^{2}(\Omega))^{n}} \\ &+ C_{7} \left| |\partial_{v} g \right| \right|_{L^{\infty}(\mathbb{R} \times \mathbb{R}_{0}^{+})} \left\| \mathcal{R}_{r} \right\|_{L((L^{2}(\Omega))^{n})} \left\| \nabla v_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \left\| \nabla c_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \\ &+ \left\| \partial_{c} f_{c} \right\|_{L^{\infty}(\mathbb{R} \times \mathbb{R}_{0}^{+})} \left\| \mathcal{R}_{r} \right\|_{L((L^{2}(\Omega))^{n})} \left\| \nabla v_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \left\| \nabla c_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \\ &+ \left\| \partial_{c} f_{c} \right\|_{L^{\infty}(\mathbb{R} \times \mathbb{R}_{0}^{+})} \left\| \mathcal{R}_{r} \right\|_{L((L^{2}(\Omega))^{n})} \left\| \nabla v_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \left\| \nabla c_{r\varepsilon} \right\|_{(L^{2}(\Omega))^{n}} \\ &+ \left\| \partial_{c} f_{c} \right\|_{L^{\infty}(\mathbb{R} \times \mathbb{R}_{0}^{+})} \left\| c_{r\varepsilon} \right\|_{L^{2}(\Omega)}^{2} \end{aligned}$$

holds for a.e. $t \in (0, T)$. Hence, Young's inequality and (3.4.15) imply that

$$\frac{1}{2}\frac{d}{dt}\|c_{r\varepsilon}(t)\|_{L^{2}(\Omega)}^{2} \leqslant C_{24}(\|\mathcal{R}_{r}\|, T)\left(1 + \|c_{r\varepsilon}(t)\|_{L^{2}(\Omega)}^{2}\right)$$
(3.4.41)

for a.e. $t \in (0, T]$ and we conclude from Gronwall's inequality that the set

$$\left\{c_{r\varepsilon} \in L^2(\Omega \times (0,T)) : c_{r\varepsilon} = \lambda \Phi(c_{r\varepsilon}) \text{ for } \lambda \in (0,1)\right\}$$

is uniformly bounded. Consequently, for all $\varepsilon \in (0, 1)$ the Schaefer's fixed-point theorem implies that Φ has a fixed point $c_{r\varepsilon}$, which together with the corresponding $v_{r\varepsilon}$, satisfies (3.4.10) in the weak-strong sense on the interval [0, T]. Since T > 0 was arbitrary, this extends to a global solution.

Step 4 (Nonnegativity of $c_{r\varepsilon}$). It remains to check that $c_{r\varepsilon}$ is nonnegative. Therefore, we take $\varphi := -(c_{r\varepsilon})_{-} = \min\{c_{r\varepsilon}, 0\}$ in (3.4.11) and use $f_c(0, \cdot) \equiv 0$, the boundedness of $G_{\varepsilon}, D_c, \partial_c f_c, \chi$, along with the Hölder and Young inequalities, which yields

$$\begin{split} \frac{1}{2} \frac{d}{dt} \| (c_{r\varepsilon})_{-} \|_{L^{2}(\Omega)}^{2} &= -\int_{\Omega} D_{c}(-(c_{r\varepsilon})_{-}, v_{r\varepsilon}) \left| \nabla(c_{r\varepsilon})_{-} \right|^{2} \, \mathrm{d}x \\ &+ \int_{\Omega} G_{\varepsilon}(\mathcal{R}_{r}(\partial_{c}g(c_{r\varepsilon}, v_{r\varepsilon})\nabla c_{r\varepsilon})) \cdot (c_{r\varepsilon})_{-} \chi(-(c_{r\varepsilon})_{-}, v_{r\varepsilon})\nabla(c_{r\varepsilon})_{-} \, \mathrm{d}x \\ &+ \int_{\Omega} G_{\varepsilon}(\mathcal{R}_{r}(\partial_{v}g(c_{r\varepsilon}, v_{r\varepsilon})\nabla v_{r\varepsilon})) \cdot (c_{r\varepsilon})_{-} \chi(-(c_{r\varepsilon})_{-}, v_{r\varepsilon})\nabla(c_{r\varepsilon})_{-} \, \mathrm{d}x \end{split}$$

$$+ \int_{\Omega} f_{c}(-(c_{r\varepsilon})_{-}, v_{r\varepsilon})(c_{r\varepsilon})_{-} dx$$

$$\leq -C_{5} \|\nabla(c_{r\varepsilon})_{-}\|_{(L^{2}(\Omega))^{n}}^{2} + \frac{2}{\varepsilon} \|\chi\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}^{+}_{0})} \|(c_{r\varepsilon})_{-}\|_{L^{2}(\Omega)} \|\nabla(c_{r\varepsilon})_{-}\|_{(L^{2}(\Omega))^{n}}$$

$$+ \|\partial_{c}f_{c}\|_{L^{\infty}(\mathbb{R}\times\mathbb{R}^{+}_{0})} \|(c_{r\varepsilon})_{-}\|_{L^{2}(\Omega)}^{2}$$

$$\leq C_{25} \|(c_{r\varepsilon})_{-}\|_{L^{2}(\Omega)}^{2}.$$

Since $c_{r\varepsilon}(0, \cdot) = c_0 \ge 0$, the Gronwall inequality implies that $(c_{r\varepsilon})_{-} \equiv 0$, i.e., that $c_{r\varepsilon} \ge 0$.

Remark 3.4.11. Observe that $c_{r\varepsilon}$ cannot be replaced by $-(c_{r\varepsilon})_{-}$ inside the nonlocal operator. This is why we introduced the flux-limitation.

Now we are ready to prove *Theorem 3.4.9*.

Proof of Theorem 3.4.9. Let T > 0.

Case $D_v > 0$: We start with the case

 $D_v > 0.$

Lemma 3.4.10 gives the existence of solutions $(c_{r\varepsilon}, v_{r\varepsilon})$ to (3.4.10). Setting $\varphi = c_{r\varepsilon}$ in (3.4.11), using the fact that $|G_{\varepsilon}(x)| \leq |x|$, we obtain similarly to (3.4.15), (3.4.40), and (3.4.41) and using Gronwall's inequality that

$$\|c_{r\varepsilon}\|_{L^{\infty}(0,T;L^{2}(\Omega))} + \|\nabla c_{r\varepsilon}\|_{L^{2}(0,T;(L^{2}(\Omega))^{n})} \leq C_{26}(\|\mathcal{R}_{r}\|,T),$$
(3.4.42)

and

. .

$$\|v_{r\varepsilon}\|_{C([0,T];H^1(\Omega))} + \|v_{r\varepsilon}\|_{L^2(0,T;H^2(\Omega))} + \|\partial_t v_{r\varepsilon}\|_{L^2(\Omega \times (0,T))} \leqslant C_{27}(\|\mathcal{R}_r\|, T)$$
(3.4.43)

where all constants are especially independent from ε . Consequently, for a.e. $t \in (0,T)$ and all $\varphi \in H^1(\Omega)$ we can estimate similarly to (3.4.40) and (3.4.41) that

$$\begin{aligned} &\langle \partial_t c_{r\varepsilon}, \varphi \rangle_{(H^1(\Omega))^*, H^1(\Omega)} \\ \leqslant & C_6 \| \nabla c_{r\varepsilon} \|_{(L^2(\Omega))^n} \| \nabla \varphi \|_{(L^2(\Omega))^n} + C_7 \| G_{\varepsilon} (\mathcal{R}_r (\partial_c g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla c_{r\varepsilon})) \|_{(L^2(\Omega))^n} \| \nabla \varphi \|_{(L^2(\Omega))^n} \\ &+ C_7 \| G_{\varepsilon} (\mathcal{R}_r (\partial_v g(c_{r\varepsilon}, v_{r\varepsilon}) \nabla v_{r\varepsilon})) \|_{(L^2(\Omega))^n} \| \nabla \varphi \|_{(L^2(\Omega))^n} + \| f_c(c_{r\varepsilon}, v_{r\varepsilon}) \|_{L^2(\Omega)} \| \varphi \|_{L^2(\Omega)} \\ \leqslant & C_{28} (\| \mathcal{R}_r \|, T) \left(1 + \| \nabla c_{r\varepsilon} \|_{(L^2(\Omega))^n} \right) \| \varphi \|_{H^1(\Omega)}. \end{aligned}$$

Integrating over (0, T), we conclude from (3.4.42) that

$$\int_{0}^{T} \|\partial_{t} c_{r\varepsilon}\|_{(H^{1}(\Omega))^{*}}^{2} dt \leq C_{29}(\|\mathcal{R}_{r}\|, T) \left(1 + \int_{0}^{T} \|\nabla c_{r\varepsilon}\|_{(L^{2}(\Omega))^{n}}^{2} dt\right) \leq C_{30}(\|\mathcal{R}_{r}\|, T). \quad (3.4.44)$$

Combining (3.4.42)–(3.4.44) and applying the Lions-Aubin lemma (*Lemma A.3.9*), the Banach-Alaoglu theorem and *Lemma A.3.8*, we conclude the existence of a pair of nonnegative functions $c_r \in L^2(0,T; H^1(\Omega)) \cap C([0,T]; L^2(\Omega))$ with $\partial_t c_r \in L^2(0,T; (H^1(\Omega))^*)$ and $v_r \in W_2^{2,1}(\Omega \times (0,T)) \cap C([0,T]; H^1(\Omega))$ such that for a sequence $\varepsilon_m \xrightarrow[m \to \infty]{} 0$ it holds that

$$c_{r\varepsilon_m} \xrightarrow[m \to \infty]{} c_r \text{ in } L^2(\Omega \times (0,T)) \text{ and a.e. in } \Omega \times (0,T),$$
 (3.4.45)

$$c_{r\varepsilon_m} \xrightarrow[m \to \infty]{} c_r \text{ in } L^2(0,T;H^1(\Omega)),$$

$$(3.4.46)$$

$$\partial_t c_{r\varepsilon_m} \xrightarrow[m \to \infty]{} \partial_t c_r \text{ in } L^2(0,T; (H^1(\Omega))^*),$$
(3.4.47)

$$v_{r\varepsilon_m} \xrightarrow[m \to \infty]{} v_r \text{ in } L^2(0,T; H^1(\Omega)) \text{ and a.e. in } \Omega \times (0,T),$$
 (3.4.48)

$$v_{r\varepsilon_m} \xrightarrow[m \to \infty]{} v_r \text{ in } L^2(0,T; H^2(\Omega)),$$
(3.4.49)

$$\partial_t v_{r\varepsilon_m} \xrightarrow[m \to \infty]{} \partial_t v_r \text{ in } L^2(\Omega \times (0,T)).$$
 (3.4.50)

Consider an arbitrary measurable set $E \subset \Omega \times (0,T)$. Using $G_{\varepsilon}(x) - x = -\varepsilon \frac{x|x|}{1+\varepsilon|x|}$, we can estimate for every component $i \in \{1, \ldots, n\}$:

$$\begin{split} & \left| \int_{E} \left(G_{\varepsilon_{m}} \left(\mathcal{R}_{r} (\partial_{c} g(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}}) \nabla c_{r\varepsilon_{m}} \right) \right) - \mathcal{R}_{r} (\partial_{c} g(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}}) \nabla c_{r\varepsilon_{m}}) \right)_{i} \, \mathrm{d}x \, \mathrm{d}t \\ \leq & \varepsilon_{m} \int_{0}^{T} \int_{\Omega} \left| \mathcal{R}_{r} (\partial_{c} g(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}}) \nabla c_{r\varepsilon_{m}} \right|^{2} \, \mathrm{d}x \, \mathrm{d}t \\ \leq & \varepsilon_{m} \| \mathcal{R}_{r} \|_{L((L^{2}(\Omega))^{n})}^{2} C_{8}^{2} \| \nabla c_{r\varepsilon_{m}} \|_{L^{2}(0,T;(L^{2}(\Omega))^{n})}^{2}, \end{split}$$

where the last term tends to 0 due to (3.4.42) as $\varepsilon_m \xrightarrow[m \to \infty]{} 0$. As the term inside the integral is moreover bounded in $L^2(\Omega \times (0,T))$ by a constant independent from ε_m , we conclude by using *Lemma A.3.3* that in $L^2(0,T;(L^2(\Omega))^n)$

$$G_{\varepsilon_m}(\mathcal{R}_r(\partial_c g(c_{r\varepsilon_m}, v_{r\varepsilon_m})\nabla c_{r\varepsilon_m})) - \mathcal{R}_r(\partial_c g(c_{r\varepsilon_m}, v_{r\varepsilon_m})\nabla c_{r\varepsilon_m}) \underset{m \to \infty}{\rightharpoonup} 0.$$
(3.4.51)

With the help of Lemma 3.2.5(ii) or 3.2.7(ii) we can rewrite that

$$\int_{0}^{T} \int_{\Omega} \mathcal{R}_{r} (\partial_{c} g(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}}) \nabla c_{r\varepsilon_{m}}) \cdot c_{r\varepsilon_{m}} \chi(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}}) \nabla \psi \, \mathrm{d}x \, \mathrm{d}t$$
$$= \int_{0}^{T} \int_{\Omega} \partial_{c} g(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}}) \nabla c_{r\varepsilon_{m}} \cdot \mathcal{R}_{r} (c_{r\varepsilon_{m}} \chi(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}}) \nabla \psi) \, \mathrm{d}x \, \mathrm{d}t.$$
(3.4.52)

The pointwise convergences from (3.4.45) and (3.4.48) together with the boundedness and continuity of $(c, v) \mapsto c\chi(c, v)$ and the dominated convergence theorem imply that

$$c_{r\varepsilon_m}\chi(c_{r\varepsilon_m}, v_{r\varepsilon_m})\nabla\psi \xrightarrow[m \to \infty]{} c_r\chi(c_r, v_r)\nabla\psi \text{ in } L^2(0, T; (L^2(\Omega))^n).$$
(3.4.53)

Then, we conclude from Lemma 3.2.5(i) or 3.2.7(i), (3.4.45), (3.4.48), the boundedness of $\partial_c g$ and the dominated convergence theorem that

$$\partial_{c}g(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}})\mathcal{R}_{r}(c_{r\varepsilon_{m}}\chi(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}})\nabla\psi) \xrightarrow[m \to \infty]{} \partial_{c}g(c_{r}, v_{r})\mathcal{R}_{r}(c_{r}\chi(c_{r}, v_{r})\nabla\psi)$$

in $L^{2}(0, T; (L^{2}(\Omega))^{n}).$ (3.4.54)

Now, combining (3.4.46) and (3.4.51)–(3.4.54) it follows due to compensated compactness (see Lemma A.3.2) that

$$\int_{0}^{T} \int_{\Omega} G_{\varepsilon_{m}}(\mathcal{R}_{r}(\partial_{c}g(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}})\nabla c_{r\varepsilon_{m}})) \cdot c_{r\varepsilon_{m}}\chi(c_{r\varepsilon_{m}}, v_{r\varepsilon_{m}})\nabla\psi \,\mathrm{d}x \,\mathrm{d}t$$
$$\xrightarrow{\rightarrow} \int_{0}^{T} \int_{\Omega} \mathcal{R}_{r}(\partial_{c}g(c_{r}, v_{r})\nabla c_{r}) \cdot c_{r}\chi(c_{r}, v_{r})\nabla\psi \,\mathrm{d}x \,\mathrm{d}t.$$

The convergence of the remaining terms in (3.4.9a) and the rest of (3.4.9) can be obtained from (3.4.45)-(3.4.50) in a way either completely analogous or very similar to the corresponding parts of the proof of *Lemma 3.4.10*. Consequently, (c_r, v_r) solves (3.4.1) in the required sense.

Case $D_v = 0$: In order to prove existence for the case

$$D_v = 0$$

consider a family of solutions (c_{rD_v}, v_{rD_v}) corresponding to $D_v \in (0, 1)$. The existence of such solutions was shown in the first part of this proof. Multiplying (3.4.9c) for v_{rD_v} by Δv_{rD_v} , integrating over Ω and using partial integration, we can estimate for a.e. $t \in (0, T)$ with Young's inequality due to the boundedness of ∇f_v that

$$\frac{1}{2} \frac{d}{dt} \| \nabla v_{rD_{v}} \|_{(L^{2}(\Omega))^{n}}^{2} + D_{v} \| \Delta v_{rD_{v}} \|_{L^{2}(\Omega)}^{2} \\
\leq \int_{\Omega} |\nabla f_{v}(c_{rD_{v}}, v_{rD_{v}})| |\nabla v_{rD_{v}}| \, \mathrm{d}x \\
\leq \| \partial_{c} f_{v} \|_{L^{\infty}(\mathbb{R}^{+}_{0} \times \mathbb{R}^{+}_{0})} \int_{\Omega} |\nabla c_{rD_{v}}| |\nabla v_{rD_{v}}| \, \mathrm{d}x + \| \partial_{v} f_{v} \|_{L^{\infty}(\mathbb{R}^{+}_{0} \times \mathbb{R}^{+}_{0})} \int_{\Omega} |\nabla v_{rD_{v}}|^{2} \, \mathrm{d}x \\
\leq C_{31} \left(\| \nabla c_{rD_{v}} \|_{(L^{2}(\Omega))^{n}}^{2} + \| \nabla v_{rD_{v}} \|_{(L^{2}(\Omega))^{n}}^{2} \right).$$
(3.4.55)

Estimating $\|c_{rD_v}\|^2_{L^2(\Omega)}$ similarly to (3.4.40) and adding $\frac{C_5C_{12}}{4C_{31}}\frac{d}{dt}\|\nabla v_{rD_v}\|^2_{(L^2(\Omega))^n}$, we conclude using (3.4.55) with Young's inequality that for a.e. $t \in (0,T)$ it holds that

$$\begin{aligned} &\frac{1}{2} \frac{d}{dt} \left(\|c_{rD_v}\|_{L^2(\Omega)}^2 + \frac{C_5 C_{12}(\|\mathcal{R}_r\|)}{2C_{31}} \|\nabla v_{rD_v}\|_{(L^2(\Omega))^n}^2 \right) \\ &+ \frac{C_5 C_{12}(\|\mathcal{R}_r\|)}{4} \|\nabla c_{rD_v}\|_{(L^2(\Omega))^n}^2 + D_v \frac{C_5 C_{12}(\|\mathcal{R}_r\|)}{2C_{31}} \|\Delta v_{rD_v}\|_{L^2(\Omega)}^2 \\ \leqslant \|\partial_c f_c\|_{L^{\infty}(\mathbb{R}^+_0 \times \mathbb{R}^+_0)} ||c_{rD_v}||_{L^2(\Omega)}^2 + C_{32}(\|\mathcal{R}_r\|) \|\nabla v_{rD_v}\|_{(L^2(\Omega))^n}^2. \end{aligned}$$

Then, Gronwall's inequality implies

$$|c_{rD_v}||_{L^{\infty}(0,T;L^2(\Omega))} \leqslant C_{33}(||\mathcal{R}_r||,T), \qquad (3.4.56)$$

$$\|\nabla c_{rD_v}\|_{L^2(0,T;(L^2(\Omega))^n)} \leqslant C_{33}(\|\mathcal{R}_r\|,T),$$
(3.4.57)

$$\|\nabla v_{rD_v}\|_{L^{\infty}(0,T;(L^2(\Omega))^n)} \leqslant C_{33}(\|\mathcal{R}_r\|,T),$$
(3.4.58)

$$D_v \|\Delta v_{rD_v}\|_{L^2(\Omega \times (0,T))} \leqslant C_{33}(\|\mathcal{R}_r\|, T)$$
(3.4.59)

for a constant $C_{33}(||\mathcal{R}_r||, T) > 0$ that is especially independent from D_v . Multiplying (3.4.9c) by v_{rD_v} we conclude with $f_v(\cdot, 0) = 0$ and the Lipschitz continuity of f_v using partial integration the estimate

$$\frac{1}{2}\frac{d}{dt}\|v_{rD_{v}}\|_{L^{2}(\Omega)}^{2} \leqslant \left|\int_{\Omega}f_{v}(c_{rD_{v}}, v_{rD_{v}})v_{rD_{v}}\,\mathrm{d}x\right| \leqslant \|\partial_{v}f_{v}\|_{L^{\infty}(\mathbb{R}^{+}_{0}\times\mathbb{R}^{+}_{0})}\|v_{rD_{v}}\|_{L^{2}(\Omega)}^{2}$$

for a.e. $t \in (0, T)$. Consequently, Gronwall's inequality implies that

$$\|v_{rD_v}\|_{L^{\infty}(0,T;L^2(\Omega))} \leqslant C_{34}(\|\mathcal{R}_r\|,T).$$
(3.4.60)

A uniform bound on $(\partial_t c_{rD_v})_{D_v}$ in $L^2(0,T; (H^1(\Omega))^*)$ follows combining (3.4.56)–(3.4.58) and estimating as above in (3.4.44). Moreover, we conclude with (3.4.59) and (3.4.60) and the Lipschitz continuity of f_v from (3.4.9c) that $(\partial_t v_{rD_v})_{D_v}$ is uniformly bounded in $L^2(\Omega \times (0,T))$. Combining this with (3.4.56) - (3.4.60) we conclude again from Lions-Aubin and Banach-Alaoglu (*Lemmas A.3.1* and *A.3.9*) and *Lemma A.3.8* that there are $c_{r0} \in L^2(0,T; H^1(\Omega)) \cap C([0,T]; L^2(\Omega))$ with $\partial_t c_{r0} \in L^2(0,T; (H^1(\Omega))^*)$ and $v_{r0} \in L^\infty(0,T; H^1(\Omega)) \cap C([0,T]; L^2(\Omega))$ with time derivative $\partial_t v_{r0} \in L^2(\Omega \times (0,T))$ s.t.

$$c_{r(D_v)_m} \xrightarrow[m \to \infty]{} c_{r0} \text{ in } L^2(\Omega \times (0,T)) \text{ and a.e. in } \Omega \times (0,T),$$
 (3.4.61)

$$c_{r(D_v)_m} \xrightarrow[m \to \infty]{} c_{r0} \text{ in } L^2(0,T; H^1(\Omega)),$$

$$\partial_t c_{r(D_v)_m} \xrightarrow[m \to \infty]{} \partial_t c_{r0} \text{ in } L^2(0,T;(H^1(\Omega))^*),$$

$$v_{r(D_v)_m} \xrightarrow[m \to \infty]{} v_{r0} \text{ in } C([0,T]; L^2(\Omega)) \text{ and a.e. in } \Omega \times (0,T),$$
 (3.4.62)

$$\nabla v_{r(D_v)_m} \stackrel{*}{\underset{m \to \infty}{\longrightarrow}} \nabla v_{r0} \text{ in } L^{\infty}(0,T;(L^2(\Omega))^n), \qquad (3.4.63)$$

$$\partial_t v_{r(D_v)_m} \xrightarrow[m \to \infty]{} \partial_t v_{r0} \text{ in } L^2(\Omega \times (0,T))$$

$$(3.4.64)$$

for a subsequence $(D_v)_m$. Observe that this time the gradients of c and v enter linearly into our equation, so that no strong convergence and no application of Lemma A.3.3 are required. From this we conclude with the dominated convergence theorem, compensated compactness (Lemma A.3.2), Lemma 3.2.5(i) and (ii) or 3.2.7(i) and (ii) similarly to above that

$$\int_{0}^{T} \int_{\Omega} \mathcal{R}_{r}(\nabla g(c_{r(D_{v})_{m}}, v_{r(D_{v})_{m}})) \cdot c_{r(D_{v})_{m}}\chi(c_{r(D_{v})_{m}}, v_{r(D_{v})_{m}})\nabla\psi \,\mathrm{d}x \,\mathrm{d}t$$

$$= \int_{0}^{T} \int_{\Omega} \nabla g(c_{r(D_{v})_{m}}, v_{r(D_{v})_{m}}) \cdot \mathcal{R}_{r}(c_{r(D_{v})_{m}}\chi(c_{r(D_{v})_{m}}, v_{r(D_{v})_{m}})\nabla\psi) \,\mathrm{d}x \,\mathrm{d}t$$

$$\xrightarrow{\rightarrow} \int_{0}^{T} \int_{\Omega} \nabla g(c_{r0}, v_{r0}) \cdot \mathcal{R}_{r}(c_{r0}\chi(c_{r0}, v_{r0})\nabla\psi) \,\mathrm{d}x \,\mathrm{d}t$$

$$= \int_{0}^{T} \int_{\Omega} \mathcal{R}_{r}(\nabla g(c_{r0}, v_{r0})) \cdot c_{r0}\chi(c_{r0}, v_{r0})\nabla\psi \,\mathrm{d}x \,\mathrm{d}t$$

holds for $\psi \in L^2(0, T; H^1(\Omega))$. The convergence of the remaining terms in the equation of $c_{r(D_v)_m}$ follow as in the proof of *Lemma 3.4.10*. Finally, we multiply (3.4.9c) by $\psi \in L^2(\Omega \times (0,T))$, integrate over $\Omega \times (0,T)$ and use partial integration to obtain

$$\begin{split} & \int_0^T \int_\Omega \partial_t v_{r(D_v)_m} \psi \, \mathrm{d}x \mathrm{d}t + (D_v)_m \int_0^T \int_\Omega \nabla v_{r(D_v)_m} \cdot \nabla \psi \, \mathrm{d}x \mathrm{d}t \\ & = \int_0^T \int_\Omega f_v(c_{r(D_v)_m}, v_{r(D_v)_m}) \psi \, \mathrm{d}x \mathrm{d}t. \end{split}$$

Then, we conclude from (3.4.58), (3.4.61), (3.4.62), and (3.4.64), the Lipschitz continuity of f_v and the dominated convergence theorem that

$$\int_0^T \int_\Omega \partial_t v_{r0} \psi \, \mathrm{d}x \, \mathrm{d}t = \int_0^T \int_\Omega f_v(c_{r0}, v_{r0}) \psi \, \mathrm{d}x \, \mathrm{d}t.$$

Hence, the fundamental lemma of calculus of variations that (c_{r0}, v_{r0}) solves (3.4.1) for $D_v = 0$ in the required sense.

3.4.3 Global existence of solutions to (3.4.1): the case of f_c dissipative

In this subsection we provide an extension of the existence theorem *Theorem 3.4.9* from *Subsection 3.4.2*.

Theorem 3.4.12. Let Assumptions 3.1.1, 3.4.1, and 3.4.3(b) hold and let r satisfy Assumptions 3.4.4(a). Set

$$q := \min\left\{2, \frac{s+1}{s}\right\}^3, \qquad q^* := \frac{q}{q-1} = \max\{2, s+1\}.$$
 (3.4.65)

³As usual, here and below the expression $\frac{s+1}{s}$ means infinity if s = 0.

Then there exists a global weak-strong solution to (3.4.1) in terms of Definition 3.4.6, with $\partial_t c_r \in L^q(0,T; (W^1_{q^*}(\Omega))^*)$ and $\varphi \in W^1_{q^*}(\Omega)$ satisfying for all T > 0 the estimates

$$||c_r||_{L^{\infty}(0,T;L^2(\Omega))} \leq C_{35}(||\mathcal{R}_r||_{L((L^2(\Omega))^n)}, T),$$

$$||\nabla c_r||_{L^2(0,T;(L^2(\Omega))^n)} \leq C_{35}(||\mathcal{R}_r||_{L((L^2(\Omega))^n)}, T),$$

$$(3.4.66)$$

$$(3.4.67)$$

$$\|\partial_t c_r\|_{L^q(0,T;(W^1_{a^*}(\Omega))^*)} \leqslant C_{35}(\|\mathcal{R}_r\|_{L((L^2(\Omega))^n)}, T),$$
(3.4.68)

 $||v_r||_{L^{\infty}(0,T;L^2(\Omega))} \leq C_{35}(||\mathcal{R}_r||_{L((L^2(\Omega))^n)},T),$ (3.4.69)

$$\|\nabla v_r\|_{L^{\infty}(0,T;(L^2(\Omega))^n)} \leqslant C_{35}(\|\mathcal{R}_r\|_{L((L^2(\Omega))^n)}, T),$$
(3.4.70)

$$\|\partial_t v_r\|_{L^2(\Omega \times (0,T))} \le C_{35}(\|\mathcal{R}_r\|_{L((L^2(\Omega))^n)}, T),$$
(3.4.71)

$$\|f_c(c_r, v_r)\|_{L^q(\Omega \times (0,T))} \leq C_{35}(\|\mathcal{R}_r\|_{L((L^2(\Omega))^n)}, T),$$
(3.4.72)

$$\|f_v(c_r, v_r)\|_{L^2(\Omega \times (0,T))} \le C_{35}(\|\mathcal{R}_r\|_{L((L^2(\Omega))^n)}, T).$$
(3.4.73)

Proof. Let T > 0. For $k \in \mathbb{N}$ set

$$f_{ck}(c,v) := f_c(c,v)\eta_k(c,v),$$

where η_k is a cut-off function:

$$\eta_k \in C_c^{\infty}(B_k^2) \quad \text{with} \quad \eta_k \equiv 1 \quad \text{in } B_{k-1}^2 \quad \text{and} \quad 0 \leq \eta_k \leq 1.$$
 (3.4.74)

Here, B_k^2 denotes the two-dimensional ball with radius k centered at the origin. The continuity of ∇f_c and (3.4.6) imply that f_{ck} has bounded derivatives. Hence, it is Lipschitz continuous due to the mean value theorem and Theorem 3.4.9 implies the existence of a solution (c_{rk}, v_{rk}) in terms of Definition 3.4.6 with $\partial_t c_{rk} \in L^2(0,T;(H^1(\Omega))^*)$ and $\varphi \in H^1(\Omega)$, which corresponds to $f_c = f_{ck}$. Our next aim is to prove that (c_{rk}, v_{rk}) satisfies the same bounds as in the statement of the theorem with some constant $C_{35}(\|\mathcal{R}_r\|_{L((L^2(\Omega))^n)}, T)$ which does not depend upon k. Set

$$C_{36}(\|\mathcal{R}_r\|) := \|\mathcal{R}_r\|_{L((L^2(\Omega))^n)}$$

Taking $\varphi := c_{rk}$ in (3.4.9a) written for c_{rk} and using Assumptions 3.4.1, 3.4.3(b), 3.4.4(a), the fact that $\mathcal{R}_r \in L((L^2(\Omega))^n)$ and the Hölder and Young inequalities, we compute

$$\begin{split} \frac{1}{2} \frac{d}{dt} \|c_{rk}\|_{L^{2}(\Omega)}^{2} &= \int_{\Omega} \left(-\left(D_{c}(c_{rk}, v_{rk}) \nabla c_{rk} - c_{rk} \chi(c_{rk}, v_{rk}) \mathcal{R}_{r}(\nabla g(c_{rk}, v_{rk})) \right) \cdot \nabla c_{rk} \\ &+ c_{rk} f_{ck}(c_{rk}, v_{rk}) \right) dx \\ &\leqslant - C_{5} \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}}^{2} + C_{7} \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} \|\mathcal{R}_{r}(\nabla g(c_{rk}, v_{rk}))\|_{(L^{2}(\Omega))^{n}} \\ &+ \int_{\Omega} (C_{10} - C_{11}c_{rk}^{1+s}) \eta_{k}(c_{rk}, v_{rk}) dx \\ &\leqslant - C_{5} \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}}^{2} + C_{7}C_{36}(\|\mathcal{R}_{r}\|) \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} \|\nabla g(c_{rk}, v_{rk})\|_{(L^{2}(\Omega))^{n}} \\ &+ C_{37} - C_{11} \int_{\Omega} c_{rk}^{1+s} \eta_{k}(c_{rk}, v_{rk}) dx \\ &\leqslant - C_{5} \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}}^{2} \\ &+ C_{7}C_{36}(\|\mathcal{R}_{r}\|) \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} \|\partial_{v}g(c_{rk}, v_{rk}) \nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} \\ &+ C_{7}C_{36}(\|\mathcal{R}_{r}\|) \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} \|\partial_{v}g(c_{rk}, v_{rk}) \nabla v_{rk}\|_{(L^{2}(\Omega))^{n}} + C_{37} \\ &- C_{11} \int_{\Omega} c_{rk}^{1+s} \eta_{k}(c_{rk}, v_{rk}) dx \end{split}$$

(3.4.67)

$$\leq -C_5 C_{12}(\|\mathcal{R}_r\|) \|\nabla c_{rk}\|_{(L^2(\Omega))^n}^2 + C_7 C_{36}(\|\mathcal{R}_r\|) \|\partial_v g\|_{L^{\infty}(\mathbb{R}_0^+ \times \mathbb{R}_0^+)} \|\nabla c_{rk}\|_{(L^2(\Omega))^n} \|\nabla v_{rk}\|_{(L^2(\Omega))^n} + C_{37} - C_{11} \int_{\Omega} c_{rk}^{1+s} \eta_k(c_{rk}, v_{rk}) dx \leq -\frac{C_5 C_{12}(\|\mathcal{R}_r\|)}{2} \|\nabla c_{rk}\|_{(L^2(\Omega))^n}^2 + C_{38}(\|\mathcal{R}_r\|) \|\nabla v_{rk}\|_{(L^2(\Omega))^n}^2 + C_{37} - C_{11} \int_{\Omega} c_{rk}^{1+s} \eta_k(c_{rk}, v_{rk}) dx.$$

$$(3.4.75)$$

Next, we estimate v_{rk} . If $D_v > 0$, then Theorem A.1.8 and Lemma A.3.8 yield as in the proof of Lemma 3.4.10 that

$$\|v_{rk}\|_{L^{\infty}(0,T;H^{1}(\Omega))}^{2} + \|v_{rk}\|_{L^{2}(0,T;H^{2}(\Omega))}^{2} + \|\partial_{t}v_{rk}\|_{L^{2}(\Omega\times(0,T))}^{2} \leq C_{39}(T)\|v_{0}\|_{H^{1}(\Omega)}^{2}.$$
(3.4.76)

Here and further in the proof we omit the dependence of constants upon D_v . If $D_v = 0$, then we get the ODE

$$\partial_t v_{rk} = f_v(c_{rk}, v_{rk}). \tag{3.4.77}$$

Hence, the assumptions on f_v and the solution components together with the chain rule imply that

$$\partial_t v_{rk} \in L^2(0, T; H^1(\Omega)).$$

Computing the gradient on both sides of (3.4.77), multiplying by ∇v_{rk} throughout, integrating over Ω , and using Assumptions 3.4.1 and the Young inequality, we obtain for a.e. $t \in (0,T)$ the estimate

$$\frac{1}{2} \frac{d}{dt} \|\nabla v_{rk}\|_{(L^{2}(\Omega))^{n}}^{2} = \int_{\Omega} \left(\partial_{v} f_{v}(c_{rk}, v_{rk}) |\nabla v_{rk}|^{2} + \partial_{c} f_{v}(c_{rk}, v_{rk}) \nabla c_{rk} \cdot \nabla v_{rk} \right) dx
\leq \|\partial_{v} f_{v}\|_{L^{\infty}(\mathbb{R}^{+}_{0} \times \mathbb{R}^{+}_{0})} \|\nabla v_{rk}\|_{(L^{2}(\Omega))^{n}}^{2}
+ \|\partial_{c} f_{v}\|_{L^{\infty}(\mathbb{R}^{+}_{0} \times \mathbb{R}^{+}_{0})} \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} \|\nabla v_{rk}\|_{(L^{2}(\Omega))^{n}}^{2}
\leq C_{40} \|\nabla v_{rk}\|_{(L^{2}(\Omega))^{n}}^{2} + C_{41} \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}}^{2}.$$
(3.4.78)

Proceeding as for estimate (3.4.41) if $D_v > 0$ and as in the second case of the proof of *Theorem 3.4.9* if $D_v = 0$ and using the Gronwall inequality yields that c_{rk} and v_{rk} satisfy estimates as (3.4.66), (3.4.67), and (3.4.69)–(3.4.71) for a constant independent from k as this is the case for all constants involved in (3.4.75), (3.4.76), and (3.4.78). Hence, the estimate

$$\int_{0}^{T} \int_{\Omega} c_{rk}^{1+s} \eta_{k}(c_{rk}, v_{rk}) \, \mathrm{d}x \, \mathrm{d}t \leqslant C_{42}(\|\mathcal{R}_{r}\|, T).$$
(3.4.79)

follows after integrating (3.4.75).

From (3.4.6) and (3.4.79), the embedding of Lebesgue spaces, and $\eta_k \in [0, 1]$ we conclude using Hölder's inequality if necessary that

$$\begin{split} \|f_{ck}(c_{rk}, v_{rk})\|_{L^{q}(\Omega \times (0,T))} &\leq C_{9} \|(1 + c_{rk}^{s}) \eta_{k}(c_{rk}, v_{rk})\|_{L^{q}(\Omega \times (0,T))} \\ &\leq C_{43}(T) + C_{44} \|c_{rk}^{s} \eta_{k}(c_{rk}, v_{rk})\||_{L^{\frac{s+1}{s}}(\Omega \times (0,T))} \\ &\leq C_{43}(T) + C_{44} \left(\int_{0}^{T} \int_{\Omega} c_{rk}^{1+s} \eta_{k}(c_{rk}, v_{rk}) \,\mathrm{d}x \,\mathrm{d}t\right)^{\frac{s}{s+1}} \leq C_{45}(\|\mathcal{R}_{r}\|, T). \end{split}$$

so that (3.4.72) holds for $f_{ck}(c_{rk}, v_{rk})$. An estimate as (3.4.73) for $f_v(c_{rk}, v_{rk})$ follows from the Lipschitz continuity of f_v , $f_v(\cdot, 0) = 0$ and the uniform in k bound on (v_{rk}) . Finally, combining Assumptions 3.4.1 with the fact that $\mathcal{R}_r \in L((L^2(\Omega))^n)$, the uniform in k bound on (∇v_{rk}) , the weak formulation (3.4.9a) for $\varphi \in W^1_{q^*}(\Omega)$, Hölder's and Young's inequality and the embedding of Lebesgue spaces yield

$$\begin{split} &|\langle \partial_{t}c_{rk}, \varphi \rangle_{(W_{q^{*}}^{1}(\Omega))^{*}, W_{q^{*}}^{1}(\Omega)}| \\ &\leq \left| \int_{\Omega} \left(D_{c}(c_{rk}, v_{rk}) \nabla c_{rk} - c_{rk} \chi(c_{rk}, v_{rk}) \mathcal{R}_{r}(\nabla g(c_{rk}, v_{rk})) \right) \cdot \nabla \varphi \, \mathrm{d}x \right| + \left| \int_{\Omega} f_{ck}(c_{rk}, v_{rk}) \varphi \, \mathrm{d}x \right| \\ &\leq C_{46}(\|\mathcal{R}_{r}\|) \left(\|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} + \|\nabla v_{rk}\|_{(L^{2}(\Omega))^{n}} \right) \|\nabla \varphi\|_{(L^{2}(\Omega))^{n}} + \|f_{ck}(c_{rk}, v_{rk})\|_{L^{q}(\Omega)} \|\varphi\|_{L^{q^{*}}(\Omega)} \\ &\leq C_{47}(\|\mathcal{R}_{r}\|, T) \left(1 + \|\nabla c_{rk}\|_{(L^{2}(\Omega))^{n}} + \|f_{ck}(c_{rk}, v_{rk})\|_{L^{q}(\Omega)} \right) \|\varphi\|_{W_{q^{*}}^{1}(\Omega)} \end{split}$$

Taking the supremum over $\|\varphi\|_{W_{q^*}^1(\Omega)} \leq 1$ and integrating the *q*th-potence over (0, T) we conclude that (3.4.68) holds for $\partial_t c_{rk}$ due to the uniform in *k* bounds on (∇c_{rk}) , and $(f_{ck}(c_{rk}, v_{rk}))$. Since (c_{rk}, v_{rk}) satisfy (3.4.66)-(3.4.73) uniformly in *k*, the Lions-Aubin lemma, Banach-Alaoglu theorem (*Lemmas A.3.1* and *A.3.9*) and *Lemma A.3.8* imply that there are $c_r \in L^2(0, T; H^1(\Omega))$ with $\partial_t c_r \in L^q(0, T; (W_{q^*}^1(\Omega))^*)$ and $v_r \in W_2^{2,1}(\Omega \times (0, T)) \cap C([0, T]; H^1(\Omega))$ if $D_v > 0$ or $v_r \in L^{\infty}(0, T; H^1(\Omega)) \cap C([0, T]; L^2(\Omega))$ with $\partial_t v_r \in L^2(\Omega \times (0, T))$ if $D_v = 0$ s.t. for a subsequence

$$c_{rk_m} \xrightarrow[m \to \infty]{} c_r \qquad \qquad \text{in } L^2(\Omega \times (0,T)) \text{ and a.e. in } \Omega \times (0,T), \qquad (3.4.80)$$

$$c_{rk_m} \stackrel{*}{\underset{m \to \infty}{\longrightarrow}} c_r \qquad \qquad \text{in } L^{\infty}(0,T;L^2(\Omega)), \qquad (3.4.81)$$

$$\nabla c_{rk_m} \underset{m \to \infty}{\longrightarrow} \nabla c_r \qquad \text{in } L^2(0, T; (L^2(\Omega))^n), \qquad (3.4.82)$$

$$\partial_t c_{rk_m} \xrightarrow{\sim} \partial_t c_r \qquad \text{in } L^q(0,T; (W^1_{q^*}(\Omega))^*), \qquad (3.4.83)$$

$$v_{rk_m} \xrightarrow{} v_r$$
 in $C([0,T]; L^2(\Omega))$ and a.e. in $\Omega \times (0,T)$, (3.4.84)

$$\nabla v_{rk_m} \stackrel{*}{\longrightarrow} \nabla v_r \qquad \text{in } L^{\infty}(0,T;(L^2(\Omega))^n), \qquad (3.4.85)$$

$$v_{rk_m} \xrightarrow[m \to \infty]{} \partial_t v_r \qquad \text{in } L^2(\Omega \times (0,T))$$

$$(3.4.86)$$

and additionally

 ∂_t

$$v_{rk_m} \xrightarrow[m \to \infty]{} v_r \text{ in } L^2(0,T; H^2(\Omega)).$$
 (3.4.87)

if $D_v > 0$. Then, due to (3.4.80) and (3.4.84), the continuity of f_c and f_v , the definition of η_k and the uniform in k bound on $(f_{ck}(c_{rk}, v_{rk}))$, the Lions lemma (*Lemma A.3.4*) implies

$$f_{ck_m}(c_{rk_m}, v_{rk_m}) \underset{m \to \infty}{\longrightarrow} f_c(c_r, v_r) \text{ in } L^q(\Omega \times (0, T)),$$
$$f_v(c_{rk_m}, v_{rk_m}) \underset{m \to \infty}{\longrightarrow} f_v(c_r, v_r) \text{ in } L^2(\Omega \times (0, T)).$$

Consequently, c_r and v_r satisfy (3.4.66)-(3.4.73) as (weak) limits of functions satisfying these inequalities and setting $X = L^2(\Omega)$ and $Y = \left(W_{q^*}^1(\Omega)\right)^*$ in Lemma A.3.5 we conclude that $c_r \in C_w([0,T]; L^2(\Omega))$ with $c_r(\cdot, 0) = c_0$ in $L^2(\Omega)$. Finally, we conclude similarly to the proof of Theorem 3.4.9 that (c_r, v_r) solve (3.4.1) in the required sense.

3.4.4 Limiting behaviour of the nonlocal model (3.4.1) as $r \to 0$

In this subsection we finally prove our main result concerning convergence for $r \to 0$.

Proof of Theorem 3.4.8. Due to (3.4.8) and Lemma 3.2.5(iv) or 3.2.7(iv), respectively, there exists a sequence $r_m \to 0$ as $m \to \infty$ such that

$$\sup_{m \in \mathbb{N}} \|\mathcal{R}_{r_m}\|_{L((L^2(\Omega))^n)} < \frac{1}{C_{13}}.$$
(3.4.88)

Since for each such r_m the Assumptions 3.4.4(a) are satisfied, Theorem 3.4.12 is applicable and yields the existence of solutions (c_{r_m}, v_{r_m}) which satisfy (3.4.66)-(3.4.73). Replacing $||\mathcal{R}_r||$ by C_{13} in $C_{35}(T, ||\mathcal{R}_r||_{L((L^2(\Omega))^n)})$ makes the constant in (3.4.66)-(3.4.73) independent of m. Using Lions-Aubin, Banach-Alaoglu (Lemmas A.3.1 and A.3.9) and Lemma A.3.5 we conclude (by possibly switching to a subsequence) that there are $c \in L^2(0,T; H^1(\Omega)) \cap C_w([0,T]; L^2(\Omega))$ with $\partial_t c \in L^q(0,T; (W^1_{q^*}(\Omega))^*)$ and $v \in W^{2,1}_2(\Omega \times (0,T)) \cap C([0,T]; H^1(\Omega))$ if $D_v > 0$ or $v \in L^{\infty}(0,T; H^1(\Omega)) \cap C([0,T]; L^2(\Omega))$ with $\partial_t v \in L^2(\Omega \times (0,T))$ if $D_v = 0$ s.t. (c_r) and (v_r) converge to c and v, respectively, in the sense of (3.4.80)-(3.4.87), i.e., especially

$$c_{r_m} \xrightarrow[m \to \infty]{} c, \quad v_{r_m} \xrightarrow[m \to \infty]{} v \quad \text{in } L^2(\Omega \times (0,T)) \text{ and a.e. in } \Omega \times (0,T), \quad (3.4.89)$$

$$\nabla c_{r_m} \underset{m \to \infty}{\rightharpoonup} \nabla c, \quad \nabla v_{r_m} \underset{m \to \infty}{\rightharpoonup} \nabla v \qquad \text{in } L^2(0, T; (L^2(\Omega))^n).$$
(3.4.90)

We conclude from (3.4.89), the continuity of χ and ∇g and the dominated convergence theorem that

$$c_{r_m}\chi(c_{r_m}, v_{r_m}) \xrightarrow[m \to \infty]{} c\chi(c, v) \qquad \qquad \text{in } L^2(\Omega \times (0, T)), \qquad (3.4.91)$$

$$\partial_c g(c_{r_m}, v_{r_m}) \xrightarrow[m \to \infty]{} \partial_c g(c, v), \quad \partial_v g(c_{r_m}, v_{r_m}) \xrightarrow[m \to \infty]{} \partial_v g(c, v) \qquad \text{in } L^2(\Omega \times (0, T)).$$
(3.4.92)

Observe that for any $\psi \in L^{\infty}(0,T; W^1_{\infty}(\Omega))$ the following estimate holds:

$$\int_{0}^{T} \int_{\Omega} \left| \partial_{c}g(c_{r_{m}}, v_{r_{m}}) \mathcal{R}_{r_{m}}(c_{r_{m}}\chi(c_{r_{m}}, v_{r_{m}})\nabla\psi) - \partial_{c}g(c, v)c\chi(c, v)\nabla\psi \right|^{2} dx dt$$

$$\leq 9 \left(\int_{0}^{T} \int_{\Omega} \left| (\partial_{c}g(c_{r_{m}}, v_{r_{m}}) - \partial_{c}g(c, v)) \mathcal{R}_{r_{m}}(c_{r_{m}}\chi(c_{r_{m}}, v_{r_{m}})\nabla\psi) \right|^{2} dx dt$$

$$+ \int_{0}^{T} \int_{\Omega} \left| \partial_{c}g(c, v) \mathcal{R}_{r_{m}}\left((c_{r_{m}}\chi(c_{r_{m}}, v_{r_{m}}) - c\chi(c, v))\nabla\psi \right) \right|^{2} dx dt$$

$$+ \int_{0}^{T} \int_{\Omega} \left| \partial_{c}g(c, v) \left(\mathcal{R}_{r_{m}}(c\chi(c, v)\nabla\psi) - c\chi(c, v)\nabla\psi \right) \right|^{2} dx dt \right). \qquad (3.4.93)$$

Now, using (3.4.88), (3.4.91), and (3.4.92) together with Lemma 3.2.5(i) and (iii) or Lemma 3.2.7(i) and (iii), respectively, we conclude that the right-hand side of (3.4.93) tends to zero, hence

$$\partial_c g(c_{r_m}, v_{r_m}) \mathcal{R}_{r_m}(c_{r_m} \chi(c_{r_m}, v_{r_m}) \nabla \psi) \xrightarrow[m \to \infty]{} \partial_c g(c, v) c \chi(c, v) \nabla \psi \qquad \text{in } L^2(0, T; (L^2(\Omega))^n).$$
(3.4.94)

An analogous convergence holds for the corresponding term involving $\partial_v g$. Finally, we obtain using Lemma 3.2.5(ii) or Lemma 3.2.7(ii), respectively, compensated compactness (Lemma A.3.2), (3.4.90) and (3.4.94) that

$$\int_0^T \int_\Omega c_{r_m} \chi(c_{r_m}, v_{r_m}) \mathcal{R}_{r_m} (\nabla g(c_{r_m}, v_{r_m})) \cdot \nabla \psi \, \mathrm{d}x \, \mathrm{d}t$$
$$= \int_0^T \int_\Omega \nabla g(c_{r_m}, v_{r_m}) \cdot \mathcal{R}_{r_m} (c_{r_m} \chi(c_{r_m}, v_{r_m}) \nabla \psi) \, \mathrm{d}x \, \mathrm{d}t$$

$$\underset{n \to \infty}{\to} \int_0^T \int_\Omega \nabla g(c, v) \cdot c\chi(c, v) \nabla \psi \, \mathrm{d}x \, \mathrm{d}t.$$

The convergence of the remaining terms follows similarly to the proof of *Theorem 3.4.9*. Hence, (c, v) solves the local system (3.4.2) in the sense of *Definition 3.4.6*.

3.5 Numerical simulations in 1D

This section is the sole work of Kevin Painter and is included for the sake of completeness.

We perform numerical simulations to investigate on the one hand the effect of differences between hitherto choices of nonlocal operators and our novel ones proposed in *Section 3.2*, and on the other hand convergence between nonlocal and local formulations. For compactness, our current study restricts to the prototypical nonlocal model for cellular adhesion (3.1.1), its reformulation as (3.4.1), and the corresponding local model (3.4.2). Thus, for (3.4.1) we take the operator form $\mathcal{R}_r = \mathcal{T}_r$, with \mathcal{T}_r as in (3.2.2). These models can be interpreted in the context of a population of cells invading an adhesion-laden ECM/tissue environment and, with this in mind, we initially concentrate cells at the centre of a one-dimensional domain $\Omega = [0, L]$ and impose an initially homogeneous ECM. Specifically, we set for the ECM

$$v_0(x) = 1, \quad x \in \Omega \tag{3.5.1}$$

and consider for the cell population a Gaussian-shaped aggregate

$$c_0(x) = \exp\left(-\alpha(x - x_c)^2\right), \quad x \in \Omega, \tag{3.5.2}$$

where we set $x_c = L/2$ or $x_c = 0$.

The numerical scheme follows that described in [67], which we refer to for details. Briefly, a Method of Lines approach is invoked whereby equations are first discretised in space (in conservative form, via a finite volume method) to yield a high-dimensional system of ODEs, which are subsequently integrated in time. Discretisation of advective terms follows a third order upwinding scheme, augmented by flux limiting to preserve positivity of solutions and the resulting scheme is (approximately) second-order accurate in space. Time integration has been performed with standard Matlab ODE solvers: our default is "ode45" with absolute and relative error tolerances set at 10^{-6} , but simulations have been compared for varying space discretisation step, ODE solver, and error tolerances. To measure the difference between two distinct solutions over time we define a distance function as follows:

$$d(u_1(x,t), u_2(x,t))(t) = \int_{\Omega} |u_1(x,t) - u_2(x,t)| \, \mathrm{d}x \, ,$$

where u_1 and u_2 denote the two solutions that are being compared.

3.5.1 Comparison of nonlocal operator representations

We first explore the correspondence between forms of nonlocal operator representation: we choose the prototypical nonlocal model for cell/matrix adhesion (3.1.1) and its reformulation (3.4.1), therefore taking for the latter the operator form $\mathcal{R}_r = \mathcal{T}_r$ with \mathcal{T}_r as in (3.2.2). In what follows,



Figure 3.1: Comparison between nonlocal formulations (3.1.1) and (3.4.1). (a-b) Cell and matrix densities for the models (3.1.1) and (3.4.1) at t = 2.5 and t = 5. (c) Difference between the solutions. For these simulations we take $\alpha = 10$, r = 1, $D_c = 0.01$, $\chi = 1$, $F_r = 2$, $f_c = 0$ and $f_v(c, v) = -cv$, along with (a-c) g(c, v) = 10v, (d-f) g(c, v) = 2.5c + 10v.

solutions to (3.1.1) are denoted c_A and v_A and those for (3.4.1) denoted c_T and v_T . For simplicity we restrict in this section to a minimalist formulation in which $D_c = \text{constant}$, $\chi = 1$, $f_c = 0$. Cell-matrix interactions are defined by $g(c, v) = S_{cc}c + S_{cv}v$ and $f_v(c, v) = -\mu cv$, where S_{cc} and S_{cv} respectively represent cell-to-cell and cell-to-matrix adhesion strengths and f_v simplistically describes (direct) proteolytic degradation of matrix by cells parametrised by degradation rate μ .

Figure 3.1 shows the computed solutions under (a-c) negligible cell-cell adhesion ($S_{cc} = 0$) and (d-f) moderate cell-cell adhesion ($S_{cc} = S_{cv}/4$). The equivalence of the two formulations is revealed through the negligible difference between solutions, with the distance magnitude attributable to the subtly distinct numerical implementation. Both simulations describe an invasion/infiltration process, in which matrix degradation by the cells generates an adhesive gradient that pulls cells into the acellular surroundings. The impact of cell-cell adhesion is manifested in the compaction of cells at the leading edge into a tight aggregate.

However, as pointed out in Section 3.2, differences in the nonlocal formulations can emerge in the vicinity of boundaries. To highlight this we consider an equivalent formulation to Figure 3.1 (a-c), but with the cells initially placed at the left boundary $(x_c = 0 \text{ in } (3.5.2))$, e.g., suggesting a tumor mass which is concentrated there and whose cells are expected to detach and migrate into the considered 1D domain, travelling from left to right. As stated earlier we impose zero-flux boundary conditions at x = 0 (and x = L), and further suppose c = v = 0 and $\nabla c = \nabla v$ in the extradomain region ($\mathbb{R} \setminus \Omega$). Representative simulations are shown in Figure 3.2. They are in agreement with our observation in Example 3.2.3. Indeed, for this scenario, in the prototypical nonlocal model (3.1.1)-(3.1.2) there is a very large adhesion velocity modulus at x = 0; the cells



Figure 3.2: (a-c) Comparison between nonlocal formulations (3.1.1) and (3.4.1) near boundaries. Model as in *Figure 3.1* (a-c), but with the cells initially concentrated at the boundary. (d-f) Comparison of the two forms of nonlocal operator corresponding to the simulations represented in (a-c). The operators are practically identical sufficiently far from the boundary, but can diverge significantly for distances < r from the boundaries.

are crowded within the tumor mass and their mutual interactions are maintained during the invasion process in a sufficiently strong manner to ensure a collective shift of the still concentrated cell aggregate, with a correspondingly strong tissue degradation in its wake. In the reformulation (3.4.1a)-(3.4.1b), rather, the adhesion magnitude at x = 0 is for the same initial condition much lower - suggesting a tumor whose cells are readier to detach and migrate individually. This results in a more diffusive spread, with accordingly less degradation of tissue, and with cell mass remaining available at the original site over a larger time span. The latter scenario is different from the former one, but it seems nevertheless reasonable, as a tumor mass would very often not move as a whole from its original location to another in a relatively short time; moreover, the active cells in a sufficiently large tumor (releasing substantial amounts of acidity) are known to preferentially adopt a migratory phenotype and perform EMT (epithelial-mesenchymal transition), see e.g., [73, 120, 125], which supports the idea of cells moving in a loose way rather than in compact, highly aggregated assemblies ⁴. As such, our simulations suggest that, within this particular function- and parameter setting, choosing the adhesion operator in the form (3.1.2)instead of (3.2.2) might possibly overestimate the tumor invasion speed and associated healthy tissue degradation, thereby predicting a spatially concentrated tumor and neglecting regions with lower cell densities which can nevertheless trigger tumor recurrence if untreated.

⁴unless environmental influences dictate conversion to a collective type of motion

3.5.2 Comparison between nonlocal and local formulation

Having compared together the original, (3.1.1), and the new, (3.4.1), nonlocal formulations, we next consider the extent to which their dynamics can be captured by the classical local formulation (3.4.2). Note that for nonlocal model simulations we will restrict to the original formulation (3.1.1), so that we can avail ourselves of an already well-established efficient (in terms of computational time) numerical scheme [67]. Here we use c_L and v_L to denote solutions to the local formulation and c_{Ar} and v_{Ar} to denote solutions to the nonlocal model with sensing radius r. We remark that a large number of related local and nonlocal models have been numerically studied to describe the invasion-type process considered here (e.g., [3, 68, 118, 121]): here the specific focus is to explore the convergence of nonlocal to local form as $r \to 0$, which, as far as we are aware, has not been systematically investigated.

As in the first test we use the initial values (3.5.1) and (3.5.2), choosing $x_c = L/2$, $\alpha = 10$ in the latter, and consider the coefficients and functions as proposed in *Example 3.4.5*. Under these choices the resultant nonlinear diffusion coefficient for the *c*-equation in the classical local formulation (compare (3.4.2a)) becomes

$$\tilde{D}_c(c,v) = \frac{a^2(1+c)^2(1+c+v)^2 - bc(1+cv)(S_{cc} + (S_{cc} - S_{cv})v)}{(1+cv)^2(1+c+v)^2}.$$
(3.5.3)

Notably, this potentially becomes negative under an injudicious combination of adhesive strengths S_{cc} , S_{cv} , and of a, b. Likewise, the actual haptotaxis sensitivity function takes the form

$$\tilde{\chi}(c,v) = b \frac{S_{cv} + (S_{cv} - S_{cc})c}{(1+cv)(1+c+v)^2}.$$
(3.5.4)

Again, depending on the relationship between S_{cc} and S_{cv} , this can become negative, which would lead to repellent haptotaxis: cells effectively moving away from regions with large ECM gradients, a rather unexpected behaviour. This suggests that cell-tissue adhesions should dominate over cell-cell adhesions,⁵ as 'usual' haptotaxis, i.e., towards the increasing tissue gradient, is known to be an essential component of cell migration, this applying to several types of cells moving through the ECM (tumor cells, mesenchymal stem cells, fibroblasts, endothelial cells, etc.) see e.g., [95, 123, 147] and references therein.

Simulations are plotted in *Figure 3.3* where we show cell densities for the local model (c_L) and nonlocal model under three sensing radii:

$c_{Ar=0.1}, c_{Ar=0.3}, c_{Ar=1.0}.$

In this first set of simulations we assume negligible cell-cell adhesion $(S_{cc} = 0)$, which automatically ensures positivity for the diffusion coefficient of the equivalent local model, $\tilde{D}_c(c, v)$. We note that matrix renewal is absent $(\mu_v = 0)$ in the left-hand column and present $(\mu_v > 0)$ in the central column. In the right-hand column we show the greater generality of the results under vastly simplified kinetics, specifically setting $f_c(c, v) = 0$ and $f_v(c, v) = -cv$ (with the other functional forms as in *Example 3.4.5*). Simulations highlight the convergence between local and nonlocal models as $r \to 0$: for r = 0.1, the solution differences become negligible. However, distinctions emerge for large r, where we can expect significant discrepancy between the solutions.

⁵An analogous behaviour was suggested by the two-scale structured population model with adhesion introduced in [56].



Figure 3.3: Convergence between nonlocal and local/classical formulations under negligible cellcell adhesion, $S_{cc} = 0$, $S_{cv} = 10$. Functional forms as proposed in *Example 3.4.5*, with modifications specified in the subfigures. (a) Solutions for r = 0.1, 0.3, 1.0 at (a1) t = 2, (a2) t = 4and (a3) t = 8; (a4) Distance between local/nonlocal solutions as a function of time. For these simulations, we take a = 0.01, b = 1, $\mu_c = 0.01$, $K_c = 2$, $\eta_c = 1$, $\mu_v = 0$, $\lambda_v = 1$. (b) Solutions for r = 0.1, 0.3, 1.0 at (b1) t = 2, (b2) t = 4 and (b3) t = 8; (b4) Distance between local/nonlocal solutions as a function of time. Parameters as in (a) except $\mu_v = 1$, $K_v = 1$. (c) Solutions for $f_c = 0$ and $f_v(c, v) = -cv$, with the other parameters as in (a).



Figure 3.4: Time restricted convergence under moderate cell-cell adhesion, $S_{cc} = 2.5$, $S_{cv} = 10$. Top row shows solutions across the full spatial region ([0, 20]), the bottom row magnifies a relevant portion for clarity. Solutions to local and nonlocal models under the functional forms proposed in *Example 3.4.5* for r = 0.01, 0.1, 0.3, 1.0 at (a) t = 3, (b) t = 3.5 and (c) t = 5. In (a) solutions to the local model continue to exist and we observe convergence between local and nonlocal formulations. In (b-c) the solutions to the local model are noncomputable. Nonlocal models, however, can destabilise into a pattern of aggregates. Parameters: a = 0.01, b = 1, $\mu_c = 0.01, K_c = 2, \eta_c = 1, \mu_v = 0, \lambda_v = 1$ and adhesion parameters as above.

This suggests that the local model fails to accurately predict the behaviour in cases where cells sample over relatively large regions of their local environment.

Next, we extend to include a degree of cell-cell adhesion, setting functions and parameters as in Figure 3.3, except now $S_{cc} > 0$. Notably this raises the possibility of a negative diffusion coefficient in the classical formulation and subsequent illposedness. Solutions under a representative set of parameters are shown in Figure 3.4. For t below some critical time we observe convergence as before, with the nonlocal formulation converging to solutions of the local model as $r \to 0$. However, continued matrix degradation further depletes v, with the result that (3.5.3) can become negative. At this point (in this case $t \approx 3.2...$) the local model becomes illposed and its solutions become incomputable (implying nonexistence of solutions). However, the nonlocal formulation appears to preserve wellposedness, consistent with previous theoretical studies where extending to a nonlocal formulation regularises a singular local model (e.g., [79]). Solutions to the nonlocal model instead destabilise into a quasi-periodic pattern of cell aggregations, maintained through the cell-cell adhesion, and with a wavelength shrinking as $r \to 0$.

Finally, we remark that convergence of solutions extends beyond the specific functional forms and, as a representative example, we consider a minimalist setting based on linear/constant forms. Specifically, we set $D_c = a$ (constant), $\chi = 1$, $f_c = 0$, $g(c, v) = S_{cc}c + S_{cv}v$ and $f_v(c, v) = -\mu cv$. In this scenario, the diffusion and haptotaxis coefficients for the classical local formulation (3.4.2) reduce to

$$\tilde{D}_c(c,v) = a - S_{cc}c \quad \text{and} \quad \tilde{\chi}(c,v) = S_{cv}.$$
(3.5.5)

Positivity is only guaranteed under appropriate parameter selection. Such a case is illustrated



Figure 3.5: Convergence between nonlocal and local/classical formulations under a set of minimalistic linear functional forms ($D_c = 0.01$, $\chi = 1$, $f_c = 0$, $g(c, v) = S_{cc}c + S_{cv}v$, $f_v(c, v) = -\mu cv$). Negligible cell-cell adhesion, $S_{cc} = 0$, $S_{cv} = 10$: solutions shown at (left) t = 2.5 and (middle) t = 5, with the distance between solutions to the nonlocal and local model shown in the right panel.

in Figure 3.5 where we assume negligible cell-cell adhesion $(S_{cc} = 0)$. Clearly, we observe convergence between the nonlocal and local formulations as $r \to 0$. Inappropriate parameter selection, however, generates backward diffusion in the local model and solutions are consequently incomputable. In all cases considered in this test the cells do not reach the boundary region where the difference between the nonlocal formulations (3.1.1) and (3.4.1) can play a role. Thus, we expect the same solution if reformulation (3.4.1) is applied instead.

3.6 Discussion

In this chapter we provide a rigorous limit procedure which links nonlocal models involving adhesion or a nonlocal form of chemotaxis gradient to their local counterparts featuring haptotaxis, respectively chemotaxis in the usual sense. As such, it closes a gap in the existing literature. Moreover, it offers a unified treatment of the two types of models and extends the previous mathematical framework to settings allowing for more general, solution dependent, coefficient functions (diffusion, tactic sensitivity, adhesion velocity, nonlocal taxis gradient, etc.). Finally, we provide simulations illustrating some of our theoretical findings in 1D.

Our reformulations in terms of \mathcal{T}_r and \mathcal{S}_r reveal the tight relationship between the nonlocal operators \mathcal{A}_r and $\mathring{\nabla}_r$ and the (local) gradient. This suggests that both nonlocal descriptions (adhesion, chemotaxis) actually encompass the dependence on the signal gradients rather than on the signal concentration/density itself, which is in line with the biological phenomenon. Indeed, through their transmembrane elements (e.g., receptors, ion channels etc.) the cells are mainly able to perceive and respond to differences in the signal at various locations or within more or less confined areas rather than measure effective signal concentrations. Along with the mentioned solution dependency of the nonlocal model coefficients, the influence of the gradient possibly reflects into contributions of the adhesion/nonlocal chemotaxis to the (nonlinear) diffusion in the local setting obtained through the limiting procedure.

The set Ω_r can be regarded as the 'domain of restricted sensing', meaning that there cells a

priori sense only what happens inside Ω , the domain of interest. The measure of this subdomain is a decreasing function of the sensing radius r. When $r \to 0$ the set Ω_r tends to cover the whole domain Ω , whereas as r increases the cells can sense at increasingly larger distances; correspondingly, Ω_r shrinks. For $r > \operatorname{diam}(\Omega)$ the restricted sensing domain is empty: everywhere in Ω the cells can perceive signals not only from any point within Ω but potentially also from the outside. In this work, however, we look at models with no-flux boundary conditions. This corresponds, e.g., to the impenetrability of the walls of a Petri dish or that of comparatively hard barriers limiting the areas populated by migrating cells, e.g., bones or cartilage material. As a result, the cells in the boundary layer $\Omega \backslash \Omega_r$ have a much reduced ability to stretch their protrusions outside Ω and thus gain little information from without. To simplify matters, we assume in this work that there is no such information or it is insufficient to trigger any change in their behaviour. In the definitions of \mathcal{T}_r and \mathcal{S}_r this corresponds to the integrands being set to zero in $\Omega \backslash \overline{\Omega_r}$.

It is important to note that for points $x \in \Omega \setminus \overline{\Omega}_r$ the influence of a signal p in a direction $y \in S_1$ is not taken into account by $\mathring{\nabla}_r$ at all if $x + ry \notin \overline{\Omega}$. If S_r is used instead, then its contribution to the average is given by

$$\tilde{y} := n \left(\int_0^1 \chi_\Omega \nabla p(x + rsy) \, \mathrm{d}s \cdot y \right) y.$$

Thus, thanks to integration w.r.t. s, the resulting vector \tilde{y} assembles the impact of those parts of the segment connecting x and x + ry which are contained in Ω . It is parallel to y, and it may have the same or the opposite orientation. In particular this means that although for a certain range of directions large parts of the sensing region of a cell are actually outside Ω , this may still strongly influence the speed and actual direction of the drift. The effect of integration w.r.t. s in \mathcal{T}_r is less obvious, since in this case the average w.r.t. y is computed over the ball B_1 . This already achieves the covering of the whole sensing region by allowing a cell to gather information about the signal not only in any direction y/|y|, but also at any distance less than r. The additional integration over the path x + rsy, $s \in [0, 1]$, appears to mean that cells at $x \in \Omega_r$ are able to measure the average of the signal gradient all along such line segment rather than its value directly at the ending point. Indeed, from a biological viewpoint this description seems to make more sense, as cells do not jump from one position to another, nor do they send out their protrusions in a discontinuous way bypassing certain space points along a chosen direction. Averages over cell paths are then averaged w.r.t. y, which finally determines the direction of population movement. Example 3.2.4 indicates that the effect of even an extremely concentrated signal gradient is mollified by averaging. This agrees with our expectations from using nonlocality. In higher dimensions $n \ge 2$, the two-stage averaging in \mathcal{T}_r (w.r.t. s and y) produces a direction field which is smooth away from the concentration point and also weakens but still keeps the singularity there. In contrast, averaging only w.r.t. y leads instead to jump discontinuities at a unit distance from the accumulation point. Moreover, we remark that without integrating w.r.t. s in $\mathcal{T}_r(\nabla)$ one cannot regain \mathcal{A}_r .

The effect observed in *Example 3.2.3* further supports the conjecture that the nonlocal operators which act directly on the signal gradients might actually be a more appropriate modelling tool. While inside the subdomain Ω_r there is no difference (recall *Lemmas 3.2.1* and *3.2.2*), inside the boundary layer $\Omega \setminus \overline{\Omega}_r$ the limiting behaviour as $r \to 0$ is qualitatively distinct. Indeed, *Example 3.2.3* shows that using, e.g., \mathcal{A}_r , leads, for $r \to 0$, to unnatural sharp singularities at the
boundary of Ω even in the absence of signal gradients, whereas this does not happen if \mathcal{T}_r is used instead. Simulations in Subsection 3.5.1 (see Figure 3.2) confirm our theoretical findings and show a substantial difference between the solutions obtained with the two nonlocal formulations involving (3.1.2) and (3.2.2), respectively. The choice (3.2.2) is motivated above all from a mathematical viewpoint (as it enables a rigorous, well-justified passage to the limit for $r \to 0$), but it also seems to make sense biologically, as our above comments and the simulations performed for the particular setting in Subsection 3.5.1 suggest.

In this chapter we have only dealt with models that include a nonlocality in the chemotaxis or cell-cell and/or cell-tissue adhesion terms and assumed the diffusion to be local. This is in line with most of the previously developed nonlocal models for cell migration, albeit they usually cover just linear diffusion. If cell-cell adhesion is present, this means that the cell flux contains the local cell gradient, as well as some averaging of it. The latter is described in our case by a suitably chosen operator \mathcal{T}_r . A possible model extension could involve a diffusion flux which is also nonlocal and has a similar form. This would mean that the cell flux is completely devoid of the local gradient. From the modelling point of view this could be seen as a population pressure acting⁶ in a nonlocal manner: each cell is sensing the population mass not only at its current position, but over a whole region (of radius r) around that location. This is actually true in vivo, where cells sample their biological environment by extending protrusions as far as several cell lengths. While cell-cell adhesions certainly play a role in this process and contribute to self-diffusion (as in the example handled in *Subsection 3.5.2*), there might be yet other ways of interaction by which the cells are able to perceive smaller or larger aggregates of their own kind. In this context one could think about replacing the local gradient by a nonlocal operator, e.g., of the form $\mathcal{T}_r(\nabla)$. However, the analysis of such a model would be considerably more involved and it is to expect that existence of solutions can be established only under rather restrictive assumptions.

 $^{^{6}\}mathrm{unlike}$ Fick's classical law which typically connects the flux over the domain boundary with the diffusion inside the domain

CHAPTER 4

Global existence of solutions to a nonlocal equation with degenerate anisotropic diffusion

This chapter was first published in Volume 543 of Journal of Mathematical Analysis and Applications in 2025.¹ The presentation has been adapted for use in this dissertation to clarify the details of the proofs and guarantee consistency of the notation.

4.1 Motivation

In this chapter we study the initial boundary value problem (IBVP)

$$\partial_t c = \nabla \nabla : (\mathbb{D}c) - \nabla \cdot (c\mathcal{A}c) + \mu c(1 - c^{r-1}) \qquad \text{in } \Omega \times (0, \infty), \tag{4.1.1a}$$

$$(\nabla \cdot (\mathbb{D}c) - c\mathcal{A}c) \cdot \nu = 0 \qquad \text{on } \partial\Omega \times (0, \infty), \qquad (4.1.1b)$$

$$c(\cdot, 0) = c_0 \qquad \qquad \text{in } \Omega, \qquad (4.1.1c)$$

where \mathcal{A} is the standard adhesion operator [5], see *Definition 4.2.1* below, $\nabla \nabla$: ($\mathbb{D}c$) is the myopic diffusion [83], see *Definition 2.1.5*, driven by a symmetric non-negative definite diffusion tensor $\mathbb{D} = \mathbb{D}(x)$, $\mu > 0$ and $r \ge 2$ are positive constants, and Ω is a smooth bounded domain in \mathbb{R}^n , $n \in \mathbb{N}$. The nonlocal reaction-diffusion-advection equation (4.1.1a) is an extension of an equation that was recently derived in [156] using a multiscale approach and corresponds to the case $\mu = 0$. It can describe the evolution of density c = c(t, x) of a cell population that disperses due to a potentially anisotropic diffusion and nonlocal adhesion, thus upgrading the original model from [5] where the diffusion term is $D\Delta c$ with D > 0 a constant. We refer to [156] for further details regarding the modelling and derivation approaches.

While the combination of adhesion with a Fickian-type diffusion has received much attention, see, e.g., [28] and references therein, the case of myopic diffusion has not been analysed so far. The few papers [77, 149–151] that have dealt with existence and long-time behaviour of solutions to problems that include both myopic diffusion and advection are restricted to versions of the model derived in [54]. It features haptotaxis, i.e., the directed movement along the local gradient of an external immovable signal, rather than the spatially nonlocal intrapopulational adhesion as in (4.1.1a). Apart from that, as a result of somewhat different underlying derivation approaches in [54] compared to [156], the advection velocity in the aforementioned haptotaxis model is

¹[50] The paper is licensed under a CC BY license.

multiplied by the diffusion tensor, whereas in (4.1.1a) this is not the case. Thus, here it is in no way 'subordinate' to the diffusion and, in particular, the adhesion term need not vanish in those areas where diffusion is absent. Finally, we observe that apart from [77] where dimensions two and three were treated, other works [149–151] only considered the one-dimensional case.

The goal of this chapter is to establish a result on global existence of solutions to (4.1.1a) equipped with no-flux boundary and initial conditions. Our approach works for $\mu > 0$, i.e., in the presence of the generalised logistic-type growth term. While it describes a biologically relevant effect (e.g., cell growth/death), our main motivation for including the source term stems from the analytical challenges that arise in the case of $\mu = 0$. In the latter scenario, since the diffusion is non-Fickian and degenerate, only mass preservation is a priori guaranteed, indicating that generally solutions need not be functions but could be measure-valued. Here we chose to avoid this possibility by including the growth term. While our analysis allows for degenerate diffusion tensors, we require the degeneracy set, i.e., the set of points where \mathbb{D} is not positive definite, to have a positive distance to the boundary of Ω and to be sufficiently low-dimensional, see condition (4.3.2f) below. This condition seems to be new in the context of degenerate diffusion. It arises from *Lemma 4.2.4* in *Subsection 4.2.2* and provides a certain balance between the degenerate diffusion and the nonlinear growth term.

The remainder of the chapter is organised in the following way. After recalling the definition of the adhesion operator \mathcal{A} in Section 4.2, we fully set-up our model and formulate our main result on existence of very weak solutions, Theorem 4.3.4, in Section 4.3. We then analyse suitably constructed approximation problems in Section 4.4. The uniform estimates that we establish there allow to apply the compactness method and prove Theorem 4.3.4 in Section 4.5. Finally, in Section 4.6 we provide a justification of our solution concept proving that regular solutions of this sort are classical.

4.2 Preliminaries

4.2.1 The adhesion operator

Recall that throughout this dissertation $\Omega \subset \mathbb{R}^n$ is a domain with smooth enough boundary (C^3 in this chapter). We recall the definition of the adhesion operator \mathcal{A} between two functional spaces from *Chapter 3* in the way that suits our needs, i.e., for a sensing radius equal to 1 in (3.1.2). This is no restriction of generality since this value can be always achieved through a suitable rescaling of the spatial variable.

Definition 4.2.1. Consider a continuous function $F : [0,1] \rightarrow [0,\infty)$. The adhesion operator is given by

$$\mathcal{A}: L^1(\Omega) \to (L^\infty(\Omega))^n, \quad \mathcal{A}u(x) := \frac{1}{|B_1|} \int_{B_1} u(x+\xi) \frac{\xi}{|\xi|} F(|\xi|) \, d\xi. \tag{4.2.1}$$

By convention $u(x+\xi) = 0$ if $x+\xi \notin \overline{\Omega}$. The operator is well-defined and bounded, see *Chapter 3*, but in contrast to *Chapter 3* we will use here the reformulation

that was observed in [156], where H is the interaction potential given by

$$H: \mathbb{R} \to [0, \infty], \quad H(x) := \frac{1}{|B_1|} \int_{\min\{|x|, 1\}}^1 F(z) \, \mathrm{d}z,$$

so that its gradient is the L^∞ function

$$\nabla H(x) := \begin{cases} -\frac{1}{|B_1|} \frac{x}{|x|} F(|x|) & \text{if } x \in B_1 \setminus \{0\}, \\ 0 & \text{if } x \in \mathbb{R}^n \setminus \overline{B_1}. \end{cases}$$

Consequently, the operator \mathcal{A} satisfies the lemma below.

Lemma 4.2.2. Let $k \in \mathbb{N}_0$ and $\alpha \in [0,1]$. Then, \mathcal{A} is a continuous operator on $C^{k,\alpha}(\overline{\Omega})$.

Proof. For k = 0 this follows from Lemma 2.3.2. Moreover, for $k \ge 1$ we conclude this combining *Theorems A.3.11* and *A.3.12* and *Lemma 2.3.2*.

4.2.2 A lemma about sets of "sufficiently small" dimension

We recall one of the (alternative) ways of defining fractional dimension from (3.5) and the subsequent discussion on p. 42 in [60].

Definition 4.2.3. (Upper box dimension) Let $K \subset \mathbb{R}^n$ be compact. For every $\delta > 0$ we denote

$$Z_{\delta}(K) := \left\{ b \in \delta \mathbb{Z}^n : |x - b|_{\infty} \leq \frac{\delta}{2} \text{ for some } x \in K \right\}.$$

The upper box dimension is the non-negative number

$$\overline{\dim}_{\mathcal{F}}(K) := \limsup_{\delta \to 0} \frac{\log_2 |Z_{\delta}(K)|}{\log_2 \delta^{-1}}.$$

Lemma 4.2.4. Let

$$3 \leqslant n \in \mathbb{N},\tag{4.2.2}$$

$$r > \frac{n}{n-2},\tag{4.2.3}$$

and $K \subset \mathbb{R}^n$ be a compact set such that

$$\overline{\dim}_{\mathcal{F}}(K) < n - \frac{2r}{r-1}.$$
(4.2.4)

Then, there exists a family $(\varphi_{\delta})_{\delta \in (0,1)}$ of functions such that for all $\delta > 0$

$$\varphi_{\delta} \in C_c^{\infty}(\mathbb{R}^n; [0, 1]), \tag{4.2.5a}$$

$$\varphi_{\delta} = 1 \qquad in \ \overline{O_{\delta\sqrt{n}}(K)}, \tag{4.2.5b}$$

$$supp(\varphi_{\delta}) \subset O_{5\delta\sqrt{n}}(K),$$
 (4.2.5c)

$$\|\nabla\varphi_{\delta}\|_{(L^{\infty}(\mathbb{R}^{n}))^{n}} \leqslant \delta^{-1}C_{1}, \qquad (4.2.5d)$$

$$\|D^2\varphi_\delta\|_{(L^\infty(\mathbb{R}^n))^{n\times n}} \leqslant \delta^{-2}C_1, \tag{4.2.5e}$$

$$\varphi_{\delta} \underset{\delta \to 0}{\xrightarrow{}} 0 \qquad a.e. \ in \ \mathbb{R}^n, \tag{4.2.5f}$$

$$\lim_{\delta \to 0} \delta^{-\frac{2r}{r-1}} |supp(\varphi_{\delta})| = 0.$$
(4.2.5g)

Proof. Let $\eta \in C_c^{\infty}([0,\infty); [0,1])$ be such that

$$\eta = \begin{cases} 1 & \text{in } [0,1], \\ 0 & \text{in } [2,\infty). \end{cases}$$

 Set

$$\varphi_{\delta}(x) := \frac{\sum_{z \in \delta \mathbb{Z}^n \cap O_{3\delta\sqrt{n}}(K)} \eta\left(\frac{|x-z|}{\delta\sqrt{n}}\right)}{\sum_{z \in \delta \mathbb{Z}^n} \eta\left(\frac{|x-z|}{\delta\sqrt{n}}\right)} \quad \text{for } x \in \mathbb{R}^n, \ \delta \in (0,1).$$
(4.2.6)

We need to check that φ_{δ} satisfies the required properties.

1. Since

$$\max_{x \in \mathbb{R}^n} \left| B_{3\delta\sqrt{n}}(x) \cap \delta\mathbb{Z}^n \right| = \max_{y \in \mathbb{R}^n} \left| B_{3\sqrt{n}}(y) \cap \mathbb{Z}^n \right| =: C_2 < \infty$$

for some $C_2 > 0$, and $\eta = 0$ in $[2, \infty)$, the sums in (4.2.6) contain at most C_2 non-zero summands.

2. Since $\mathbb{R}^n = O_{\delta\sqrt{n}}(\delta\mathbb{Z}^n)$ and $\eta = 1$ in [0, 1], the denominator in (4.2.6) is never zero, and

$$\sum_{z \in \delta \mathbb{Z}^n} \eta\left(\frac{|x-z|}{\delta \sqrt{n}}\right) \ge 1.$$
(4.2.7)

- 3. By 1. and 2. and the assumptions on η , the function φ_{δ} is well-defined and belongs to $C_c^{\infty}(\mathbb{R}^n; [0, 1]).$
- 4. Since $\eta = 0$ in $[2, \infty)$, the numerator and the denominator coincide for $x \in O_{\delta\sqrt{n}}(K)$ both having the value

$$\sum_{z \in \delta \mathbb{Z}^n \cap O_{2\delta\sqrt{n}}(x)} \eta\left(\frac{|x-z|}{\delta\sqrt{n}}\right),\,$$

hence $\varphi_{\delta} = 1$ there.

5. Since $\eta = 0$ in $[2, \infty)$,

$$\sup (\varphi_{\delta}) \subset \overline{O_{2\delta\sqrt{n}} \left(\delta \mathbb{Z}^{n} \cap O_{3\delta\sqrt{n}}(K)\right)}$$

$$\subset \overline{O_{5\delta\sqrt{n}}(K)}$$

$$\subset \overline{O_{6\delta\sqrt{n}}(Z_{\delta}(K))}.$$

$$(4.2.8)$$

Combining (4.2.4) and (4.2.8), we obtain

$$\log_2\left(\delta^{-\frac{2r}{r-1}}|\operatorname{supp}(\varphi_{\delta})|\right) \leq \log_2\left(\delta^{-\frac{2r}{r-1}}|Z_{\delta}(K)|(6\delta\sqrt{n})^n|B_1|\right)$$
$$\leq \log_2\left(C_3\delta^{n-\frac{2r}{r-1}}|Z_{\delta}(K)|\right)$$
$$= \log_2 C_3 + \log_2 \delta^{-1}\left(\frac{\log_2|Z_{\delta}(K)|}{\log_2 \delta^{-1}} - \left(n - \frac{2r}{r-1}\right)\right)$$
$$\xrightarrow{\delta \to 0} - \infty, \tag{4.2.9}$$

so (4.2.5g) holds.

6. We compute

$$\nabla \varphi_{\delta}(x) = \frac{1}{\delta\sqrt{n}} \left(\frac{\sum_{z \in \delta\mathbb{Z}^{n} \cap O_{3\delta\sqrt{n}}(K)} \operatorname{sign}(x-z)\eta'\left(\frac{|x-z|}{\delta\sqrt{n}}\right)}{\sum_{z \in \delta\mathbb{Z}^{n}} \eta\left(\frac{|x-z|}{\delta\sqrt{n}}\right)} - \frac{\sum_{z \in \delta\mathbb{Z}^{n} \cap O_{3\delta\sqrt{n}}(K)} \eta\left(\frac{|x-z|}{\delta\sqrt{n}}\right) \sum_{z \in \delta\mathbb{Z}^{n}} \operatorname{sign}(x-z)\eta'\left(\frac{|x-z|}{\delta\sqrt{n}}\right)}{\left(\sum_{z \in \delta\mathbb{Z}^{n}} \eta\left(\frac{|x-z|}{\delta\sqrt{n}}\right)\right)^{2}} \right).$$
(4.2.10)

Combining 1., 2. and (4.2.10) and the assumptions on η , we obtain (4.2.5d). Differentiating again and using the same argument yields (4.2.5e).

7. The convergence (4.2.5f) is a direct consequence of (4.2.5b) and (4.2.5c) and $\overline{\dim}_{\mathcal{F}}(K) < n$.

4.3 Problem setting and main result

We make the following assumptions on the coefficients and other parameters.

Assumptions 4.3.1.

$$n \ge 3, \ r > \frac{n}{n-2}, \ r \ge 2,$$

$$F \in C([0,1]; [0,\infty)),$$

$$\mu > 0,$$

$$c_0 \in L^r(\Omega),$$
(4.3.1)

and

$$\mathbb{D} := (d_{ij})_{i,j=1,\dots,n} \in C(\overline{\Omega}; \mathbb{R}^{n \times n}), \tag{4.3.2a}$$

$$\mathbb{D}(x) \text{ symmetric for } x \in \overline{\Omega}, \tag{4.3.2b}$$

$$\nabla \cdot \mathbb{D} \in \left(L_{loc}^{\infty}(\{\mathbb{D} > 0\}) \right)^n, \tag{4.3.2c}$$

$$\mathbb{D} \ge 0, \tag{4.3.2d}$$

$$a := dist(\partial\Omega, \{\mathbb{D} \ge 0\}) > 0, \tag{4.3.2e}$$

$$\overline{\dim}_{\mathcal{F}}(\{\mathbb{D} \ge 0\}) < n - \frac{2r}{r-1}.$$
(4.3.2f)

Recall that for $\mathbb{D} \geqslant 0$ the above sets are defined as

 $\{\mathbb{D} > 0\} = \{x \in \overline{\Omega} : y^T \mathbb{D}(x)y > 0 \,\forall y \in \mathbb{R}^n\}$

and

$$\{\mathbb{D} \ge 0\} := \{x \in \overline{\Omega} : \exists y \in \mathbb{R}^n \text{ s.t. } y^T \mathbb{D}(x)y = 0\}.$$

To illustrate condition (4.3.2f), we consider two special cases where it is satisfied.

Example 4.3.2. 1. For a finite set $\{\mathbb{D} \ge 0\} = \{a_1, \dots, a_K\}$ with $K \in \mathbb{N}$ we can estimate

$$|Z_{\delta}(\{\mathbb{D} \ge 0\})| = \left| \left\{ b \in \delta \mathbb{Z}^n : |a_i - b|_{\infty} \le \frac{\delta}{2} \text{ for some } i = 1, \dots, K \right\} \right| \le 2K$$

Then,

$$\overline{\dim}_{\mathcal{F}}(\{\mathbb{D} \ge 0\}) = \limsup_{\delta \to 0} \frac{\log_2 |Z_{\delta}(\{\mathbb{D} \ge 0\})|}{\log_2 \delta^{-1}} \leqslant \limsup_{\delta \to 0} \frac{\log_2(2K)}{\log_2 \delta^{-1}} = 0.$$

2. Consider a sequence $(a_k)_{k \in \mathbb{N}}$ in Ω with $\lim_{k \to \infty} a_k = \tilde{a} \in \Omega$, i.e., for all $\delta > 0$ there is $K(\delta) \in \mathbb{N}$ s.t. $|a_k - \tilde{a}| < \frac{\delta}{2}$ for all $k \ge K(\delta)$. We assume that the sequence converges fast enough to its limit in the sense that there is $b < n - \frac{2r}{r-1}$ s.t.

$$\limsup_{\delta \to 0} \frac{\log_2(K(\delta))}{\log_2 \delta^{-1}} = b.$$
(4.3.3)

If $\{\mathbb{D} \ge 0\} = \{\tilde{a}, a_1, a_2, \dots\}$, then

$$|Z_{\delta}(\{\mathbb{D} \ge 0\})| \le 2(K(\delta) + 1). \tag{4.3.4}$$

Combining (4.3.3) and (4.3.4), we arrive at (4.3.2f).

We define solutions to the IBVP (4.1.1) as follows.

Definition 4.3.3. We call a function $c \in L^r_{loc}(\overline{\Omega} \times [0, \infty))$ a <u>global very weak solution</u> to (4.1.1) if it satisfies

$$-\int_{0}^{\infty} \int_{\Omega} c\partial_{t}\eta \, dx \, dt - \int_{\Omega} c_{0}\eta(\cdot,0) \, dx$$

=
$$\int_{0}^{\infty} \int_{\Omega} c\mathbb{D} : D^{2}\eta \, dx \, dt + \int_{0}^{\infty} \int_{\Omega} c(\mathcal{A}c) \cdot \nabla\eta \, dx \, dt + \mu \int_{0}^{\infty} \int_{\Omega} c(1-c^{r-1})\eta \, dx \, dt \qquad (4.3.5)$$

for all

$$\eta \in C^{2,1}_c(\overline{\Omega} \times [0,\infty)) \qquad s.t. \ \nabla \eta \cdot (\mathbb{D}\nu) \equiv 0 \ on \ \partial \Omega \times (0,\infty)$$

The main result of this chapter concerns with the existence of such solutions.

Theorem 4.3.4. Let Assumptions 4.3.1 hold. Then, there is a very weak solution

$$c \in L^{\infty}(0,\infty; L^{1}(\Omega)) \cap L^{r}_{loc}(\overline{\Omega} \times [0,\infty))$$

to (4.1.1) in the sense of Definition 4.3.3.

Remark 4.3.5. The very weak formulation (4.3.5) is obtained by multiple partial integration that shifts all spatial derivatives to the test function. This choice of formulation exploits the structure of the myopic diffusion. At the same time, it avoids including terms such as $\nabla \cdot (\mathbb{D}c)$ or ∇c , for which it is likely not possible to obtain a priori bounds in the whole domain Ω due to a combination of the degeneracy of the diffusion tensor \mathbb{D} and the diffusion being myopic.

Furthermore, we show in *Section* 4.6 that sufficiently smooth very weak solutions are classical solutions to (4.1.1). This justifies our solution concept.

4.4 Approximate problems

4.4.1 Construction of a regular matrix family $(\mathbb{D}_{\varepsilon})$

We begin by constructing a family of regular matrices \mathbb{D}_{ε} that approximate \mathbb{D} in a suitable fashion. For such diffusion matrices, existence of regular solutions can be directly concluded from known results. Unlike [77], we impose uniform L^{∞} boundedness of the divergence for $(\mathbb{D}_{\varepsilon})$ instead of convergence in an L^p space for finite p and make no additional restrictions such as, e.g., vanishing normal trace, because our analysis does not require this.

Lemma 4.4.1. Let Ω be Lipschitz. For $\varepsilon_1 > 0$ small enough, there is a sequence of symmetric matrices $(\mathbb{D}_{\varepsilon})_{\varepsilon \in (0,\varepsilon_1)} = (((d_{ij})_{\varepsilon})_{i,j=1,...,n})_{\varepsilon \in (0,\varepsilon_1)} \subset C^{\infty}(\overline{\Omega}; \mathbb{R}^{n \times n})$ s.t.

$$\|\mathbb{D}_{\varepsilon}\|_{(L^{\infty}(\Omega))^{n\times n}} \leqslant \varepsilon + \|\mathbb{D}\|_{(L^{\infty}(\Omega))^{n\times n}} \quad \text{for } \varepsilon \in (0, \varepsilon_{1}), \tag{4.4.1}$$

$$\mathbb{D}_{\varepsilon} \ge \varepsilon \qquad \qquad for \ \varepsilon \in (0, \varepsilon_1), \tag{4.4.2}$$

$$\|\mathbb{D}_{\varepsilon} - \mathbb{D}\|_{C(\overline{\Omega};\mathbb{R}^{n\times n})} \to 0 \text{ as } \varepsilon \to 0.$$

$$(4.4.3)$$

Further, for every relatively open

$$B \subset \subset \{\mathbb{D} > 0\}$$

there exists some $\varepsilon_2(B) \in (0, \varepsilon_1]$ such that

$$\|\nabla \cdot \mathbb{D}_{\varepsilon}\|_{(L^{\infty}(B))^{n}} \leqslant C_{4}(B) \qquad for \ \varepsilon \in (0, \varepsilon_{2}(B)).$$

$$(4.4.4)$$

Proof. We take a standard approach, arguing in a manner similar to the proof of Theorem 3 in Section 5.3.3 in [58].

Step 1. We need some preparation before we can proceed with a regularisation of \mathbb{D} . It concerns the domain Ω . Let $x_0 \in \partial \Omega$. Since Ω has a Lipschitz boundary, there exist $\gamma \in C^{0,1}(\mathbb{R}^{n-1};\mathbb{R})$ and $\rho \in (0, a)$ s.t. (after relabeling and reorientation of the axes if necessary)

$$\Omega \cap B_{\rho}(x_0) = \left\{ y \in B_{\rho}(x_0) : y_n > \gamma(y') \right\},$$
(4.4.5)

$$\partial\Omega \cap B_{\rho}(x_0) = \left\{ y \in B_{\rho}(x_0) : y_n = \gamma(y') \right\}.$$

$$(4.4.6)$$

Due to $\rho < a$, the set $\Omega \cap B_{\rho}(x_0)$ does not intersect $\{\mathbb{D} \ge 0\}$. Let us check that

$$\Omega \cap B_{\frac{\rho}{2}}(x_0) + \varepsilon(L+1)e_n + B_{\varepsilon}(0) \subset \Omega \cap B_{\rho}(x_0)$$
(4.4.7)

for

$$\varepsilon \leqslant \frac{\rho}{2(L+2)},\tag{4.4.8}$$

where L is a Lipschitz constant for γ in

$$B^{n-1}_{\rho}(x'_0) := \left\{ z' \in \mathbb{R}^{n-1}: \ |z' - x'_0| < \rho \right\}.$$

Let $y \in \Omega \cap B_{\frac{\rho}{2}}(x_0)$ and $z \in B_{\varepsilon}(\varepsilon(L+1)e_n) = \varepsilon(L+1)e_n + B_{\varepsilon}(0)$. Due to (4.4.8), it holds that

$$|y+z-x_0| \leqslant |y-x_0| + |z-\varepsilon(L+1)e_n| + \varepsilon(L+1)$$

As $z \in B_{\varepsilon}(\varepsilon(L+1)e_n)$ implies that $|z'| < \varepsilon$ and $z_n > \varepsilon(L+1) - \varepsilon = \varepsilon L$, the Lipschitz continuity of γ and (4.4.5) together imply

$$(y' + z') \leq \gamma(y') + L|z'|$$

$$< y_n + \varepsilon L$$

$$< y_n + z_n - \varepsilon L + \varepsilon L$$

$$= y_n + z_n.$$
(4.4.10)

Combining (4.4.5), (4.4.9), and (4.4.10), we arrive at (4.4.7).

 γ

Due to compactness of $\overline{\Omega}$, we can find some constants $\rho \in (0, a)$, L > 0, and $K \in \mathbb{N}$ and points $x_k \in \partial\Omega$ and $z_k \in S_1(0)$, $k \in \{1, \ldots, K\}$, such that

$$\Omega \cap B_{\frac{\rho}{2}}(x_k) + B_{\varepsilon}(\varepsilon(L+1)z_k) \subset \Omega \cap B_{\rho}(x_k) \quad \text{for } k \in \{1, \dots, K\}, \ \varepsilon \in \left(0, \frac{\rho}{2(L+2)}\right)$$
(4.4.11)

and

$$\partial\Omega\subset\bigcup_{k=1}^KB_{\frac{\rho}{2}(x_k)}$$

Let

$$A_0 := \overline{\Omega} \setminus \left(\bigcup_{n=1}^K B_{\frac{\rho}{2}(x_k)} \right).$$

This set is compact and satisfies $\rho_0 := \operatorname{dist}(A_0, \partial \Omega) > 0$. We set

$$\varepsilon_1 := \min\left\{\frac{\rho}{2(L+2)}, \frac{\rho_0}{3}\right\}.$$

By Lemma A.3.10, there is a partition of unity $\{\psi_k\}_{k=0}^K$ subordinate to the open covering

$$O_{\frac{\rho_0}{2}}(A_0), \ \{B_{\frac{\rho}{2}}(x_k)\}_{k=1}^K$$
(4.4.12)

of $\overline{\Omega}$, i.e., a set of functions that satisfies

$$\psi_0 \in C_c^{\infty}(O_{\frac{\rho_0}{2}}(A_0)),$$
 (4.4.13a)

$$\psi_k \in C_c^{\infty}(B_{\frac{\rho}{2}}(x_k)) \qquad \text{for } k \in \{1, \dots, K\}, \tag{4.4.13b}$$

$$0 \leqslant \psi_k \leqslant 1 \qquad \qquad \text{for } k \in \{0, \dots, K\}, \tag{4.4.13c}$$

$$\sum_{k=0}^{K} \psi_k = 1 \qquad \text{in } \overline{\Omega}. \tag{4.4.13d}$$

Step 2. Now we can proceed with the construction of approximations for \mathbb{D} . For $\varepsilon \in (0, \varepsilon_1)$ and $x \in \overline{\Omega}$ we set

$$\mathbb{D}_{\varepsilon}(x) := \varepsilon I_n + \psi_0(x)(\varsigma_{\varepsilon} * \mathbb{D})(x) + \sum_{k=1}^K \psi_k(x)(\varsigma_{\varepsilon} * \mathbb{D})(x + \varepsilon(L+1)z_k)$$
(4.4.14)

$$=\varepsilon I_n + \psi_0(x) \int_{B_{\varepsilon}(0)} \mathbb{D}(x-y)\varsigma_{\varepsilon}(y) \,\mathrm{d}y + \sum_{k=1}^K \psi_k(x) \int_{B_{\varepsilon}(0)} \mathbb{D}(x+\varepsilon(L+1)z_k-y)\varsigma_{\varepsilon}(y) \,\mathrm{d}y.$$

We set $\mathbb{D}_{\varepsilon} =: ((d_{ij})_{\varepsilon})_{i,j=1,\dots,n}$. Obviously, $\mathbb{D}_{\varepsilon} \in C^{\infty}(\overline{\Omega}, \mathbb{R}^{n \times n})$, is symmetric in $\overline{\Omega}$, and $\mathbb{D}_{\varepsilon} \ge \varepsilon$. We estimate for $w \in \mathbb{R}^n$ and $x \in \overline{\Omega}$ that

$$\begin{split} |\mathbb{D}_{\varepsilon}(x)w| \leqslant \varepsilon |w| + \psi_{0}(x) \int_{B_{\varepsilon}(0)} |\mathbb{D}(x-y)|_{2} |w|\varsigma_{\varepsilon}(y) \, \mathrm{d}y \\ &+ \sum_{k=1}^{K} \psi_{k}(x) \int_{B_{\varepsilon}(0)} |\mathbb{D}(x+\varepsilon(L+1)z_{k}-y)|_{2} |w|\varsigma_{\varepsilon}(y) \, \mathrm{d}y \\ \leqslant \varepsilon |w| + \sum_{k=0}^{K} \psi_{k}(x) ||\mathbb{D}||_{(L^{\infty}(\Omega))^{n \times n}} |w| \int_{B_{\varepsilon}(0)} \varsigma_{\varepsilon}(y) \, \mathrm{d}y \end{split}$$

and taking the supremum over |w| = 1 and $x \in \overline{\Omega}$ obtain (4.4.1).

Further, thanks to (4.4.11) we can exploit uniform continuity of \mathbb{D} in $\overline{\Omega}$, yielding

$$\frac{\max_{x\in\overline{\Omega\cap B_{\frac{\rho}{2}(x_k)}}} |(\varsigma_{\varepsilon} * d_{ij})(x+\varepsilon(L+1)z_k) - d_{ij}(x)| \qquad (4.4.15)$$

$$= \max_{x\in\overline{\Omega\cap B_{\frac{\rho}{2}(x_k)}}} \left| \int_{B_{\varepsilon}(0)} (d_{ij}(x+\varepsilon(L+1)z_k-y) - d_{ij}(x))\varsigma_{\varepsilon}(y) dy \right| \qquad (4.4.16)$$

$$\leq \max_{(x,y)\in\overline{\Omega\cap B_{\frac{\rho}{2}(x_k)}} \times \overline{B_{\varepsilon}(0)}} |(d_{ij}(x+\varepsilon(L+1)z_k-y) - d_{ij}(x))| \xrightarrow{\varepsilon \to 0} 0 \qquad (4.4.16)$$

for $k \in \{1, \ldots, K\}$ and $i, j \in \{1, \ldots, n\}$. Since $\overline{O_{\frac{\rho_0}{2}}(A_0)} \subset \Omega$, due to Lemma 2.3.4 we also have

$$\varsigma_{\varepsilon} * d_{ij} \underset{\varepsilon \to 0}{\longrightarrow} d_{ij} \quad \text{in } C\left(\overline{O_{\frac{\rho_0}{2}}(A_0)}\right).$$
(4.4.17)

Combining (4.4.13), (4.4.14), (4.4.16), and (4.4.17) and using the equivalence of the $|\cdot|_2$ and $|\cdot|_{\infty}$ matrix norms, we arrive at

 $\mathbb{D}_{\varepsilon} \underset{\varepsilon \to 0}{\to} \mathbb{D} \quad \text{in } C(\overline{\Omega}; \mathbb{R}^{n \times n}).$

Step 3. It remains to verify (4.4.4). Let $B \subset \subset \{D > 0\}$ relatively open. We set

$$\varepsilon_2(B) := \min\left\{\frac{\rho}{2(L+3)}, \frac{\rho_0}{4}, \frac{1}{2(L+2)} \operatorname{dist}(\overline{B}, \{\mathbb{D} \ge 0\})\right\} < \varepsilon_1.$$

Assume

$$B \subset B_{\frac{\rho}{2}}(x_k) \tag{4.4.18}$$

for some $k \in \{1, \ldots, K\}$. Then,

$$\overline{B} + \overline{B_{\varepsilon}(0)} + \varepsilon(L+1)z_k \subset \overline{O_{\varepsilon(L+2)}(B)} \cap \overline{\Omega} \subset \{\mathbb{D} > 0\} \cap B_{\rho}(x_k) \quad \text{for } \varepsilon \in (0, \varepsilon_2(B)).$$
(4.4.19a)

Let $\varphi \in C_c^{\infty}(B \setminus \partial \Omega)$. By the definition of weak divergence we have

$$\int_{\mathbb{R}^n} \nabla \cdot (\varsigma_{\varepsilon} * \mathbb{D})(x + \varepsilon(L+1)z_k)\varphi \, \mathrm{d}x = -\int_{\mathbb{R}^n} (\varsigma_{\varepsilon} * \mathbb{D})(x + \varepsilon(L+1)z_k)\nabla\varphi(x) \, \mathrm{d}x.$$

Now, using integration by substitution and the symmetry of ς_ε we compute

$$\int_{\mathbb{R}^n} (\varsigma_{\varepsilon} * \mathbb{D})(x + \varepsilon(L+1)z_k) \nabla \varphi(x) \, \mathrm{d}x = \int_{\mathbb{R}^n} \int_{B_{\varepsilon}(0)} \mathbb{D}(x + \varepsilon(L+1)z_k - y)\varsigma_{\varepsilon}(y) \nabla \varphi(x) \, \mathrm{d}y \, \mathrm{d}x$$

$$= \int_{\mathbb{R}^n} \mathbb{D}(x) \int_{B_{\varepsilon}(0)} \varsigma_{\varepsilon}(y) \nabla \varphi(x - \varepsilon(L+1)z_k - y) \, \mathrm{d}y \, \mathrm{d}x$$
$$= \int_{\mathbb{R}^n} \mathbb{D}(x) (\varsigma_{\varepsilon} * \nabla \varphi) (x - \varepsilon(L+1)z_k) \, \mathrm{d}x$$
$$= \int_{\mathbb{R}^n} \mathbb{D}(x) \nabla ((\varsigma_{\varepsilon} * \varphi) (x - \varepsilon(L+1)z_k)) \, \mathrm{d}x.$$

Again by the definition of weak divergence and (4.4.19a) we obtain

$$\int_{\mathbb{R}^n} \mathbb{D}(x) \nabla ((\varsigma_{\varepsilon} * \varphi)(x - \varepsilon(L+1)z_k)) \, \mathrm{d}x = -\int_{\mathbb{R}^n} \nabla \cdot \mathbb{D}(x)(\varsigma_{\varepsilon} * \varphi)(x - \varepsilon(L+1)z_k) \, \mathrm{d}x$$
$$= -\int_{O_{\varepsilon(L+2)}(B) \cap \Omega} \nabla \cdot \mathbb{D}(x)(\varsigma_{\varepsilon} * \varphi)(x - \varepsilon(L+1)z_k) \, \mathrm{d}x$$

so that

$$\left| \int_{\mathbb{R}^{n}} \nabla \cdot (\varsigma_{\varepsilon} * \mathbb{D}) (x + \varepsilon (L+1) z_{k}) \varphi \, \mathrm{d}x \right|_{\infty} \leq \| \nabla \cdot \mathbb{D} \|_{(L^{\infty}(O_{\varepsilon(L+2)}(B) \cap \Omega))^{n}} \| \varphi \|_{L^{1}(B)} \leq \| \nabla \cdot \mathbb{D} \|_{(L^{\infty}(O_{\varepsilon_{2}(B)(L+2)}(B) \cap \Omega))^{n}} \| \varphi \|_{L^{1}(B)}.$$
(4.4.20)

Consequently, due to the density of $C_c^{\infty}(B \setminus \partial \Omega)$ in $L^1(B \setminus \partial \Omega) = L^1(B)$ and as $(L^1(B))^* = L^{\infty}(B)$, the estimate

$$\|\nabla \cdot (\varsigma_{\varepsilon} * \mathbb{D})(\cdot + \varepsilon(L+1)z_k))\|_{(L^{\infty}(B))^n} \leq \sqrt{n} \|\nabla \cdot \mathbb{D}\|_{(L^{\infty}(O_{\varepsilon_2(B)(L+2)}(B) \cap \Omega))^n}$$
(4.4.21)

for $\varepsilon \in (0, \varepsilon_2(B))$ follows.

On the other hand, if $B \subset \subset \{\mathbb{D} > 0\}$ and $B \subset O_{\frac{\rho_0}{2}}(A_0)$, then

$$\overline{B} + \overline{B_{\varepsilon}(0)} \subset \overline{O_{\varepsilon}(B)} \subset \{\mathbb{D} > 0\} \cap O_{\frac{5\rho_0}{6}}(A_0) \text{ for } \varepsilon \in (0, \varepsilon_2(B)).$$

A similar argument as above with $\varphi \in C_c^{\infty}(B)$ works for $\varsigma_{\varepsilon} * \mathbb{D}$, yielding

$$\|\nabla \cdot (\varsigma_{\varepsilon} * \mathbb{D})\|_{(L^{\infty}(B))^{n}} \leq \sqrt{n} \|\nabla \cdot \mathbb{D}\|_{(L^{\infty}(O_{\varepsilon_{2}(B)}(B)))^{n}}$$

$$(4.4.22)$$

for $\varepsilon \in (0, \varepsilon_2(B))$.

Now, consider an arbitrary $B \subset \{\mathbb{D} > 0\}$. Then, $B = \bigcup_{k=1}^{K} \left(B \cap B_{\frac{\rho}{2}}(x_k)\right) \bigcup \left(B \cap O_{\frac{\rho_0}{2}}(A_0)\right)$. Combining (4.4.1), (4.4.13), (4.4.14), (4.4.21), and (4.4.22) and using the product rule we obtain

$$\begin{aligned} \|\nabla \cdot \mathbb{D}_{\varepsilon}\|_{(L^{\infty}(B))^{n}} &\leqslant \sqrt{n} \|\nabla \cdot D\|_{(L^{\infty}(O_{\varepsilon_{2}(B \cap O_{\frac{\rho_{0}}{2}}(A_{0}))}(B \cap O_{\frac{\rho_{0}}{2}}(A_{0}))))^{n}} \\ &+ \sqrt{n} \sum_{k=1}^{K} \|\nabla \cdot \mathbb{D}\|_{(L^{\infty}(O_{\varepsilon_{2}(B \cap B_{\frac{\rho}{2}}(x_{k}))(L+2)}(B \cap B_{\frac{\rho}{2}}(x_{k})) \cap \Omega))^{n}} + C_{5} =: C_{4}(B) \end{aligned}$$

for $\varepsilon \in (0, \varepsilon_2(B))$.

4.4.2 Existence of a global classical solution to the approximate problem

Let $\alpha \in (0, \frac{1}{2})$. Since $C_c^{\infty}(\Omega)$ is dense in $L^2(\Omega)$, there is a sequence $(c_{0\varepsilon})_{\varepsilon \in (0,\varepsilon_1)} \subset C^{2+\alpha}(\overline{\Omega})$ and a constant C_6 satisfying

$$(\nabla \cdot (\mathbb{D}_{\varepsilon} c_{0\varepsilon}) - c_{0\varepsilon} \mathcal{A} c_{0\varepsilon}) \cdot \nu = 0 \quad \text{on } \partial\Omega,$$

$$c_{0\varepsilon} \underset{\varepsilon \to 0}{\longrightarrow} c_0 \qquad \qquad \text{in } L^2(\Omega), \qquad (4.4.23)$$

$$\|c_{0\varepsilon}\|_{L^2(\Omega)} \leq C_6$$
 for all $\varepsilon \in (0, \varepsilon_1)$.

With the help of the diffusion tensors constructed in *Subsection 4.4.1* we formulate for $\varepsilon \in (0, \varepsilon_1)$ the approximate problems

$$\partial_t c_{\varepsilon} = \nabla \nabla : (\mathbb{D}_{\varepsilon} c_{\varepsilon}) - \nabla \cdot (c_{\varepsilon} \mathcal{A} c_{\varepsilon}) + \mu c_{\varepsilon} (1 - c_{\varepsilon}^{r-1}) \quad \text{in } \Omega \times [0, \infty), \tag{4.4.24a}$$

$$(\nabla \cdot (\mathbb{D}_{\varepsilon}c_{\varepsilon}) - c_{\varepsilon}\mathcal{A}c_{\varepsilon}) \cdot \nu = 0 \qquad \text{on } \partial\Omega \times [0,\infty), \qquad (4.4.24b)$$

$$c_{\varepsilon}(\cdot, 0) = c_{0\varepsilon} \qquad \qquad \text{in } \Omega. \tag{4.4.24c}$$

The subsequent Lemmas 4.4.2 and 4.4.3 provide local and then global existence of classical solutions to (4.4.24).

Lemma 4.4.2. For $\varepsilon \in (0, \varepsilon_1)$ there is a maximal existence time $T_{max,\varepsilon} \in (0, \infty]$ and a nonnegative classical solution $c_{\varepsilon} \in C^{2+\alpha,1+\frac{\alpha}{2}}(\overline{\Omega} \times [0, T_{max,\varepsilon}))$ of (4.4.24). It holds that either $T_{max,\varepsilon} = \infty$ or $T_{max,\varepsilon} < \infty$ and

$$\lim_{t \nearrow T_{max,\varepsilon}} \|c_{\varepsilon}(\cdot,t)\|_{C^{2+\alpha}(\overline{\Omega})} = \infty.$$
(4.4.25)

Proof. The proof is based on a standard fixed-point argument.

Let $\varepsilon \in (0, \varepsilon_1)$ and $\alpha \in (0, \frac{1}{2})$, as above, $T \in (0, 1)$ small enough (to be determined later), and

$$M := \|c_{0\varepsilon}\|_{C^{1+\alpha}(\overline{\Omega})} + 1.$$

We define the set

$$S:=\left\{\overline{c}\in C^{1+\alpha,\frac{1+\alpha}{2}}(\overline{\Omega}\times[0,T]):\,\overline{c}\geqslant 0,\,\|\overline{c}\|_{C^{1+\alpha,\frac{1+\alpha}{2}}(\overline{\Omega}\times(0,T))}\leqslant M\right\}.$$

For $\overline{c} \in S$ we consider the linearised IBVP

$$\partial_t c_{\varepsilon} = \nabla \nabla : (\mathbb{D}_{\varepsilon} c_{\varepsilon}) - \nabla \cdot (c_{\varepsilon} \mathcal{A} \overline{c}) + \mu c_{\varepsilon} (1 - \overline{c}^{r-1}) \qquad \text{in } \Omega \times [0, \infty), \tag{4.4.26a}$$

$$(\nabla \cdot (\mathbb{D}_{\varepsilon} c_{\varepsilon}) - c_{\varepsilon} \mathcal{A}\overline{c}) \cdot \nu = 0 \qquad \text{in } \partial\Omega \times [0, \infty), \qquad (4.4.26b)$$

$$c_{\varepsilon} = c_{0\varepsilon} \qquad \qquad \text{in } \Omega \times \{0\}. \tag{4.4.26c}$$

Due to Lemmas 2.2.3, 4.2.2, and 4.4.1, the coefficient functions and the initial data are smooth enough and satisfy the compatibility condition, allowing us to conclude from Theorem A.1.6 with

$$\begin{aligned} a_{ij} &:= (d_{ij})_{\varepsilon}, \\ a_i &:= -\sum_{j=1}^n ((d_{ij})_{\varepsilon} + (d_{ji})_{\varepsilon})_{x_j} + (\mathcal{A}\overline{c})_i, \\ a &:= -\sum_{i,j=1}^n ((d_{ij})_{\varepsilon})_{x_j x_i} + \nabla \cdot (\mathcal{A}\overline{c}) - \mu(1 - \overline{c}^{r-1}), \\ b_i &:= \sum_{j=1}^n (d_{ji})_{\varepsilon} \nu_j, \\ b &:= \sum_{i,j=1}^n ((d_{ij})_{\varepsilon})_{x_j} \nu_i - (\mathcal{A}\overline{c}) \cdot \nu, \\ f &:= 0 \end{aligned}$$

for i, j = 1, ..., n that (4.4.26) has a unique solution $c_{\varepsilon} \in C^{2+\alpha, 1+\frac{\alpha}{2}}(\overline{\Omega} \times [0, T])$ satisfying

$$\|c_{\varepsilon}\|_{C^{2+\alpha,1+\frac{\alpha}{2}}(\overline{\Omega}\times[0,T])} \leqslant C_{7}(M)\|c_{0\varepsilon}\|_{C^{2+\alpha}(\overline{\Omega})} =: C_{8}$$

for some constant $C_7(M) > 0$ independent of the choice of \overline{c} . From Theorem A.1.11 we conclude that $c_{\varepsilon} \ge 0$ in $\overline{\Omega} \times [0, T]$. Lemma 2.2.9(ii) implies the estimate

$$\begin{split} \|c_{\varepsilon}\|_{C^{1+\alpha,\frac{1+\alpha}{2}}(\overline{\Omega}\times[0,T])} &\leqslant \|c_{\varepsilon} - c_{0\varepsilon}\|_{C^{1+\alpha,\frac{1+\alpha}{2}}(\overline{\Omega}\times[0,T])} + \|c_{0\varepsilon}\|_{C^{1+\alpha}(\overline{\Omega})} \\ &\leqslant 2\max\{1, K_{I}(\alpha)\}T^{\frac{\alpha}{2}}\|c_{\varepsilon}\|_{C^{2+\alpha,1+\frac{\alpha}{2}}(\overline{\Omega}\times[0,T])} + \|c_{0\varepsilon}\|_{C^{1+\alpha}(\overline{\Omega})} \\ &\leqslant 2\max\{1, K_{I}(\alpha)\}T^{\frac{\alpha}{2}}C_{8} + \|c_{0\varepsilon}\|_{C^{1+\alpha}(\overline{\Omega})}, \end{split}$$

where $K_I(\alpha)$ denotes the embedding constant from $W^1_{\infty}(\Omega)$ into $C^{\alpha}(\overline{\Omega})$ from Lemma 2.2.8(ii). Hence, taking

$$T \leqslant \left(\frac{1}{2\max\{1, K_I(\alpha)\}C_8}\right)^{\frac{2}{\alpha}}$$

we ensure that $c_{\varepsilon} \in S$. Consequently, the operator

$$F: S \to S, \quad \bar{c} \mapsto c_{\varepsilon}$$

$$(4.4.27)$$

is well-defined and, due to $C^{2+\alpha,1+\frac{\alpha}{2}}(\overline{\Omega}\times[0,T]) \hookrightarrow G^{1+\alpha,\frac{1+\alpha}{2}}(\overline{\Omega}\times[0,T])$, a compact self-map. Moreover, the continuous dependence of c_{ε} from the coefficients (that follows with Lemma 4.2.2 and Theorem A.1.6) implies that F is a continuous and compact operator. Now, Schauder's fixed-point theorem (Theorem A.2.2) applies, providing a fixed-point $c_{\varepsilon} \in C^{1+\alpha,\frac{1+\alpha}{2}}(\overline{\Omega}\times[0,T])$ of F that is also in $C^{2+\alpha,1+\frac{\alpha}{2}}(\overline{\Omega}\times[0,T])$ and a classical solution to (4.4.24) on $\overline{\Omega}\times[0,T]$.

Extending the solution to its maximal existence time $T_{max,\varepsilon}$, it holds that either $T_{max,\varepsilon} = \infty$ or $T_{max,\varepsilon} < \infty$ and (4.4.25).

Next, we verify global existence for (4.4.24).

Lemma 4.4.3. For $\varepsilon \in (0, \varepsilon_1)$ there is a nonnegative classical solution $c_{\varepsilon} \in C^{2+\alpha, 1+\frac{\alpha}{2}}(\overline{\Omega} \times [0, \infty))$ of (4.4.24).

Proof. We follow a standard approach which is based on excluding the possibility of (4.4.25). Let $\varepsilon \in (0, \varepsilon_1)$ and assume $T_{max,\varepsilon} < \infty$, so that c_{ε} is a solution to (4.4.24) in $\overline{\Omega} \times [0, T_{max,\varepsilon})$. Integrating the first equation in (4.4.24) over Ω and using partial integration, we conclude from Gronwall's inequality that

$$\|c_{\varepsilon}\|_{L^{\infty}(0,T_{max,\varepsilon};L^{1}(\Omega))} \leqslant e^{\mu T_{max,\varepsilon}} \|c_{0\varepsilon}\|_{L^{1}(\Omega)} =: C_{9}(T_{max,\varepsilon}).$$

$$(4.4.28)$$

Consider an arbitrary $p \in (1, \infty)$. We multiply (4.4.24a) by pc_{ε}^{p-1} , integrate over Ω and use partial integration to obtain

$$\begin{split} \frac{d}{dt} \int_{\Omega} c_{\varepsilon}^{p} \, \mathrm{d}x &= -p \int_{\Omega} \left(\mathbb{D}_{\varepsilon} \nabla c_{\varepsilon} + \nabla \cdot \mathbb{D}_{\varepsilon} c_{\varepsilon} - c_{\varepsilon} \mathcal{A} c_{\varepsilon} \right) \cdot \nabla c_{\varepsilon}^{p-1} \, \mathrm{d}x + \mu p \int_{\Omega} c_{\varepsilon}^{p} (1 - c_{\varepsilon}^{r-1}) \, \mathrm{d}x \\ &= -p(p-1) \int_{\Omega} c_{\varepsilon}^{p-2} (\mathbb{D}_{\varepsilon} \nabla c_{\varepsilon}) \cdot \nabla c_{\varepsilon} + c_{\varepsilon}^{p-1} (\nabla \cdot \mathbb{D}_{\varepsilon} - \mathcal{A} c_{\varepsilon}) \cdot \nabla c_{\varepsilon} \, \mathrm{d}x \\ &+ \mu p \int_{\Omega} c_{\varepsilon}^{p} (1 - c_{\varepsilon}^{r-1}) \, \mathrm{d}x. \end{split}$$

Due to $\mathbb{D}_{\varepsilon} \ge \varepsilon$ and the boundedness of its divergence, the fact that $\nabla c_{\varepsilon}^{\frac{p}{2}} = \frac{p}{2} c_{\varepsilon}^{\frac{p}{2}-1} \nabla c_{\varepsilon}$, (4.4.28) and Young's inequality we conclude that

$$\begin{split} & \frac{d}{dt} \int_{\Omega} c_{\varepsilon}^{p} \, \mathrm{d}x + \frac{4(p-1)}{p} \varepsilon \int_{\Omega} |\nabla c_{\varepsilon}^{\frac{p}{2}}|^{2} \, \mathrm{d}x \\ \leqslant & 2(p-1) \int_{\Omega} |\nabla c_{\varepsilon}^{\frac{p}{2}}| c_{\varepsilon}^{\frac{p}{2}} \left(\|\nabla H\|_{(L^{\infty}(B_{1}))^{n}} \|c_{\varepsilon}\|_{L^{1}(\Omega)} + \|\nabla \cdot \mathbb{D}_{\varepsilon}\|_{(L^{\infty}(\Omega))^{n}} \right) \, \mathrm{d}x + \mu p \int_{\Omega} c_{\varepsilon}^{p} \, \mathrm{d}x \\ \leqslant & \frac{4(p-1)}{p} \varepsilon \int_{\Omega} |\nabla c_{\varepsilon}^{\frac{p}{2}}|^{2} \, \mathrm{d}x + p C_{10}(p, T_{max,\varepsilon}) \int_{\Omega} c_{\varepsilon}^{p} \, \mathrm{d}x, \end{split}$$

where

$$C_{10}(p,T_{max,\varepsilon}) := \frac{p-1}{\varepsilon} \left(\|\nabla H\|_{(L^{\infty}(B_1))^n}^2 C_9(T_{max,\varepsilon})^2 + \|\nabla \cdot \mathbb{D}_{\varepsilon}\|_{(L^{\infty}(\Omega))^n}^2 \right) + \mu.$$

Hence, it follows again from Gronwall's inequality that

$$\|c_\varepsilon\|_{L^\infty(0,T_{max,\varepsilon};L^p(\Omega))}\leqslant e^{pC_{10}(p,T_{max,\varepsilon})T_{max,\varepsilon}}\|c_{0\varepsilon}\|_{L^p(\Omega)}.$$

The coefficients of (4.4.24) are regular enough s.t. Theorems A.1.1 and A.1.4 and Remark A.1.2 with

$$a_{ij} = (d_{ij})_{\varepsilon}, a_i = \sum_{j=1}^n ((d_{ij})_{\varepsilon})_{x_j} - (\mathcal{A}c_{\varepsilon})_i, a = \mu(c_{\varepsilon}^{r-1} - 1), f = 0$$

for i, j = 1, ..., n imply that

$$c_{\varepsilon} \in L^{2}(0, T_{max,\varepsilon}; H^{1}(\Omega)) \cap C([0, T_{max,\varepsilon}]; L^{2}(\Omega)) \cap L^{\infty}(\Omega \times (0, T_{max,\varepsilon}))$$

and uniquely solves (4.4.24) on $\Omega \times (0, T_{max,\varepsilon})$ in the weak sense from *Remark A.1.2*. Now, Theorem A.1.12 with

$$a(x, t, c, \nabla c) = \mathbb{D}_{\varepsilon} \nabla c + \nabla \cdot \mathbb{D}_{\varepsilon} c - c\mathcal{A}c_{\varepsilon},$$

$$b(x, t, c) = \mu c(c^{r-1} - 1)$$
(4.4.29)

applies and yields

$$c_{\varepsilon} \in C^{\gamma, \frac{\gamma}{2}}(\overline{\Omega} \times [0, T_{max, \varepsilon}])$$

for some $\gamma \in (0, \alpha)$. Next, using the Hölder continuity of the coefficients (which holds especially due to *Lemmas 2.2.3, 4.2.2,* and 4.4.1) again and the fact that the initial data are smooth enough and satisfy the compatibility condition, *Theorem A.1.13* with the coefficients from (4.4.29) implies that there are $C_{11}(\varepsilon) > 0$ and $\delta \in (0, \gamma)$ s.t.

$$\|c_{\varepsilon}\|_{C^{1+\delta,\frac{1+\delta}{2}}(\overline{\Omega}\times[0,T_{max,\varepsilon}])} \leqslant C_{11}(\varepsilon).$$

Hence, as above Theorem A.1.6 implies that

$$c_{\varepsilon} \in C^{2+\delta,1+\frac{\delta}{2}}(\overline{\Omega} \times [0,T_{max,\varepsilon}]) \hookrightarrow C^{1+\alpha,\frac{1+\alpha}{2}}(\overline{\Omega} \times [0,T_{max,\varepsilon}]).$$

Finally, applying *Theorem A.1.6* one more time we obtain

$$c_{\varepsilon} \in C^{2+\alpha,1+\frac{\alpha}{2}}(\overline{\Omega} \times [0,T_{max,\varepsilon}])$$

contradicting (4.4.25).

4.5 Existence of a very weak solution to the original problem

In this section, we show the convergence of a suitably chosen sequence of the classical solutions to (4.4.24) to a very weak solution to (4.1.1) in the sense of *Definition 4.3.3*. We start with some basic uniform estimates of (c_{ε}) .

Lemma 4.5.1. For all $\varepsilon \in (0, \varepsilon_1)$ it holds that

$$\|c_{\varepsilon}\|_{L^{\infty}(0,\infty;L^{1}(\Omega))} \leqslant C_{12},\tag{4.5.1}$$

$$\|c_{\varepsilon}\|_{L^{r}(\Omega\times(0,T))} \leq C_{13} + T\mu C_{12} \qquad for \ T > 0.$$
(4.5.2)

Proof. Consider a constant $C_{14} > 0$ and

$$C_{15} := \max_{x \in [0, \ r - \sqrt[r]{\frac{C_{14}}{r\mu} + \frac{1}{r}}]} |\mu x^r - (C_{14} + \mu)x|.$$

Then, $C_{15} - C_{14}x - (\mu x - \mu x^r) \ge 0$ holds for $x \ge 0$. Integrating (4.4.24a) over Ω , by parts where necessary, using the boundary condition, the assumption $r \ge 2$, and the boundedness of Ω , we obtain

$$\frac{d}{dt} \int_{\Omega} c_{\varepsilon} \, \mathrm{d}x = \mu \int_{\Omega} c_{\varepsilon} - c_{\varepsilon}^{r} \, \mathrm{d}x \qquad (4.5.3)$$
$$\leqslant C_{15} |\Omega| - C_{14} \int_{\Omega} c_{\varepsilon} \, \mathrm{d}x.$$

Consequently, Lemma A.1.20 implies (4.5.1). Integrating (4.5.3) over (0,T) and using (4.5.1) immediately yields (4.5.2).

Next, we establish some uniform estimates for derivatives of c_{ε} . For the spatial gradient, the estimates hold away from the degeneracy set of \mathbb{D} .

Lemma 4.5.2. For any T > 0 and relatively open $B \subset \subset \{\mathbb{D} > 0\}$ there is a constant $C_{16}(B,T)$ s.t.

$$\|c_{\varepsilon}\|_{L^{2}(0,T;H^{1}(B))} \leq C_{16}(B,T), \qquad (4.5.4)$$

$$\|c_{\varepsilon}\|_{L^{r+1}(B\times(0,T))} \leqslant C_{16}(B,T) \tag{4.5.5}$$

for all $\varepsilon \in (0, \varepsilon_2(B))$.

Furthermore, let q be a number that satisfies

$$q \in \left(1, \frac{n}{n-1}\right). \tag{4.5.6}$$

Then, for any T > 0 there exists some constant $C_{17}(T) > 0$ s.t.

$$\|\partial_t c_{\varepsilon}\|_{L^1(0,T;W_q^{-2}(\Omega))} \le C_{17}(T)$$
(4.5.7)

for $\varepsilon \in (0, \varepsilon_1)$.

Proof. Let $T \in (0, \infty)$ and consider a relatively open set

$$B \subset \subset \{\mathbb{D} > 0\}.$$

Then, there exists a sufficiently small number $\alpha > 0$ such that

$$\hat{B} := O_{\alpha}(B) \subset \mathbb{R}^n \setminus \{\mathbb{D} \ge 0\}.$$

Due to the uniform continuity of \mathbb{D} and our choice of B,

$$\delta := \inf_{x \in \hat{B}} \min_{z \in S_1(0)} z^T \mathbb{D}(x) z > 0.$$

$$(4.5.8)$$

In what follows we leave out the dependence on δ and \hat{B} in the constants as they are determined by B. Let $\varphi \in C_c^{\infty}(\mathbb{R}^n \setminus \{D \ge 0\})$ be s.t.

$$\varphi \begin{cases} = 1 & \text{on } B, \\ \in [0,1] & \text{on } \hat{B} \backslash B, \\ = 0 & \text{on } \mathbb{R}^n \backslash \hat{B}. \end{cases}$$
(4.5.9)

Let $\varepsilon \in (0, \varepsilon_2(B))$, where $\varepsilon_2(B)$ is from Lemma 4.4.1. We multiply (4.4.24a) by $c_{\varepsilon}\varphi^2$, integrate over Ω by parts where necessary, use the no-flux boundary condition, and obtain

$$\frac{1}{2} \frac{d}{dt} \int_{\Omega} c_{\varepsilon}^{2} \varphi^{2} dx$$

$$= -\int_{\Omega} \left(\mathbb{D}_{\varepsilon} \nabla c_{\varepsilon} + \nabla \cdot \mathbb{D}_{\varepsilon} c_{\varepsilon} - c_{\varepsilon} \mathcal{A} c_{\varepsilon} \right) \cdot \left(\varphi^{2} \nabla c_{\varepsilon} + 2c_{\varepsilon} \varphi \nabla \varphi \right) dx + \mu \int_{\Omega} c_{\varepsilon}^{2} \varphi^{2} (1 - c_{\varepsilon}^{r-1}) dx$$

$$\leq -\int_{\Omega} (\nabla c_{\varepsilon})^{T} \mathbb{D}_{\varepsilon} \varphi^{2} \nabla c_{\varepsilon} dx + 2 \left| \int_{\Omega} (\varphi \nabla c_{\varepsilon})^{T} \mathbb{D}_{\varepsilon} (c_{\varepsilon} \nabla \varphi) dx \right|$$

$$+ \left| \int_{\Omega} (\nabla \cdot \mathbb{D}_{\varepsilon} - \mathcal{A} c_{\varepsilon}) \cdot \left(c_{\varepsilon} \varphi^{2} \nabla c_{\varepsilon} + 2c_{\varepsilon}^{2} \varphi \nabla \varphi \right) dx \right| + \mu \int_{\Omega} c_{\varepsilon}^{2} \varphi^{2} dx - \mu \int_{\Omega} c_{\varepsilon}^{r+1} \varphi^{2} dx. \quad (4.5.10)$$

The matrix \mathbb{D}_{ε} is symmetric and positive-definite. Hence, it defines a scalar product on \mathbb{R}^n , and with the Cauchy-Schwartz and Young's inequalities and (4.4.1) we obtain

$$2\left|\int_{\Omega} \left(\varphi \nabla c_{\varepsilon}\right)^{T} \mathbb{D}_{\varepsilon} \left(c_{\varepsilon} \nabla \varphi\right) \, \mathrm{d}x\right| \leq 2\sqrt{\int_{\Omega} \left(\varphi \nabla c_{\varepsilon}\right)^{T} \mathbb{D}_{\varepsilon} \varphi \nabla c_{\varepsilon} \, \mathrm{d}x} \sqrt{\int_{\Omega} \left(c_{\varepsilon} \nabla \varphi\right)^{T} \mathbb{D}_{\varepsilon} c_{\varepsilon} \nabla \varphi \, \mathrm{d}x} \\ \leq \frac{1}{4} \int_{\Omega} \left(\nabla c_{\varepsilon}\right)^{T} \mathbb{D}_{\varepsilon} \varphi^{2} \nabla c_{\varepsilon} \, \mathrm{d}x + 4 \|\nabla \varphi\|_{(L^{\infty}(\mathbb{R}^{n}))^{n}}^{2} C_{18} \int_{\Omega} c_{\varepsilon}^{2} \, \mathrm{d}x,$$

$$(4.5.11)$$

where $C_{18} := \varepsilon_1 + \|\mathbb{D}\|_{(L^{\infty}(\Omega))^{n \times n}}$ is some uniform upper bound on $\|\mathbb{D}_{\varepsilon}\|_{(L^{\infty}(\Omega))^{n \times n}}$ for $\varepsilon \in (0, \varepsilon_1)$ that exists due to (4.4.1). Moreover, combining (4.4.4), Lemma 4.5.1, and Hölder's and Young's inequalities, we can estimate

$$\begin{aligned} \left| \int_{\Omega} \left(\nabla \cdot \mathbb{D}_{\varepsilon} - \mathcal{A}c_{\varepsilon} \right) \cdot \left(c_{\varepsilon}\varphi^{2}\nabla c_{\varepsilon} + 2c_{\varepsilon}^{2}\varphi\nabla\varphi \right) \, \mathrm{d}x \right| \\ &\leq \left(\| \nabla \cdot \mathbb{D}_{\varepsilon} \|_{(L^{\infty}(\hat{B}))^{n}} + \| \nabla H \|_{(L^{\infty}(B_{1}))^{n}} \| c_{\varepsilon} \|_{L^{\infty}(0,\infty;L^{1}(\Omega))} \right) \int_{\Omega} \left| c_{\varepsilon}\varphi^{2}\nabla c_{\varepsilon} \right| + 2 \left| c_{\varepsilon}^{2}\varphi\nabla\varphi \right| \, \mathrm{d}x \\ &\leq \frac{\delta}{2} \int_{\Omega} \left| \varphi\nabla c_{\varepsilon} \right|^{2} \, \mathrm{d}x + C_{19}(B) \int_{\Omega} c_{\varepsilon}^{2} \, \mathrm{d}x. \end{aligned}$$

$$(4.5.12)$$

Combining (4.5.10)–(4.5.12) and rearranging the terms leads due to (4.5.8) and (4.5.9) to

$$\frac{1}{2}\frac{d}{dt}\int_{\Omega}c_{\varepsilon}^{2}\varphi^{2}\,\mathrm{d}x + \frac{\delta}{4}\int_{\Omega}|\nabla c_{\varepsilon}|^{2}\varphi^{2}\,\mathrm{d}x + \mu\int_{\Omega}c_{\varepsilon}^{r+1}\varphi^{2}\,\mathrm{d}x \leqslant C_{20}(B)\int_{\Omega}c_{\varepsilon}^{2}\,\mathrm{d}x.$$
(4.5.13)

We integrate (4.5.13) over (0,T) and obtain due to $r \ge 2$, Lemma 4.5.1, (4.5.9), the uniform boundedness of the initial values, and using Hölder's inequality that

$$\frac{1}{2} \|c_{\varepsilon}\varphi\|_{L^{\infty}(0,T;L^{2}(\Omega))}^{2} + \frac{\delta}{4} \|\varphi\nabla c_{\varepsilon}\|_{L^{2}(0,T;(L^{2}(\Omega))^{n})}^{2} + \mu \|c_{\varepsilon}\|_{L^{r+1}(B\times(0,T))}^{r+1} \leqslant C_{21}(B,T), \quad (4.5.14)$$

which yields (4.5.4) and (4.5.5).

Let $\psi \in W^{\circ}_{\frac{q}{q-1}}(\Omega)$. Due to our choice of q Lemma 2.2.8(*iii*) implies that

$$\psi \in W^{\overset{\circ}{2}}_{\frac{q}{q-1}}(\Omega) \subset \left\{ \psi \in C^{1}(\overline{\Omega}) : \psi = 0, \, \nabla \psi = 0 \text{ on } \partial \Omega \right\}.$$

$$(4.5.15)$$

Hence, multiplying (4.4.24a) by ψ and using partial integration once or twice where necessary we arrive at

$$\int_{\Omega} \partial_t c_{\varepsilon} \psi \, \mathrm{d}x = \int_{\Omega} c_{\varepsilon} \mathbb{D}_{\varepsilon} : D^2 \psi \, \mathrm{d}x + \int_{\Omega} c_{\varepsilon} (\mathcal{A}c_{\varepsilon}) \cdot \nabla \psi \, \mathrm{d}x + \mu \int_{\Omega} c_{\varepsilon} (1 - c_{\varepsilon}^{r-1}) \psi \, \mathrm{d}x.$$

Using Lemmas 2.1.2 and 2.2.8 and (4.3.1), (4.4.1), (4.5.1), (4.5.2), (4.5.6), and (4.5.15), Hölder's inequality and the embedding of Lebesgue spaces we obtain the estimate

$$\begin{split} \left| \int_{\Omega} \partial_t c_{\varepsilon} \psi \, \mathrm{d}x \right| &\leqslant \|\mathbb{D}_{\varepsilon}\|_{(L^{\infty}(\Omega))^{n \times n}} \sum_{i,j=1}^n \int_{\Omega} |c_{\varepsilon}| |\psi_{x_i x_j}| \, \mathrm{d}x + \|\nabla H\|_{(L^{\infty}(B_1))^n} \|c_{\varepsilon}\|_{L^1(\Omega)} \int_{\Omega} |c_{\varepsilon} \nabla \psi| \, \mathrm{d}x \\ &+ \mu \int_{\Omega} (c_{\varepsilon} + c_{\varepsilon}^r) ||\psi| \, \mathrm{d}x \\ &\leqslant C_{18} \sum_{i,j=1}^n \|\psi_{x_i x_j}\|_{L^{\frac{q}{q-1}}(\Omega)} \|c_{\varepsilon}\|_{L^q(\Omega)} + \|\nabla H\|_{(L^{\infty}(B_1))^n} C_{12}^2 \|\nabla \psi\|_{(L^{\infty}(\Omega))^n} \\ &+ \mu (C_{12} + \|c_{\varepsilon}\|_{L^r(\Omega)}^r) \|\psi\|_{L^{\infty}(\Omega)} \\ &\leqslant C_{22} \left(\|c_{\varepsilon}\|_{L^r(\Omega)}^r + \|c_{\varepsilon}\|_{L^r(\Omega)} + 1 \right) \|\psi\|_{W^{\frac{2}{q}}_{\frac{q}{q-1}}(\Omega)}. \end{split}$$

We conclude

$$\|\partial_t c_{\varepsilon}\|_{W_q^{-2}(\Omega)} \leq C_{22} \left(\|c_{\varepsilon}\|_{L^r(\Omega)}^r + \|c_{\varepsilon}\|_{L^r(\Omega)} + 1 \right)$$

$$(4.5.16)$$

Integration over (0, T) together with (4.5.2) yield (4.5.7).

With the obtained estimates at hand we can now proceed to establishing convergence.

Lemma 4.5.3. There exist $c \in L^{\infty}(0, \infty; L^1(\Omega))$ and a sequence $(\varepsilon_k) \subset (0, \varepsilon_1), \varepsilon_k \to 0, s.t.$

$$c_{\varepsilon_k} \xrightarrow[k \to \infty]{} c \qquad in \ L^1_{loc}(\overline{\Omega} \times [0, \infty)),$$

$$(4.5.17)$$

a.e. in
$$\Omega \times (0, \infty)$$
. (4.5.18)

Proof. Let T > 0, $q \in (1, \frac{n}{n-1})$ and consider a relatively open set $B \subset \{D > 0\}$. Thanks to estimates (4.5.4) and (4.5.7), the dense embeddings

$$H^1(B) \hookrightarrow \hookrightarrow L^2(B) \hookrightarrow W_q^{-2}(\Omega),$$

where the latter holds due to our choice of q, and the Lions-Aubin lemma (*Lemma A.3.9*), every subsequence $(c_{\varepsilon_j})_{j\in\mathbb{N}}$ has a subsequence that converges in $L^2(B \times (0,T))$, and it can be chosen such that it converges a.e. in $B \times (0,T)$. Observe that since $\{\mathbb{D} > 0\}$ is relatively open in the compact set $\overline{\Omega}$, there exists a sequence $(B_i)_{i\in\mathbb{N}}$ with $B_i \subset B_{i+1}$ of relatively open sets such that $B_i \subset \subset \{\mathbb{D} > 0\}$ and $\{\mathbb{D} > 0\} = \bigcup_{i=1}^{\infty} B_i$. In view of this, we have

$$\{\mathbb{D} > 0\} \times (0, \infty) = \bigcup_{i=1}^{\infty} B_i \times (0, i).$$

Together with a diagonal argument this description in the form of a countable union allows to conclude from the above that there exist some $c \in L^2_{loc}(\{\mathbb{D} > 0\} \times [0, \infty))$ and a sequence $(\varepsilon_k) \subset (0, \varepsilon_1)$ that converges to zero and is such that

$$c_{\varepsilon_k} \xrightarrow[k \to \infty]{} c$$
 a.e. in $\{\mathbb{D} > 0\} \times (0, \infty)$. (4.5.19)

Since by (4.3.2f) the degeneracy set $\{\mathbb{D} \ge 0\}$ has the *n*-dimensional Lebesgue measure zero, (4.5.19) is equivalent to

$$c_{\varepsilon_k} \xrightarrow[k \to \infty]{} c$$
 a.e. in $\Omega \times (0, \infty)$. (4.5.20)

Furthermore, due to (4.5.2) and $r \ge 2$, the sequence $(c_{\varepsilon_k})_{k \in \mathbb{N}}$ is uniformly integrable on $\Omega \times (0, T)$ for all T > 0 by the de la Vallée-Poussin theorem (*Theorem A.3.6*). Now

$$c_{\varepsilon_k} \xrightarrow[k \to \infty]{} c \text{ in } L^1(\Omega \times (0,T)) \quad \text{for all } T > 0$$

and $c \in L^1_{loc}(\overline{\Omega} \times [0, \infty))$ follow with (4.5.20) and Vitali's lemma (*Lemma A.3.7*). This implies that for a.e. $t \in (0, \infty)$ we have

$$c_{\varepsilon_k} \xrightarrow[k \to \infty]{} c \text{ in } L^1(\Omega).$$

Hence, $c \in L^{\infty}(0, \infty; L^1(\Omega))$ due to (4.5.1).

In preparation for the proof of existence of a very weak solution to (4.1.1) we still need one more lemma that allows us to handle the nonlinear part of the reaction term in (4.4.24).

Lemma 4.5.4. Let $(\varepsilon_k)_{k \in \mathbb{N}}$ be as in Lemma 4.5.3. Then, for all $T \in (0, \infty)$ it holds that

$$c^r_{\varepsilon_k} \xrightarrow[k \to \infty]{} c^r \qquad in \ L^1(\Omega \times (0,T))$$

$$(4.5.21)$$

and $c \in L^r_{loc}(\overline{\Omega} \times [0,\infty)).$

Proof. Let $T \in (0, \infty)$ and consider the sequence $(c_{\varepsilon_k})_{k \in \mathbb{N}}$ from Lemma 4.5.3. Fatou's lemma together with estimate (4.5.2) from Lemma 4.5.1 imply that

$$\int_0^T \int_\Omega c^r \, \mathrm{d}x \, \mathrm{d}t \leq \liminf_{k \to \infty} \int_0^T \int_\Omega c^r_{\varepsilon_k} \, \mathrm{d}x \, \mathrm{d}t \leq C_{23}(T),$$

and so $c \in L^r(\Omega \times (0,T))$.

Due to Lemma 4.2.4 and assumption (4.3.2f), there exists a family of functions $(\varphi_{\delta})_{\delta \in (0,1)} \subset C_c^{\infty}(\mathbb{R}^n; [0,1])$ satisfying (4.2.5) for $K = \{\mathbb{D} \ge 0\}$. We adopt the splitting

$$c_{\varepsilon_k}^r = c_{\varepsilon_k}^r (1 - \varphi_\delta) + c_{\varepsilon_k}^r \varphi_\delta \tag{4.5.22}$$

and next study the convergence of each of the terms separately.

Step 1 (Convergence of the first term). For $\delta \in (0, 1)$ we set

$$B_{\delta} := \{ \mathbb{D} > 0 \} \setminus \overline{O_{\delta \sqrt{n}}(\{ \mathbb{D} \ge 0 \})}$$

Obviously, B_{δ} is relatively open and satisfies $B_{\delta} \subset \{\mathbb{D} > 0\}$. Arguing similar to the proof of Lemma 4.5.3, we conclude with (4.5.5) in Lemma 4.5.2, (4.5.18) in Lemma 4.5.3, the de la Vallée-Poussin theorem and Vitali's lemma (Theorem A.3.6 and Lemma A.3.7) that

$$c_{\varepsilon_k}^r \xrightarrow[k \to \infty]{} c^r \quad \text{in } L^1(B_\delta \times (0,T)).$$
 (4.5.23)

Since $\varphi_{\delta} = 1$ outside of B_{δ} due to (4.2.5b), (4.5.23) yields

$$c_{\varepsilon_k}^r (1 - \varphi_{\delta}) \underset{k \to \infty}{\longrightarrow} c^r (1 - \varphi_{\delta}) \qquad \text{in } L^1(\Omega \times (0, T)) \text{ for all } \delta \in (0, 1).$$

$$(4.5.24)$$

Furthermore, the integrability of c^r together with (4.2.5f) and the uniform boundedness of (φ_{δ}) allow to conclude using the dominated convergence theorem that

$$c^r(1-\varphi_\delta) \underset{\delta \to 0}{\longrightarrow} c^r \quad \text{in } L^1(\Omega \times (0,T)).$$
 (4.5.25)

Step 2 (Convergence of the second term and conclusion). Due to (4.3.2e) and (4.2.5c), we have

$$\operatorname{supp}(\varphi_{\delta}) \subset O_{5\delta\sqrt{n}}(\{\mathbb{D} \ge 0\}) \subset \subset \Omega$$

for $\delta \in (0, 1)$ sufficiently small. For such δ , we multiply (4.4.24a) by φ_{δ} and integrate over Ω , once/twice by parts where necessary, so as to shift all spatial derivatives to φ_{δ} . Using *Lemmas 2.1.2, 4.2.4*, and 4.5.1 and (4.4.1) and Hoelder's inequality, we estimate as follows:

$$\begin{split} &\frac{d}{dt} \int_{\Omega} c_{\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + \mu \int_{\Omega} c_{\varepsilon_{k}}^{r} \varphi_{\delta} \, \mathrm{d}x \\ \leqslant &\mu \int_{\Omega} c_{\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + \int_{\Omega} |c_{\varepsilon_{k}} \mathbb{D}_{\varepsilon_{k}} : D^{2} \varphi_{\delta}| \, \mathrm{d}x + \int_{\Omega} |c_{\varepsilon_{k}} \mathcal{A} c_{\varepsilon_{k}} \cdot \nabla \varphi_{\delta}| \, \mathrm{d}x \\ \leqslant &\mu \int_{\Omega} c_{\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + n^{2} \|\mathbb{D}_{\varepsilon_{k}}\|_{(L^{\infty}(\Omega))^{n \times n}} \|D^{2} \varphi_{\delta}\|_{(L^{\infty}(\Omega))^{n \times n}} \int_{\{D^{2} \varphi_{\delta} \neq 0\}} c_{\varepsilon_{k}} \, \mathrm{d}x \\ &+ \|c_{\varepsilon_{k}}\|_{L^{\infty}(0,\infty;L^{1}(\Omega))} \|\nabla H\|_{(L^{\infty}(B_{1}))^{n}} \|\nabla \varphi_{\delta}\|_{(L^{\infty}(\Omega))^{n}} \int_{\{\nabla \varphi_{\delta} \neq 0\}} c_{\varepsilon_{k}} \, \mathrm{d}x \\ \leqslant &\mu \int_{\Omega} c_{\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + (\|D^{2} \varphi_{\delta}\|_{(L^{\infty}(\Omega))^{n \times n}} + \|\nabla \varphi_{\delta}\|_{(L^{\infty}(\Omega))^{n}}) C_{24} \int_{\mathrm{supp}(\varphi_{\delta})} c_{\varepsilon_{k}} \, \mathrm{d}x \\ \leqslant &\mu \int_{\Omega} c_{\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + \delta^{-2} |\mathrm{supp}(\varphi_{\delta})|^{1 - \frac{1}{r}} C_{25} \|c_{\varepsilon_{k}}\|_{L^{r}(\Omega)}. \end{split}$$

We conclude from Gronwall's and Hölder's inequalities and Lemma 4.5.1 that

$$\int_{\Omega} c_{\varepsilon_{k}}(\cdot, T)\varphi_{\delta} \, \mathrm{d}x + \mu \int_{0}^{T} \int_{\Omega} c_{\varepsilon_{k}}^{r} \varphi_{\delta} \, \mathrm{d}x \, \mathrm{d}t$$

$$\leq e^{\mu T} \left(\int_{\Omega} c_{0\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + \delta^{-2} |\mathrm{supp}(\varphi_{\delta})|^{1-\frac{1}{r}} C_{25} \int_{0}^{T} \|c_{\varepsilon_{k}}\|_{L^{r}(\Omega)} \, \mathrm{d}t \right)$$

$$\leq e^{\mu T} \left(\int_{\Omega} c_{0\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + \delta^{-2} |\mathrm{supp}(\varphi_{\delta})|^{1-\frac{1}{r}} T^{1-\frac{1}{r}} C_{25} \|c_{\varepsilon_{k}}\|_{L^{r}(\Omega \times (0,T))} \right)$$

$$\leq e^{\mu T} \left(\int_{\Omega} c_{0\varepsilon_{k}} \varphi_{\delta} \, \mathrm{d}x + \delta^{-2} |\mathrm{supp}(\varphi_{\delta})|^{1-\frac{1}{r}} C_{26}(T) \right) \quad \text{for } t \in (0,T). \tag{4.5.26}$$

Combining (4.4.23) and (4.5.26), we find that

$$\limsup_{k \to \infty} \int_0^T \int_\Omega c_{\varepsilon_k}^r \varphi_\delta \, \mathrm{d}x \, \mathrm{d}t \leqslant \mu^{-1} e^{\mu T} \left(\int_\Omega c_0 \varphi_\delta \, \mathrm{d}x + \delta^{-2} |\mathrm{supp}(\varphi_\delta)|^{1-\frac{1}{r}} C_{26}(T) \right)$$
(4.5.27)

for $\delta \in (0, 1)$. Due to the integrability of c_0 , (4.2.5f), the uniform boundedness of (φ_{δ}) , and the dominated convergence theorem we have

$$\lim_{\delta \to 0} \int_{\Omega} c_0 \varphi_\delta \, \mathrm{d}x = 0. \tag{4.5.28}$$

Together, (4.2.5g), (4.5.27), and (4.5.28) yield

$$c_{\varepsilon_k}^r \varphi_{\delta} \xrightarrow[k \to \infty \delta \to 0]{} 0 \quad \text{in } L^1(\Omega \times (0,T)).$$
 (4.5.29)

Finally, combining (4.5.22), (4.5.24), (4.5.25), and (4.5.29), we arrive at (4.5.21).

Remark 4.5.5. The assumptions $n \ge 3$ and $r > \frac{n}{n-2}$ from (4.3.1) are only required in the proof of Lemma 4.5.4. Together with (4.3.2f), they ensure the existence of the φ_{δ} s due to Lemma 4.2.4.

Finally, we can prove our main result on the existence of a very weak solution to the original IBVP (4.1.1).

Proof of Theorem 4.3.4. Consider the sequence $(c_{\varepsilon_k})_{k\in\mathbb{N}}$ from Lemmas 4.5.3 and 4.5.4 and let $\eta \in C_c^{2,1}(\overline{\Omega} \times [0,\infty))$ with $\nabla \eta \cdot (\mathbb{D}\nu) = 0$ on $\partial \Omega \times (0,\infty)$. Then, there is $T \in (0,\infty)$ s.t. $\eta \equiv 0$ for $t \geq T$. We multiply (4.4.24a) by η and integrate over $\Omega \times (0,\infty)$, once or twice by parts where necessary, using the boundary condition on η as well as (4.4.24b), and for all $k \in \mathbb{N}$ conclude that

$$-\int_{0}^{\infty} \int_{\Omega} c_{\varepsilon_{k}} \partial_{t} \eta \, \mathrm{d}x \, \mathrm{d}t - \int_{\Omega} c_{0\varepsilon_{k}} \eta(\cdot, 0) \, \mathrm{d}x$$

$$= \int_{0}^{\infty} \int_{\Omega} c_{\varepsilon_{k}} \mathbb{D}_{\varepsilon_{k}} : D^{2} \eta \, \mathrm{d}x \, \mathrm{d}t - \int_{0}^{\infty} \int_{\partial\Omega} c_{\varepsilon_{k}} \nabla \eta \cdot (\mathbb{D}_{\varepsilon_{k}} \nu) \, \mathrm{d}\sigma(x) \, \mathrm{d}t$$

$$+ \int_{0}^{\infty} \int_{\Omega} c_{\varepsilon_{k}} (\mathcal{A}c_{\varepsilon_{k}}) \cdot \nabla \eta \, \mathrm{d}x \, \mathrm{d}t + \mu \int_{0}^{\infty} \int_{\Omega} c_{\varepsilon_{k}} (1 - c_{\varepsilon_{k}}^{r-1}) \eta \, \mathrm{d}x \, \mathrm{d}t.$$
(4.5.30)

We first address convergence of (c_{ε_k}) on $\partial\Omega \times (0,T)$. Observe that since $O_{a/2}(\partial\Omega) \cap \Omega$ is open and precompact in $\{\mathbb{D} > 0\}$, we can make use of the uniform boundedness of $(c_{\varepsilon_k})_{k\in\mathbb{N}}$ in $L^2(0,T; H^1(O_{a/2}(\partial\Omega) \cap \Omega))$ due to (4.5.4) and convergence (4.5.17) and the Banach-Alaoglu theorem, yielding

$$c_{\varepsilon_k} \underset{k \to \infty}{\longrightarrow} c \quad \text{in } L^2(0,T; H^1(O_{a/2}(\partial\Omega) \cap \Omega)).$$

$$(4.5.31)$$

Using the continuity of the trace operator, we conclude with (4.5.31) that

$$c_{\varepsilon_k} \underset{k \to \infty}{\longrightarrow} c \quad \text{in } L^2(\partial \Omega \times (0,T)).$$
 (4.5.32)

Now convergences (4.4.3), (4.4.23), (4.5.17), (4.5.21), and (4.5.32) and continuity of the operator $\mathcal{A}: L^1(\Omega) \to (L^{\infty}(\Omega))^n$ together with compensated compactness (*Lemma A.3.2*) allow to pass to the limit in each term in (4.5.30), yielding (4.3.5).

4.6 Smooth very weak solutions are classical

In this final section we provide a justification for the very weak formulation (4.3.5). We show that as in the case of Neumann boundary conditions for elliptic equations (see e.g., Theorem 2.2.2.5 in [72]) it holds for smooth \mathbb{D} that any sufficiently smooth very weak solution in terms of *Definition 4.3.3* is also a classical solution to (4.1.1). Theorem 4.6.1. In addition to Assumptions 4.3.1, let

$$\mathbb{D} \in C^{2}(\overline{\Omega}; \mathbb{R}^{n \times n}),$$

$$c \in C^{2,1}(\overline{\Omega} \times (0, \infty)) \cap C(\overline{\Omega} \times [0, \infty)),$$

$$c_{0} \in C(\overline{\Omega}).$$
(4.6.1)

Then, if c is a solution to (4.1.1) in the sense of Definition 4.3.3, then it solves this IBVP in the classical sense.

Proof. Let

$$\eta \in C_c^{2,1}(\overline{\Omega} \times [0,\infty)) \qquad \text{s.t. } \nabla \eta \cdot (\mathbb{D}\nu) = 0 \text{ on } \partial\Omega \times (0,\infty).$$

$$(4.6.2)$$

Then, there is $T \in (0, \infty)$ s.t. $\eta \equiv 0$ for $t \ge T$. Integrating by parts on both sides of (4.3.5), once or twice where necessary, using the information about η on $\partial \Omega \times (0, \infty)$, yields

$$\int_{0}^{\infty} \int_{\Omega} \partial_{t} c\eta \, dx \, dt + \int_{\Omega} (c(\cdot, 0) - c_{0})\eta(0) \, dx$$

$$= -\int_{0}^{\infty} \int_{\Omega} \nabla \cdot (c\mathbb{D}) \cdot \nabla \eta \, dx \, dt - \int_{0}^{\infty} \int_{\Omega} \nabla \cdot (c(\mathcal{A}c))\eta \, dx \, dt$$

$$+ \mu \int_{0}^{\infty} \int_{\Omega} c(1 - c^{r-1})\eta \, dx \, dt + \int_{0}^{\infty} \int_{\partial\Omega} c(\mathcal{A}c) \cdot \nu\eta \, d\sigma(x) \, dt$$

$$= \int_{0}^{\infty} \int_{\Omega} \left(\nabla \nabla : (c\mathbb{D}) - \nabla \cdot (c(\mathcal{A}c)) + \mu c(1 - c^{r-1}) \right) \eta \, dx \, dt$$

$$- \int_{0}^{\infty} \int_{\partial\Omega} \left(\nabla \cdot (\mathbb{D}c) - c(\mathcal{A}c) \right) \cdot \nu\eta \, d\sigma(x) \, dt.$$
(4.6.3)

For the subset of $\eta \in C_c^{2,1}(\Omega \times (0,\infty))$ it holds that $\eta(0) = 0$. Moreover, for such η the boundary integral in (4.6.3) vanishes. Thus, the fundamental lemma of calculus of variations applies and yields that c satisfies

$$\partial_t c = \nabla \nabla : (\mathbb{D}(x)c) - \nabla \cdot (c\mathcal{A}c) + \mu c(1 - c^{r-1})$$

pointwise in $\Omega \times (0, \infty)$. Considering the subset of $\eta \in C_c^{2,1}(\Omega \times [0, \infty))$ again the fundamental lemma of calculus of variations that c satisfies the initial condition (4.1.1c) in $\overline{\Omega}$. Now we can conclude from (4.6.3) that for all η satisfying (4.6.2) it holds that

$$\int_0^\infty \int_{\partial\Omega} \left(\nabla \cdot (\mathbb{D}c) - c(\mathcal{A}c) \right) \cdot \nu \eta \, \mathrm{d}\sigma(x) \, \mathrm{d}t = 0.$$
(4.6.4)

Finally, we consider η of the form $\eta(x,t) = \eta_1(x)\eta_2(t)$, where $\eta_1 \in C^2(\overline{\Omega})$ satisfies $\nabla \eta_1 \cdot (\mathbb{D}\nu) = 0$ on $\partial\Omega$ and $\eta_2 \in C_c^1([0,\infty))$. Applying the fundamental lemma of calculus of variations w.r.t. to time integration in (4.6.4) yields

$$\int_{\partial\Omega} \left(\nabla \cdot (\mathbb{D}c) - c(\mathcal{A}c) \right) \cdot \nu \eta_1 \, \mathrm{d}\sigma(x) = 0 \qquad \text{for all } t \in (0,T).$$
(4.6.5)

The boundary condition (4.1.1b) now follows with (4.6.5), Lemma 4.6.2 below and the embedding $H^1(\Omega) \hookrightarrow L^2(\partial\Omega)$.

Lemma 4.6.2. Let $\mathbb{D} \in C^2(\overline{\Omega}; \mathbb{R}^{n \times n})$ symmetric, $\mathbb{D} \ge 0$ with $\{\mathbb{D} \ge 0\} \cap \partial\Omega = \emptyset$. Then, the set

$$\{\eta \in C^2(\overline{\Omega}) : \nabla \eta \cdot (\mathbb{D}\nu) = 0 \text{ on } \partial\Omega\}$$

is dense in $H^1(\Omega)$.

Proof. Since \mathbb{D} is continuous and positive definite on $\partial\Omega$, it is positive definite in some open neighbourhood of $\partial\Omega$. Choose some symmetric $\mathbb{B} \in C_c^{\infty}(\Omega; \mathbb{R}^{n \times n})$, $\mathbb{B} \ge 0$, and positive definite in an open neighbourhood of $\{\mathbb{D} \ge 0\}$. Then,

$$\tilde{\mathbb{D}} := \mathbb{D} + \mathbb{B}$$

satisfies

$$\mathbb{D} = \tilde{\mathbb{D}}$$

in some open neighbourhood of $\partial\Omega$, is symmetric and there is some $\delta > 0$ s.t.

$$y^T \tilde{\mathbb{D}}(x) y \ge \delta |y|^2$$
 for all $x \in \overline{\Omega}, y \in \mathbb{R}^n$.

On $H^1(\Omega)$, consider the scalar product

$$\langle f,g \rangle := \lambda \int_{\Omega} fg \,\mathrm{d}x + \int_{\Omega} (\nabla f)^T \tilde{\mathbb{D}} \nabla g \,\mathrm{d}x$$

$$(4.6.6)$$

for some $\lambda > 0$. Since \tilde{D} is smooth and positive definite in $\overline{\Omega}$, the norm induced by (4.6.6) is equivalent to the standard norm on $H^1(\Omega)$. Set

$$E := \{ \eta \in C^2(\overline{\Omega}) : \quad \nabla \eta \cdot (\mathbb{D}\nu) = 0 \text{ on } \partial \Omega \}$$
$$= \{ \eta \in C^2(\overline{\Omega}) : \quad \nabla \eta \cdot (\tilde{\mathbb{D}}\nu) = 0 \text{ on } \partial \Omega \}.$$

We thus need to verify that the orthogonal complement of E w.r.t. the above scalar product in $H^1(\Omega)$ only contains the zero vector. Assume the contrary, i.e., that

$$\overline{E}^{\perp} = \{ \xi \in H^1(\Omega) : \quad \langle \xi, \eta \rangle = 0 \text{ for all } \eta \in \overline{E} \} \neq \{ 0 \}.$$

Let $\xi \in \overline{E}^{\perp}$ and $\xi \neq 0$. Then, as $\partial \Omega$ is sufficiently smooth, there is a sequence $(\xi_n)_{n \in \mathbb{N}} \subset C^{\infty}(\overline{\Omega})$ s.t.

$$\xi_n \xrightarrow[n \to \infty]{} \xi \text{ in } H^1(\Omega). \tag{4.6.7}$$

Consider the sequence of elliptic BVPs

$$-\nabla \cdot (\tilde{\mathbb{D}} \nabla u_n) + \lambda u_n = \xi_n \qquad \text{in } \Omega, \tag{4.6.8a}$$

$$\nabla u_n \cdot (\mathbb{D}\nu) = 0 \quad \text{on } \partial\Omega.$$
 (4.6.8b)

Lemma A.1.19 implies that for sufficiently large $\lambda > 0$ there exists a unique solution $u_n \in C^2(\overline{\Omega})$ to (4.6.8) for all $n \in \mathbb{N}$, and $(u_n)_{n \in \mathbb{N}}$ is uniformly bounded in $H^2(\Omega)$. Consequently, due to the continuity of the trace operator, the embeddings $H^2(\Omega) \hookrightarrow H^1(\Omega)$, $H^2(\Omega) \hookrightarrow H^1(\partial\Omega)$ and the Banach-Alaoglu theorem there exist a sequence $(n_l)_{l \in \mathbb{N}}$ and some $u \in H^2(\Omega)$ s.t.

$$u_{n_l} \xrightarrow{\sim} u \qquad \text{in } H^2(\Omega), \tag{4.6.9}$$

$$u_{n_l} \xrightarrow{} u \qquad \text{in } H^1(\Omega),$$

$$(4.6.10)$$

$$u_{n_l} \underset{l \to \infty}{\longrightarrow} u \qquad \text{in } H^1(\partial\Omega).$$
 (4.6.11)

From (4.6.7) and (4.6.9)–(4.6.11) and as $(u_n) \subset E$ it follows that $u \in \overline{E}$ and is a strong L^2 solution to the BVP

$$-\nabla \cdot (\tilde{\mathbb{D}}\nabla u) + \lambda u = \xi \qquad \text{in } \Omega \tag{4.6.12a}$$

$$\nabla u \cdot (\tilde{\mathbb{D}}\nu) = 0 \quad \text{on } \partial\Omega. \tag{4.6.12b}$$

Multiplying (4.6.12a) by ξ and integrating by parts using (4.6.12b) and the symmetry of $\tilde{\mathbb{D}}$ then yields

$$0 = \langle u, \xi \rangle = \lambda \int_{\Omega} u\xi \, \mathrm{d}x + \int_{\Omega} (\nabla u)^T \tilde{\mathbb{D}} \nabla \xi \, \mathrm{d}x = \int_{\Omega} \xi^2 \, \mathrm{d}x.$$

This shows that $\xi \equiv 0$, contradicting the above assumption. Therefore, $\overline{E} = H^1(\Omega)$, as required.

Part II

Nonlocal models with nonlocality in the reaction term

On a mathematical model for cancer invasion with repellent pH-taxis and nonlocal intraspecific interaction

This chapter was first published in Volume 75 of Zeitschrift für Angewandte Mathematik und Physik in 2024.¹ The presentation has been adapted for use in this dissertation to clarify the details of the proofs and guarantee consistency of the notation.

5.1 Motivation

Migration, proliferation, and differentiation of cells are influenced by biochemical and biophysical characteristics of their surroundings, which they perceive by way of transmembrane units like ion channels, receptors, etc. Increasing experimental evidence suggests that cells are able to sense such cues not only where they are, but also at larger distances, up to several cell diameters around their current position [71, 90, 130]. This led to mathematical models accounting for various types of nonlocalities, most of them addressing cell-cell and/or cell-matrix adhesions; we refer to the review article [28] and references therein. The settings typically involve reaction, diffusion and drift terms, whereby the latter contain an integral operator to characterize the so-called adhesion velocity over the interaction range. In *Chapter 3* was performed a rigorous passage from a cellmatrix adhesion model to a reaction-diffusion-haptotaxis equation when the sensing radius is becoming infinitesimally small, thus recovering the local PDE formulation from that featuring the mentioned nonlocality. The remote sensing of signals by cells affects, however, not only motility, but also proliferation, growth, and phenotypic switch, either directly - by occupancy of transmembrane units on cellular extensions like cytonemes and folopodia and subsequently initiated signaling pathways, or in an indirect manner - as effects of altered migratory and aggregation behavior. Models involving reaction-diffusion equations with nonlocal source terms have been proposed in various contexts, including biological and ecological ones, see e.g., [87, 145] and references therein for rather generic settings, [13, 15, 129] for chemotaxis systems, and [113, 136, 137] for equations dedicated to tumor growth. We refer to [28, 87, 145] for some reviews of model classes addressing this type of nonlocality.

As far as growth and migration of cell populations are concerned, the reaction-diffusion models with nonlocal source terms

$$u_t = \nabla \cdot (D\nabla u) + F(u) \tag{5.1.1}$$

 $^{^{1}[49]}$ The paper is licensed under a Creative Commons Attribution 4.0 International License.

typically feature $F(u) = \mu J * u(1 - u)$ to describe nonlocal stimulation of growth (see e.g., [137, 145]), or $F(u) = \mu u^{\alpha}(1 - J * u^{\beta})$, which characterizes competition between (bunches of) cells for available resources in their surroundings, attempting, e.g., to prevent overcrowding. In the context of (tumor) cell migration such models have been handled e.g., in [136], where intra- and interspecific nonlocal interactions led to an ODE-PDE system for the interplay between cancer cells peforming linear diffusion and haptotaxis with the extracellular matrix being (nonlocally) degraded by the cells and remodeled with the mentioned growth limitation. We also refer to [28, 99] for short reviews of models with source terms of this type and therewith associated mathematical challenges.

In this chapter we propose and analyze a model for tumor cell migration involving myopic diffusion, repellent pH-taxis, and a nonlocal source term of the competition type mentioned above. The cross-diffusion system is obtained upon starting from the mesoscopic description of cell migration via a kinetic transport equation for the space-time distribution function of cells sharing some velocity regime. An appropriate upscaling relying on diffusion dominance then leads to the effective macroscopic equation for the cancer cell density, with precisely specified diffusion and drift coefficients. The remaining of this chapter is structured as follows: Section 5.2 contains the model deduction with the mentioned upscaling. Section 5.3 is dedicated to the mathematical analysis of the obtained nonlocal macroscopic system, in terms of global existence, uniqueness, and boundedness of a solution to a simplified version of the problem. In Section 5.4 we study the asymptotic behavior. Section 5.5 offers a 1D study of pattern formation for the equations handled in Section 5.3, but only involving constant motility coefficients. In Section 5.6 we provide numerical simulations to illustrate the qualitative behavior of solutions to the investigated nonlocal problem. Section 5.7 contains a discussion of the results.

5.2 Modeling

In this section we start from a mesoscopic description of cell migration and intrapopulation interactions and deduce (in a non-rigorous way) effective equations on the macroscopic scale of cell population dynamics. The deduction closely follows that in [99], however extends it, by accounting here for the repellent effects of acidity eventually leading on the population scale to chemorepellent pH-taxis.

Tumor migration and spread are typically assessed on the macroscopic scale of the cancer cell population via biomedical imaging. The involved processes are, however, highly complex and originate at the lower levels of cell aggregates sharing -beside time-space dynamics- one or several further traits (e.g., velocity, phenotypic state or other so called 'activity variables'), down to microscopic events on individual cells. This multiscale character of cell migration can be captured (at least partially) by models within the kinetic theory of active particles (KTAP) framework formulated by Bellomo et al. (see e.g., [8, 11] and references therein). Starting from kinetic transport equations (KTEs), a large variety of (spatially) local and nonlocal models have been proposed and various kinds of upscaling and moment closure methods have been performed in order to deduce their macroscopic limits which enable a mathematically more efficient handling, see e.g., [25, 27, 29–33, 42, 53–55, 78, 91, 92, 105, 155]. The obtained macroscopic equations carry in the coefficients of their motility and source terms some of the traits from the mesoscale

on which KTEs were formulated. Those coefficients are no longer 'guessed' as in the case of stating reaction-diffusion-taxis directly on the population level and the diffusion is often of the 'myopic' type, involving a drift correction. We will perform here a diffusion-dominated upscaling of mesoscale dynamics.

We will use the following notations:

- p = p(t, x, v): distribution function of cells having at time t and position $x \in \mathbb{R}^n$ the velocity $v \in V$;
- $V = [s_1, s_2] \times \mathbb{S}^{n-1}$: velocity space. Thereby, s_1, s_2 denote the minimum, respectively the maximum speed of a cell, $\theta \in \mathbb{S}^{n-1}$ represents the cell direction;
- $u(t,x) = \int_V p(t,x,v) dv$: macroscopic cell density;
- h(t, x): concentration of protons. This is a macroscopic quantity throughout this note.

The kinetic transport equation (KTE)

$$p_t + v \cdot \nabla_x p = \mathcal{L}[h]p + \tilde{\mu}\mathcal{I}[p, p]$$
(5.2.1)

characterizes the mesoscopic dynamics of the considered cell population. This is the framework set in [116], which assumes that changes in p are due to velocity jumps accompanied by reorientations dictated by a turning kernel contained in the operator $\mathcal{L}[h]$.

The first term on the right-hand side of (5.2.1) represents the so-called turning operator. The second term describes growth/decay of cells due to intraspecific proliferative/competitive interactions, while $\tilde{\mu} > 0$ is the constant interaction rate.² With a small constant $\varepsilon > 0$ relating to the cell size and to the distance at which cells can sense signals in their proximity, we will assume that $\tilde{\mu} = \varepsilon^2 \mu$. This means that cells have a much higher preference to motility (in particular, to changing direction) than to interaction and crowding.

We assume that the turning operator is of the form

$$\mathcal{L}[h](p) = \int_{V} \left(T[h](v, v')p(t, x, v') - T[h](v', v)p(t, x, v) \right) \mathrm{d}v',$$
(5.2.2)

with the turning rate $T[h](v, v') \ge 0$ chosen such that the reorientation is a Poisson process with rate

$$\lambda[h] = \int_V T[h](v, v') \,\mathrm{d}v,$$

hence such that $T[h]/\lambda[h]$ is a kernel giving the probability density for a change of the velocity regime of a cell from v' to v. In particular, this means that $\mathcal{L}[h]$ is preserving mass. The reorientation of cells depends on the acidity of their environment (expressed by the concentration h of protons).

In the following we assume that the turning rate has an asymptotic expansion of the form

$$T[h] = T_0[h] + \varepsilon T_1[h] + O(\varepsilon^2), \qquad (5.2.3)$$

²We could actually consider $\tilde{\mu}$ to be a function of x and/or t (but not of derivatives w.r.t. these variables) and even of h. The latter would allow us to account e.g., for the unfavorable effect of acidity on the proliferation of tumor cells. The deduction done here works then exactly in the same way. In fact, our analysis in *Section 5.3* is performed in the case where such h-dependence is considered.

thus the turning operator admits itself an expansion

$$\mathcal{L}[h](p) := L_0[h](p) + \varepsilon L_1[h](p) + O(\varepsilon^2), \qquad (5.2.4)$$

where $L_0[h]$ and $L_1[h]$ are linear operators,

$$L_i[h](p)(t, x, v) = \int_V [T_i[h](v, v')p(t, x, v') - T_i[h](v', v)p(t, x, v)] \,\mathrm{d}v', \qquad i = 0, 1.$$
(5.2.5)

For \mathcal{I} we consider as in [99] the form

$$\mathcal{I}[p,p](t,x,v) = \frac{p^{\alpha}(t,x,v)}{\int_{V} M^{\alpha}(x,v) \,\mathrm{d}v} - \frac{1}{\int_{V} M^{\alpha+\beta}(x,v) \,\mathrm{d}v} p^{\alpha}(t,x,v) \int_{\Omega} J(x,x') p^{\beta}(t,x',v) \,\mathrm{d}x',$$
(5.2.6)

where: $\alpha, \beta > 0$ are constants, J(x, x') is a function weighting the interactions between (bunches of) cells sharing the same velocity regime within a bounded domain $\Omega \subset \mathbb{R}^n$. We assume that J depends on the distance between interacting (clusters of) cells and take J(x, x') = J(x - x'), also requiring J to satisfy

$$\int_{V} J(x) \,\mathrm{d}x = 1, \tag{5.2.7}$$

$$\inf_{B_{\operatorname{diam}(\Omega)}(0)} J \ge \eta \quad \text{for some } \eta > 0.$$
(5.2.8)

We also assume that there exists a bounded velocity distribution M(x, v) > 0 such that:

- 1. $\int_V M(x, v) dv = 1$, i.e., M is a kernel w.r.t. v.
- 2. $\int_V v M(x, v) \, dv = 0$, i.e., the flow produced by the equilibrium distribution M(v) vanishes.
- 3. The rate $T_0[h](v, v')$ satisfies the detailed balance equation

$$T_0[h](v,v')M(v') = T_0[h](v',v)M(v).$$

4. The turning rate $T_0[h](v, v')$ is bounded and there exists $\sigma > 0$ such that

$$T_0[h](v,v') \ge \sigma M(x,v), \text{ for all } (v,v') \in V \times V, \ x \in \mathbb{R}^n, \ t > 0.$$

The following lemma summarizing the properties of the operator $-L_0$ can be easily verified (see e.g., [10, 25]).

Lemma 5.2.1. Let $L_0[h]$ be the operator defined in (5.2.5). Then $-L_0[h]$ has the following properties:

(i) $-L_0[h]$ is positive definite w.r.t. the scalar product and the associated norm in the weighted space $L^2(V, \frac{dv}{M(x,v)})$, and self-adjoint: for all $p, \zeta \in L^2(V, \frac{dv}{M(x,v)})$ it holds that

$$\int_{V} L_0[h](p)(v) \frac{\zeta(v)}{M(v)} \, dv = \int_{V} L_0[h](\zeta)(v) \frac{p(v)}{M(v)} \, dv$$

(ii) For $\phi \in L^2(V, \frac{dv}{M(x,v)})$, the equation $L_0[h](\zeta) = \phi$ has a unique solution $\zeta \in L^2(V, \frac{dv}{M(x,v)})$ satisfying³ $\bar{\zeta} = 0$ if and only if $\bar{\phi} = 0$.

³Here and in the remaining of this section we use the notation $\bar{\zeta} := \int_V \zeta(v) \, dv$ for any V-integrable function ζ (hence also $u = \bar{p}$).

- (iii) Ker $L_0[h] = span (M(v)).$
- (iv) The equation $L_0[h](\psi) = vM(v)$ has a unique solution $\psi(v) =: L_0[h]^{-1}(vM(v))$ (this is actually a pseudoinverse).

Example 5.2.2. Consider $T_0[h](v, v') := \lambda_0[h]M(v)$, with $\lambda[h] \ge \lambda_0[h] > 0$ for any h. This obviously satisfies the properties 3. and 4. in our above assumption. With this choice,

$$L_0[h](p) = \lambda_0[h](M(v)u - p)$$
(5.2.9)

and it is straightforward to see that this operator satisfies the properties in Lemma 5.2.1 and the function ψ in *(iv)* becomes $\psi(v) = -vM(v)/\lambda_0[h]$ if $\psi \in (\text{span } (M(v)))^{\perp}$.⁴

Equation (5.2.1) is supplemented with the macroscopic PDE for proton concentration:

$$h_t = D_H \Delta h + g(u, h), \tag{5.2.10}$$

where $D_H > 0$ is the diffusion constant and g(u, h) represents production by tumor cells and uptake (e.g., by blood capillaries - not explicitly modeled in this note) or decay. As such, gwill have to be bounded; moreover, when there is no or very less acid its production is turned on and sustained, whereas a high proton concentration exceeding some upper threshold level H is leading to a drop in h, by enhanced (more or less passive) uptake by surrounding tissues and vasculature and/or by ceased expression, due to hypoxia-induced apoptosis of (too crowded) tumor cells. More details on the concrete assumptions made about g(u, h) are provided at the beginning of Sections 5.3 and 5.5. A concrete choice of g is given at the beginning of Section 5.6.

We also consider initial conditions for p and h:

$$p(0, x, v) = p_0(x, v), \qquad h(0, x) = h_0(x), \quad x \in \Omega \subseteq \mathbb{R}^n, \ v \in V.$$
 (5.2.11)

Together with these, equations (5.2.1) and (5.2.10) form a meso-macro system describing the dynamics of the (mesoscopic) cell distribution in response to acidity in the extracellular space.

We perform a parabolic scaling to obtain the diffusion limit of the KTE (5.2.1). This means that we rescale the time and space variables as follows:

$$\hat{t} = \varepsilon^2 t, \quad \hat{x} = \varepsilon x.$$

Subsequently we will drop the '^' symbol and the ε -dependency of the solution p^{ε} to the resulting KTE, in order not to complicate the writing. Then, (5.2.1) becomes

$$\varepsilon p_t + v \cdot \nabla_x p = \frac{1}{\varepsilon} \mathcal{L}[h] p + \mu \varepsilon \mathcal{I}[p, p].$$
(5.2.12)

Now consider the decomposition (Chapman-Enskog expansion)

$$p(t, x, v) = F(u)(t, x, v) + \varepsilon p^{\perp}(t, x, v), \qquad (5.2.13)$$

with $\int_V p^{\perp}(t, x, v) dv = 0$, thus $p^{\perp} \in (\text{span } (M(v)))^{\perp}$, and $F(u) \in \text{span } (M(v))$ such that $\int_V F(u) dv = u$. A natural choice is F(u)(t, x, v) := M(x, v)u(t, x), which we will subsequently adopt.

⁴This is actually the case even if T_0 has a more general form (depending only on v and not on v') without having to satisfy condition 2.

Then observe that

$$\mathcal{I}[p,p] = \mathcal{I}[M(v)u + \varepsilon p^{\perp}, M(v)u + \varepsilon p^{\perp}] = \mathcal{I}[M(v)u, M(v)u] + O(\varepsilon)$$

and (5.2.12) becomes

$$\partial_t (M(v)u) + \varepsilon \partial_t p^\perp + \frac{1}{\varepsilon} v \cdot \nabla_x (M(v)u) + v \cdot \nabla_x p^\perp$$

= $\frac{1}{\varepsilon} L_0[h](p^\perp) + \frac{1}{\varepsilon} L_1[h](M(v)u) + L_1[h](p^\perp) + \mu \mathcal{I}[M(v)u, M(v)u] + O(\varepsilon).$ (5.2.14)

Let $P: L^2(V, \frac{\mathrm{d}v}{M(v)}) \to \mathrm{Ker} \ L_0[h]$ be the projection operator. Then

$$P(\phi) = M(v)\overline{\phi}, \qquad \phi \in L^2(V, \frac{\mathrm{d}v}{M(v)}).$$

It is easy to verify that the following lemma holds (see, e.g., [10]).

Lemma 5.2.3. The projection operator P has the following properties:

(i)
$$(I - P)(M(v)u) = P(p^{\perp}) = 0.$$

(ii) $(I - P)(v \cdot \nabla_x(M(v)u)) = v \cdot \nabla_x(M(v)u).$
(iii) $(I - P)(L_0[h](M(v)u)) = L_0[h](M(v)u)$ and $(I - P)(L_1[h](M(v)u)) = L_1[h](M(v)u).$
(iv) $(I - P)(L_1[h](p^{\perp})) = L_1[h](p^{\perp}).$

If we now apply I - P to (5.2.14) we get

$$\varepsilon \partial_t p^{\perp} + \frac{1}{\varepsilon} v \cdot \nabla_x (Mu) + (I - P)(v \cdot \nabla_x p^{\perp})$$

$$= \frac{1}{\varepsilon} L_0[h](p^{\perp}) + \frac{1}{\varepsilon} L_1[h](Mu) + L_1[h](p^{\perp}) + \mu \mathcal{I}[Mu, Mu] + O(\varepsilon).$$
(5.2.15)

Integrating (5.2.14) w.r.t. v gives (at leading order) the macroscopic PDE⁵

$$u_t + \int_V v \cdot \nabla_x p^{\perp} \, \mathrm{d}v = \mu \int_V \mathcal{I}[Mu, Mu] \, \mathrm{d}v.$$
 (5.2.16)

On the other hand, from (5.2.15) we obtain (again at leading order)

$$L_0[h](p^{\perp}) = v \cdot \nabla_x(Mu) - L_1[h](Mu).$$
(5.2.17)

Since $\int_V L_1[h](Mu) dv = 0$, we see that the integral w.r.t. v of the right-hand side in (5.2.17) vanishes, so we can pseudo-invert $L_0[h]$ to obtain

$$p^{\perp} = L_0[h]^{-1} \Big(v \cdot \nabla_x(Mu) - L_1[h](Mu) \Big).$$
(5.2.18)

Plugging this into (5.2.16) gives

$$u_t + \int_V v \cdot \nabla_x \left(L_0[h]^{-1} (v \cdot \nabla_x(Mu)) - L_0[h]^{-1} (L_1[h](Mu)) \right) = \mu \int_V \mathcal{I}[Mu, Mu] \, \mathrm{d}v. \quad (5.2.19)$$

⁵involving nonlocalities w.r.t. velocity

For the right-hand side in (5.2.19) we have

$$\mu \int_{V} \mathcal{I}[Mu, Mu] \, \mathrm{d}v = \mu u^{\alpha} (1 - J * u^{\beta}).$$

For the first transport term on the left-hand side we compute

$$\begin{split} \int_{V} v \cdot \nabla_{x} \Big(L_{0}[h]^{-1} (v \cdot \nabla_{x}(Mu)) \Big) \, \mathrm{d}v &= \nabla_{x} \cdot \Big(\frac{1}{\lambda_{0}[h]} \nabla_{x} \cdot \Big(\int_{V} v \otimes vM(v) \, \mathrm{d}v \, u \Big) \Big) \\ &= -\nabla_{x} \cdot \Big(\frac{1}{\lambda_{0}[h]} \nabla_{x} \cdot (\mathbb{D} \, u) \Big), \end{split}$$

where we applied the observations made at the end of Example 5.2.2 and denoted by

$$\mathbb{D}(x) := \int_{V} v \otimes v M(x, v) \, \mathrm{d}v$$

the diffusion tensor of tumor cells.

For the second transport term on the left-hand side of (5.2.19) we have

$$\begin{split} &- \int_{V} v \cdot \nabla_{x} \left(L_{0}[h]^{-1} (L_{1}[h](M(v)u)) \right) \mathrm{d}v \\ &= - \nabla_{x} \cdot \int_{V} v L_{0}[h]^{-1} (L_{1}[h](M(v)u)) \mathrm{d}v \\ &= - \nabla_{x} \cdot \int_{V} v M(v) \frac{1}{M(v)} L_{0}[h]^{-1} (L_{1}[h](M(v)u)) \mathrm{d}v \\ &= - \nabla_{x} \cdot \int_{V} L_{0}[h](\psi(v)) \frac{1}{M(v)} L_{0}[h]^{-1} (L_{1}[h](M(v)u)) \mathrm{d}v \\ &= - \nabla_{x} \cdot \left(\int_{V} \frac{\psi(v)}{M(v)} L_{1}[h](M(v)) \mathrm{d}v u \right) \\ &= \nabla_{x} \cdot \left(u \Gamma[h] \right), \end{split}$$

where we used the fact that L_0 is self-adjoint, $\psi(v) = -vM(v)$ is its pseudo-inverse, and the notation

$$\Gamma[h](x) := \frac{1}{\lambda_0[h]} \int_V v L_1[h](M(x,v)) \,\mathrm{d}v.$$

With the above calculations (5.2.19) becomes

$$u_t - \nabla_x \cdot \left(\frac{1}{\lambda_0[h]} \nabla_x \cdot (\mathbb{D} \ u)\right) + \nabla_x \cdot \left(u\Gamma[h]\right) = \mu u^{\alpha}(1 - J \ast u^{\beta}).$$
(5.2.20)

To specify $\Gamma[h]$ we consider⁶ $T_1[h](v, v') := -a(h)v \cdot \nabla h + b(h)v' \cdot \nabla h$ with $a, b \ge 0$. Then we compute

$$\int_{V} v L_1[h](M(x,v)) \,\mathrm{d}v = -a(h) \frac{s_2^{n+2} - s_1^{n+2}}{n(n+2)} |\mathbb{S}^{n-1}| I_n \nabla h - \frac{b(h)}{|V|} \mathbb{D} \nabla h,$$

recalling that $V = [s_1, s_2] \times \mathbb{S}^{n-1}$, thus $|V| = \frac{s_2^n - s_1^n}{n} |\mathbb{S}^{n-1}|$. With the notation

$$\mathbb{T}(x) := a(h) \frac{s_2^{n+2} - s_1^{n+2}}{n(n+2)} |\mathbb{S}^{n-1}| \ I_n + \frac{b(h)}{|V|} \mathbb{D}$$

we obtain

$$\Gamma[h](x) = -\frac{1}{\lambda_0[h]} \mathbb{T}(x) \nabla h,$$

 $^{^{6}}$ a similar choice has been proposed in [25]

which leads to the macroscopic PDE

$$u_t = \nabla_x \cdot \left(\frac{1}{\lambda_0[h]} \nabla_x \cdot (\mathbb{D}(x)u)\right) + \nabla_x \cdot \left(\frac{u}{\lambda_0[h]} \mathbb{T}(x) \nabla h\right) + \mu u^{\alpha} (1 - J * u^{\beta}).$$
(5.2.21)

The particular choice $\lambda_0[h] := 1$, a(h) := 0, b(h) := |V| leads to the first equation in (5.3.1).

The first term on the right-hand side of (5.2.21) represents (myopic) diffusion, the second one characterizes repellent chemotaxis, away from increasing gradients of proton concentration,⁷ while the last is a source term accounting for tumor cell growth enhanced or limited by intraspecific interactions. Thereby, the growth rate μ can also depend on the proton concentration h, as will actually be the case in the subsequent sections. The requirements it has to satisfy are biologically motivated: cancer cells are able to survive and divide at far lower pH than normal cells and tissue; this gives them an advantage in using resources, thus enabling and even enhancing proliferation under mildly acidic conditions. However, when the proton concentration surpasses a certain critical level⁸ the environmental conditions for cell division are so unfavorable, that tumor cells are arrested in their cycle and cease proliferation. As growth rates cannot be negative we account for an environmental-mediated decay by way of the (nonlocal) competition term, as the crowded tumor environment is the main source of acidity.

The above deduction of a macroscopic reaction-diffusion-taxis is merely formal; the nonlinear source term prevents applying the proof of the rigorous derivation from [25]. The following section will be dedicated to proving global existence and boundedness of nonnegative solutions to the coupled PDE system for u and h obtained on the macrolevel by considering the above much simplified forms of the coefficient functions λ_0, a, b . The previous calculations were made for $x \in \mathbb{R}^n$, however we can restrict to a bounded domain $\Omega \subset \mathbb{R}^n$ upon proceeding as in [33, 42, 124] and assuming no-flux of cells or protons through the boundary.

5.3 Mathematical analysis

Let $\Omega \subset \mathbb{R}^n$ be a bounded domain with smooth enough boundary and outer unit normal ν . We consider the model

$$\begin{aligned} u_t &= \nabla \nabla : \left(\mathbb{D}(x)u \right) + \nabla \cdot \left(\mathbb{D}(x)u\nabla h \right) + \mu(h)u^{\alpha}(1 - J * u^{\beta}) & \text{in } \Omega \times (0, \infty), \\ h_t &= D_H \Delta h + g(u, h) & \text{in } \Omega \times (0, \infty), \\ (\mathbb{D}(x)\nabla u + \nabla \cdot \mathbb{D}(x)u + \mathbb{D}(x)u\nabla h) \cdot \nu &= \nabla h \cdot \nu = 0 & \text{on } \partial\Omega \times (0, \infty), \\ u(\cdot, 0) &= u_0, h(\cdot, 0) = h_0 & \text{in } \Omega, \end{aligned}$$
(5.3.1)

where u denotes the cell density and h the acid concentration. We assume that our diffusion tensor $\mathbb{D} = (d_{ij})_{i,j=1,...,n}$ satisfies $d_{ij} \in C^1(\overline{\Omega})$. Moreover, \mathbb{D} satisfies the uniform parabolicity and boundedness condition, i.e., there are $B_1, B_2 > 0$ such that for all $\xi \in \mathbb{R}^n$ and $x \in \overline{\Omega}$ it holds that

$$B_1|\xi|^2 \leqslant \sum_{i,j=1}^n d_{ij}(x)\xi_j\xi_i \leqslant B_2|\xi|^2.$$
(5.3.2)

⁷as in [31, 33, 89, 91, 92] we call this a repellent pH-taxis

⁸denoted by H in the assumptions at the beginning of Section 5.3

The exponents $\alpha, \beta \ge 1$ satisfy (as in [99])

$$\alpha < \begin{cases} 1+\beta, & n = 1, 2, \\ 1+\frac{2\beta}{n}, & n > 2. \end{cases}$$
(5.3.3)

On the remaining functions and parameters we make the subsequent assumptions:

- $\cdot u_0 \in C(\overline{\Omega}) \text{ and } u_0 \ge 0,$
- · $h_0 \in W^1_{\infty}(\Omega)$ and $0 \leq h_0 \leq H$, $h_0 \neq H$, where H is a positive constant,
- · μ is Lipschitz-continuous with constant L_{μ} , satisfying $0 \leq \mu$ and $0 < \delta \leq \mu(h)$ for $h \leq H$,
- $g \in C^1(\mathbb{R}^+_0 \times \mathbb{R}^+_0) \text{ with } \nabla g \in (L^\infty(\mathbb{R}^+_0 \times \mathbb{R}^+_0))^2, \ 0 \leq g(u,0) \leq G \text{ and } g(u,H) \leq 0 \text{ for } u \in \mathbb{R}^+_0,$
- · $J \in L^p(B)$ for $B := B_{\operatorname{diam}(\Omega)}(0)$ and some $p \in (1, \infty)$ and $0 < \eta \leq J$,
- $D_H > 0.$

5.3.1 Local existence in an approximate problem

The Stone-Weierstraß theorem implies that there is a sequence $(u_{0l})_{k\in\mathbb{N}} \subset C^{0,1}(\overline{\Omega}), u_{0l} \ge 0$ and

$$u_{0l} \xrightarrow[l \to \infty]{} u_0 \text{ in } C(\Omega) \tag{5.3.4}$$

and a sequence of diffusion tensors $(\mathbb{D}_l)_{l \in \mathbb{N}}$ with $\mathbb{D}_l = (d_{lij})_{i,j=1,\dots,n}$ s.t. $d_{lij} \in C^{2+\vartheta}(\overline{\Omega})$ for $\vartheta \in (0,1)$ and

$$\mathbb{D}_{l \to \infty} \mathbb{D} \text{ in } C^{1}(\overline{\Omega}; \mathbb{R}^{n \times n}).$$
(5.3.5)

Moreover, \mathbb{D}_l satisfies the uniform parabolicity condition for all $l \in \mathbb{N}$, i.e., there are $D_1 \in (0, B_1)$ and $D_2 \in (B_2, \infty)$ such that for all $\xi \in \mathbb{R}^n$, $x \in \overline{\Omega}$ and $l \in \mathbb{N}$ it holds that

$$D_1|\xi|^2 \leqslant \sum_{i,j=1}^n d_{lij}(x)\xi_j\xi_i \leqslant D_2|\xi|^2.$$
(5.3.6)

For $l \in \mathbb{N}$ we consider the approximate problem

$$\begin{aligned}
\partial_t u_l &= \nabla \nabla : \left(\mathbb{D}_l(x) u_l \right) + \nabla \cdot \left(\mathbb{D}_l(x) u_l \nabla h_l \right) + \mu(h_l) u_l^{\alpha} (1 - J * u_l^{\beta}) & \text{in } \Omega \times (0, \infty), \\
\partial_t h_l &= D_H \Delta h_l + g(u_l, h_l) & \text{in } \Omega \times (0, \infty), \\
(\mathbb{D}_l(x) \nabla u_l + \nabla \cdot \mathbb{D}_l(x) u_l + \mathbb{D}_l(x) u_l \nabla h_l) \cdot \nu &= \nabla h_l \cdot \nu = 0 & \text{on } \partial \Omega \times (0, \infty), \\
u_l(\cdot, 0) &= u_{0l}, h(\cdot, 0) = h_0 & \text{in } \Omega.
\end{aligned}$$
(5.3.7)

Lemma 5.3.1. For all $l \in \mathbb{N}$ there are $T_{max} > 0$ and a weak solution (u_l, h_l) of (5.3.7) with $u_l \ge 0$ s.t. $u_l \in C(\overline{\Omega} \times [0, T])) \cap L^2(0, T; H^1(\Omega))$ and $h_l \in L^{\infty}(0, T; W^1_{\infty}(\Omega)) \cap W^{2,1}_2(\Omega \times (0, T))$ for all $T \in (0, T_{max})$ and (u_l, h_l) satisfies

$$-\int_{0}^{T}\int_{\Omega}u_{l}\eta_{t} dx dt + \int_{0}^{T}\int_{\Omega}\left(\nabla \cdot \mathbb{D}_{l}u_{l} + \mathbb{D}_{l}\nabla u_{l} + \mathbb{D}_{l}u_{l}\nabla h_{l}\right) \cdot \nabla\eta dx dt$$
$$=\int_{0}^{T}\int_{\Omega}\mu(h_{l})u_{l}^{\alpha}(1-J*u_{l}^{\beta})\eta dx dt + \int_{\Omega}u_{0l}(x)\eta(x,0) dx,$$
(5.3.8)

for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$ and

$$\partial_t h_l = D_H \Delta h_l + g(u_l, h_l) \ a.e. \ in \ \Omega \times (0, T_{max}), \tag{5.3.9a}$$

$$\nabla h_l \cdot \nu = 0 \qquad a.e. \ on \ \partial\Omega \times (0, T_{max}), \tag{5.3.9b}$$

$$h_l(0) = h_0$$
 in $H^1(\Omega)$. (5.3.9c)

Moreover, it holds either $T_{max} = \infty$ or $T_{max} < \infty$ and

$$\lim_{t \nearrow T_{max}} \left(\|u_l(\cdot, t)\|_{L^{\infty}(\Omega)} + \|h_l(\cdot, t)\|_{W^1_{\infty}(\Omega)} \right) = \infty.$$
(5.3.10)

Proof. Fix $l \in \mathbb{N}$. We set $M := \|u_{0l}\|_{L^{\infty}(\Omega)} < \infty$. For h < 0 and $\overline{u} \ge 0$ extend the coefficients by

$$g(\overline{u},h) := 2g(\overline{u},0) - g(\overline{u},-h)$$
 and $\mu(h) := \mu(-h)$.

We show the existence of a solution (u_l, h_l) of (5.3.7) in the sense of (5.3.8) and (5.3.9a)-(5.3.9c) by showing the existence of a fixed-point of the operator F introduced below similarly to [138]. Namely, we define for some small enough $T \in (0, 1)$ the set

$$S := \{ \overline{u} \in L^{\infty}(\Omega \times (0,T)) : 0 \leq \overline{u} \leq M+1 \text{ a.e. in } \Omega \times (0,T) \}.$$

For $\overline{u} \in S$ we consider the IBVPs

$$\begin{aligned} \partial_t u_l &= \nabla \nabla : (\mathbb{D}_l(x)u_l) + \nabla \cdot (\mathbb{D}_l(x)u_l\nabla h_l) + \mu(h_l)\overline{u}^{\alpha-1}(1 - J * \overline{u}^\beta)u_l & \text{ in } \Omega \times (0,T), \\ (\mathbb{D}_l(x)\nabla u_l + \nabla \cdot \mathbb{D}_l(x)u_l + \mathbb{D}_l(x)u_l\nabla h_l) \cdot \nu &= 0 & \text{ on } \partial\Omega \times (0,T), \\ u_l(\cdot,0) &= u_{0l} & \text{ in } \Omega, \end{aligned}$$

and

$$\begin{cases} \partial_t h_l = D_H \Delta h_l + g(\overline{u}, h_l) & \text{in } \Omega \times (0, T), \\ \nabla h_l \cdot \nu = 0 & \text{on } \partial \Omega \times (0, T), \\ h_l(\cdot, 0) = h_0 & \text{in } \Omega. \end{cases}$$
(5.3.12)

Here, T can be chosen independent of \overline{u} .

Let $q > \max\{2, n\}$. Consider the space

$$X := \left\{ \overline{h} \in L^{\infty}(0,T; W^1_q(\Omega)) : |\overline{h}| \leq C_1 \|h_0\|_{W^1_q(\Omega)} + 1 \right\},$$

where C_1 depends on the Sobolev embedding constant from Lemma 2.2.8(ii), the constant from Poincaré inequality and from the constants in Lemma A.1.18(ii) and (iii). For $\overline{h} \in X$ we set

$$\Psi(\overline{h})(t) := e^{tD_H\Delta}h_0 + \int_0^t e^{(t-s)D_H\Delta}g(\overline{u},\overline{h})\,\mathrm{d}s.$$

Using estimates from Lemma A.1.18 it follows that Ψ defines a contraction on X for small enough T. With a Banach fixed-point argument in X similar to [85] we conclude that there is a unique $h_l \in X$ that satisfies

$$h_l(t) = \Psi(h_l)(t) = e^{tD_H\Delta}h_0 + \int_0^t e^{(t-s)D_H\Delta}g(\overline{u}, h_l) \,\mathrm{d}s.$$
(5.3.13)
Moreover, h_l is the unique weak solution of (5.3.12) in the sense of *Theorem A.1.1*, i.e., for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$ it holds that

$$-\int_0^T \int_\Omega h_l \eta_t \,\mathrm{d}x \,\mathrm{d}t + D_H \int_0^T \int_\Omega \nabla h_l \cdot \nabla \eta \,\mathrm{d}x \,\mathrm{d}t = \int_0^T \int_\Omega g(\bar{u}, h_l) \eta \,\mathrm{d}x \,\mathrm{d}t + \int_\Omega h_0(x) \eta(x, 0) \,\mathrm{d}x.$$
(5.3.14)

Estimating as in Lemma 5.3.3 below it follows from Lemma A.1.18(ii) and (iii) that

$$\|\nabla h_l\|_{L^{\infty}(0,T;(L^{\infty}(\Omega))^n)} \leqslant C_2.$$
(5.3.15)

Moreover, from Theorem A.1.8 applied to the equation in non-divergence form we conclude that

$$\|h_l\|_{W_2^{2,1}(\Omega \times (0,T))} \leqslant C_3 \tag{5.3.16}$$

and solves (5.3.12) in the sense of (5.3.9a) - (5.3.9c). Moreover, the continuity of h_l follows from Remark A.1.2 and Theorem A.1.12 with

$$a(\nabla h_l) := D_H \nabla h_l, \ b(x, t, h_l) := -g(\overline{u}, h_l)$$
(5.3.17)

due the embedding of $W^1_{\infty}(\Omega)$ into some Hölder space on $\overline{\Omega}$ from Lemma 2.2.8(ii). Now, Theorems A.1.1 and A.1.4 with

$$a_{ij}(x) := d_{lij}(x),$$

$$a_i(x,t) := \sum_{j=1}^n \left((d_{lij})_{x_j} + d_{lij}(h_l)_{x_j} \right),$$

$$a(x,t) := -\mu(h_l)\overline{u}^{\alpha-1}(1 - J * \overline{u}^{\beta}),$$

$$f := 0$$
(5.3.18)

(that are due to (5.3.5), (5.3.15), Lemma 2.3.2 and the Lipschitz continuity of μ all bounded in $L^{\infty}(\Omega \times (0,T))$ by constants independent from \overline{u}) imply that there is a unique weak solution $u_l \in C([0,T]; L^2(\Omega)) \cap L^2(0,T; H^1(\Omega))$ of (5.3.11) satisfying

$$-\int_{0}^{T}\int_{\Omega}u_{l}\eta_{t}\,\mathrm{d}x\,\mathrm{d}t + \int_{0}^{T}\int_{\Omega}\left(\nabla\cdot\mathbb{D}_{l}u_{l} + \mathbb{D}_{l}\nabla u_{l} + \mathbb{D}_{l}u_{l}\nabla h_{l}\right)\cdot\nabla\eta\,\mathrm{d}x\,\mathrm{d}t$$
$$=\int_{0}^{T}\int_{\Omega}\mu(h_{l})\overline{u}^{\alpha-1}(1-J*\overline{u}^{\beta})u_{l}\eta\,\mathrm{d}x\,\mathrm{d}t + \int_{\Omega}u_{0l}(x)\eta(x,0)\,\mathrm{d}x \tag{5.3.19}$$

for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$ and $||u_l||_{L^{\infty}(\Omega \times (0,T))} \leq C_4$. Moreover, we conclude from *Remark A.1.2* and *Theorem A.1.12* with

$$a(x,t,u,\nabla u) := \mathbb{D}_l \nabla u + \nabla \cdot \mathbb{D}_l u + \mathbb{D}_l u \nabla h_l,$$

$$b(x,t,u) := -\mu(h_l)\overline{u}^{\alpha-1}(1-J*\overline{u}^{\beta})u \qquad (5.3.20)$$

requiring the conditions of the theorem especially due to (5.3.6) and the boundedness of the coefficients that

$$\|u_l\|_{C^{\kappa,\frac{\kappa}{2}}(\overline{\Omega}\times[0,T])} \leqslant C_5 \tag{5.3.21}$$

for some $\kappa \in (0, 1)$ and C_5 depending on l. Hence, we can estimate for $x \in \Omega$ and $t \in (0, T)$ that

$$u_{l}(x,t) = u_{0l}(x) + t^{\frac{\kappa}{2}} \frac{u_{l}(x,t) - u_{0l}(x)}{t^{\frac{\kappa}{2}}} \leqslant M + T^{\frac{\kappa}{2}}C_{5} \leqslant M + 1$$

holds for a small enough T. We conclude as for (5.3.26) below with $(u_l)_- := \max\{0, -u_l\}$ that for $t \in (0, T)$ it holds that

$$\frac{1}{2} \| (u_l)_{-}(t) \|_{L^2(\Omega)}^2 + \frac{D_1}{2} \int_0^t \| \nabla (u_l)_{-} \|_{(L^2(\Omega))^n}^2 \, \mathrm{d}s \leqslant C_6 \int_0^t \| (u_l)_{-} \|_{L^2(\Omega)}^2 \, \mathrm{d}s.$$

Then, Gronwall's inequality implies that $u_l \ge 0$ and $u_l \in S \cap C^{\kappa, \frac{\kappa}{2}}(\overline{\Omega} \times [0, T])$. Note that C_2 , C_3 , C_4 and C_5 and consequently also the choice of T are independent from \overline{u} and k. Hence, the operator

$$F: S \mapsto S, \quad \overline{u} \mapsto u_l,$$

where u_l solves (5.3.11) for \overline{u} in the sense of (5.3.19), is well-defined. Moreover, due to the compact embedding $C^{\kappa,\frac{\kappa}{2}}(\overline{\Omega} \times [0,T]) \hookrightarrow C(\overline{\Omega} \times [0,T])$ that is a consequence of the Arzelà-Ascoli theorem, F maps bounded sets on precompact ones. To apply Schauder's fixed-point theorem (*Theorem A.2.2*) it remains to show that F is closed and, consequently, a compact operator. Consider a sequence $(\overline{u}_m)_{m\in\mathbb{N}}$ s.t.

$$\overline{u}_m \underset{m \to \infty}{\to} \overline{u} \text{ in } L^{\infty}(\Omega \times (0,T)), \tag{5.3.22}$$

$$u_{lm} := F(\overline{u}_m) \underset{m \to \infty}{\to} u_l \text{ in } L^{\infty}(\Omega \times (0,T)).$$
(5.3.23)

We want to show that $F(\overline{u}) = u_l$.

Let h_{lm} be the solution of (5.3.12) that corresponds to \overline{u}_m for $m \in \mathbb{N}$. From the equation of form (5.3.19) for u_{lm} we conclude with Lemma A.1.3 that for a.e. $t \in (0,T)$ it holds that

$$\frac{1}{2} \|u_{lm}(\cdot,t)\|_{L^{2}(\Omega)}^{2} + \int_{0}^{t} \int_{\Omega} (\nabla \cdot \mathbb{D}_{l} u_{lm} + \mathbb{D}_{l} \nabla u_{lm} + \mathbb{D}_{l} u_{lm} \nabla h_{lm}) \cdot \nabla u_{lm} \, \mathrm{d}x \, \mathrm{d}s$$

$$= \int_{0}^{t} \int_{\Omega} \mu(h_{lm}) \overline{u}_{m}^{\alpha-1} (1 - J * \overline{u}_{m}^{\beta}) u_{lm}^{2} \, \mathrm{d}x \, \mathrm{d}s + \frac{1}{2} \|u_{0l}\|_{L^{2}(\Omega)}^{2}.$$
(5.3.24)

Using Hölder's and Young's inequalities, (5.3.5) and (5.3.15), we estimate

$$\left| \int_{\Omega} \mathbb{D}_{l} u_{lm} \nabla h_{lm} \cdot \nabla u_{lm} \, \mathrm{d}x \right| \leq C_{2} \|\mathbb{D}_{l}\|_{(L^{\infty}(\Omega))^{n \times n}} \|u_{lm}\|_{L^{2}(\Omega)} \|\nabla u_{lm}\|_{(L^{2}(\Omega))^{n}} \\ \leq C_{7} \|u_{lm}\|_{L^{2}(\Omega)}^{2} + \frac{D_{1}}{2} \|\nabla u_{lm}\|_{(L^{2}(\Omega))^{n}}^{2}.$$
(5.3.25)

Inserting this into (5.3.24) and using (5.3.5) and (5.3.6), Lemma 2.3.2, Young's inequality, the boundedness of h_l and the Lipschitz-continuity of μ , we conclude that for a.e. $t \in (0, T)$ it holds

$$\frac{1}{2} \|u_{lm}(\cdot,t)\|_{L^2(\Omega)}^2 + \frac{D_1}{4} \int_0^t \|\nabla u_{lm}\|_{(L^2(\Omega))^n}^2 \,\mathrm{d}s \leqslant C_8 \int_0^t \|u_{lm}\|_{L^2(\Omega)}^2 + C_9.$$
(5.3.26)

From Gronwall's inequality it follows that $\|\nabla u_{lm}\|_{L^2(0,T;(L^2(\Omega))^n)} \leq C_{10}(T)$ for all $m \in \mathbb{N}$. Combining this with the fact that $h_{lm} \in X$, (5.3.15) and (5.3.16) we conclude from the Lions-Aubin lemma (*Lemma A.3.9*) and the Banach-Alaoglu theorem (*Lemma A.3.1*) that there is $h_l \in L^{\infty}(0,T; W^1_{\infty}(\Omega)) \cap W^{2,1}_2(\Omega \times (0,T))$ s.t. for a subsequence

$$\nabla u_{lm_o} \xrightarrow[o \to \infty]{} \nabla u_l \text{ in } L^2(0,T; (L^2(\Omega))^n), \tag{5.3.27}$$

$$h_{lm_o} \xrightarrow[q \to \infty]{} h_l \text{ in } L^2(0, T; H^2(\Omega)), \tag{5.3.28}$$

$$h_{lm_o} \xrightarrow[o \to \infty]{} h_l \text{ in } L^2(0,T; H^1(\Omega)) \text{ and a.e. in } \Omega \times (0,T),$$
 (5.3.29)

$$\partial_t h_{lm_o} \underset{o \to \infty}{\longrightarrow} \partial_t h_l \text{ in } L^2(\Omega \times (0,T)).$$
 (5.3.30)

Therefore, due to (5.3.22) and (5.3.28)–(5.3.30), the Lipschitz-continuity of g, the dominated convergence theorem and the fundamental lemma of calculus of variations it follows that h_l is a solution of (5.3.12) in the sense of (5.3.9a)-(5.3.9c). Moreover, (5.3.5), (5.3.22), (5.3.23), (5.3.27), and (5.3.29), the Lipschitz-continuity of μ and the dominated convergence theorem imply that u_l is a solution of (5.3.11) in the sense of (5.3.8), and therefore, $F(\overline{u}) = u_l$ and F is a compact operator. Consequently, by Schauder's fixed-point theorem we obtain the existence of a fixedpoint u_l of F, that satisfies for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$ the weak formulation (5.3.8).

Finally, for such pair property (5.3.10) follows from a standard extensibility argument.

Theorem 5.3.2. There is $T_{max} \in (0, \infty]$ and a solution (u_l, h_l) of (5.3.7) with $0 \leq u_l$ and $0 \leq h_l < H$ and

$$u_l, h_l \in C(\overline{\Omega} \times [0, T_{max})) \cap C^{2,1}(\overline{\Omega} \times (0, T_{max})).$$
(5.3.31)

The solution is unique if $p \ge \frac{2n}{n+2}$ for $n \ge 3$ and $p \in (1, \infty)$ for n = 1, 2.

Proof. 1. Regularity: Let $l \in \mathbb{N}$, $0 < T_1 < T_{max}$ and consider the weak solution (u_l, h_l) from Lemma 5.3.1. Again from Remark A.1.2 and Theorem A.1.12 with a and b as in (5.3.17) and (5.3.20), respectively, it follows that $u_l \in C^{\lambda, \frac{\lambda}{2}}(\overline{\Omega} \times (0, T_1])$ and $h_l \in C^{\lambda, \frac{\lambda}{2}}(\overline{\Omega} \times [0, T_1])$ for some $\lambda \in (0, 1)$. Combining this with the Lipschitz continuity of g, Theorem A.1.5 with $a_{ii} := D_H$ and f := -g(u, h) and all other coefficients equal to zero implies $h_l \in C^{2+\lambda, 1+\frac{\lambda}{2}}(\Omega \times (0, T_1))$.

Let $t_0 \in (0, T_!)$. We consider $\xi \in C^{\infty}(\mathbb{R}, [0, 1])$ satisfying $\xi \equiv 0$ on $(-\infty, \frac{t_0}{2}]$ and $\xi \equiv 1$ on $[t_0, \infty)$. Then, $\tilde{h}(x, t) := h_l(x, t)\xi(t) \in W_2^{2,1}(\Omega \times (0, T_1))$ is a strong solution of the IBVP

$$\begin{cases} \partial_t \tilde{h} = D_H \Delta \tilde{h} + g(u_l, h_l) \xi + h_l \xi' & \text{in } \Omega \times (0, T_1), \\ \partial_\nu \tilde{h} = 0 & \text{on } \partial \Omega \times (0, T_1), \\ \tilde{h}(\cdot, 0) = 0 & \text{in } \Omega. \end{cases}$$

Due to the Lipschitz continuity of g and the boundedness of u_l on $\overline{\Omega} \times [0, T_1]$, Theorem A.1.13 with

$$a(\nabla h_{lk}) := D_H \nabla h_{lk},$$

$$b(x,t) := -g(u_l(x,t), h_l(x,t))\xi(t) - h_l(x,t)\xi'(t)$$

especially implies that $\nabla \tilde{h} \in C(\overline{\Omega} \times [0, T_1], \mathbb{R}^n)$ and consequently, h_l is a classical solution of the nonhomogeneous heat equation in (5.3.7) on $\Omega \times (0, T_1)$. Finally, using again the Lipschitz continuity of g and the Hölder continuity of u_l and h_l , Theorem A.1.7 implies that \tilde{h} is also in $C^{2,1}(\overline{\Omega} \times [0, T_1])$ which leads to $h_l \in C^{2,1}(\overline{\Omega} \times (0, T_{max}))$. Analogously, $u_l \in C^{2,1}(\overline{\Omega} \times (0, T_{max}))$ follows.

The boundedness by H of h_l follows applying Proposition A.1.14 on $\Omega \times (0, T_1)$ using the boundedness of h_l on the closure of this set, the fact that $h \in W_2^{2,1}(\Omega \times (0, T_1))$, our assumptions on gand the estimate

$$\partial_t (h_l - H) - D_H \Delta(h_l - H) = g(u_l, h_l) = \partial_h g(\zeta)(h_l - H) + g(u_l, H) \leqslant \partial_h g(\zeta)(h_l - H)$$

that holds for all $(x, t) \in \Omega \times (0, T_1)$ for some $\zeta(x, t) \in (h_l(x, t), H)$ due to the mean value theorem. Analogously, the nonnegativity of h follows.

2. Uniqueness: Let $p \ge \frac{2n}{n+2}$ for $n \ge 3$ or $p \in (1, \infty)$ for n = 1, 2. With an ansatz similar to [13] we want to show the uniqueness of the solution. Assume that there are two solutions $(u_1, h_1), (u_2, h_2)$ of (5.3.7) for $l \in \mathbb{N}$ with the regularity from (5.3.31). The functions h_1 and h_2 satisfy

$$\partial_t (h_1 - h_2) = D_H \Delta (h_1 - h_2) + g(u_1, h_1) - g(u_2, h_2)$$

in $\Omega \times (0, T_1)$. We multiply this equation by $h_1 - h_2$ and integrate over Ω . Then, using partial integration, the Lipschitz continuity of g, and Young's inequality we conclude from Gronwall's inequality that

$$\|h_1 - h_2\|_{L^{\infty}(0,t;L^2(\Omega))}, \|\nabla(h_1 - h_2)\|_{L^2(0,t;(L^2(\Omega))^n)} \leq C_{11}(T_1)\|u_1 - u_2\|_{L^2(\Omega \times (0,t))}$$
(5.3.32)

for $t \in (0, T_1)$. Moreover, we can rewrite

$$\partial_{t}(u_{1} - u_{2}) = \nabla \nabla : (\mathbb{D}_{l}(u_{1} - u_{2})) + \nabla \cdot (\mathbb{D}_{l}(u_{1} \nabla h_{1} - u_{2} \nabla h_{2})) + \mu(h_{1})u_{1}^{\alpha}(1 - J * u_{1}^{\beta}) - \mu(h_{2})u_{2}^{\alpha}(1 - J * u_{2}^{\beta}) = \nabla \nabla : (\mathbb{D}_{l}(u_{1} - u_{2})) + \nabla \cdot (\mathbb{D}_{l}(u_{1} - u_{2}) \nabla h_{1}) + \nabla \cdot (\mathbb{D}_{l}u_{2} \nabla (h_{1} - h_{2})) + (\mu(h_{1}) - \mu(h_{2}))u_{1}^{\alpha}(1 - J * u_{1}^{\beta}) + \mu(h_{2})(u_{1}^{\alpha} - u_{2}^{\alpha})(1 - J * u_{1}^{\beta}) + \mu(h_{2})u_{2}^{\alpha}J * (u_{2}^{\beta} - u_{1}^{\beta}).$$
(5.3.33)

With the boundedness of u_2 on $\Omega \times (0, T_1)$ by some $C_{12}(T_1, l) > 0$, Hölder's inequality, the mean value theorem and the Sobolev embedding from Lemma 2.2.8(i) we estimate

$$\begin{aligned} |J*(u_{2}^{\beta}-u_{1}^{\beta})| &\leq \beta C_{12}^{\beta-1}(T_{1},l) \int_{\Omega} |J(x-y)| |u_{1}(y) - u_{2}(y)| \,\mathrm{d}y \\ &\leq \beta C_{12}^{\beta-1}(T_{1},l) \|J\|_{L^{p}(B)} \|u_{1} - u_{2}\|_{L^{\frac{p}{p-1}}(\Omega)} \leq C_{13}(T_{1},l) \|u_{1} - u_{2}\|_{H^{1}(\Omega)}. \end{aligned}$$
(5.3.34)

Again, we multiply (5.3.33) by $u_1 - u_2$ and integrate over Ω for $t \in (0, T_1)$. Then, using partial integration together with Young's and Hölder's inequality, the mean value theorem, the Lipschitz continuity of μ , Lemma 2.3.2(i), the boundedness of u_1 and u_2 , the boundedness of ∇h_1 by some $C_{14}(T_1, l) > 0$ and (5.3.6) and (5.3.34), it follows that

$$\begin{split} &\frac{1}{2} \frac{d}{dt} \|u_1 - u_2\|_{L^2(\Omega)}^2 + D_1 \|\nabla(u_1 - u_2)\|_{(L^2(\Omega))^n}^2 \\ \leqslant &- \int_{\Omega} \nabla \cdot \mathbb{D}_l (u_1 - u_2) \nabla(u_1 - u_2) + (u_1 - u_2) (\mathbb{D}_l \nabla h_1) \cdot \nabla(u_1 - u_2) \\ &+ u_2 (\mathbb{D}_l \nabla (h_1 - h_2)) \cdot \nabla(u_1 - u_2) \, dx \\ &+ \int_{\Omega} [(\mu(h_1) - \mu(h_2)) u_1^{\alpha} (1 - J * u_1^{\beta}) + \mu(h_2) (u_1^{\alpha} - u_2^{\alpha}) (1 - J * u_1^{\beta}) \\ &+ \mu(h_2) u_2^{\alpha} J * (u_2^{\beta} - u_1^{\beta})] (u_1 - u_2) \, dx \\ \leqslant (\|\nabla \cdot \mathbb{D}_l\|_{(L^{\infty}(\Omega))^n} + \|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}} \|\nabla h_1\|_{(L^{\infty}(\Omega))^n}) \int_{\Omega} |u_1 - u_2| |\nabla(u_1 - u_2)| \, dx \\ &+ \|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}} C_{12}(T_1, l) \int_{\Omega} |\nabla(h_1 - h_2)| |\nabla(u_1 - u_2)| \, dx \\ &+ L_{\mu} C_{12}^{\alpha}(T_1, l) (1 + \|J\|_{L^1(B)} C_{12}^{\beta}(T_1, l)) \int_{\Omega} |h_1 - h_2| |u_1 - u_2| \, dx \end{split}$$

$$+ \|\mu\|_{L^{\infty}(0,H)} \alpha C_{12}^{\alpha-1}(T_{1},l)(1+\|J\|_{L^{1}(B)}C_{12}^{\beta}(T_{1},l)) \int_{\Omega} |u_{1}-u_{2}|^{2} dx$$

$$+ \|\mu\|_{L^{\infty}(0,H)}C_{12}^{\alpha}(T_{1},l)C_{13}(T_{1},l)\|u_{1}-u_{2}\|_{H^{1}(\Omega)} \int_{\Omega} |u_{1}-u_{2}| dx$$

$$\leq C_{15}(T_{1},l) \left(\|u_{1}-u_{2}\|_{L^{2}(\Omega)}^{2}+\|h_{1}-h_{2}\|_{L^{2}(\Omega)}^{2}+\|\nabla(h_{1}-h_{2})\|_{(L^{2}(\Omega))^{n}}^{2}\right)$$

$$+ D_{1}\|\nabla(u_{1}-u_{2})\|_{(L^{2}(\Omega))^{n}}^{2}.$$

Integrating over (0,t) for $t \in (0,T_1)$ and using (5.3.32) we conclude that for a.e. $t \in (0,T_1)$ it holds that

$$\begin{split} \|u_1 - u_2\|_{L^2(\Omega)}^2 \leqslant & C_{16}(T_1, l) \left(\int_0^t \|u_1 - u_2\|_{L^2(\Omega)}^2 + \|\nabla(h_1 - h_2)\|_{(L^2(\Omega))^n}^2 + \|h_1 - h_2\|_{L^2(\Omega)}^2 \right) \\ \leqslant & C_{17}(T_1, l) \int_0^t \|u_1 - u_2\|_{L^2(\Omega)}^2 \, \mathrm{d}t. \end{split}$$

Consequently, combining this with Gronwall's inequality and (5.3.32) implies that $u_1 \equiv u_2$ and $h_1 \equiv h_2$ a.e. on $\Omega \times (0, T_1)$.

5.3.2 Global existence and boundedness of *u* in the approximate problem

Lemma 5.3.3. There is a positive constant C_{18} independent from $l \ s.t.$

$$\|\nabla h_l\|_{L^{\infty}(0,T_{max};(L^{\infty}(\Omega))^n)} \leq C_{18}$$

holds for all $l \in \mathbb{N}$.

Proof. Let $l \in \mathbb{N}$. We have shown in Lemma 5.3.1 that h_l satisfies

$$h_l(t) = \Psi(h_l)(t) = e^{tD_H\Delta}h_0 + \int_0^t e^{(t-s)D_H\Delta}g(u_l, h_l) \,\mathrm{d}s$$

for $t \in (0, T_{max})$. With Lemma A.1.18(ii) and (iii) and (5.3.13) we estimate that

$$\begin{split} \|\nabla h_l(t)\|_{(L^q(\Omega))^n} \leqslant & K_{12} e^{-\lambda_1 D_H t} \|\nabla h_0\|_{(L^q(\Omega))^n} \\ &+ K_{11} |\Omega|^{\frac{1}{q}} \int_0^t \left(1 + \frac{1}{(D_H(t-s))^{\frac{1}{2}}}\right) e^{-\lambda_1 D_H(t-s)} \|g(u_l, h_l)\|_{L^{\infty}(\mathbb{R}^+_0 \times \mathbb{R}^+_0)} \, \mathrm{d}s \end{split}$$

holds for all $q \in (1, \infty)$, where λ_1 is the first positive eigenvalue of $-\Delta$ on Ω with Neumann boundary condition. Using the properties of g, the uniform boundedness of (h_l) and Hölder's inequality, we conclude that

$$\begin{split} \|\nabla h_{l}(t)\|_{(L^{q}(\Omega))^{n}} \leqslant & K_{12}|\Omega|^{\frac{1}{q}} \|\nabla h_{0}\|_{(L^{\infty}(\Omega))^{n}} \\ &+ K_{11}|\Omega|^{\frac{1}{q}} (\|\partial_{h}g\|_{L^{\infty}(\mathbb{R}^{+}_{0}\times\mathbb{R}^{+}_{0})}H + G) \int_{0}^{t} \left(1 + \frac{1}{(D_{H}(t-s))^{\frac{1}{2}}}\right) e^{-\lambda_{1}D_{H}(t-s)} \,\mathrm{d}s \\ \leqslant & K_{12}|\Omega|^{\frac{1}{q}} \|\nabla h_{0}\|_{(L^{\infty}(\Omega))^{n}} + \frac{K_{11}|\Omega|^{\frac{1}{q}} (\|\partial_{h}g\|_{L^{\infty}(\mathbb{R}^{+}_{0}\times\mathbb{R}^{+}_{0})}H + G)}{D_{H}} \left(\frac{1}{\lambda_{1}} + \frac{\sqrt{\pi}}{\sqrt{\lambda_{1}}}\right). \end{split}$$

Consequently,

$$\|\nabla h_l\|_{(L^{\infty}(\Omega))^n} = \lim_{q \to \infty} \|\nabla h_l\|_{(L^q(\Omega))^n}$$

$$\leq K_{12} \|\nabla h_0\|_{(L^{\infty}(\Omega))^n} + \frac{K_{11}(\|\partial_h g\|_{L^{\infty}(\mathbb{R}^+_0 \times \mathbb{R}^+_0)}H + G)}{D_H} \left(\frac{1}{\lambda_1} + \frac{\sqrt{\pi}}{\sqrt{\lambda_1}}\right) =: C_{18}$$

We will show the global boundedness of u_l as in the proof of Theorem 1.1 in [99]. For a more detailed proof of the boundedness of the corresponding solution see *Chapter 6*, where the same method will be used.

Lemma 5.3.4. For all $l \in \mathbb{N}$ and $q \in [1, \infty)$ it holds that $u_l \in L^{\infty}(0, T_{max}; L^q(\Omega))$.

Proof. Let $l \in \mathbb{N}$ and $q > \max\{1, \beta + \alpha - 1\}$. Due to (5.3.31) the terms in the estimates below are well-defined for a.e. $t \in (0, T_{max})$. Multiplying the first equation of (5.3.7) by qu_l^{q-1} , integrating over Ω and using partial integration, we obtain

$$\frac{d}{dt} \int_{\Omega} u_l^q \, \mathrm{d}x = q \int_{\Omega} \nabla \cdot \left(\mathbb{D}_l \nabla u_l + (\nabla \cdot \mathbb{D}_l) u_l + \mathbb{D}_l u_l \nabla h_l \right) u_l^{q-1} + \mu(h_l) u_l^{q-1+\alpha} (1 - J * u_l^{\beta}) \, \mathrm{d}x$$

$$= -q(q-1) \int_{\Omega} u_l^{q-2} (\mathbb{D}_l \nabla u_l) \cdot \nabla u_l + u_l^{q-1} (\nabla \cdot \mathbb{D}_l) \cdot \nabla u_l + u_l^{q-1} (\mathbb{D}_l \nabla h_l) \cdot \nabla u_l \, \mathrm{d}x$$

$$+ q \int_{\Omega} \mu(h_l) u_l^{q-1+\alpha} (1 - J * u_l^{\beta}) \, \mathrm{d}x.$$
(5.3.35)

Using the uniform parabolicity of \mathbb{D}_l and $\nabla u^{\frac{q}{2}} = \frac{q}{2}u^{\frac{q}{2}-1}\nabla u$, we estimate

$$\begin{aligned} q(q-1)\int_{\Omega}u_l^{q-2}(\mathbb{D}_l\nabla u_l)\cdot\nabla u_l\,\mathrm{d}x &= \frac{4(q-1)}{q}\sum_{i,j=1}^n\int_{\Omega}d_{lij}\left(u_l^{\frac{q}{2}}\right)_{x_i}\left(u_l^{\frac{q}{2}}\right)_{x_j}\,\mathrm{d}x\\ &\geqslant \frac{4(q-1)}{q}D_1\int_{\Omega}|\nabla u_l^{\frac{q}{2}}|^2\,\mathrm{d}x. \end{aligned}$$

Further, due to Young's inequality and Lemma 5.3.3 we obtain the estimate

$$\begin{split} q(q-1) \left| \int_{\Omega} u_l^{q-1} (\nabla \cdot \mathbb{D}_l) \cdot \nabla u_l + u_l^{q-1} (\mathbb{D}_l \nabla h_l) \cdot \nabla u_l \, \mathrm{d}x \right| \\ \leqslant & 2(q-1) \left(\|\nabla \cdot \mathbb{D}_l\|_{(L^{\infty}(\Omega))^n} + \|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}} \|\nabla h_l\|_{(L^{\infty}(\Omega))^n} \right) \int_{\Omega} u_l^{\frac{q}{2}} |\nabla u_l^{\frac{q}{2}}| \, \mathrm{d}x \\ \leqslant & \frac{2(q-1)}{q} D_1 \int_{\Omega} |\nabla u_l^{\frac{q}{2}}|^2 \, \mathrm{d}x \\ & + \frac{q(q-1)}{D_1} \left(\|\nabla \cdot \mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}}^2 + \|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}}^2 C_{18}^2 \right) \int_{\Omega} u_l^q \, \mathrm{d}x. \end{split}$$

Inserting these estimates into (5.3.35) and using our assumptions on μ and J, the boundedness of h_l , and

$$\int_{\Omega} u_l^q \,\mathrm{d}x \leqslant \int_{\Omega} u_l^{q+\alpha-1} \,\mathrm{d}x + |\Omega|, \tag{5.3.36}$$

it follows that

$$\frac{d}{dt} \int_{\Omega} u_l^q \, \mathrm{d}x + \frac{2(q-1)}{q} D_1 \int_{\Omega} |\nabla u_l^{\frac{q}{2}}|^2 \, \mathrm{d}x + q\eta \delta \int_{\Omega} u_l^{q-1+\alpha} \, \mathrm{d}x \int_{\Omega} u_l^{\beta} \, \mathrm{d}x$$

$$\leq q C_{19}(q,l) \left(\int_{\Omega} u_l^{q-1+\alpha} \, \mathrm{d}x + |\Omega| \right),$$
(5.3.37)

where

$$C_{19}(q,l) := \frac{q-1}{D_1} \left(\|\nabla \cdot \mathbb{D}_l\|_{(L^{\infty}(\Omega))^n}^2 + \|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}}^2 C_{18}^2 \right) + \|\mu\|_{L^{\infty}(0,H)}.$$

Adding $qC_{19}(q,l) \|u_l\|_{L^q}^q$ on both sides of (5.3.37) and using (5.3.36) one more time, we obtain

$$\frac{d}{dt} \int_{\Omega} u_l^q \, \mathrm{d}x + qC_{19}(q,l) \int_{\Omega} u_l^q \, \mathrm{d}x + 2\frac{q-1}{q} D_1 \int_{\Omega} |\nabla u_l^{\frac{q}{2}}|^2 \, \mathrm{d}x + q\delta\eta \int_{\Omega} u_l^{q-1+\alpha} \, \mathrm{d}x \int_{\Omega} u_l^{\beta} \, \mathrm{d}x \\
\leqslant 2qC_{19}(q,l) \left(\int_{\Omega} u_l^{q-1+\alpha} \, \mathrm{d}x + |\Omega| \right).$$
(5.3.38)

It follows from Lemma A.4.1 with $K_{18} = \frac{2C_{19}}{D_1}$ and $K_{22} = \frac{2C_{19}}{\delta \eta}$ that

1

$$\begin{split} 2qC_{19}(q,l) \int_{\Omega} u_l^{q-1+\alpha} \leqslant & 2\frac{q-1}{q} D_1 \int_{\Omega} |\nabla u_l^{\frac{q}{2}}|^2 \,\mathrm{d}x + q\delta\eta \int_{\Omega} u_l^{q-1+\alpha} \,\mathrm{d}x \int_{\Omega} u_l^{\beta} \,\mathrm{d}x \\ &+ 2qC_{19}(q,l) K_{23}(q,l), \end{split}$$

where

$$\begin{split} K_{23}(q,l) &:= \left(2 \left(\frac{2K_{21}^2 q^2 C_{19}(q,l)}{(q-1)D_1} \right)^{\frac{q+\alpha-1-\beta}{q-\alpha+1+\beta-2\frac{q+\alpha-1+\beta}{s}}} + K_{24}(q)^{\frac{q+\alpha-1-\beta}{q-\frac{q+\alpha-1+\beta}{s}}} \right)^{\frac{q-\alpha+1+\beta-\frac{2(q+\alpha-1+\beta)}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}} \\ & \cdot \left(\frac{2C_{19}(q,l)}{\delta\eta} \right)^{\frac{q-2(q+\alpha-1)}{\beta+1-\alpha-\frac{2\beta}{s}}} K_{24}(q)^{\frac{q+\alpha-1-\beta}{q-\frac{q+\alpha-1+\beta}{s}}} \end{split}$$

with

$$K_{21} := 2K_S(s)(1 + 2K_P),$$

$$K_{24}(q) := 4K_S(s)|\Omega|^{\frac{1}{2} - \frac{q}{q + \alpha - 1 + \beta}}.$$

Here, $K_S(s) > 0$ denotes the Sobolev embedding constant from $H^1(\Omega)$ into $L^s(\Omega)$ from Lemma 2.2.8(i), $K_P > 0$ the constant from the Poincaré inequality, and

$$s \begin{cases} = \infty, & n = 1, \\ \in \left(\frac{2(q+\alpha-1+\beta)}{q-\alpha+1+\beta}, \infty\right), & n = 2, \\ = \frac{2n}{n-2}, & n > 2. \end{cases}$$
(5.3.39)

Hence, for $t \in (0, T_{max})$ we conclude that

$$\frac{d}{dt} \|u_l\|_{L^q(\Omega)}^q + qC_{19}(q,l)\|u_l\|_{L^q(\Omega)}^q \leq 2qC_{19}(q,l)(K_{23}(q,l) + |\Omega|)$$
(5.3.40)

and obtain for $t \in (0, T_{max})$ from (5.3.40) and setting $K_{14} = qC_{19}$ and $K_{15} = 2(K_{23} + |\Omega|)$ in Lemma A.1.20 the upper bound

$$\|u_{l}(\cdot,t)\|_{L^{q}(\Omega)} \leq \sqrt[q]{2K_{23}(q,l)+2|\Omega|+\|u_{0l}\|_{L^{q}(\Omega)}^{q}} \leq \sqrt[q]{2K_{23}(q,l)+|\Omega|\left(2+\|u_{0l}\|_{L^{\infty}(\Omega)}^{q}\right)}.$$
(5.3.41)

Remark 5.3.5. As in [99] we cannot directly conclude from *Lemma 5.3.4* that u_l is bounded on $\Omega \times (0, T_{max})$ as

$$\lim_{q \to \infty} \sqrt[q]{2K_{23}(q,l)} + |\Omega| \left(2 + \|u_{0l}\|_{L^{\infty}(\Omega)}^{q}\right) = \infty$$

due to

$$\left(\left(\frac{q^2 C_{19}(q,l)}{q-1} \right)^{\frac{q+\alpha-1-\beta}{q-\alpha+1+\beta-2\frac{q+\alpha-1+\beta}{s}} \cdot \frac{q-\alpha+1+\beta-\frac{2(q+\alpha-1+\beta)}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}} \right)^{\frac{1}{q}} \\ \ge \left(\|\mu\|_{L^{\infty}(0,H)} q \right)^{\frac{1}{q} \cdot \frac{q+\alpha-1-\beta}{\beta+1-\alpha-\frac{2\beta}{s}}} = \left(\left(\|\mu\|_{L^{\infty}(0,H)} q \right)^{1+\frac{\alpha-1-\beta}{q}} \right)^{\frac{1}{\beta+1-\alpha-\frac{2\beta}{s}}} \xrightarrow[q \to \infty]{} \infty.$$

Theorem 5.3.6. For all $l \in \mathbb{N}$ there is a unique bounded and nonnegative solution (u_l, h_l) of (5.3.7) consisting of nonnegative functions

$$u_l, h_l \in C(\overline{\Omega} \times [0, \infty)) \cap C^{2,1}(\overline{\Omega} \times (0, \infty)).$$

Thereby, $h_l < H$ and there is some $C_{20}(u_{0l}) > 0$ s.t. $u_l \leq C_{20}(u_{0l})$ and

$$C_{20}(u_{0l}) \xrightarrow[l \to \infty]{} C_{21}. \tag{5.3.42}$$

Moreover, for K > 1 and some 'small' enough choice of parameters of type (5.3.47) and (5.3.48) below, it holds that

$$\|u_l\|_{L^{\infty}(\Omega\times(0,\infty))} \leqslant K \max\left\{1, \|u_{0l}\|_{L^{\infty}(\Omega)}, \left(4K_S(s)|\Omega|^{-\frac{1}{2}}\right)^{\frac{1-\frac{2}{s}}{(1-\frac{1}{s})(\beta+1-\alpha-\frac{2\beta}{s})}} \left(\frac{2}{\delta\eta}\right)^{\frac{1-\frac{2}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}}\right\}.$$
(5.3.43)

If Ω is convex, the constant $K_S(s)$ is explicitly given in Remark 6.3.4 in Chapter 6.

Proof. Let $l \in \mathbb{N}$. We proceed with a Moser iteration as in Step 2 of the proof of Theorem 1.1. in [99].

Set $q_k := 2^k + a$ with $a := \frac{2(s-1)(\alpha-1)}{s-2}$ for $k \in \mathbb{N}$ large enough s.t. $q_k > \max\{1, \beta + \alpha - 1\}$ holds. As in Step 2 of Lemma 6.3.2 we obtain for $t \in (0, T_{max})$ the estimate

$$\frac{d}{dt} \|u_l\|_{L^{q_k}(\Omega)}^{q_k} + q_k C_{19}(q_k, l) \|u_l\|_{L^{q_k}(\Omega)}^{q_k} \le 2q_k C_{19}(q_k, l) C_{22}(q_k, l) \max\left\{1, \|u_l\|_{L^{q_{k-1}}}^{2q_{k-1}}\right\}, \quad (5.3.44)$$

where

$$C_{22}(q_k, l) := 2\left(\frac{2K_{21}^2 q_k^2 C_{19}(q_k, l)}{(q_k - 1)D_1}\right)^{\frac{s}{s-2}} + 2\left(\max\{4K_S(s), 1\}\max\{1, |\Omega|^{-\frac{1}{2}}\}\right)^{\alpha+1} + |\Omega|.$$

Due to (5.3.5), there is $C_{23} > 0$ s.t. $\|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}}, \|\nabla \cdot \mathbb{D}_l\|_{(L^{\infty}(\Omega))^n} \leq C_{23}$ for all $l \in \mathbb{N}$. We can further estimate using the definition of q_k that

$$C_{19}(q_k, l) \leq \frac{q_k - 1}{D_1} C_{23}^2 (1 + C_{18}^2) + \|\mu\|_{L^{\infty}(0, H)},$$

$$\leq \left(\frac{1 + a}{D_1} C_{23}^2 (1 + C_{18}^2) + \|\mu\|_{L^{\infty}(0, H)}\right) 2^k$$
(5.3.45)

and consequently,

$$\begin{split} C_{22}(q_k, l) \leqslant & 2\left(\frac{2K_{21}^2}{D_1}2^k(1+a)\left(\frac{1+a}{D_1}C_{23}^2(1+C_{18}^2) + \|\mu\|_{L^{\infty}(0,H)}\right)2^k\right)^{\frac{s}{s-2}} \\ & + 2\left(\max\{4K_S(s), 1\}\max\{1, |\Omega|^{-\frac{1}{2}}\}\right)^{\alpha+1} + |\Omega| \end{split}$$

$$\leq 2^{2k\frac{s}{s-2}}C_{24}$$

for

$$\begin{split} C_{24} :=& 2\left(2\frac{K_{21}^2}{D_1}(1+a)\left(\frac{1+a}{D_1}C_{23}^2\left(1+C_{18}^2\right)+\|\mu\|_{L^\infty(0,H)}\right)\right)^{\frac{s}{s-2}} \\ &+ 2\left(\max\{4K_S(s),1\}\max\{1,|\Omega|^{-\frac{1}{2}}\}\right)^{\alpha+1}+|\Omega|. \end{split}$$

For $k \in \mathbb{N}$ and $t \in (0, T_{max})$ we set

$$y_k(t) := \|u_l(\cdot, t)\|_{L^{q_k}(\Omega)}^{q_k}.$$

Inserting this into (5.3.44) we obtain

$$y'_{k}(t) + q_{k}C_{19}(q_{k}, l)y_{k}(t) \leq 2q_{k}C_{19}(q_{k}, l)2^{2k\frac{s}{s-2}}C_{24}\max\left\{1, \left(\int_{\Omega} u_{l}^{q_{k-1}}\right)^{2}\right\}.$$

Moreover, we estimate that

$$\|u_{0l}\|_{L^{q_k}(\Omega)}^{q_k} \leq \|u_{0l}\|_{L^{\infty}(\Omega)}^{q_k}|\Omega| = \|u_{0l}\|_{L^{\infty}(\Omega)}^{2^k}\|u_{0l}\|_{L^{\infty}(\Omega)}^{a}|\Omega|.$$

Hence, from Lemma A.4.3 with $c_k = q_k C_{19}(q_k, l)$, $\bar{a} = 2C_{24}$ and $D = 2\frac{s}{s-2}$ it follows that for $k \ge m \ge 1$ large enough (s.t. $2C_{24}2^{2\frac{s}{s-2}m} \ge 1$) and $t \in (0, T_{max})$ it holds that

$$\begin{split} \int_{\Omega} u_l^{q_k} \, \mathrm{d}x \leqslant & (4C_{24})^{2^{k-m+1}} 2^{\frac{2s}{s-2} \left(2(2^{k-m}-1) + m2^{k-m+1} - k \right)} \\ & \cdot \max \left\{ \sup_{t \geqslant 0} \left(\int_{\Omega} u_l^{q_{m-1}} \right)^{2^{k-m+1}}, \|u_{0l}\|_{L^{\infty}(\Omega)}^{2^k} (\|u_{0l}\|_{L^{\infty}(\Omega)}^a |\Omega|)^{2^{k-m}}, 1 \right\}. \end{split}$$

Consequently, for $t \in (0, T_{max})$ and $m \ge 1$ large enough it holds that

$$\|u_{l}\|_{L^{\infty}(\Omega)} = \lim_{k \to \infty} \|u_{l}\|_{L^{q_{k}}(\Omega)}$$

$$\leq (4C_{24})^{2^{-m+1}} 2^{\frac{2s(1+m)}{(s-2)2^{m-1}}} \max\left\{ \sup_{t \ge 0} \left(\int_{\Omega} u_{l}^{q_{m-1}} \right)^{2^{-m+1}}, \|u_{0l}\|_{L^{\infty}(\Omega)} (\|u_{0l}\|_{L^{\infty}(\Omega)}^{a} |\Omega|)^{2^{-m}}, 1 \right\}$$

$$=: C_{25}(m, l).$$
(5.3.46)

Due to (5.3.41) and (5.3.45) there is $C_{20}(m, u_{0l}) > 0$ s.t. $||u_l||_{L^{\infty}(\Omega \times (0, T_{max}))} \leq C_{20}(m, u_{0l})$ for all $l \in \mathbb{N}$. Together with (5.3.4) this implies the existence of $C_{21}(m)$ s.t. $C_{20}(m, u_{0l}) \rightarrow C_{21}(m)$ for $l \rightarrow \infty$.

Consequently, u_l is bounded on $\overline{\Omega} \times [0, T_{max})$. Combining this with the boundedness of h_l , Lemma 5.3.3 and (5.3.10) in Theorem 5.3.2, $T_{max} = \infty$ follows.

We proceed as in Step 3 of the proof of Theorem 1.1 in [99]. First, we fix some m and choose our parameters sufficiently 'small' s.t.

$$\frac{2K_{21}^2}{D_1} \left(\frac{q_{m-1}}{D_1} \left(\|\nabla \cdot \mathbb{D}_l\|_{(L^{\infty}(\Omega))^n}^2 + \|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}}^2 C_{18}^2 \right) + \|\mu\|_{L^{\infty}(0,H)} \right)$$

$$\leq \frac{2K_{21}^2}{D_1} \left(\frac{q_{m-1}}{D_1} C_{23}^2 \left(1 + C_{18}^2 \right) + \|\mu\|_{L^{\infty}(0,H)} \right) < \frac{1}{q_{m-1}^2}$$
(5.3.47)

and

$$C_{19}(q_{m-1}) \leqslant \frac{q_{m-1}-1}{D_1} C_{23}^2 \left(1+C_{18}^2\right) + \|\mu\|_{L^{\infty}(0,H)} \leqslant 1$$
(5.3.48)

are satisfied. This depends on our choice of \mathbb{D} , μ , g, h_0 , and D_H . Consequently, we conclude again from (5.3.41) that

$$\begin{split} & \int_{\Omega} u_l^{q_{m-1}} \, \mathrm{d}x \\ \leqslant 5 \max \left\{ \left(2 \left(\frac{1}{q_{m-1} - 1} \right)^{\frac{q_{m-1} + \alpha - 1 - \beta}{q_{m-1} - \alpha + \beta + 1 - \frac{2(q_{m-1} + \alpha - 1 + \beta)}{s}} \right. \\ & + K_{24}(q_{m-1})^{\frac{q_{m-1} + \alpha - 1 - \beta}{q_{m-1} - \frac{q_{m-1} + \alpha - 1 + \beta}{s}} \right)^{\frac{q_{m-1} - \alpha + 1 + \beta - \frac{2(q_{m-1} + \alpha - 1 + \beta)}{s}}{\beta + 1 - \alpha - \frac{2\beta}{s}}} \\ & \cdot \left(\frac{2}{\delta \eta} \right)^{\frac{q_{m-1} - \frac{2(q_{m-1} + \alpha - 1)}{s}}{\beta + 1 - \alpha - \frac{2\beta}{s}}}, K_{24}(q_{m-1})^{\frac{q_{m-1} - \alpha - 1 - \beta}{q_{m-1} - \frac{q_{m-1} + \alpha - 1 + \beta}{s}}}, |\Omega| \left(2 + \|u_{0l}\|_{L^{\infty}(\Omega)}^{q_{m-1}} \right) \right\} \end{split}$$

=:H(m).

With
$$(4C_{24})^{2^{-m+1}} 2^{\frac{2s(1+m)}{(s-2)2^{m-1}}} \to 1$$
 and $\left(\|u_{0l}\|_{L^{\infty}(\Omega)}^{a} |\Omega| \right)^{2^{-m}} \to 1$ for $m \to \infty$ and

$$\lim_{m \to \infty} H(m)^{\frac{1}{2^{m-1}}} = \max\left\{ \left(4K_{S}(s)|\Omega|^{-\frac{1}{2}} \right)^{\frac{1-\frac{2}{s}}{(1-\frac{1}{s})(\beta+1-\alpha-\frac{2\beta}{s})}} \left(\frac{2}{\delta\eta} \right)^{\frac{1-\frac{2}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}}, \|u_{0l}\|_{L^{\infty}(\Omega)}, 1 \right\}$$

we conclude from (5.3.46) that

$$\|u_l\|_{L^{\infty}(\Omega\times(0,\infty))} \leqslant \max\left\{1, \|u_{0l}\|_{L^{\infty}(\Omega)}, \left(4K_S(s)|\Omega|^{-\frac{1}{2}}\right)^{\frac{1-\frac{2}{s}}{(1-\frac{1}{s})(\beta+1-\alpha-\frac{2\beta}{s})}} \left(\frac{2}{\delta\eta}\right)^{\frac{1-\frac{2}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}}\right\}.$$

There are obviously no parameters satisfying (5.3.47) and (5.3.48) for all m. But for any K > 1 we find 'small' enough parameters (satisfying (5.3.47) and (5.3.48) for some large enough m) such that (5.3.43) holds.

5.3.3 Global existence and boundedness in the original problem

Theorem 5.3.7. There is a bounded and nonnegative weak solution (u,h) of (5.3.1) s.t. for all T > 0 it holds that $u \in C([0,T]; L^2(\Omega)) \cap L^2(0,T; H^1(\Omega))$ with $\partial_t u \in L^2(0,T; H^{-1}(\Omega))$ and $h \in C([0,T]; H^1(\Omega)) \cap W_2^{2,1}(\Omega \times (0,T)) \cap L^{\infty}(0,T; W_{\infty}^1(\Omega))$ and for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$ the functions u and h satisfy

$$-\int_{0}^{T}\int_{\Omega}u\eta_{t}\,dx\,dt + \int_{0}^{T}\int_{\Omega}\left(\nabla\cdot\mathbb{D}u + \mathbb{D}\nabla u + \mathbb{D}u\nabla h\right)\cdot\nabla\eta\,dx\,dt$$
$$=\int_{0}^{T}\int_{\Omega}\mu(h)u^{\alpha}(1-J*u^{\beta})\eta\,dx\,dt + \int_{\Omega}u_{0}(x)\eta(x,0)\,dx,$$
(5.3.49)

and

$$\partial_t h = D_H \Delta h + g(u, h) \quad a.e. \text{ in } \Omega \times (0, T)$$
(5.3.50)

$$\nabla h \cdot \nu = 0 \qquad a.e. \ in \ \partial\Omega \times (0,T), \qquad (5.3.51)$$

$$h(0) = h_0$$
 in $H^1(\Omega)$. (5.3.52)

Moreover, $u \leq C_{21}$, $h \leq H$ in $\Omega \times (0, \infty)$. For $t_0 \in (0, \infty)$ there are constants $\gamma(t_0) \in (0, 1)$ and $C_{26}(t_0) > 0$ s.t. $u, h \in C^{\gamma, \frac{\gamma}{2}}(\overline{\Omega} \times [t_0, \infty))$ and for all $t \in [t_0, \infty)$ it holds that

$$\|u\|_{C^{\gamma,\frac{\gamma}{2}}(\overline{\Omega}\times[t,t+1])}, \|h\|_{C^{\gamma,\frac{\gamma}{2}}(\overline{\Omega}\times[t,t+1])} \leqslant C_{26}.$$
(5.3.53)

For the parameter choice from Theorem 5.3.6 u satisfies (5.3.43).

The solution (u, h) is unique for $p \ge \frac{2n}{n+2}$ for $n \ge 3$ and $p \in (1, \infty)$ for n = 1, 2.

Proof. Let $\varphi \in H^1(\Omega)$ and T > 0. Obviously, for each $l \in \mathbb{N}$ the function u_l satisfies

$$\int_{\Omega} \partial_t u_l \varphi \, \mathrm{d}x = -\int_{\Omega} (\mathbb{D}_l \nabla u_l + \nabla \cdot \mathbb{D}_l u_l + \mathbb{D}_l u_l \nabla h_l) \cdot \nabla \varphi \, \mathrm{d}x + \int_{\Omega} \mu(h_l) u_l^{\alpha} (1 - J * u_l^{\beta}) \, \mathrm{d}x,$$
(5.3.54)

for $t \in (0, \infty)$ as it is a classical solution. Due to Theorem A.1.8, h_l satisfies

$$\|h_l\|_{W_2^{2,1}(\Omega\times(0,T))} \leqslant C_{27}(T), \tag{5.3.55}$$

where $C_{27}(T) > 0$ is independent from l due to the properties of g and the uniform boundedness of $(h_l)_l$ and $(u_l)_l$.

There is a constant C_{23} independent from l s.t. $\|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}}, \|\nabla \cdot \mathbb{D}_l\|_{(L^{\infty}(\Omega))^n} \leq C_{23}$ for all $l \in \mathbb{N}$ due to (5.3.5). Setting $\varphi = u_l$ in (5.3.54) and using (5.3.6), Hölder's inequality, the continuity of μ and the uniform boundedness of $(h_l)_l, (u_l)_l$ from *Theorem 5.3.6*, we can estimate that

$$\frac{1}{2}\frac{d}{dt}\|u_l\|_{L^2(\Omega)}^2 + D_1\|\nabla u_l\|_{(L^2(\Omega))^n}^2 \leqslant C_{23}\|u_l\|_{L^2(\Omega)}(1+C_{18})\|\nabla u_l\|_{(L^2(\Omega))^n} + C_{28}.$$

Consequently, Young's inequality and integration over (0, T) lead

$$\|\nabla u_l\|_{L^2(0,T;(L^2(\Omega))^n)} \le C_{29}(T)$$

for all $l \in \mathbb{N}$. Similarly (from (5.3.54) for $\varphi \in H_0^1(\Omega)$) it follows that

$$\|\partial_t u_l\|_{L^2(0,T;H^{-1}(\Omega))} \le C_{30}(T)$$

for all $l \in \mathbb{N}$. Putting this together with the uniform boundedness of $(h_l)_l, (u_l)_l, (5.3.55),$ Lemma 5.3.3, the Lions-Aubin lemma (respectively, with $H^1(\Omega) \hookrightarrow L^2(\Omega) \hookrightarrow H^{-1}(\Omega)$ and $H^2(\Omega) \hookrightarrow H^1(\Omega) \hookrightarrow L^2(\Omega))$, the Banach-Alaoglu theorem, Lemma A.3.8 and Lemma A.3.1 with $L^{\infty}(0,T; (L^{\infty}(\Omega))^n) = (L^1(0,T; (L^1(\Omega))^n))^*$ we conclude that there are $u \in C([0,T]; L^2(\Omega)) \cap$ $L^2(0,T; H^1(\Omega))$ with $\partial_t u \in L^2(0,T; H^{-1}(\Omega))$ and $h \in C([0,T]; H^1(\Omega)) \cap W_2^{2,1}(\Omega \times (0,T)) \cap$ $L^{\infty}(0,T; W_{\infty}^1(\Omega))$ s.t. (after switching to a subsequence if necessary)

$$\begin{aligned} u_l &\xrightarrow{\rightarrow} u & \text{in } L^2(\Omega \times (0,T)) \text{ and pointwise a.e.,} \\ u_l &\xrightarrow{\rightarrow} u & \text{in } L^2(0,T;H^1(\Omega)), \\ \partial_t u_l &\xrightarrow{\rightarrow} \partial_t u & \text{in } L^2(0,T;H^{-1}(\Omega)), \\ h_l &\xrightarrow{\rightarrow} h & \text{in } L^2(0,T;H^1(\Omega)) \text{ and pointwise a.e.,} \\ h_l &\xrightarrow{\rightarrow} h & \text{in } L^2(0,T;H^2(\Omega)), \\ \partial_t h_l &\xrightarrow{\rightarrow} \partial_t h & \text{in } L^2(\Omega \times (0,T)), \end{aligned}$$
(5.3.57)

$$\nabla h_l \stackrel{*}{\underset{l \to \infty}{\longrightarrow}} \nabla h \text{ in } L^{\infty}(0, T; (L^{\infty}(\Omega))^n).$$
(5.3.58)

The uniform boundedness of $(u_l)_l$ from (5.3.42) together with (5.3.56) and the dominated convergence theorem imply

$$J\ast u_l^\beta \underset{l \to \infty}{\to} J\ast u^\beta \text{ a.e. in } \Omega \times (0,T).$$

From this using the above convergences, the dominated convergence theorem, the uniform boundedness of $(u_l)_l$, $(h_l)_l$ and $(\nabla h_l)_l$ from Lemma 5.3.3 and Theorem 5.3.6, the Lipschitz-continuity of μ and g, compensated compactness (Lemma A.3.2) and (5.3.5) and the fundamental lemma of variational calculus, it follows as in Theorem 6.4.4 that (u, h) solves (5.3.1) in the required sense. Moreover, $u \in C([0, T]; L^2(\Omega))$ holds due to Theorem A.1.1.

The a.e. boundedness and nonnegativity of u and h follow from the pointwise convergence and the uniform boundedness and nonnegativity of $(u_l)_l$ and $(h_l)_l$. Uniqueness follows similarly to *Theorem 5.3.2* using *Lemma A.1.3*.

Finally, the global boundedness of u, h, (5.3.5), Lemmas 2.3.2 and 5.3.3 and the Lipschitz continuity of μ and g, Theorem A.1.12 and Remark A.1.2 with a and b, respectively, chosen similar to Lemma 5.3.1 imply (5.3.53).

5.4 Long time behavior

We consider the long time behavior of our solution under the additional assumptions that we make from now on:

- the domain Ω is convex,
- there are $h^* \in [0, H]$ and constants $C_H > 0$ and $C_U \ge 0$ s.t.

$$g(u,h)(h-h^*) \leqslant -C_H(h-h^*)^2 + C_U u^{\alpha-1} (u^\beta - U)^2$$
(5.4.1)

for $0 \leq h \leq H$ and $0 \leq u \leq U^{\frac{1}{\beta}}$, where $U^{\frac{1}{\beta}}$ is some upper bound on u_l for all $l \in \mathbb{N}$ (that exists and is independent from l, due to *Theorem 5.3.6*),

- · we extend J by 0 to $\mathbb{R}^n \setminus B$ and assume $J = \frac{1}{U}\tilde{J}$ for a kernel $\tilde{J} \in L^1(\mathbb{R}^n)$ with norm $\|\tilde{J}\|_{L^1(\mathbb{R}^n)} = 1$,
- the parameters

$$C_B := \frac{1}{4D_1} \|\mu\|_{L^{\infty}(0,H)} \, (\operatorname{diam}(\Omega)\beta)^2 \, U^{\frac{\alpha-1}{\beta}}, \tag{5.4.2}$$

$$C_A := \frac{C_U C_{23}^2 \beta^2 U}{4\delta \eta |\Omega| D_H D_1} - 1 - C_B, \qquad (5.4.3)$$

where $\|\mathbb{D}_l\|_{(L^{\infty}(\Omega))^{n \times n}} \leq C_{23}$ for all $l \in \mathbb{N}$ (due to (5.3.5)), satisfy $C_A^2 > 4C_B$, $C_A < 0$, $C_B \in (0,1)$ and

$$C_B < -\frac{C_A}{2} + \sqrt{\frac{C_A^2}{4} - C_B},$$
 (5.4.4)

 $u_0 \neq 0, u_0 \leq U^{\frac{1}{\beta}} \text{ and } \int_{\Omega} \ln(u_0) \, \mathrm{d}x < \infty.$

Moreover, let

$$M := \left\{ \mathbb{D} \in \left(C^{2+\vartheta}(\bar{\Omega}) \right)^{n \times n} : \left(\nabla \cdot \mathbb{D} \right) \cdot \nu = 0 \text{ on } \partial\Omega, \, \nabla \cdot \left(\nabla \cdot \mathbb{D} \right) = 0 \text{ on } \Omega \right\}.$$

We assume that \mathbb{D} is in the closure of M in the $C^1(\overline{\Omega}, \mathbb{R}^{n \times n})$ -norm and that the sequence $(\mathbb{D}_l)_{l \in \mathbb{N}}$ from Section 5.3 is the sequence in M that approaches \mathbb{D} .

Remark 5.4.1. The inequality (5.4.1) implies that such h^* is unique and $g(U^{\frac{1}{\beta}}, h^*) = 0$ holds.

We proceed by combining the methods from [91, 99].

Lemma 5.4.2. It holds that

$$\int_{0}^{\infty} \int_{\Omega} u^{\alpha - 1} (u^{\beta} - U)^{2} \, dx \, dt, \\ \int_{0}^{\infty} \int_{\Omega} |h - h^{*}|^{2} \, dx \, dt < \infty.$$
 (5.4.5)

Proof. Let $l \in \mathbb{N}$ and consider the global classical solution to (5.3.7) from *Theorem 5.3.6*. We conclude from *Proposition A.1.10* and the assumption $u_0 \neq 0$ that $u_l > 0$ holds in $\Omega \times (0, \infty)$.

As in [99] we define $a(s) := \frac{s}{\beta} - \frac{U}{\beta} \ln(s) + \frac{U}{\beta} (\ln(U) - 1)$ with $a(s) \ge 0$ for $s \in (0, \infty)$. By multiplying the equation for u_l in (5.3.7) by $u_l^{\beta-1} - Uu_l^{-1}$, integrating over Ω and using partial integration, we obtain

$$\begin{split} \frac{d}{dt} \int_{\Omega} a(u_l^{\beta}) \, \mathrm{d}x &= \int_{\Omega} \partial_t u_l (u_l^{\beta-1} - U u_l^{-1}) \, \mathrm{d}x \\ &= \int_{\Omega} \nabla \cdot \left(\mathbb{D}_l \nabla u_l + \nabla \cdot \mathbb{D}_l u_l + \mathbb{D}_l u_l \nabla h_l \right) \left(u_l^{\beta-1} - U u_l^{-1} \right) \, \mathrm{d}x \\ &+ \int_{\Omega} \mu(h_l) u_l^{\alpha} (1 - J * u_l^{\beta}) (u_l^{\beta-1} - U u_l^{-1}) \, \mathrm{d}x \\ &= - \int_{\Omega} \left(\mathbb{D}_l \nabla u_l + \nabla \cdot \mathbb{D}_l u_l + \mathbb{D}_l u_l \nabla h_l \right) \left((\beta - 1) u_l^{\beta} + U \right) \cdot \frac{\nabla u_l}{u_l^2} \, \mathrm{d}x \\ &+ \frac{1}{U} \int_{\Omega} \mu(h_l) u_l^{\alpha-1} \left(U - \tilde{J} * u_l^{\beta} \right) \left(u_l^{\beta} - U \right) \, \mathrm{d}x. \end{split}$$

Due to $\mathbb{D}_l \in M$ using partial integration again leads to

$$\begin{split} &\int_{\Omega} \nabla \cdot \mathbb{D}_{l} u_{l} ((\beta - 1) u_{l}^{\beta} + U) \cdot \frac{\nabla u_{l}}{u_{l}^{2}} \, \mathrm{d}x \\ &= \int_{\Omega} \nabla \cdot \mathbb{D}_{l} ((\beta - 1) u_{l}^{\beta - 1} + U u_{l}^{-1}) \cdot \nabla u_{l} \, \mathrm{d}x \\ &= \int_{\Omega} (\nabla \cdot \mathbb{D}_{l}) \cdot \nabla \left(\frac{\beta - 1}{\beta} u_{l}^{\beta} + U \ln(u_{l}) \right) \, \mathrm{d}x \\ &= -\int_{\Omega} \nabla \cdot (\nabla \cdot \mathbb{D}_{l}) \left(\frac{\beta - 1}{\beta} u_{l}^{\beta} + U \ln(u_{l}) \right) \, \mathrm{d}x + \int_{\partial \Omega} \left(\frac{\beta - 1}{\beta} u_{l}^{\beta} + U \ln(u_{l}) \right) (\nabla \cdot \mathbb{D}_{l}) \cdot \nu \, \mathrm{d}\sigma(x) = 0. \end{split}$$

Hence, we can estimate using the positivity of u_l and (5.3.6) that

$$\frac{d}{dt} \int_{\Omega} a(u_l^{\beta}) \, \mathrm{d}x + U \int_{\Omega} \left(\frac{\nabla u_l}{u_l} \right)^T \mathbb{D}_l \frac{\nabla u_l}{u_l} \, \mathrm{d}x$$

$$\leq \int_{\Omega} \left((\beta - 1) u_l^{\beta} + U \right) \left| (\nabla h_l)^T \mathbb{D}_l \frac{\nabla u_l}{u_l} \right| \, \mathrm{d}x + \frac{1}{U} \int_{\Omega} \mu(h_l) u_l^{\alpha - 1} \left(U - \tilde{J} * u_l^{\beta} \right) \left(u_l^{\beta} - U \right) \, \mathrm{d}x.$$
(5.4.6)

Then, we proceed similarly to the proof of Proposition 3.1 in [99] and obtain using $u_l \leq U^{\frac{1}{\beta}}$ and Young's inequality that

$$\begin{split} &\int_{\Omega} \mu(h_{l}(x)) u_{l}^{\alpha-1}(x) (U - \tilde{J} * u_{l}^{\beta}(x)) (u_{l}^{\beta}(x) - U) \, \mathrm{d}x \\ &= \int_{\Omega} \mu(h_{l}(x)) u_{l}^{\alpha-1}(x) \left(U \int_{\mathbb{R}^{n}} \tilde{J}(y) \, \mathrm{d}y - \int_{\Omega} \tilde{J}(x - y) u_{l}^{\beta}(y) \, \mathrm{d}y \right) \left(u_{l}^{\beta}(x) - U \right) \, \mathrm{d}x \\ &\leq \int_{\Omega} \mu(h_{l}(x)) u_{l}^{\alpha-1}(x) \left(\int_{\Omega} \tilde{J}(x - y) (U - u_{l}^{\beta}(y)) \, \mathrm{d}y \right) \left(u_{l}^{\beta}(x) - U \right) \, \mathrm{d}x \\ &\leq - \int_{\Omega} \mu(h_{l}(x)) u_{l}^{\alpha-1}(x) \int_{\Omega} \tilde{J}(x - y) (u_{l}^{\beta}(x) - U)^{2} \, \mathrm{d}y \, \mathrm{d}x \\ &+ \int_{\Omega} \mu(h_{l}(x)) u_{l}^{\alpha-1}(x) \int_{\Omega} \tilde{J}(x - y) (u_{l}^{\beta}(x) - u_{l}^{\beta}(y)) (u_{l}^{\beta}(x) - U) \, \mathrm{d}y \, \mathrm{d}x \\ &\leq - (1 - \varepsilon) \int_{\Omega} \int_{\Omega} \mu(h_{l}(x)) u_{l}^{\alpha-1}(x) \tilde{J}(x - y) (u_{l}^{\beta}(x) - U)^{2} \, \mathrm{d}y \, \mathrm{d}x \\ &+ \frac{1}{4\varepsilon} \|\mu\|_{L^{\infty}(0,H)} U^{\frac{\alpha-1}{\beta}} \int_{\Omega} \int_{\Omega} \tilde{J}(x - y) (u_{l}^{\beta}(x) - u_{l}^{\beta}(y))^{2} \, \mathrm{d}y \, \mathrm{d}x \end{split}$$
(5.4.7)

for

$$\varepsilon \in \left(\max\left\{ -\frac{C_A}{2} - \sqrt{\frac{C_A^2}{4} - C_B}; C_B \right\}, \min\left\{ -\frac{C_A}{2} + \sqrt{\frac{C_A^2}{4} - C_B}; 1 \right\} \right) \subset (C_B, 1),$$

where the interval on the right-hand side is nonempty due to our assumptions on C_A and C_B . Moreover, due to the convexity of Ω and using the uniform boundedness of $(u_l)_l$ by $U^{\frac{1}{\beta}}$ we conclude from *Lemma A.4.5* the estimate

$$\begin{split} \int_{\Omega} \int_{\Omega} \tilde{J}(x-y) (u_l^{\beta}(x) - u_l^{\beta}(y))^2 \, \mathrm{d}y \, \mathrm{d}x &\leq \operatorname{diam}(\Omega)^2 \int_{\Omega} |\nabla(u_l^{\beta})|^2 \, \mathrm{d}x \\ &\leq \left(\operatorname{diam}(\Omega)\beta U\right)^2 \int_{\Omega} \frac{|\nabla u_l|^2}{u_l^2} \, \mathrm{d}x. \end{split}$$

Now, inserting this in (5.4.7), using our assumptions on J and μ , and the uniform boundedness of $(u_l)_l$, we conclude

$$\frac{1}{U} \int_{\Omega} \mu(h_l(x)) u_l^{\alpha-1}(x) (U - \tilde{J} * u_l^{\beta}(x)) (u_l^{\beta}(x) - U) \, \mathrm{d}x$$

$$\leq \frac{1}{\varepsilon} D_1 U C_B \int_{\Omega} \frac{|\nabla u_l|^2}{u_l^2} \, \mathrm{d}x - (1 - \varepsilon) \delta\eta |\Omega| \int_{\Omega} u_l^{\alpha-1} (u_l^{\beta} - U)^2 \, \mathrm{d}x$$

Inserting this into (5.4.6) and using (5.3.6) and Young's inequality, it follows that

$$\frac{d}{dt} \int_{\Omega} a(u_l^{\beta}) \,\mathrm{d}x + D_1 U \frac{\varepsilon - C_B}{\varepsilon} \int_{\Omega} \frac{|\nabla u_l|^2}{u_l^2} \,\mathrm{d}x + (1 - \varepsilon) \delta\eta |\Omega| \int_{\Omega} u_l^{\alpha - 1} (u_l^{\beta} - U)^2 \,\mathrm{d}x$$

$$\leq \int_{\Omega} \left((\beta - 1) u_l^{\beta} + U \right) \left| (\nabla h_l)^T \mathbb{D}_l \frac{\nabla u_l}{u_l} \right| \,\mathrm{d}x$$

$$\leq D_1 U \frac{\varepsilon - C_B}{\varepsilon} \int_{\Omega} \frac{|\nabla u_l|^2}{u_l^2} \,\mathrm{d}x + \frac{\varepsilon C_{23}^2 \beta^2 U}{4D_1 (\varepsilon - C_B)} \int_{\Omega} |\nabla h_l|^2 \,\mathrm{d}x.$$
(5.4.8)

Multiplying the equation for h_l by $h_l - h^*$, integrating over Ω and using (5.4.1), we obtain

$$\frac{1}{2}\frac{d}{dt}\int_{\Omega}(h_l - h^*)^2 \,\mathrm{d}x + D_H \int_{\Omega} |\nabla h_l|^2 \,\mathrm{d}x \leq -C_H \int_{\Omega}(h - h^*)^2 \,\mathrm{d}x + C_U \int_{\Omega} u_l^{\alpha - 1}(u_l^{\beta} - U)^2 \,\mathrm{d}x.$$

Further, we multiply this by $C_{31} := \frac{\varepsilon C_{23}^2 \beta^2 U}{4D_1 D_H (\varepsilon - C_B)}$ and add it to (5.4.8) to obtain

$$\frac{d}{dt} \left(\int_{\Omega} a(u_l^{\beta}) \, \mathrm{d}x + \frac{1}{2} C_{31} \int_{\Omega} (h_l - h^*)^2 \, \mathrm{d}x \right)$$
(5.4.9)

$$+C_H C_{31} \int_{\Omega} (h_l - h^*)^2 \,\mathrm{d}x + ((1 - \varepsilon)\delta\eta |\Omega| - C_U C_{31}) \int_{\Omega} u_l^{\alpha - 1} (u_l^{\beta} - U)^2 \,\mathrm{d}x \le 0.$$
(5.4.10)

Due to our assumptions on C_A, C_B and our choice of ε it holds that $(1 - \varepsilon)\delta\eta |\Omega| - C_U C_{31} > 0$. Consequently, for $T \in (0, \infty)$ it holds that

$$\begin{split} &\int_{\Omega} a(u_l^{\beta}(T)) \,\mathrm{d}x + \frac{1}{2} C_{31} \int_{\Omega} (h_l(T) - h^*)^2 \,\mathrm{d}x + C_H C_{31} \int_0^T \int_{\Omega} (h_l - h^*)^2 \,\mathrm{d}x \,\mathrm{d}t \\ &+ ((1 - \varepsilon) \delta \eta |\Omega| - C_U C_{31}) \int_0^T \int_{\Omega} u_l^{\alpha - 1} (u_l^{\beta} - U)^2 \,\mathrm{d}x \,\mathrm{d}t \\ &\leqslant \int_{\Omega} a(u_0^{\beta}) + \frac{1}{2} C_{31} \int_{\Omega} (h_0 - h^*)^2 \,\mathrm{d}x, \end{split}$$

where the right-hand side is finite due to our additional assumption on u_0 . Hence, using the uniform boundedness of $(u_l)_l$ and $(h_l)_l$, their pointwise a.e. convergences from (5.3.56) and (5.3.57) and the dominated convergence theorem we conclude

$$C_{H}C_{31} \int_{0}^{T} \int_{\Omega} (h - h^{*})^{2} dx dt + ((1 - \varepsilon)\delta\eta |\Omega| - C_{U}C_{31}) \int_{0}^{T} \int_{\Omega} u^{\alpha - 1}(u^{\beta} - U)^{2} dx dt$$

$$\leq \int_{\Omega} a(u_{0}^{\beta}) + \frac{1}{2}C_{31} \int_{\Omega} (h_{0} - h^{*})^{2} dx$$

.4.5) follows. \Box

and (5.4.5) follows.

Now we can conclude uniform convergence as in Lemma 3.10 in [139]:

Theorem 5.4.3. It holds that

$$\lim_{t \to \infty} \|u(\cdot, t) - c\|_{L^{\infty}(\Omega)} = \lim_{t \to \infty} \|h(\cdot, t) - h^*\|_{L^{\infty}(\Omega)} = 0,$$
(5.4.11)

where $c \in \{0, U^{\frac{1}{\beta}}\}$ if $\alpha > 1$ and $c = U^{\frac{1}{\beta}}$ if $\alpha = 1$.

Proof. Due to (5.3.53) in *Theorem 5.3.7*, u and h are uniformly continuous in $\overline{\Omega} \times (1, \infty)$ and we can conclude from Lemmas 2.2.3, 5.4.2, and A.4.6 that

$$\lim_{t \to \infty} \|u(\cdot, t)^{\frac{\alpha - 1}{2}} (u^{\beta}(\cdot, t) - U)\|_{L^{\infty}(\Omega)} = \lim_{t \to \infty} \|h(\cdot, t) - h^{*}\|_{L^{\infty}(\Omega)} = 0.$$
(5.4.12)

Consider $\alpha > 1$ and set

$$\varepsilon := \min\left\{\frac{U^{\frac{1}{\beta}}}{8}, \left(1 - \left(\frac{7}{8}\right)^{\beta}\right) \left(\frac{U^{\frac{1}{\beta}}}{8}\right)^{\beta + \frac{\alpha - 1}{2}}\right\}.$$

Due to the uniform continuity of u we can consider w.l.o.g. a sequence $(t_k)_{k\in\mathbb{N}} \subset (1,\infty)$ with $\lim_{k\to\infty} t_k = \infty$ satisfying for all $x \in \overline{\Omega}$ and $k \in \mathbb{N}$ the estimate

$$|u(x, t_{k+1}) - u(x, t_k)| < \varepsilon.$$
(5.4.13)

Moreover, (5.4.12) implies that there is $K(\varepsilon) \in \mathbb{N}$ s.t.

$$\|u(\cdot,t_k)^{\frac{\alpha-1}{2}}(u^{\beta}(\cdot,t_k)-U)\|_{L^{\infty}(\Omega)}<\varepsilon$$

and consequently, due to our choice of ε above $u(x,t_k) \in [0, \frac{1}{8}U^{\frac{1}{\beta}}) \cup (\frac{7}{8}U^{\frac{1}{\beta}}, U^{\frac{1}{\beta}}]$ for $k \ge K(\varepsilon)$. Hence, either $u(x,t_k) \in [0, \frac{1}{8}U^{\frac{1}{\beta}})$ for all $x \in \Omega$ and $k \ge K(\varepsilon)$ or $u(x,t_k) \in (\frac{7}{8}U^{\frac{1}{\beta}}, U^{\frac{1}{\beta}}]$ for all $x \in \Omega$ and $k \ge K(\varepsilon)$ due to (5.4.13). If $u(x,t_k) \in [0, \frac{1}{8}U^{\frac{1}{\beta}})$ for all $x \in \Omega$ and $k \ge K(\varepsilon)$ then (5.4.12) implies that

$$\|u(\cdot,t_k)\|_{L^{\infty}(\Omega)} \leqslant \left(\frac{\|u(\cdot,t_k)^{\frac{\alpha-1}{2}}(u^{\beta}(\cdot,t_k)-U)\|_{L^{\infty}(\Omega)}}{U\left(1-\left(\frac{1}{8}\right)^{\beta}\right)}\right)^{\frac{2}{\alpha-1}} \underset{k \to \infty}{\longrightarrow} 0.$$

Analogously, the other convergence in the case $u(x, t_k) \in \left(\frac{7}{8}U^{\frac{1}{\beta}}, U^{\frac{1}{\beta}}\right]$ follows.

5.5 Pattern formation: a 1D study

We want to investigate pattern formation in our model in 1D (see [110]). For this aim we adapt some of the assumptions on our functions and parameters:

- · $J \in L^1(\mathbb{R}), \ J(x) = J(-x)$ for $x \in \mathbb{R}$ and $\int_{\mathbb{R}} J(x) dx = 1$, whereas we drop the condition that $0 < \eta \leq J$;
- $\cdot d \in \mathbb{R}$ constant;
- there is exactly one $h^* > 0$ with $g(1, h^*) = 0$, moreover, $\mu(h^*) > 0$, $\partial_u g(1, h^*) \ge 0$ and $\partial_h g(1, h^*) < 0$ for this h^* . This means that when the cancer cells are at their carrying capacity (corresponding to an acidity level h^*), the production of protons is increasing with the cell mass and decreasing with enhancing proton concentration. Indeed, crowded tumor cells are highly hypoxic, and a too acidic environment leads to quiescence or necrosis, thus reducing proton expression. Moreover, we assume that $\mu'(h^*) < 0$, thus the growth rate is decreasing with the proton concentration in the neighborhood of the critical value h^* .
- w.l.o.g. we consider $\Omega = (-a, a)$ for $a \in \mathbb{R}$.

We define the convolution over \mathbb{R} as $J \circledast u(x) := \int_{\mathbb{R}} J(x-y)u(y) \, dy$. Hence, we consider the model

$$u_t = du_{xx} + d(uh_x)_x + \mu(h)u^{\alpha}(1 - J \circledast u^{\beta}) \quad \text{in } \Omega \times (0, \infty),$$

$$h_t = D_H h_{xx} + g(u, h) \quad \text{in } \Omega \times (0, \infty),$$

$$u_x = h_x = 0 \quad \text{on } \partial\Omega \times (0, \infty),$$

$$u(\cdot, 0) = u_0, h(\cdot, 0) = h_0 \quad \text{in } \Omega.$$
(5.5.1)

5.5.1 Stability in the local model without diffusion and taxis

We start by establishing the equilibria of the non-spatial local model that corresponds to (5.5.1), i.e.,

$$\begin{cases} \partial_t u = \mu(h)u^{\alpha}(1-u^{\beta}), \\ \partial_t h = g(u,h). \end{cases}$$
(5.5.2)

The biologically more interesting one is given by $(u^*, h^*) = (1, h^*)$, where h^* is the unique solution of g(1, h) = 0. The corresponding characteristic equation of the Jacobian in $(1, h^*)$ is given by

$$\lambda^2 + (\beta\mu(h^*) - \partial_h g(1, h^*))\lambda - \beta\mu(h^*)\partial_h g(1, h^*) = 0.$$

The corresponding eigenvalues are

$$\lambda_1 = -\beta \mu(h^*)$$
 and $\lambda_2 = \partial_h g(1, h^*)$

and both have negative real parts due to the assumption $\partial_h g(1, h^*) < 0$. Hence, the steady state $(1, h^*)$ is stable in this case.

5.5.2 Stability in the local model with diffusion and taxis

We continue by adding again the diffusion and taxis terms to the local model (5.5.2), i.e.,

$$\begin{cases} \partial_t u = du_{xx} + d(uh_x)_x + \mu(h)u^{\alpha}(1-u^{\beta}), \\ \partial_t h = D_H h_{xx} + g(u,h). \end{cases}$$

Adapting the ansatz from [119] we consider perturbations of $(1, h^*)$ of the form $u = 1 + \varepsilon \bar{u}(k)$ and $h = h^* + \varepsilon \bar{h}(k)$, where $\bar{u}(k) = \tilde{u}e^{\lambda(k)t}\cos(kx)$ and $\bar{h}(k) = \tilde{h}e^{\lambda(k)t}\cos(kx)$ for $\tilde{u}, \tilde{h} \in \mathbb{R}$, wavenumber $k \in \mathbb{N}$ and $|\varepsilon| << 1$. Here, $\lambda(k)$ denotes some eigenvalue of the corresponding characteristic equation. As in [115] we use the fact that $\frac{e^{ikx} + e^{-ikx}}{2} = \cos(kx)$ to ensure that our perturbations are real for real λ .

Inserting these u and h into our model and linearizing about the steady state $(1, h^*)$, we obtain

$$\begin{cases} \lambda(k)\bar{u} = -dk^{2}\bar{u} - dk^{2}\bar{h} - \beta\mu(h^{*})\bar{u}, \\ \lambda(k)\bar{h} = -D_{H}k^{2}\bar{h} + \partial_{u}g(1,h^{*})\bar{u} + \partial_{h}g(1,h^{*})\bar{h}. \end{cases}$$
(5.5.3)

The corresponding eigenvalues are given by

$$\lambda_{1,2}(k) = \frac{\operatorname{tr}(J_{u,h}(k)) \pm \sqrt{\operatorname{tr}(J_{u,h}(k))^2 - 4\operatorname{det}(J_{u,h}(k))}}{2},$$

where we denote by $J_{u,h}$ the Jacobian of the right-hand side of system (5.5.3) at $(1, h^*)$ and its determinant and trace are, respectively, given by

$$\operatorname{tr}(J_{u,h}(k)) = -(d+D_H)k^2 - \beta\mu(h^*) + \partial_h g(1,h^*) < 0,$$

$$\operatorname{det}(J_{u,h}(k)) = dD_H k^4 + (d(\partial_u g(1,h^*) - \partial_h g(1,h^*)) + \beta\mu(h^*)D_H)k^2 - \beta\mu(h^*)\partial_h g(1,h^*) > 0.$$

Hence, the equilibrium $(1, h^*)$ is stable. The local model does not lead to any Turing type patterns.

5.5.3 Stability in the nonlocal model

We consider u and h as in the previous section and linearize the convolution term about $(1, h^*)$ similarly to [119]. Hence, inserting u in the convolution term and using the symmetry of J, we compute that

$$J \circledast u^{\beta}(x) \approx \int_{\mathbb{R}} J(x-y)(1+\beta \varepsilon \tilde{u} e^{\lambda(k)t} \cos(ky)) \,\mathrm{d}y$$

$$= 1 + \beta \varepsilon \tilde{u} e^{\lambda(k)t} \frac{1}{2} \int_{\mathbb{R}} J(z) e^{ik(x-z)} + J(-z) e^{-ik(x-z)} dz$$
$$= 1 + \beta \varepsilon \tilde{u} e^{\lambda(k)t} \cos(kx) \int_{\mathbb{R}} J(z) e^{-ikz} dz$$
$$= 1 + \varepsilon \beta \bar{u} (2\pi)^{\frac{1}{2}} \hat{J}(k).$$

Here, \hat{J} denotes the Fourier transform of J. Hence, linearizing system (5.5.1), we obtain

$$\begin{cases} \lambda(k)\bar{u} = -dk^2\bar{u} - dk^2\bar{h} - \beta\mu(h^*)(2\pi)^{\frac{1}{2}}\hat{J}(k)\bar{u}, \\ \lambda(k)\bar{h} = -D_Hk^2\bar{h} + \partial_u g(1,h^*)\bar{u} + \partial_h g(1,h^*)\bar{h}. \end{cases}$$
(5.5.4)

The corresponding eigenvalues are as above given by

$$\lambda_{1,2}(k) = \frac{\operatorname{tr}(J_{u,h}(k)) \pm \sqrt{\operatorname{tr}(J_{u,h}(k))^2 - 4 \operatorname{det}(J_{u,h}(k))}}{2},$$

where we denote by $J_{u,h}$ the Jacobian of the right-hand side in (5.5.4) at (1, h^*) and its trace and determinant are given by

$$\operatorname{tr}(J_{u,h})(k) = -(d+D_H)k^2 - \beta\mu(h^*)(2\pi)^{\frac{1}{2}}\hat{J}(k) + \partial_h g(1,h^*), \\ \det(J_{u,h})(k) = dD_H k^4 + (d(\partial_u g(1,h^*) - \partial_h g(1,h^*)) + \beta\mu(h^*)(2\pi)^{\frac{1}{2}}\hat{J}(k)D_H)k^2 \\ - \beta\mu(h^*)(2\pi)^{\frac{1}{2}}\hat{J}(k)\partial_h g(1,h^*).$$

The sign of the real part of the eigenvalues is ambiguous here and depends especially on the sign of $\hat{J}(k)$, which depends on k. As above, we have stability here if

$$\operatorname{tr} J_{u,h}(k) < 0 \text{ and } \det J_{u,h}(k) > 0$$
 (5.5.5)

for all $k = \frac{\pi}{a} z$, where $z \in \mathbb{Z}$. We make this restriction due to our boundary condition $u_x = h_x = 0$.

Now, we are looking for a critical k_c (that is not necessarily of the form $\frac{\pi}{a}z$) depending on our choice of parameters, where we distinguish as in [115] the occurrence of Turing instabilities in the case $\text{Im}(\lambda(k_c)) = 0$ for some arbitrary critical k_c , Hopf instabilities in the case $\text{Im}(\lambda(0)) \neq 0$, and wave instabilities in the case $\text{Im}(\lambda(k_c)) \neq 0$ for some critical $k_c \neq 0$. If \hat{J} is symmetric it suffices to consider only positive k_c .

A Turing bifurcation can occur if we find k_c such that

$$\det(J_{u,h})(k_c) = 0 \text{ and } \operatorname{tr}(J_{u,h})(k_c) < 0.$$

Now, rewriting these conditions we conclude that the equality

$$\hat{J}(k_c) = -d \frac{k_c^2}{\beta(2\pi)^{\frac{1}{2}} \mu(h^*)} \left(1 + \frac{\partial_u g(1, h^*)}{D_H k_c^2 - \partial_h g(1, h^*)} \right)$$
(5.5.6)

and the inequality

$$\frac{\partial_h g(1,h^*) - (d+D_H)k_c^2}{\beta \mu(h^*)(2\pi)^{\frac{1}{2}}} < \hat{J}(k_c)$$
(5.5.7)

have to hold for one or several critical k_c in a set K_c , whereas (5.5.5) holds for all $k \notin K_c$ that are of the form $\frac{\pi}{a}z$. Such k_c exist depending on the choice of parameters, on the functions μ and g, and especially on the sign of the Fourier transform of J. Moreover, due to our assumptions the terms on the right-hand side of (5.5.6) and on the left-hand side of (5.5.7) are negative and tend to $-\infty$ for $k \to \pm \infty$. On the other hand, a Hopf or a wave instability can occur if we find k_c such that

$$tr(J_{u,h})(k_c) = 0$$
 and $det J_{u,h}(k_c) > 0$,

whereas (5.5.5) holds for all k that do not satisfy this and are of the form $\frac{\pi}{a}z$. Hence, a Hopf instability occurs if

$$\frac{\partial_h g(1,h^*)}{\mu(h^*)\beta(2\pi)^{\frac{1}{2}}} = \hat{J}(0) \text{ and } \hat{J}(0) > 0,$$
(5.5.8)

whereas (5.5.5) holds for all $k \neq 0$. On the other hand, a wave instability occurs if

$$\frac{\partial_h g(1,h^*) - (d+D_H)k_c^2}{\mu(h^*)\beta(2\pi)^{\frac{1}{2}}} = \hat{J}(k_c)$$
(5.5.9)

and

$$-d\frac{k_c^2}{\beta(2\pi)^{\frac{1}{2}}\mu(h^*)}\left(1+\frac{\partial_u g(1,h^*)}{D_H k_c^2 - \partial_h g(1,h^*)}\right) < \hat{J}(k_c)$$
(5.5.10)

holds for one or several $k_c \neq 0$, whereas (5.5.5) holds for all other k that are of the form $\frac{\pi}{a}z$ and do not satisfy the above equality and inequality.

From the above considerations we conclude that the occurrence of a Turing, Hopf or wave instability depends on the concrete choice of J, as we need to find suitable k of the form $\frac{\pi}{a}z$. If the Fourier transform \hat{J} is nonnegative, no Turing patterns occur due to (5.5.6). More precisely, the determined patterns are only of Turing-like type as they are induced by the nonlocality and not the diffusion.

Example 5.5.1. We explore the occurrence of Turing-like patterns in the nonlocal model (5.5.1) for the uniform kernel $J_U(x) = \frac{1}{2}\chi_{[-1,1]}(x)$ and the logistic kernel $J_L(x) = \frac{1}{2+e^x+e^{-x}}$ in the domain $\Omega = (-5,5)$ for the parameters and functions g(u,h) = u(1-h), $\mu(h) = \frac{\mu}{1+h}$, with a constant $\mu > 0$ and $d = D_H = 1$. The steady state is given by $(u^*, h^*) = (1,1)$ with $\partial_u g(1,1) = 0$, $\partial_h g(1,1) = -1 < 0$ and $\mu(1) = \frac{\mu}{2}$.

The Fourier transform of the uniform kernel J_U is given by

$$\hat{J}_U(k) = \frac{\sin(k)}{\sqrt{2\pi}k}$$
 (5.5.11)

for $k \in \mathbb{R}$. Its sign is ambiguous. Inserting this into (5.5.6) and (5.5.7) we conclude that

$$-\frac{2k_c^2}{\beta\mu} = \frac{\sin(k_c)}{k_c}$$
(5.5.12)

and

$$-\frac{2+4k_c^2}{\beta\mu} < \frac{\sin(k_c)}{k_c}$$

have to be satisfied for some k_c for the occurrence of Turing-like patterns, where the second condition is a direct consequence of the first. Equation (5.5.12) has a solution if $\beta\mu$ are larger than approximately 168,4. Hence, for such $\beta\mu$ Turing-like patterns occur.

Moreover, the Fourier transform of the logistic kernel J_L is given by

$$\hat{J}_L(k) = \frac{\sqrt{\pi k}}{\sqrt{2}\sinh(k\pi)} > 0$$
 (5.5.13)

for $k \in \mathbb{R}$. Consequently, no Turing-like patterns occur for this kernel

In Section 5.6 we will perform numerical simulations for this choice of functions and kernels.

Remark 5.5.2. If there is a steady state of the form $(0, h^{**})$ for some $h^{**} > 0$ and $\partial_h g(0, h^{**}) \leq 0$, then this equilibrium is stable in the case with diffusion, taxis and nonlocal term. If, on the other hand, $\partial_h g(0, h^{**}) > 0$, this steady state is unstable already in the case without diffusion and taxis. This case is, however, unrealistic for the biological problem investigated here. Indeed, the proton expression by hypoxic cells is much reduced and there must be at least some very weak acid buffering, lest all cells (and surrounding tissue) become apoptotic.

Likewise, the steady state $(1, h^*)$ is unstable already in the case without diffusion and taxis if $\partial_h g(1, h^*) > 0$. This situation may occur at least in a transient manner, e.g., when the cells can still extrude protons while their environment is quite acidic and if the cells are at their carrying capacity and the proton buffering is relatively low. That can lead, e.g., to a choice of the form $g(u, h) = u + uh - \gamma h^2$ with $\gamma \leq 4/5$.

5.6 Numerical simulations

In this section we perform numerical simulations of system (5.5.1), in order to illustrate the solution behavior. The equations are discretized by using the algorithm in [111] similarly to *Section 6.5*; the motility terms were discretized with finite differences (centered for the diffusion, upwind for the drift). The initial conditions are as in [99]:

$$u_0(x) = \begin{cases} e^{-(x-x_l)^2}, & \text{for } x_l < x \le 0\\ e^{-x_l^2}(1-\frac{x}{x_r}), & \text{for } 0 < x \le x_r \end{cases}, \quad \text{with } x_l = -5, \ x_r = 5.$$

Unless otherwise stated we take g(u,h) = u(1-h), $\mu(h) = \frac{\mu}{1+h}$, with $\mu > 0$ a constant and $d = D_H = 1$.

In a first test we took $\beta = \mu = 1$, along with the logistic kernel $J_L(x) = \frac{1}{2+e^x+e^{-x}}$ (see, e.g., [98]) and the uniform kernel $J_U(x;\rho) = \frac{1}{2\rho}\chi_{[-\rho,\rho]}$. The first two columns of Figure 5.1 show simulation results for $\alpha = 2$, which is the 'limit value' in (5.3.3). The solution ceased (in finite time) to exist for sufficiently large α in each of these situations ($\alpha \sim 6.25$ and $\alpha \sim 8.2$, respectively), uexhibiting strong aggregation near the initial bulk of cells, cf. last two columns in Figure 5.1. This behavior was also observed for increasing values of μ , with the difference of singularities already occuring for smaller α values.

Increasing the values of μ and β leads to patterns, the shape of which depends decisively on the interaction kernel J and also on the values of α and d. Figure 5.2 shows 1D space-time patterns of the cell density u for $\beta = 20$, $\mu = 100$, and several combinations of α and J. The results for the proton concentration h are not shown, as there are only small quantitative differences between the respective cases. Figure 5.2 suggests that, irrespective of the chosen kernel⁹, higher cooperative intraspecific interactions (larger α values) or slower diffusion delay the invasion of cells in the whole region, leading instead to enhanced proliferation. On the long run the cells

 $^{^{9}}$ We performed simulations with several other kernels, including the so-called 'Mexican hat' (also known as Ricker wavelet, see e.g., [52, 153] for its use in related, but different contexts), cosine, and Epanechnikov.



Figure 5.1: Simulation results for (5.5.1) with $\beta = \mu = 1$. First two columns: $\alpha = 2$, 3rd column: $\alpha = 6.2$, last column: $\alpha = 8.15$. Uniform kernel used with $\rho = 1$.

tend to fill the whole space and remain at their carrying capacity. This behavior endorses the results in *Section 5.4* and is particularly well visible for the logistic kernel, which satisfies all conditions in the proofs of the theoretical results of *Sections 5.3* and *5.4*; the process is much slower when a uniform kernel is used, however it has eventually the same outcome. The last row in *Figure 5.2* exhibits the situation of a cell diffusion which is much slower than that of protons. The effect is a delayed filling of the space with cells (and produced protons) and a later formation of the patterns observed in the upper rows. The asymptotic behavior is similar, only it takes longer for the solution to reach the respective states.

To assess the effect of nonlocality we performed simulations with the source term in the *u*-equation of (5.5.1) replaced by $\mu(h)u^{\alpha}(1-u^{\beta})$. The results are shown in Figure 5.3. The first two columns illustrate the case with the same source term for proton concentration as above, namely g(u, h) = u(1-h), for which no patterns seem to develop (we tried several combinations of parameters, including those used for the patterns in Figure 5.2). In fact, decreasing the value of ρ in the uniform kernel $J_U(x; \rho)$ eventually leads to the local version of the system. The plots in the leftmost column were produced with $d = D_H$, while those in the middle column used $d \ll D_H$. The behavior of u and h is the same, with the difference of the second case inferring a slower spread of cells and protons. The last column in Figure 5.2 already shows the tendency of disappearing patterns when approaching the local case. The last column of Figure 5.3 shows the case where the source term in the h-equation is replaced by $g(u, h) = u + uh - \gamma h^2$, as proposed in Remark 5.5.2.¹⁰

No patterns for u were observed for the local model, which, together with the simulations performed for intermediary values of ρ , suggests that the patterns are driven by the nonlocality of cell-cell interactions, more precisely by intraspecific competition. The simulations also confirm the long time behavior of the system, even in the local case.

¹⁰We tried several other source terms satisfying the conditions in *Remark 5.5.2*, e.g., g(u, h) = uh/(1 + uh + h), all resulting in the same qualitative behavior.



Figure 5.2: Simulation results for (5.5.1) with $\beta = 20$ and $\mu = 100$. Upper row: $\alpha = 2$, lower row: $\alpha = 10$. First column *J* logistic, other columns *J* uniform: 2nd column: $\rho = 1$, 3rd column: $\rho = 0.6$, 4th column: $\rho = 0.05$. Upper rows: $d = D_H = 1$, last row: $d \ll D_H$.



Figure 5.3: Simulation results for (5.5.1) with local source term $\mu(h)u^{\alpha}(1-u^{\beta})$ replacing the one in the equation for u. Left and middle column: g(u,h) = u(1-h) with $d = D_H$ and $d \ll D_H$, respectively. Right column: $g(u,h) = u + uh - \gamma h^2$, $d = D_H$.

5.7 Discussion

In this chapter we investigated a model describing pH-tactic behavior of cells with nonlocal source terms. As such, it extends the one in [99], which studied the Fisher-KPP equation with nonlocal intraspecific competition with various powers of the solution. In contrast to [99] we handled here a problem in a bounded domain, and the population dynamics was coupled to that of the proton concentration, which also led to a taxis term. The proof of our results concerning global well-posedness and long time behavior relied, however, to a substantial extent on the methods in [99]. We also dealt here with space-dependent tensor coefficients in the motility terms, which involve myopic rather than Fickian diffusion. The dissipative effect of the repellent pH-taxis contributed to reducing some of the difficulties in the analysis - as long as the required conditions on the functions involved in the system are satisfied.

Among the relatively few existing models with nonlocal source terms, the one in [136] is closely related, however it features several differences: the cells perform attractive haptotaxis towards gradients of extracellular matrix (ECM), the nonlocal source terms are contained in both equations, do not involve any powers, and the Fickian diffusion of cells has a constant coefficient. Our model requires less regularity for the interaction kernel and the motility coefficients involve a tensor and are more general. On the other hand, the nonexploding solution behavior is favorized in our case by repellent chemotaxis. We also provided an informal model deduction and an assessment of the long time solution behavior. The analysis done in [113] for a model with standard motility and with nonlocal source terms as in [136], but with one or two species performing chemotaxis towards the same attractant imposes certain requirements on the forcing term of the latter, mainly in order to obtain the asymptotic behavior of the cell-related solution components. Our condition (5.4.1) imposed for similar purposes on the source term of the tactic signal looks rather differently. The attraction-repulsion chemotaxis models considered in [129] have closer similarities with our setting, as far as the nonlocal intraspecific interactions are concerned. Major differences occur through our system only featuring two equations, in the source terms of the chemical cues, and in the motility terms: the latter involve in our case the space-dependent tensor $\mathbb{D}(x)$ and myopic diffusion, while the nonlocal reaction term in the proton dynamics is more general. We also prove an explicit long time behavior of both solution components and provide a short analysis of space-time patterns (in 1D), along with numerical simulations.

Our preliminary analysis in Section 5.5 and the simulation results in Section 5.6 suggest that patterns occur only in the nonlocal model, are not of Turing type, and seem to be driven by the nonlocal source terms and influenced by the chosen kernel and the combination of parameters in the nonlocal term. This is in line with the pattern behavior observed in [99] and with other works concerning reaction-diffusion problems with nonlocal intra- and/or interspecific competition, cf. e.g., [64, 74, 119, 131, 142, 153]. Those works involved more or less similar source terms and no taxis, however the repellent pH-taxis contained in our model does not seem to have a relevant influence on the patterns.

Open problems relate to a thorough study of patterns depending on the interplay between the parameters α , β , μ and the influence of the kernel J. Moreover, the well-posedness, asymptotic and blow-up behavior, along with patterning are largely unknown in the case of a degenerating motility tensor - the less so in combination with myopic diffusion and/or other types of taxis.

Indeed, these can lead in the local case to very complex issues even in 1D, as shown e.g., in [149, 151].

On a PDE-ODE-PDE model for two interacting cell populations under the influence of an acidic environment and with nonlocal intra- and interspecific growth limitation

This chapter is based on the article "On a PDE-ODE-PDE model for two interacting cell populations under the influence of an acidic environment and with nonlocal intra- and interspecific growth limitation".¹ The presentation has been adapted for use in this dissertation to clarify the details of the proofs and guarantee consistency of the notation.

6.1 Motivation

Tumor heterogeneity is a well established fact [75]. The neoplastic tissue is -among otherscomposed of several cell phenotypes, all of which are related to the stage within the cell cycle. To simply, of this vast variety we only consider here two phenotypes: active and quiescent cells. The former are supposed to be motile and proliferate, while the latter just infer transitions toward or from activity. While competing with their active counterparts, quiescent cells can also be degraded. Furthermore, the advancement through the cell cycle and the corresponding phenotypic switch is influenced, inter alia, by biochemical factors in the peritumoral space, see [75] and references therein. In particular, pH regulation is a key feature in tumor cell cycle progression, which it can delay or even inhibit [18, 61, 62, 126].

The interactions of cells with their environment occur not only locally, but cells can perceive their surroundings in a far more extensive manner, by way of protrusions like cytonemes/filopodia/ invadopodia, tunneling nanotubes etc. [24, 112, 130, 144]. This motivated the introduction of mathematical models for cell migration, proliferation, and spread. Most of them are of the reaction-diffusion-transport type, with spatial nonlocalities occuring in the advection terms, mainly to model cell-cell and/or cell-tissue adhesions, or nonlocal taxis see e.g., *Chapters 3* and 4 and [5, 23, 43, 81, 117, 156], or in the source terms, to describe intraspecific interactions over a whole sensing range as in *Chapter 5* and [99]. We refer to e.g., [4, 114, 135] for settings also involving nonlocal interspecific competition in different, but related contexts, where the focus is on global stability and pattern issues. The work [136] also considered spatially nonlocal interspecific interactions, but of cancer cells with extracellular matrix and both species featured such

¹[48] The article is available online under https://doi.org/10.48550/arXiv.2409.12657.

terms. For a recent review on nonlocal models for cell migration see [28]; for more comprehensive reviews of nonlocal models in a broader context refer to [51, 87].

Nonlocal models can be obtained, thus far still in a non-rigorous manner, from space- or velocityjump descriptions on the mesoscopic level (also including the kinetic theory of active particles framework [11]), possibly also accounting for microscale dynamics like binding of transmembrane units to soluble or unsoluble ligands. We refer to *Chapter 5* and [21, 44, 99, 156] for such deductions.

The remainder of this chapter is structured as follows: in Section 6.2 we present the model consisting of a PDE-ODE-PDE system, along with requirements for the involved parameters and functions. Sections 6.3 and 6.4 are dedicated to proving global existence of a nonnegative weak solution to the system, in the sense specified therein. In Section 6.5 we perform numerical simulations in 1D within various scenarios, to get some insight into boundedness and patterning behavior under the influence of different choices of relevant parameters, interaction kernels, and phenotypic switch triggered by acidity. Finally, Section 6.6 provides some concluding remarks and an outlook.

6.2 Model

In the following u and w represent the densities of active and of quiescent cells, respectively, whereas h is the concentration of protons in the extracellular space. By 'active' we mean here cells which are migrating and proliferating. On the other hand, 'quiescent' means cells which only interact with their active counterparts and with the environment, without moving nor being able to proliferate. We consider the IBVP in a bounded domain $\Omega \subset \mathbb{R}^n$ having a sufficiently regular boundary $\partial \Omega$ with no-flux boundary conditions

$$\begin{cases} \partial_{t}u = \nabla \cdot (\psi(w,h)\nabla u) + \mu_{1}u^{\alpha} \left(1 - J_{1}(x,h) * u^{\beta} - J_{2}(x,h) * w^{\gamma}\right) \\ + \tilde{\mu}_{3}(h)F(w) & \text{in } \Omega \times (0,\infty), \\ \partial_{t}w = \mu_{2}(h)(1-w)u - \mu_{3}(h)F(w) & \text{in } \Omega \times (0,\infty), \\ \partial_{t}h = D_{H}\Delta h + g(u,w) - \lambda h & \text{in } \Omega \times (0,\infty), \\ \partial_{\nu}u = \partial_{\nu}h = 0 & \text{on } \partial\Omega \times (0,\infty), \\ u(\cdot,0) = u_{0}, w(\cdot,0) = w_{0}, h(\cdot,0) = h_{0} & \text{in } \Omega. \end{cases}$$
(6.2.1)

The first term on the right-hand side of the first PDE in (6.2.1) describes nonlinear diffusion of active cells. The diffusion coefficient ψ can thereby depend on w and h: a large amount of w-cells can increase the population pressure, thus leading to faster diffusion; too many quiescent cells would, however, impede migration (e.g., due to lack of space). Large h-values are also supposed to enhance motility, as the active cells tend to leave such areas faster than more favorable places. The next term describes proliferation of active cells, which is limited by spatially nonlocal intraand interspecific interactions. As in *Chapter 5* and [99] we consider the exponents α , β , γ in the weak Allee effect and the competition/crowding terms with the interaction kernels J_1 and J_2 . The latter can be seen as weighting the influence of either interactions on the dynamics of u and over a whole region. This description enables a more flexible characterization of the interaction

121

strengths and is related to the size of u- and w-cell clusters exchanging information with (bunches of) active cells. Eventually, the last term describes phenotypic switch from quiescent to active cells; this transition is happening with a certain saturation and its rate $\tilde{\mu}_3$ depends on the concentration h of protons. Indeed, less acidic environments favor exit from the quiescent phase and advancement towards activity [22, 140].

The second equation in (6.2.1) is an ODE describing the dynamics of quiescent cells. These are supposed to be non-motile, to be produced by active cells with a rate μ_2 which depends on the acidity in the peritumoral space, and to infer a transition to activity, again with an aciditydependent rate μ_3 , which might differ from $\tilde{\mu}_3$. We also include a kind of acidity-triggered competition between active and quiescent cells; it might have an own rate, but to keep the number of model coefficients as low as possible we take it to be $\mu_2(h)$, too.

The third equation in system (6.2.1) is again a reaction-diffusion PDE and models the dynamics of proton concentration h. Protons are very small in comparison with cells and accordingly able to diffuse quite fastly. They are produced by both tumor cell phenotypes (primarily by active cells and to a lesser amount by quiescent ones) and infer natural decay (e.g., by proton buffering). Concrete choices of motility, transition, and proliferation coefficients will be provided in Section 6.5.

This setting extends our macroscopic model from *Chapter 5* in the sense that we consider here two interacting populations, the dynamics of both being influenced by that of the acidity in their surroundings. Instead of the tumor diffusion tensor depending only on space we have here a dependency on two of the solution components, however we do not include any repellent pH-taxis, but focus instead on the nonlocal interactions and on the phenotypic switch. It also extends the model in [136], where the two interacting species are not influenced by a third one, the diffusion of cells is of the linear type, and there are no transitions from one species to the other, although all interactions therein are nonlocal in space.

The model can be obtained in a way similar to the meso-to-macro deduction performed in *Chapter 5*, if the dynamics of w and h is given as in the second and third equations of (6.2.1), respectively. Although it is not clear how to obtain nonlinear diffusion in general, this can be achieved if the diffusion coefficient is only depending on macroscopic quantities other than u. If only linear diffusion is considered, then the method provides a space-dependent (myopic) diffusion tensor of u-cells, which by an adequate choice of the cell velocity distribution leads to classical Fickian diffusion.

Moreover, we make the following assumptions on involved parameters and functions:

 $\alpha, \beta, \gamma \ge 1$ satisfy

$$\alpha < \begin{cases} 1 + \beta, & n = 1, 2, \\ 1 + \frac{2\beta}{n}, & n > 2, \end{cases}$$
(6.2.2)

 $\cdot \ \mu_1, D_H, \lambda > 0,$

 $\psi \in C^1(\mathbb{R}^+_0 \times \mathbb{R}^+_0)$ with derivatives $\partial_w \psi$, $\partial_h \psi$ that are Lipschitz continuous on $[0,1]^2$ and

$$\psi(w,h) \ge \delta > 0 \text{ for } h, w \in [0,1], \tag{6.2.3}$$

- · $\mu_2, \mu_3 \in C^1(\mathbb{R}^+_0)$ with $\partial_h \mu_2, \partial_h \mu_3 \in L^\infty(\mathbb{R}^+_0)$, $\tilde{\mu}_3$ Lipschitz with Lipschitz constant $L_{\tilde{\mu}_3} \ge 0$, $\mu_2, \mu_3, \tilde{\mu}_3 \ge 0$,
- F(w) = w or $F(w) = \frac{w}{1+w}$ and set $\tilde{F}(w) = 1$ if F(w) = w and $\tilde{F}(w) = \frac{1}{1+w}$ if $F(w) = \frac{w}{1+w}$,
- g is Lipschitz continuous on $\mathbb{R}_0^+ \times \mathbb{R}_0^+$ with constant $L_g > 0$ and satisfies

$$0 \leqslant g(u, w) \leqslant G \tag{6.2.4}$$

for $G \in (0, \infty)$ s.t. $\frac{G}{\lambda} \leq 1$,

• for i = 1, 2 and $B := B_{\operatorname{diam}(\Omega)}(0)$ it holds that

 $J_i(x, \cdot)$ is Lipschitz continuous on \mathbb{R}^+_0 for $x \in B$ with constant $L_{J_i}(x) \ge 0$, (6.2.5a)

$$L_{J_i}, J_i(\cdot, 0) \in L^{p_i}(B) \text{ for some } p_i \in (1, \infty), \tag{6.2.5b}$$

$$J_2 \ge 0, \ J_1 \ge \eta > 0 \text{ for } 0 \le h \le 1, \tag{6.2.5c}$$

 $u_0 \in C(\overline{\Omega}), w_0, h_0 \in H^1(\Omega) \text{ and } 0 \leq u_0, h_0, w_0 \leq 1.$

These assumptions are primarily made out of technical reasons, in order to support the analysis in *Sections 6.3* and *6.4*, however most of them are reasonable from the application viewpoint: all parameters should be nonnegative and the interactions should involve at least one cell on either side; the diffusion of active and motile cells should be nondegenerate; there should be an effective, but uniformly limited production of protons, which should not dominate the natural decay in a too substantial manner; the interaction kernels should be nonnegative and there should be genuine intraspecific interactions, while the proton concentration remains reasonably bounded, and the initial conditions should be nonnegative and uniformly bounded.

6.3 Global existence of a classical solution to an approximate problem

Let $\vartheta \in (0,1)$. There are sequences of initial values $(u_{0\varepsilon})_{\varepsilon \in (0,1)}$, $(w_{0\varepsilon})_{\varepsilon \in (0,1)}$, $(h_{0\varepsilon})_{\varepsilon \in (0,1)}$ in $C^{2+\vartheta}(\overline{\Omega})$ s.t.

$$0 \leqslant u_{0\varepsilon}, w_{0\varepsilon}, h_{0\varepsilon} \leqslant 1, \tag{6.3.1a}$$

$$\partial_{\nu} u_{0\varepsilon} = \partial_{\nu} w_{0\varepsilon} = \partial_{\nu} h_{0\varepsilon} = 0 \text{ on } \partial\Omega,$$

$$u_{0\varepsilon} \xrightarrow[\varepsilon \to 0]{} u_0 \text{ in } C(\overline{\Omega}),$$
 (6.3.1b)

$$w_{0\varepsilon} \xrightarrow[\varepsilon \to 0]{} w_0, \ h_{0\varepsilon} \xrightarrow[\varepsilon \to 0]{} h_0 \text{ in } H^1(\Omega).$$
 (6.3.1c)

Throughout this chapter we consider for $\varepsilon \in (0, 1)$ the approximate IBVP

$$\begin{aligned}
\partial_{t}u_{\varepsilon} &= \nabla \cdot (\psi(w_{\varepsilon}, h_{\varepsilon})\nabla u_{\varepsilon}) + \mu_{1}u_{\varepsilon}^{\alpha} \left(1 - J_{1}(x, h_{\varepsilon}) * u_{\varepsilon}^{\beta} - J_{2}(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma}\right) \\
&+ \tilde{\mu}_{3}(h_{\varepsilon})F(w_{\varepsilon}) & \text{in } \Omega \times (0, \infty), \\
\partial_{t}w_{\varepsilon} &= \varepsilon \Delta w_{\varepsilon} + \mu_{2}(h_{\varepsilon})(1 - w_{\varepsilon})u_{\varepsilon} - \mu_{3}(h_{\varepsilon})F(w_{\varepsilon}) & \text{in } \Omega \times (0, \infty), \\
\partial_{t}h_{\varepsilon} &= D_{H}\Delta h_{\varepsilon} + g(u_{\varepsilon}, w_{\varepsilon}) - \lambda h_{\varepsilon} & \text{in } \Omega \times (0, \infty), \\
\partial_{\nu}u_{\varepsilon} &= \partial_{\nu}w_{\varepsilon} = \partial_{\nu}h_{\varepsilon} = 0 & \text{on } \partial\Omega \times (0, \infty), \\
u_{\varepsilon}(\cdot, 0) &= u_{0\varepsilon}, w_{\varepsilon}(\cdot, 0) = w_{0\varepsilon}, h_{\varepsilon}(\cdot, 0) = h_{0\varepsilon} & \text{in } \Omega.
\end{aligned}$$
(6.3.2)

1

We show local existence of a solution with a fixed-point argument.

Lemma 6.3.1. For all $\varepsilon \in (0,1)$ there is $T_{max,\varepsilon} \in (0,\infty]$ and a solution $(u_{\varepsilon}, w_{\varepsilon}, h_{\varepsilon})$ of (6.3.2) in $\left(C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T_{max,\varepsilon}))\right)^3$ with $0 \le u_{\varepsilon}$ and $0 \le w_{\varepsilon}, h_{\varepsilon} \le 1$ s.t. either $T_{max,\varepsilon} = \infty$ or $T_{max,\varepsilon} < \infty$ and

$$\lim_{t \nearrow T_{max,\varepsilon}} \left(\|u_{\varepsilon}(\cdot,t)\|_{C^{2+\vartheta}(\overline{\Omega})} + \|w_{\varepsilon}(\cdot,t)\|_{C^{2+\vartheta}(\overline{\Omega})} + \|h_{\varepsilon}(\cdot,t)\|_{C^{2+\vartheta}(\overline{\Omega})} \right) = \infty.$$
(6.3.3)

Proof. Let $\varepsilon \in (0, 1)$ and $T \in (0, 1)$ small enough. For h < 0 we set

$$\mu_2(h) := \mu_2(-h), \ \mu_3(h) := \mu_3(-h), \ \tilde{\mu}_3(h) := \tilde{\mu}_3(-h).$$

We will perform a fixed-point argument in

$$S := \left\{ \left(\overline{u}, \overline{w} \right) \in \left(C^{\vartheta, \frac{\vartheta}{2}} (\overline{\Omega} \times [0, T]) \right)^2 : \overline{u}, \overline{w} \ge 0, \ \left\| \overline{u} \right\|_{C^{\vartheta, \frac{\vartheta}{2}} (\overline{\Omega} \times [0, T])} + \left\| \overline{w} \right\|_{C^{\vartheta, \frac{\vartheta}{2}} (\overline{\Omega} \times [0, T])} \le M + 1 \right\}$$

for $M := \|u_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} + \|w_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} + 1$. For $(\overline{u}, \overline{w}) \in S$, we consider the three decoupled IBVPs

$$\begin{cases} \partial_{t}u_{\varepsilon} = \nabla \cdot \left(\psi(w_{\varepsilon}, h_{\varepsilon})\nabla u_{\varepsilon}\right) - \mu_{1}\overline{u}^{\alpha-1} \left(J_{1}(x, h_{\varepsilon}) * \overline{u}^{\beta} + J_{2}(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma}\right) u_{\varepsilon} \\ + \mu_{1}\overline{u}^{\alpha} + \tilde{\mu}_{3}(h_{\varepsilon})\tilde{F}(\overline{w})w_{\varepsilon} & \text{in } \Omega \times (0, \infty), \\ \partial_{\nu}u_{\varepsilon} = 0 & \text{on } \partial\Omega \times (0, \infty)0, \\ u_{\varepsilon}(\cdot, 0) = u_{0\varepsilon} & \text{in } \Omega, \end{cases}$$

$$(6.3.4)$$

$$\begin{aligned}
\partial_t w_{\varepsilon} &= \varepsilon \Delta w_{\varepsilon} + \mu_2 (h_{\varepsilon}) (1 - w_{\varepsilon}) \overline{u} - \mu_3 (h_{\varepsilon}) \tilde{F}(\overline{w}) w_{\varepsilon} & \text{ in } \Omega \times (0, \infty), \\
\partial_{\nu} w_{\varepsilon} &= 0 & \text{ on } \partial \Omega \times (0, \infty), \\
w_{\varepsilon} (\cdot, 0) &= w_{0\varepsilon} & \text{ in } \Omega,
\end{aligned}$$
(6.3.5)

and

$$\begin{cases} \partial_t h_{\varepsilon} = D_H \Delta h_{\varepsilon} + g(\overline{u}, \overline{w}) - \lambda h_{\varepsilon} & \text{in } \Omega \times (0, \infty), \\ \partial_{\nu} h_{\varepsilon} = 0 & \text{on } \partial \Omega \times (0, \infty), \\ h_{\varepsilon}(\cdot, 0) = h_{0\varepsilon} & \text{in } \Omega. \end{cases}$$
(6.3.6)

We start with (6.3.6). Due to the Hölder continuity of \overline{u} and \overline{w} and the Lipschitz continuity of g we can apply *Theorem A.1.6* with coefficients

$$a_{ii} := D_H, a_i := 0, a := \lambda, b_i := \nu_i, b := 0, f := g(\overline{u}, \overline{w})$$

for $i \in \{1, \ldots, n\}$ to (6.3.6) and obtain a unique solution $h_{\varepsilon} \in C^{2+\vartheta, 1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0, T])$ satisfying

$$\|h_{\varepsilon}\|_{C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} \leqslant C_1\left(\|g(\overline{u},\overline{w})\|_{C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} + \|h_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})}\right) \leqslant C_2\left(M,\|h_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})}\right).$$

Moreover, due to the Lipschitz continuity of μ_2, μ_3 , the Hölder continuity of $\overline{u}, \overline{w}$ and Lemma 2.2.3 we conclude again from *Theorem A.1.6* with the coefficients

$$a_{ii} := \varepsilon, \ a_i := 0, \ a := \mu_2(h_\varepsilon)\overline{u} + \mu_3(h_\varepsilon)\widetilde{F}(\overline{w}), \ b_i := \nu_i, \ b := 0, \ f := \mu_2(h_\varepsilon)\overline{u}$$

for $i \in \{1, \ldots, n\}$ that there is a unique solution $w_{\varepsilon} \in C^{2+\vartheta, 1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0, T])$ to (6.3.5) satisfying

$$\|w_{\varepsilon}\|_{C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} \leq C_{3} \left(\|\mu_{2}(h_{\varepsilon})\overline{u}\|_{C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} + \|w_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})} \right)$$

$$\leq C_{4} \left(M, \|w_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})}, \|h_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})} \right).$$
(6.3.7)

Then, we can estimate

$$0 \leq \mu_2(h_{\varepsilon})\overline{u} = (w_{\varepsilon})_t - \varepsilon \Delta w_{\varepsilon} + \left(\mu_2(h_{\varepsilon})\overline{u} + \mu_3(h_{\varepsilon})\tilde{F}(\overline{w})\right) w_{\varepsilon}$$

and

$$0 \leqslant g(\bar{u}, \bar{w}) = (h_{\varepsilon})_t - D_H \Delta h_{\varepsilon} + \lambda h_{\varepsilon} \leqslant G \leqslant \lambda$$

due to (6.2.4). Hence, from a parabolic comparison principle (*Theorem A.1.9*) and due to (6.3.1a) it follows that $0 \leq w_{\varepsilon}$ and $0 \leq h_{\varepsilon} \leq 1$. Further, we set $v_{\varepsilon} := 1 - w_{\varepsilon}$ and estimate

$$0 \leqslant \mu_3(h_{\varepsilon}) \tilde{F}(\overline{w}) w_{\varepsilon} = (v_{\varepsilon})_t - \varepsilon \Delta v_{\varepsilon} + \mu_2(h_{\varepsilon}) \overline{u} v_{\varepsilon}$$

and combining this with (6.3.1a) we conclude that $v_{\varepsilon} \ge 0$ and consequently, $w_{\varepsilon} \le 1$. Now, we set

$$\begin{split} a_{ii} &:= \psi(w_{\varepsilon}, h_{\varepsilon}), \\ a_i &:= -\partial_w \psi(w_{\varepsilon}, h_{\varepsilon})(w_{\varepsilon})_{x_i} - \partial_h \psi(w_{\varepsilon}, h_{\varepsilon})(h_{\varepsilon})_{x_i}, \\ a &:= \mu_1 \overline{u}^{\alpha - 1} \left(J_1(x, h_{\varepsilon}) * \overline{u}^{\beta} + J_2(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma} \right), \\ b_i &:= \nu_i, \ b &:= 0, \\ f &:= \mu_1 \overline{u}^{\alpha} + \tilde{\mu}_3(h_{\varepsilon}) \tilde{F}(\overline{w}) w_{\varepsilon} \end{split}$$

for $i \in \{1, \ldots, n\}$ to apply again *Theorem A.1.6* to the equation corresponding to (6.3.4) in nondivergence form. From this theorem due to the Lipschitz continuity of $\partial_h \psi$ and $\partial_w \psi$ together with the bounds and the Hölder continuity of w_{ε} and h_{ε} and its gradients, the Lipschitz continuity of $\tilde{\mu}_3$ and *Lemmas 2.2.3* and *2.3.2* it follows that for $\kappa := \min\{1, \alpha - 1\}\vartheta$ there is a unique solution $u_{\varepsilon} \in C^{2+\kappa, 1+\frac{\kappa}{2}}(\overline{\Omega} \times [0, T])$ to (6.3.4) that satisfies

$$\begin{aligned} \|u_{\varepsilon}\|_{C^{2+\kappa,1+\frac{\kappa}{2}}(\overline{\Omega}\times[0,T])} &\leqslant C_{5} \left(\|a_{ii}\|_{C^{\kappa,\frac{\kappa}{2}}(\overline{\Omega}\times[0,T])}, \|a_{i}\|_{C^{\kappa,\frac{\kappa}{2}}(\overline{\Omega}\times[0,T])}, \|a\|_{C^{\kappa,\frac{\kappa}{2}}(\overline{\Omega}\times[0,T])}, \|b_{i}\|_{C^{1+\kappa,\frac{1+\kappa}{2}}} \right) \\ & \cdot \left(\|f\|_{C^{\kappa,\frac{\kappa}{2}}(\overline{\Omega}\times[0,T])} + \|u_{0\varepsilon}\|_{C^{2+\kappa}(\overline{\Omega})} \right) \\ &\leqslant C_{6} \left(M, \|u_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})}, \|w_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})}, \|h_{0\varepsilon}\|_{C^{2+\vartheta}(\overline{\Omega})} \right) \end{aligned}$$
(6.3.8)

due to the embedding of Hölder spaces. Further, we estimate

$$\begin{aligned} &(u_{\varepsilon})_{t} - \psi(w_{\varepsilon}, h_{\varepsilon}) \Delta u_{\varepsilon} - (\partial_{h} \psi(w_{\varepsilon}, h_{\varepsilon}) \nabla h_{\varepsilon} + \partial_{w} \psi(w_{\varepsilon}, h_{\varepsilon}) \nabla w_{\varepsilon}) \cdot \nabla u_{\varepsilon} \\ &+ \mu_{1} \overline{u}^{\alpha - 1} \left(J_{1}(x, h_{\varepsilon}) * \overline{u}^{\beta} + J_{2}(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma} \right) u_{\varepsilon} \\ &= \mu_{1} \overline{u}^{\alpha} + \tilde{\mu}_{3}(h_{\varepsilon}) \tilde{F}(\overline{w}) w_{\varepsilon} \ge 0 \end{aligned}$$

and conclude from the comparison principle in *Theorem A.1.9* that $u_{\varepsilon} \ge 0$. Now, we estimate with (6.3.7) and (6.3.8) and *Lemma 2.2.9(iii)*:

$$\begin{split} & \|u_{\varepsilon}\|_{C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} + \|w_{\varepsilon}\|_{C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} \\ \leqslant & \|u_{\varepsilon} - u_{0\varepsilon}\|_{C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} + \|w_{\varepsilon} - w_{0\varepsilon}\|_{C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega}\times[0,T])} + \|u_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} + \|w_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} \end{split}$$

$$\leq 2 \max\{1, K_{I}(\vartheta)\} T^{\frac{1}{2}\min\{2-\vartheta, 1+\kappa\}} \|u_{\varepsilon}\|_{C^{2+\kappa, 1+\frac{\kappa}{2}}(\overline{\Omega} \times [0,T])} \\ + 2 \max\{1, K_{I}(\vartheta)\} T^{\frac{1}{2}\min\{2-\vartheta, 1+\vartheta\}} \|w_{\varepsilon}\|_{C^{2+\vartheta, 1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])} + \|u_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} + \|h_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} \\ \leq 2 \max\{1, K_{I}(\vartheta)\} T^{\frac{1}{2}}(C_{6}+C_{4}) + \|u_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} + \|w_{0\varepsilon}\|_{C^{\vartheta}(\overline{\Omega})} \leq M+1$$

for

$$T \leq \left(\frac{1}{2\max\{1, K_I(\vartheta)\}(C_6 + C_4)}\right)^2$$

where $K_I(\vartheta) > 0$ denotes the constant from the continuous embedding of $W^1_{\infty}(\Omega)$ into $C^{\vartheta}(\overline{\Omega})$ from Lemma 2.2.8(ii). Hence, $(u_{\varepsilon}, w_{\varepsilon}) \in S$ and the operator

$$K: S \to S, \, (\overline{u}, \overline{w}) \mapsto (u_{\varepsilon}, w_{\varepsilon})$$

is well-defined. Due to the continuous dependence of the solution on the coefficients in *The*orem A.1.6 the operator K is continuous. Moreover, (6.3.7) and (6.3.8) imply that K maps bounded subsets of $(C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T]))^2$ on bounded subsets of $(C^{2+\kappa,1+\frac{\kappa}{2}}(\overline{\Omega} \times [0,T]))^2$. Hence, from the compact embedding $C^{2+\kappa,1+\frac{\kappa}{2}}(\overline{\Omega} \times [0,T]) \hookrightarrow C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])$ we conclude that K is a compact operator. Schauder's fixed-point theorem (*Theorem A.2.2*) implies that K has a fixed-point $(u_{\varepsilon}, w_{\varepsilon})$ in $(C^{\vartheta,\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T]))^2$, where additionally $u_{\varepsilon} \in C^{2+\kappa,1+\frac{\kappa}{2}}(\overline{\Omega} \times [0,T])$ and $w_{\varepsilon} \in C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])$ as was shown above. Applying *Theorem A.1.6* to u_{ε} again but with

$$a := 0, f := \mu_1 u_{\varepsilon}^{\alpha} \left(1 - J_1(x, h_{\varepsilon}) * u_{\varepsilon}^{\beta}(x, t) - J_2(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma} \right) + \tilde{\mu}_3(h_{\varepsilon}) F(w_{\varepsilon})$$

we conclude that also $u_{\varepsilon} \in C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])$. Finally, (6.3.3) follows extending the solution to its maximal existence time $T_{max,\varepsilon}$.

We show the global boundedness of our solution by adapting the estimates from Step 1 and 2 of the proof of Theorem 1.1 in [99] similar to *Chapter 5*.

Lemma 6.3.2. There is $C_7 > 0$ s.t. $||u_{\varepsilon}||_{L^{\infty}(\Omega \times (0,T_{max,\varepsilon}))} \leq C_7$ for all $\varepsilon \in (0,1)$.

Proof.

Step 1. Let $\varepsilon \in (0,1)$ and $q > \max\{1, \beta + \alpha - 1\}$. Consider $t \in (0, T_{max,\varepsilon})$. We multiply the first equation of (6.3.2) by qu_{ε}^{q-1} and integrate over Ω to obtain

$$\begin{split} \frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{q} \, \mathrm{d}x &= -q(q-1) \int_{\Omega} \psi(w_{\varepsilon}, h_{\varepsilon}) \left| \nabla u_{\varepsilon} \right|^{2} u_{\varepsilon}^{q-2} \, \mathrm{d}x \\ &+ q \mu_{1} \int_{\Omega} u_{\varepsilon}^{q+\alpha-1} \left(1 - J_{1}(x, h_{\varepsilon}) * u_{\varepsilon}^{\beta} - J_{2}(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma} \right) \, \mathrm{d}x \\ &+ q \int_{\Omega} \tilde{\mu}_{3}(h_{\varepsilon}) F(w_{\varepsilon}) u_{\varepsilon}^{q-1} \, \mathrm{d}x \end{split}$$

using partial integration. Hence, we conclude from (6.2.3) and (6.2.5c), the continuity of $\tilde{\mu}_3$, the boundedness of h_{ε} , the fact that $F(w_{\varepsilon}) \leq 1$ for $w_{\varepsilon} \leq 1$ and Young's inequality that

$$\begin{split} & \frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{q} \, \mathrm{d}x + \frac{4(q-1)}{q} \delta \int_{\Omega} \left| \nabla u_{\varepsilon}^{\frac{q}{2}} \right|^{2} \, \mathrm{d}x + q \mu_{1} \eta \int_{\Omega} u_{\varepsilon}^{\beta} \, \mathrm{d}x \int_{\Omega} u_{\varepsilon}^{q+\alpha-1} \, \mathrm{d}x \\ \leqslant q \mu_{1} \int_{\Omega} u_{\varepsilon}^{q+\alpha-1} \, \mathrm{d}x + q \| \tilde{\mu}_{3} \|_{L^{\infty}(0,1)} \int_{\Omega} u_{\varepsilon}^{q-1} \, \mathrm{d}x. \end{split}$$

Setting $C_8 := \mu_1 + \|\tilde{\mu}_3\|_{L^{\infty}(0,1)}$ and adding $qC_8 \|u_{\varepsilon}\|_{L^q(\Omega)}^q$ on both sides of the above equation and using Young's inequality we arrive at

$$\frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{q} dx + \frac{4(q-1)}{q} \delta \int_{\Omega} \left| \nabla u^{\frac{q}{2}} \right|^{2} dx + q\mu_{1}\eta \int_{\Omega} u_{\varepsilon}^{\beta} dx \int_{\Omega} u_{\varepsilon}^{q+\alpha-1} dx + qC_{8} \int_{\Omega} u_{\varepsilon}^{q} dx$$

$$\leq 2qC_{8} \left(\int_{\Omega} u_{\varepsilon}^{q+\alpha-1} dx + |\Omega| \right).$$
(6.3.9)

From Lemma A.4.1 it follows for $K_{18} = \frac{C_8}{\delta}$ and $K_{22} = \frac{2C_8}{\mu_1 \eta}$ that

$$\int_{\Omega} u_{\varepsilon}^{q+\alpha-1} \,\mathrm{d}x \leqslant \frac{2(q-1)}{q^2 C_8} \delta \int_{\Omega} |\nabla u_{\varepsilon}^{\frac{q}{2}}|^2 \,\mathrm{d}x + \frac{\mu_1 \eta}{2C_8} \int_{\Omega} u_{\varepsilon}^{\beta} \,\mathrm{d}x \int_{\Omega} u_{\varepsilon}^{q+\alpha-1} \,\mathrm{d}x + K_{23}\left(q\right), \qquad (6.3.10)$$

where

$$s \begin{cases} = \infty, & n = 1, \\ \in \left(\frac{2(q+\alpha-1+\beta)}{q-\alpha+1+\beta}, \infty\right), & n = 2, \\ = \frac{2n}{n-2}, & n > 2, \end{cases}$$

$$K_{23}(q) := \left(2 \left(\frac{K_{21}^2 q^2 C_8}{(q-1)\delta} \right)^{\frac{q+\alpha-1-\beta}{q-\alpha+1+\beta-2\frac{q+\alpha-1+\beta}{s}}} + K_{24}(q)^{\frac{q+\alpha-\beta-1}{q-\frac{q+\alpha-1+\beta}{s}}} \right)^{\frac{q-\alpha+1+\beta-\frac{2(q+\alpha-1+\beta)}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}} \cdot \left(\frac{2C_8}{\mu_1\eta} \right)^{\frac{q-2(q+\alpha-1)}{s}} + K_{24}(q)^{\frac{q+\alpha-\beta-1}{q-\frac{q+\alpha-1+\beta}{s}}},$$

and

$$K_{21} := 2K_S \left(1 + 2K_P \right),$$

$$K_{24}(q) := 4K_S |\Omega|^{\frac{1}{2} - \frac{q}{q + \alpha - 1 + \beta}}.$$

Here, K_S denotes the embedding constant from $H^1(\Omega)$ into $L^s(\Omega)$ from Lemma 2.2.8(i) and K_P denotes the constant from the Poincaré inequality. We will leave out the dependence of the constants on s. Hence, inserting (6.3.10) into (6.3.9) we obtain

$$\frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{q} \,\mathrm{d}x + qC_{8} \int_{\Omega} u_{\varepsilon}^{q} \,\mathrm{d}x \leq 2qC_{8} \left(K_{23} \left(q \right) + |\Omega| \right)$$

and conclude from Lemma A.1.20 with $K_{14} = qC_8$ and $K_{15} = 2(K_{23}(q) + |\Omega|)$ that

$$\|u_{\varepsilon}\|_{L^{q}(\Omega)} \leq \sqrt[q]{2(K_{23}(q) + |\Omega|) + \|u_{0\varepsilon}\|_{L^{q}(\Omega)}^{q}} \leq \sqrt[q]{2K_{23}(q) + |\Omega| \left(2 + \|u_{0\varepsilon}\|_{L^{\infty}(\Omega)}^{q}\right)} \xrightarrow[q \to \infty]{(6.3.11)}$$

due to

$$\begin{split} & \left(\left(\frac{q^2}{q-1}\right)^{\frac{q+\alpha-1-\beta}{q-\alpha+1+\beta-2\frac{q+\alpha-1+\beta}{s}} \cdot \frac{q-\alpha+1+\beta-\frac{2(q+\alpha-1+\beta)}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}} \right)^{\frac{1}{q}} \\ & \geqslant q^{\frac{1}{q} \cdot \frac{q+\alpha-1-\beta}{\beta+1-\alpha-\frac{2\beta}{s}}} = \left(q^{1+\frac{\alpha-1-\beta}{q}}\right)^{\frac{1}{\beta+1-\alpha-\frac{2\beta}{s}}} \xrightarrow[q \to \infty]{} \infty. \end{split}$$

Due to (6.3.1a) we can also find an upper bound independent from ε , namely

$$\|u_{\varepsilon}\|_{L^{q}(\Omega)} \leqslant \sqrt[q]{2K_{23}(q) + 3|\Omega|}.$$
(6.3.12)

Step 2. We proceed with a Moser iteration. Set $a := \frac{2(s-1)(\alpha-1)}{s-2}$ and $q_k := 2^k + a$ for $k \in \mathbb{N}$ large enough s.t. $q_k > \max\{1, \beta + \alpha - 1\}$. Then, using Lemma A.4.1 with $K_{18} = \frac{C_8}{\delta}$ and $r = r_k := \frac{2q_{k-1}}{q_k}$ we conclude from (6.3.9) that

$$\frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{q_{k}} dx + \frac{4(q_{k}-1)}{q_{k}} \delta \int_{\Omega} \left| \nabla u_{\varepsilon}^{\frac{q_{k}}{2}} \right|^{2} dx + q_{k} \mu_{1} \eta \int_{\Omega} u_{\varepsilon}^{\beta} dx \int_{\Omega} u_{\varepsilon}^{q_{k}+\alpha-1} dx + q_{k} C_{8} \int_{\Omega} u_{\varepsilon}^{q_{k}} dx$$

$$\leq 2q_{k} C_{8} \left(\int_{\Omega} u_{\varepsilon}^{q_{k}+\alpha-1} dx + |\Omega| \right)$$

$$\leq \frac{4(q_{k}-1)}{q_{k}^{2}} \delta \int_{\Omega} |\nabla u_{\varepsilon}^{\frac{q_{k}}{2}}|^{2} dx$$

$$+ 2q_{k} C_{8} \left((2C_{9}(k) + C_{10}(k)) \left\| u_{\varepsilon}^{\frac{q_{k}}{2}} \right\|_{L^{r_{k}}(\Omega)}^{2r_{k} \frac{2(q_{k}+\alpha-1)}{q_{k}} - q_{k}} + C_{10}(k) + |\Omega| \right), \quad (6.3.13)$$

where

$$C_{9}(k) := \left(\frac{K_{21}^{2}q_{k}^{2}C_{8}}{(q_{k}-1)\delta}\right)^{\frac{2q_{k-1}-2(q_{k}+\alpha-1)}{2q_{k-1}\left(\frac{2}{s}-1\right)+2(\alpha-1)}}$$
$$C_{10}(k) := K_{20}(k)^{\frac{2q_{k-1}-2(q_{k}+\alpha-1)}{\frac{2q_{k}-1}{s}-q_{k}}},$$
$$K_{20}(k) := 4K_{S}(s)|\Omega|^{\frac{q_{k-1}-q_{k}}{2q_{k-1}}}.$$

,

We know from $Lemma \ A.4.2$ that

$$\frac{\frac{2(q_k + \alpha - 1)}{s} - q_k}{2q_{k-1}\left(\frac{2}{s} - 1\right) + 2(\alpha - 1)} = 1$$

and consequently,

$$\|u_{\varepsilon}^{\frac{q_k}{2}}\|_{L^{r_k}(\Omega)}^{2r_k\frac{2(q_k+\alpha-1)}{s}-q_k} = \left(\int_{\Omega} u_{\varepsilon}^{q_{k-1}} \,\mathrm{d}x\right)^2.$$

Inserting this into (6.3.13) we conclude that

$$\frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{q_{k}} \, \mathrm{d}x + q_{k} C_{8} \int_{\Omega} u_{\varepsilon}^{q_{k}} \, \mathrm{d}x$$

$$\leq 2q_{k} C_{8} \left(2C_{9}(k) + 2C_{10}(k) + |\Omega|\right) \max\left\{1, \left(\int_{\Omega} u_{\varepsilon}^{q_{k-1}} \, \mathrm{d}x\right)^{2}\right\}.$$
(6.3.14)

Moreover, Lemma A.4.2 implies that

$$\frac{2q_{k-1} - 2(q_k + \alpha - 1)}{2q_{k-1}\left(\frac{2}{s} - 1\right) + 2(\alpha - 1)} = \frac{s}{s-2} \text{ and } \frac{2q_{k-1} - 2(q_k + \alpha - 1)}{\frac{2q_{k-1}}{s} - q_k} \leqslant \alpha + 1.$$

Further, we can compute that

$$\frac{q_{k-1}-q_k}{2q_{k-1}} \in \left(-\frac{1}{2},0\right).$$

Hence, we can estimate

$$2C_9(k) + 2C_{10}(k) + |\Omega| = 2\left(\frac{K_{21}^2 q_k^2 C_8}{(q_k - 1)\delta}\right)^{\frac{s}{s-2}} + 2\left(4K_S |\Omega|^{\frac{q_{k-1} - q_k}{2q_{k-1}}}\right)^{\frac{2q_{k-1} - 2(q_k + \alpha - 1)}{\frac{2q_{k-1} - 2(q_k + \alpha - 1)}{s}} + |\Omega|$$

$$\leq 2\left(\frac{K_{21}^2 C_8}{\delta} 2^k (1+a)\right)^{\frac{s}{s-2}} + 2\left(4\max\left\{1, K_S\right\}\max\left\{1, |\Omega|^{-\frac{1}{2}}\right\}\right)^{\alpha+1} + |\Omega|$$

$$\leq \frac{\bar{a}}{2} 2^{\frac{s}{s-2}k},$$

where

$$\bar{a} := 2\left(2\left(\frac{K_{21}^2 C_8}{\delta}(1+a)\right)^{\frac{s}{s-2}} + 2\left(4\max\{1, K_S\}\max\{1, |\Omega|^{-\frac{1}{2}}\}\right)^{\alpha+1} + |\Omega|\right)$$

Inserting this into (6.3.14) we obtain

$$\frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{q_k} \, \mathrm{d}x + q_k C_8 \int_{\Omega} u_{\varepsilon}^{q_k} \, \mathrm{d}x \leqslant q_k C_8 \bar{a} 2^{\frac{s}{s-2}k} \max\left\{ 1, \sup_{t \ge 0} \left(\int_{\Omega} u_{\varepsilon}^{q_{k-1}} \, \mathrm{d}x \right)^2 \right\}.$$

For $k \ge 1$ we can estimate that $\int_{\Omega} u_{0\varepsilon}^{q_k} dx \le ||u_{0\varepsilon}||_{L^{\infty}(\Omega)}^{q_k} |\Omega| \le |\Omega|$ due to (6.3.1a). Hence, Lemma A.4.3 with $c_k = q_k C_8$ and $D = \frac{s}{s-2}$ implies that for $k \ge m \ge 1$ large enough, i.e., s.t. $\bar{a}2^{\frac{s}{s-2}m} \ge 1$, it holds that

$$\left(\int_{\Omega} u_{\varepsilon}^{q_{k}} \mathrm{d}x\right)^{\frac{1}{q_{k}}} \leq \left(2\bar{a}\right)^{\frac{2^{k-m+1}-1}{q_{k}}} 2^{\frac{s}{s-2}\left(2(2^{k-m}-1)+m2^{k-m+1}-k\right)} \\ \cdot \max\left\{\sup_{t\geq 0}\left(\int_{\Omega} u_{\varepsilon}^{q_{m-1}} \mathrm{d}x\right)^{\frac{2^{k-m+1}}{q_{k}}}, |\Omega|^{\frac{2^{k-m}}{q_{k}}}, 1\right\}.$$

For $k \to \infty$ we obtain

$$\|u_{\varepsilon}\|_{L^{\infty}(\Omega)} \leq (2\bar{a})^{\frac{1}{2^{m-1}}} 2^{\frac{s(m+1)}{(s-2)2^{m-1}}} \max\left\{\sup_{t\geq 0} \left(\int_{\Omega} u_{\varepsilon}^{q_{m-1}} \,\mathrm{d}x\right)^{\frac{1}{2^{m-1}}}, |\Omega|^{2^{-m}}, 1\right\}$$
(6.3.15)

in $(0, T_{max,\varepsilon})$. We already know from Step 1 that the right-hand side is bounded above by a constant independent from ε . Consequently, $u_{\varepsilon} \in L^{\infty}(\Omega \times (0, T_{max,\varepsilon}))$ is bounded above by a constant independent from ε .

We can also perform a quasi-maximum principle as in Step 3 of the proof of Theorem 1.1 in [99].

Corollary 6.3.3. We find K > 1 and 'small' enough parameters s.t.

$$\|u_{\varepsilon}\|_{L^{\infty}(\Omega\times(0,T_{max,\varepsilon}))} \leqslant K \max\left\{1, \left(\left(4K_{S}|\Omega|^{-\frac{1}{2}}\right)^{\frac{1}{1-\frac{1}{s}}} \frac{2}{\eta}\left(1+\frac{\|\tilde{\mu}_{3}\|_{L^{\infty}(0,1)}}{\mu_{1}}\right)\right)^{\frac{1-\frac{2}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}}\right\}.$$
(6.3.16)

Proof. We want to consider the limit $m \to \infty$ in (6.3.15) for 'small' enough parameters. We already know from (6.3.12) in the proof of Lemma 6.3.2 that for $t \in (0, T_{max,\varepsilon})$ it holds that

$$\int_{\Omega} u_{\varepsilon}^{q_{m-1}} \,\mathrm{d}x \leqslant 2K_{23} \left(q_{m-1} \right) + 3|\Omega|.$$

We fix some m and assume that our parameters are 'small' enough s.t.

$$C_8 = \mu_1 + \|\tilde{\mu}_3\|_{L^{\infty}(0,1)} \leq \frac{\delta}{K_{21}^2 q_{m-1}^2}$$
(6.3.17)

holds. Then, we can estimate

$$\begin{split} & \int_{\Omega} u_{\varepsilon}^{q_{m-1}} \, \mathrm{d}x \\ \leqslant & 7 \max \left\{ \left(2 \left(\frac{1}{2^{m-1} + a - 1} \right)^{\frac{q_{m-1} + \alpha - 1 - \beta}{q_{m-1} - \alpha + 1 + \beta - 2} \frac{q_{m-1} + \alpha - 1 + \beta}{s}}{q_{m-1} - \alpha + 1 + \beta - 2} \right)^{\frac{q_{m-1} + \alpha - 1 + \beta}{s}} \\ & + K_{24}(q_{m-1})^{\frac{q_{m-1} - \frac{q_{m-1} + \alpha - \beta - 1}{q_{m-1} - \frac{q_{m-1} + \alpha - 1 + \beta}{s}}} \right)^{\frac{q_{m-1} - \alpha + 1 + \beta - 2(q_{m-1} + \alpha - 1 + \beta)}{\beta + 1 - \alpha - \frac{2\beta}{s}}} \\ & \cdot \left(\frac{2}{\eta} \left(1 + \frac{\|\tilde{\mu}_{3}\|_{L^{\infty}(0, 1)}}{\mu_{1}} \right) \right)^{\frac{q_{m-1} - \frac{2(q_{m-1} + \alpha - 1)}{s}}{\beta + 1 - \alpha - \frac{2\beta}{s}}}}, \\ & K_{24}(q_{m-1})^{\frac{q_{m-1} + \alpha - \beta - 1}{q_{m-1} - \frac{q_{m-1} + \alpha - 1 + \beta}{s}}}, |\Omega| \right\} =: H(m). \end{split}$$

Finally, we conclude with

$$\lim_{m \to \infty} (H(m))^{\frac{1}{2^{m-1}}} = \max\left\{ 1, \left(\left(4K_S |\Omega|^{-\frac{1}{2}} \right)^{\frac{1}{1-\frac{1}{s}}} \frac{2}{\eta} \left(1 + \frac{\|\tilde{\mu}_3\|_{L^{\infty}(0,1)}}{\mu_1} \right) \right)^{\frac{1-\frac{2}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}} \right\},\\ \lim_{m \to \infty} (2\bar{a})^{\frac{1}{2^{m-1}}} 2^{\frac{s(m+1)}{(s-2)2^{m-1}}} = 1$$

and $|\Omega|^{2^{-(m+1)}} \xrightarrow[m \to \infty]{} 1$ from (6.3.15) that

$$\|u_{\varepsilon}\|_{L^{\infty}(\Omega\times(0,T_{max,\varepsilon}))} \leqslant \max\left\{1, \left(\left(4K_{S}|\Omega|^{-\frac{1}{2}}\right)^{\frac{1}{1-\frac{1}{s}}} \frac{2}{\eta}\left(1+\frac{\|\tilde{\mu}_{3}\|_{L^{\infty}(0,1)}}{\mu_{1}}\right)\right)^{\frac{1-\frac{2}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}}\right\}.$$

Obviously, we do not find parameters satisfying (6.3.17) for m tending to infinity. Nevertheless, for any K > 1 we find an m^* depending only on K s.t. if (6.3.17) is satisfied for m^* then, (6.3.16) holds.

In the following remark we give an exact formula for the Sobolev constant K_S from Lemma 2.2.8(i) that only depends on the domain Ω and the dimension n to get an impression of the upper bound of u_{ε} .

Remark 6.3.4. If Ω is convex the upper bound from *Corollary 6.3.3* can be given in terms of K and our parameters as due to *Lemma A.4.4* the Sobolev embedding constant $K_S(s)$ is given by

$$K_{S}(s) = \begin{cases} \max\left\{1, \frac{\operatorname{diam}(\Omega)|V|}{|\Omega|}^{\frac{1}{2}}\right\} & n = 1, \\ \sqrt{2} \max\left\{|\Omega|^{\frac{1}{s} - \frac{1}{2}}, \frac{\operatorname{diam}(\Omega)^{1 + \frac{s+2}{s}} \pi^{\frac{s+2}{2s}} \pi^{\frac{s+2}{2s}} \frac{\Gamma\left(\frac{s-2}{2s}\right)}{\Gamma\left(\frac{s+2}{2s}\right)}\right\} \sqrt{\frac{\Gamma\left(\frac{2}{s}\right)}{\Gamma\left(2^{s-1}\right)}} \left(\frac{\Gamma(2)}{\Gamma(1)}\right)^{\frac{s-2}{2s}} & n = 2, \quad (6.3.18) \\ \sqrt{2} \max\left\{|\Omega|^{-\frac{1}{n}}, \frac{\operatorname{diam}(\Omega)^{n} \pi^{\frac{n-1}{2}}}{n|\Omega|} \frac{\Gamma\left(\frac{1}{2}\right)}{\Gamma\left(\frac{n-1}{2}\right)}\right\} \sqrt{\frac{\Gamma\left(\frac{n-2}{2}\right)}{\Gamma\left(\frac{n+2}{2}\right)}} \left(\frac{\Gamma(n)}{\Gamma\left(\frac{n}{2}\right)}\right)^{\frac{1}{n}} & n \ge 3, \end{cases}$$

where $V := \bigcup_{x \in \Omega} \Omega_x$ and $\Omega_x := \{y - x : y \in \Omega\}$ for $x \in \Omega$, and Γ denotes the Gamma function given by $\Gamma(x) = \int_0^\infty t^{x-1} e^{-t} dt$ for x > 0.

Global existence of our solution follows from the last lemma.

Theorem 6.3.5. For
$$\varepsilon \in (0,1)$$
 there is a bounded global solution $(u_{\varepsilon}, w_{\varepsilon}, h_{\varepsilon})$ of (6.3.2) in $\left(C^{2+\vartheta, 1+\frac{\vartheta}{2}}\left(\overline{\Omega} \times [0,\infty)\right)\right)^3$ satisfying $0 \le u_{\varepsilon} \le C_7$ and $0 \le w_{\varepsilon}, h_{\varepsilon} \le 1$.

Proof. Let $\varepsilon \in (0,1)$. Lemmas 6.3.1 and 6.3.2 imply that $u_{\varepsilon}, w_{\varepsilon}, h_{\varepsilon} \in L^{\infty}(\Omega \times (0, T_{max,\varepsilon}))$. Assume $T_{max,\varepsilon} < \infty$. Then, $g(u_{\varepsilon}, w_{\varepsilon}) \in L^{\infty}(\Omega \times (0, T_{max}))$ follows from the Lipschitz continuity of g. Putting together Theorem A.1.1, the boundedness of h_{ε} , Theorem A.1.12 with

$$a(\nabla h_{\varepsilon}) := D_H \nabla h_{\varepsilon}, \ b(x, t, h_{\varepsilon}) := \lambda h_{\varepsilon} - g(u_{\varepsilon}, w_{\varepsilon})$$

we conclude that there is $\kappa_1 \in (0, 1)$ s.t.

$$\|h_{\varepsilon}\|_{C^{\kappa_1,\frac{\kappa_1}{2}}(\overline{\Omega}\times[0,T_{max,\varepsilon}])} \leqslant C_{11}.$$
(6.3.19)

Analogously, setting

$$a(\nabla w_{\varepsilon}) := \varepsilon \nabla w_{\varepsilon}, \ b(x, t, w_{\varepsilon}) := \mu_3(h_{\varepsilon})F(w_{\varepsilon}) + \mu_2(h_{\varepsilon})(w_{\varepsilon} - 1)u_{\varepsilon}$$

in *Theorem A.1.12* we conclude using the boundedness of our solution and the Lipschitz continuity of μ_2 and μ_3 on [0,1] that there is $\kappa_2 \in (0,1)$ s.t.

$$\|w_{\varepsilon}\|_{C^{\kappa_2,\frac{\kappa_2}{2}}(\overline{\Omega}\times[0,T_{max,\varepsilon}])} \leqslant C_{12}.$$
(6.3.20)

Moreover, due to the boundedness of our solution, the continuity of ψ , (6.2.3) and (6.2.5b) and setting

$$\begin{aligned} a(x,t,\nabla u_{\varepsilon}) &:= \psi(w_{\varepsilon},h_{\varepsilon})\nabla u_{\varepsilon},\\ b(x,t,u_{\varepsilon}) &:= \mu_{1}u_{\varepsilon}^{\alpha}\left(-1 + J_{1}(x,h_{\varepsilon}) * u_{\varepsilon}^{\beta} + J_{2}(x,h_{\varepsilon}) * w_{\varepsilon}^{\gamma}\right) - \tilde{\mu}_{3}(h_{\varepsilon})F(w_{\varepsilon}) \end{aligned}$$

in Theorem A.1.12 it follows analogously that there is $\kappa_3 \in (0, 1)$ s.t.

$$\|u_{\varepsilon}\|_{C^{\kappa_3,\frac{\kappa_3}{2}}(\overline{\Omega}\times[0,T_{max,\varepsilon}])} \leqslant C_{13}.$$
(6.3.21)

Finally, it follows as in Lemma 6.3.1 (applying Theorem A.1.6 twice to every function if necessary) that $u_{\varepsilon}, w_{\varepsilon}, h_{\varepsilon} \in C^{2+\vartheta, 1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0, T_{max,\varepsilon}])$. This contradicts (6.3.3). Hence, $T_{max,\varepsilon} = \infty$ follows.

We show the uniqueness of this solution as in *Chapter 5*, where we need to restrict p_1, p_2 in (6.2.5b) if $n \ge 3$.

Lemma 6.3.6. Assume that p_1, p_2 from (6.2.5b) satisfy $p_1, p_2 \ge \frac{2n}{n+2}$ if $n \ge 3$ and $p_1, p_2 \in (1, \infty)$ as before if n = 1, 2 then the classical solution from Theorem 6.3.5 is unique.

Proof. Let $\varepsilon \in (0,1)$ and $T \in (0,\infty)$. Assume that there are two solutions (u_1, w_1, h_1) and (u_2, w_2, h_2) in $\left(C^{2+\vartheta, 1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,\infty))\right)^3$ to (6.3.2). Then, we obtain after subtracting the equations for h_1 and h_2 from another that

$$(h_1 - h_2)_t = D_H \Delta(h_1 - h_2) + g(u_1, w_1) - g(u_2, w_2) - \lambda(h_1 - h_2)$$
holds in $\Omega \times (0,T)$. Now, we multiply the above equation with $h_1 - h_2$, integrate over Ω and obtain for $t \in (0,T)$ using partial integration, the Lipschitz continuity of g and Young's inequality that

$$\frac{1}{2} \frac{d}{dt} \int_{\Omega} |h_1 - h_2|^2 \, dx + \lambda \int_{\Omega} |h_1 - h_2|^2 \, dx + D_H \int_{\Omega} |\nabla(h_1 - h_2)|^2 \, dx$$

$$= \int_{\Omega} (g(u_1, w_1) - g(u_2, w_2))(h_1 - h_2) \, dx$$

$$\leq 2L_g \int_{\Omega} |u_1 - u_2| |h_1 - h_2| \, dx + 2L_g \int_{\Omega} |w_1 - w_2| |h_1 - h_2| \, dx$$

$$\leq C_{14} \left(\int_{\Omega} |u_1 - u_2|^2 \, dx + \int_{\Omega} |w_1 - w_2|^2 \, dx \right) + \lambda \int_{\Omega} |h_1 - h_2|^2 \, dx.$$
(6.3.22)

Subtracting the equations for w_1 and w_2 from another we obtain the equation

$$\begin{split} (w_1 - w_2)_t = &\varepsilon \Delta(w_1 - w_2) + \mu_2(h_1)(1 - w_1)u_1 - \mu_2(h_2)(1 - w_2)u_2 \\ &+ \mu_3(h_2)F(w_2) - \mu_3(h_1)F(w_1) \\ = &\varepsilon \Delta(w_1 - w_2) + \mu_2(h_1)(1 - w_1)(u_1 - u_2) + \mu_2(h_1)(w_2 - w_1)u_2 \\ &+ (\mu_2(h_1) - \mu_2(h_2))(1 - w_2)u_2 + \mu_3(h_2)(w_2 - w_1)\tilde{F}(w_1)\tilde{F}(w_2) \\ &+ (\mu_3(h_2) - \mu_3(h_1))F(w_1) \end{split}$$

that holds in $\Omega \times (0,T)$ and conclude analogously to above using the Lipschitz continuity of μ_2 and μ_3 and the boundedness of the solutions that for $t \in (0,T)$ it holds that

$$\frac{1}{2} \frac{d}{dt} \int_{\Omega} |w_{1} - w_{2}|^{2} dx + \varepsilon \int_{\Omega} |\nabla(w_{1} - w_{2})|^{2} dx$$

$$\leq ||\mu_{2}||_{L^{\infty}(0,1)} \int_{\Omega} |u_{1} - u_{2}||w_{1} - w_{2}| dx + (||\mu_{2}||_{L^{\infty}(0,1)}C_{7} + ||\mu_{3}||_{L^{\infty}(0,1)}) \int_{\Omega} |w_{1} - w_{2}|^{2} dx$$

$$+ (||\mu_{2}'||_{L^{\infty}(0,1)}C_{7} + ||\mu_{3}'||_{L^{\infty}(0,1)}) \int_{\Omega} |h_{1} - h_{2}||w_{1} - w_{2}| dx$$

$$\leq C_{15} \left(\int_{\Omega} |u_{1} - u_{2}|^{2} dx + \int_{\Omega} |w_{1} - w_{2}|^{2} dx + \int_{\Omega} |h_{1} - h_{2}|^{2} dx \right).$$
(6.3.23)

Further, we obtain by subtracting the equations for u_1 and u_2 from another that

$$\begin{aligned} (u_{1} - u_{2})_{t} = \nabla \cdot (\psi(w_{1}, h_{1}) \nabla u_{1} - \psi(w_{2}, h_{2}) \nabla u_{2}) + \mu_{1} \left(u_{1}^{\alpha} - u_{2}^{\alpha} + u_{2}^{\alpha} J_{1}(x, h_{2}) * u_{2}^{\beta} \\ &- u_{1}^{\alpha} J_{1}(x, h_{1}) * u_{1}^{\beta} + u_{2}^{\alpha} J_{2}(x, h_{2}) * w_{2}^{\gamma} - u_{1}^{\alpha} J_{2}(x, h_{1}) * w_{1}^{\gamma} \right) \\ &+ \tilde{\mu}_{3}(h_{1}) F(w_{1}) - \tilde{\mu}_{3}(h_{2}) F(w_{2}) \\ = \nabla \cdot (\psi(w_{1}, h_{1}) \nabla (u_{1} - u_{2}) + (\psi(w_{1}, h_{1}) - \psi(w_{2}, h_{2})) \nabla u_{2}) + \mu_{1}(u_{1}^{\alpha} - u_{2}^{\alpha}) \\ &+ \mu_{1} \left(u_{2}^{\alpha} J_{1}(x, h_{2}) * u_{2}^{\beta} - u_{1}^{\alpha} J_{1}(x, h_{1}) * u_{1}^{\beta} + u_{2}^{\alpha} J_{2}(x, h_{2}) * w_{2}^{\gamma} - u_{1}^{\alpha} J_{2}(x, h_{1}) * w_{1}^{\gamma} \right) \\ &+ (\tilde{\mu}_{3}(h_{1}) - \tilde{\mu}_{3}(h_{2})) F(w_{1}) + \tilde{\mu}_{3}(h_{2})(w_{1} - w_{2}) \tilde{F}(w_{1}) \tilde{F}(w_{2}) \end{aligned}$$

$$(6.3.24)$$

holds in $\Omega \times (0, T)$. Using (6.2.5a), (6.2.5b), the boundedness of u_1, u_2 , the mean value theorem and Hölder's inequality and the Sobolev embedding from Lemma 2.2.8(i) we can estimate on $\Omega \times (0, T)$ that

$$\begin{aligned} & \left| J_1(x,h_2) * u_2^{\beta} - J_1(x,h_1) * u_1^{\beta} \right| \\ \leq & \left| (J_1(x,h_2) - J_1(x,h_1)) * u_2^{\beta} \right| + \left| J_1(x,h_1) * (u_1^{\beta} - u_2^{\beta}) \right| \end{aligned}$$

$$\leq C_{7}^{\beta} \int_{\Omega} L_{J_{1}}(x-y) |h_{1}(y) - h_{2}(y)| \, \mathrm{d}y + \beta C_{7}^{\beta-1} \int_{\Omega} J_{1}(x-y,h_{1}(y)) |u_{1}(y) - u_{2}(y)| \, \mathrm{d}y \\ \leq C_{7}^{\beta} \|L_{J_{1}}\|_{L^{p_{1}}(B)} \|h_{1} - h_{2}\|_{L^{\frac{p_{1}}{p_{1}-1}}(\Omega)} + \beta C_{7}^{\beta-1} (\|L_{J_{1}}\|_{L^{p_{1}}(B)} + \|J_{1}(\cdot,0)\|_{L^{p_{1}}(B)}) \|u_{1} - u_{2}\|_{L^{\frac{p_{1}}{p_{1}-1}}(\Omega)} \\ \leq C_{16} \left(\|h_{1} - h_{2}\|_{H^{1}(\Omega)} + \|u_{1} - u_{2}\|_{H^{1}(\Omega)} \right).$$

From this we obtain using (6.2.5a), (6.2.5b), the boundedness of u_1, u_2 , the mean value theorem, Lemma 2.3.2(i) and Hölder's and Young's inequality that

$$\begin{split} &\int_{\Omega} \left| u_{2}^{\alpha} J_{1}(x,h_{2}) \ast u_{2}^{\beta} - u_{1}^{\alpha} J_{1}(x,h_{1}) \ast u_{1}^{\beta} \right| |u_{1} - u_{2}| \, \mathrm{d}x \\ \leqslant \alpha C_{7}^{\alpha-1} \| J_{1}(\cdot,h_{2}) \ast u_{2}^{\beta} \|_{L^{\infty}(\Omega \times (0,T))} \| u_{1} - u_{2} \|_{L^{2}(\Omega)}^{2} \\ &+ C_{16} C_{7}^{\alpha} \left(\| h_{1} - h_{2} \|_{H^{1}(\Omega)} + \| u_{1} - u_{2} \|_{H^{1}(\Omega)} \right) \| u_{1} - u_{2} \|_{L^{1}(\Omega)}^{2} \\ \leqslant C_{17} \left(D_{H}^{-1}, \delta^{-1} \right) \left(\| h_{1} - h_{2} \|_{L^{2}(\Omega)}^{2} + \| u_{1} - u_{2} \|_{L^{2}(\Omega)}^{2} \right) + \frac{D_{H}}{2} \| \nabla (h_{1} - h_{2}) \|_{(L^{2}(\Omega))^{n}}^{2} \\ &+ \frac{\delta}{2} \| \nabla (u_{1} - u_{2}) \|_{(L^{2}(\Omega))^{n}}^{2}. \end{split}$$

$$(6.3.25)$$

Analogously, we obtain

$$\int_{\Omega} |u_{2}^{\alpha} J_{2}(x,h_{2}) * w_{2}^{\gamma} - u_{1}^{\alpha} J_{2}(x,h_{1}) * w_{1}^{\gamma}| |u_{1} - u_{2}| dx$$

$$\leq C_{18} \left(D_{H}^{-1}, \varepsilon^{-1} \right) \left(\|u_{1} - u_{2}\|_{L^{2}(\Omega)}^{2} + \|h_{1} - h_{2}\|_{L^{2}(\Omega)}^{2} + \|w_{1} - w_{2}\|_{L^{2}(\Omega)}^{2} \right)$$

$$+ \frac{D_{H}}{2} \|\nabla(h_{1} - h_{2})\|_{(L^{2}(\Omega))^{n}}^{2} + \varepsilon \|\nabla(w_{1} - w_{2})\|_{(L^{2}(\Omega))^{n}}^{2}.$$
(6.3.26)

Multiplying (6.3.24) by $u_1 - u_2$, integrating over Ω , using partial integration, (6.2.3), the Lipschitz continuity of ψ and $\tilde{\mu}_3$, the boundedness of the solutions and of ∇u_2 on $\Omega \times (0,T)$, the mean value theorem, (6.3.25), (6.3.26) and Young's inequality we conclude that

$$\begin{aligned} &\frac{1}{2} \frac{d}{dt} \int_{\Omega} |u_{1} - u_{2}|^{2} dx + \delta \int_{\Omega} |\nabla(u_{1} - u_{2})|^{2} dx \\ \leqslant \|\nabla u_{2}\|_{L^{\infty}(0,T;(L^{\infty}(\Omega))^{n})} \int_{\Omega} \left(\|\partial_{w}\psi\|_{L^{\infty}((0,1)^{2})} |w_{1} - w_{2}| + \|\partial_{h}\psi\|_{L^{\infty}((0,1)^{2})} |h_{1} - h_{2}| \right) |\nabla(u_{1} - u_{2})| dx \\ &+ \mu_{1} \alpha C_{7}^{\alpha - 1} \int_{\Omega} |u_{1} - u_{2}|^{2} dx + \mu_{1} \int_{\Omega} |u_{2}^{\alpha} J_{1}(x, h_{2}) * u_{2}^{\beta} - u_{1}^{\alpha} J_{1}(x, h_{1}) * u_{1}^{\beta}| |u_{1} - u_{2}| dx \\ &+ \mu_{1} \int_{\Omega} |u_{2}^{\alpha} J_{2}(x, h_{2}) * w_{2}^{\gamma} - u_{1}^{\alpha} J_{2}(x, h_{1}) * w_{1}^{\gamma}| |u_{1} - u_{2}| dx \\ &+ L_{\tilde{\mu}_{3}} \int_{\Omega} |h_{1} - h_{2}| |u_{1} - u_{2}| dx + \|\tilde{\mu}_{3}\|_{L^{\infty}(0,1)} \int_{\Omega} |w_{1} - w_{2}| |u_{1} - u_{2}| dx \\ \leqslant C_{19}(T) \left(\|u_{1} - u_{2}\|_{L^{2}(\Omega)}^{2} + \|h_{1} - h_{2}\|_{L^{2}(\Omega)}^{2} + \|w_{1} - w_{2}\|_{L^{2}(\Omega)}^{2} \right) \\ &+ \delta \|\nabla(u_{1} - u_{2})\|_{(L^{2}(\Omega))^{n}}^{2} + D_{H} \|\nabla(h_{1} - h_{2})\|_{L^{2}(\Omega)}^{2} + \varepsilon \|\nabla(w_{1} - w_{2})\|_{(L^{2}(\Omega))^{n}}^{2}. \end{aligned}$$

$$(6.3.27)$$

Adding up (6.3.22), (6.3.23) and (6.3.27) we conclude that for $t \in (0, T)$ it holds that

$$\frac{1}{2}\frac{d}{dt}\left(\|u_1 - u_2\|_{L^2(\Omega)}^2 + \|h_1 - h_2\|_{L^2(\Omega)}^2 + \|w_1 - w_2\|_{L^2(\Omega)}^2\right)$$

$$\leq (C_{14} + C_{15} + C_{19})\left(\|u_1 - u_2\|_{L^2(\Omega)}^2 + \|h_1 - h_2\|_{L^2(\Omega)}^2 + \|w_1 - w_2\|_{L^2(\Omega)}^2\right)$$

Finally, we obtain $u_1 \equiv u_2$, $h_1 \equiv h_2$ and $w_1 \equiv w_2$ on $\overline{\Omega} \times [0, T]$ from Gronwall's inequality. As this holds for all $T \in (0, \infty)$ uniqueness of the classical solution to (6.3.2) follows.

6.4 Existence of a weak solution to the original problem

Definition 6.4.1. By a weak solution to (6.2.1) we mean a tuple (u, w, h) of nonnegative bounded functions s.t. for all $T \in (0, \infty)$ it holds that $u \in L^2(0, T; H^1(\Omega)) \cap C([0, T]; L^2(\Omega))$, $w \in L^{\infty}(0, T; H^1(\Omega)) \cap C([0, T]; L^2(\Omega))$ with $\partial_t w \in L^2(\Omega \times (0, T))$ and $h \in W_2^{2,1}(\Omega \times (0, T)) \cap$ $C([0, T]; H^1(\Omega))$ and they satisfy

$$-\int_{0}^{T}\int_{\Omega}u\partial_{t}\eta\,dx\,dt - \int_{\Omega}u_{0}\eta(\cdot,0)\,dx$$

= $-\int_{0}^{T}\int_{\Omega}\psi(w,h)\nabla u\cdot\nabla\eta\,dx\,dt + \mu_{1}\int_{0}^{T}\int_{\Omega}u^{\alpha}\left(1 - J_{1}(x,h)*u^{\beta} - J_{2}(x,h)*w^{\gamma}\right)\eta\,dx\,dt$
+ $\int_{0}^{T}\int_{\Omega}\tilde{\mu}_{3}(h)F(w)\eta\,dx\,dt,$ (6.4.1)

for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$ and

$$w_t = \mu_2(h)(1-w)u - \mu_3(h)F(w) \ a.e. \ in \ \Omega \times (0,T), \tag{6.4.2}$$

$$h_t = D_H \Delta h + g(u, w) - \lambda h \qquad a.e. \text{ in } \Omega \times (0, T),$$

$$\partial_{\nu} h = 0 \qquad a.e. \text{ on } \partial\Omega \times (0, T),$$

$$w(\cdot, 0) = w_0, h(\cdot, 0) = h_0 \qquad a.e. \text{ in } \Omega.$$

$$(6.4.3)$$

Lemma 6.4.2. There are u, w, h with bounds $0 \le u \le C_7$ and $0 \le w, h \le 1$ on $\overline{\Omega} \times [0, \infty)$ s.t. for all $T \in (0,T)$ it holds that $u \in L^2(0,T; H^1(\Omega)), w \in L^\infty(0,T; H^1(\Omega)) \cap C([0,T]; L^2(\Omega))$ and $h \in W_2^{2,1}(\Omega \times (0,T))$ and for a subsequence

$$u_{\varepsilon_k} \underset{k \to \infty}{\longrightarrow} u \qquad \qquad in \ L^2(0,T; H^1(\Omega)), \tag{6.4.4a}$$

$$u_{\varepsilon_k} \xrightarrow[k \to \infty]{} u$$
 in $L^2(\Omega \times (0,T))$ and a.e. in $\Omega \times (0,T)$, (6.4.4b)

$$w_{\varepsilon_k} \underset{k \to \infty}{\to} w \qquad in \ C([0,T]; L^2(\Omega)) \ and \ a.e. \ in \ \Omega \times (0,T),, \qquad (6.4.4c)$$

$$w_{\varepsilon_k} \stackrel{*}{\underset{k \to \infty}{\longrightarrow}} \nabla w \qquad \qquad in \ L^{\infty}(0, T; (L^2(\Omega))^n), \tag{6.4.4d}$$

$$\partial_t w_{\varepsilon_k} \underset{k \to \infty}{\rightharpoonup} \partial_t w \qquad \qquad in \ L^2(\Omega \times (0,T)), \tag{6.4.4e}$$

$$h_{\varepsilon_k} \underset{k \to \infty}{\rightharpoonup} h \qquad \qquad in \ L^2(0,T; H^2(\Omega)), \tag{6.4.4f}$$

$$h_{\varepsilon_k} \xrightarrow[k \to \infty]{} h$$
 in $L^2(0,T; H^1(\Omega))$ and a.e. in $\Omega \times (0,T)$, (6.4.4g)

Proof. Let T > 0. First, *Theorem A.1.8* implies that

 ∇

$$\|h_{\varepsilon}\|_{W_{2}^{2,1}(\Omega\times(0,T))} \leqslant C_{20}\left(\|g(u_{\varepsilon},w_{\varepsilon})\|_{L^{2}(\Omega\times(0,T))} + \|h_{0\varepsilon}\|_{H^{1}(\Omega)}\right).$$
(6.4.5)

Due to the Lipschitz continuity of g, the uniform boundedness of (u_{ε}) and (w_{ε}) and the convergence in (6.3.1c) the right-hand side of (6.4.5) is uniformly bounded for $\varepsilon \in (0, 1)$. Hence, Lions-Aubin (with $H^2(\Omega) \hookrightarrow H^1(\Omega) \hookrightarrow L^2(\Omega)$) and Banach-Alaoglu imply the existence of $h \in W_2^{2,1}(\Omega \times (0,T))$ and a subsequence s.t. (6.4.4f) - (6.4.4h) hold and from Lemma 6.3.1 it follows for a.e. $(x,t) \in \Omega \times (0,T)$ that $0 \leq h(x,t) = \lim_{k \to \infty} h_{\varepsilon_k}(x,t) \leq 1$.

Next, we want to obtain a uniform estimate on the norm of ∇u_{ε} . Therefore, we multiply the equation for u_{ε} in (6.3.2) by u_{ε} , integrate over Ω and use partial integration to obtain

$$\begin{split} \frac{1}{2} \frac{d}{dt} \int_{\Omega} u_{\varepsilon}^{2} \, \mathrm{d}x &= -\int_{\Omega} \psi(w_{\varepsilon}, h_{\varepsilon}) |\nabla u_{\varepsilon}|^{2} \, \mathrm{d}x + \mu_{1} \int_{\Omega} u_{\varepsilon}^{\alpha+1} \left(1 - J_{1}(x, h_{\varepsilon}) * u_{\varepsilon}^{\beta} - J_{2}(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma} \right) \, \mathrm{d}x \\ &+ \int_{\Omega} \tilde{\mu}_{3}(h_{\varepsilon}) F(w_{\varepsilon}) u_{\varepsilon} \, \mathrm{d}x. \end{split}$$

Then, we can estimate using (6.2.3), Lemma 2.3.2(i), the uniform boundedness of (u_{ε}) , (w_{ε}) , (h_{ε}) , $\alpha \ge 1$, the continuity of $\tilde{\mu}_3$ that

$$\begin{split} &\frac{1}{2}\frac{d}{dt}\int_{\Omega}u_{\varepsilon}^{2}\,\mathrm{d}x+\delta\int_{\Omega}|\nabla u_{\varepsilon}|^{2}\,\mathrm{d}x\\ \leqslant &\mu_{1}C_{7}^{\alpha-1}\left(1+C_{7}^{\beta}(\|L_{J_{1}}\|_{L^{1}(B)}+\|J_{1}(\cdot,0)\|_{L^{1}(B)})+\|\mathbf{L}_{J_{2}}\|_{L^{1}(B)}+\|J_{2}(\cdot,0)\|_{L^{1}(B)}\right)\int_{\Omega}u_{\varepsilon}^{2}\,\mathrm{d}x\\ &+\|\tilde{\mu}_{3}\|_{L^{\infty}(0,1)}C_{7}|\Omega| \end{split}$$

Hence, we conclude from Gronwall's inequality that for all $\varepsilon \in (0, 1)$ it holds that

$$\|\nabla u_{\varepsilon}\|_{L^{2}(0,T;(L^{2}(\Omega))^{n})} \leq C_{21}(T).$$
(6.4.6)

Further, we multiply the equation for u_{ε} in (6.3.2) by $\varphi \in H_0^1(\Omega)$ and obtain obtain using partial integration, the Hölder inequality, the continuity of ψ , the uniform boundedness of (u_{ε}) , (w_{ε}) and (h_{ε}) , Lemma 2.3.2(i) and the Lipschitz continuity of $\tilde{\mu}_3$ that

$$\begin{split} \left| \int_{\Omega} \partial_t u_{\varepsilon} \varphi \, \mathrm{d}x \right| &\leq \int_{\Omega} \left| \psi(w_{\varepsilon}, h_{\varepsilon}) \nabla u_{\varepsilon} \cdot \nabla \varphi \right| \mathrm{d}x + \mu_1 \int_{\Omega} \left| u_{\varepsilon}^{\alpha} \left(1 - J_1(x, h_{\varepsilon}) * u_{\varepsilon}^{\beta} - J_2(x, h_{\varepsilon}) * w_{\varepsilon}^{\gamma} \right) \varphi \right| \mathrm{d}x \\ &+ \int_{\Omega} \left| \tilde{\mu}_3(h_{\varepsilon}) F(w_{\varepsilon}) \varphi \right| \mathrm{d}x \\ &\leq \| \psi \|_{L^{\infty}((0,1)^2)} \| \nabla u_{\varepsilon} \|_{(L^2(\Omega))^n} \| \nabla \varphi \|_{(L^2(\Omega))^n} \\ &+ \left(\mu_1 C_7^{\alpha} \left(1 + C_7^{\beta} (\| L_{J_1} \|_{L^1(B)} + \| J_1(\cdot, 0) \|_{L^1(B)} \right) + \| L_{J_2} \|_{L^1(B)} + \| J_2(\cdot, 0) \|_{L^1(B)} \right) \\ &+ \| \tilde{\mu}_3 \|_{L^{\infty}(0,1)} \right) \| \varphi \|_{L^1(\Omega)} \\ &\leq C_{22} (\| \nabla u_{\varepsilon} \|_{(L^2(\Omega))^n} + 1) \| \varphi \|_{H^1(\Omega)}. \end{split}$$

Hence,

$$\|\partial_t u_{\varepsilon}\|_{H^{-1}(\Omega)} \leq C_{22}(\|\nabla u_{\varepsilon}\|_{(L^2(\Omega))^n} + 1)$$

and we conclude from (6.4.6) that

$$\|\partial_t u_{\varepsilon}\|_{L^2(0,T;H^{-1}(\Omega))} \leqslant C_{23}(T).$$
 (6.4.7)

Combining (6.4.6) and (6.4.7) with the uniform boundedness of (u_{ε}) we conclude from Lions-Aubin (Lemma A.3.9) with $H^1(\Omega) \hookrightarrow L^2(\Omega) \hookrightarrow H^{-1}(\Omega)$ and Banach-Alaoglu that there is $u \in L^2(0,T; H^1(\Omega))$ and a subsequence s.t. (6.4.4a) and (6.4.4b) hold. Moreover, due to the pointwise convergence it holds a.e. in $\Omega \times (0,T)$ that $0 \leq u(x,t) \leq \lim_{k \to \infty} u_{\varepsilon_k}(x,t) \leq C_7$.

To obtain a uniform estimate on the norm of ∇w_{ε} we multiply the equation for w_{ε} by Δw_{ε} and obtain after integration over Ω and partial integration due to our boundary condition on w_{ε} that

$$\frac{1}{2}\frac{d}{dt}\int_{\Omega}|\nabla w_{\varepsilon}|^{2}\,\mathrm{d}x = \int_{\Omega}(\nabla w_{\varepsilon})_{t}\cdot\nabla w_{\varepsilon}\,\mathrm{d}x = -\int_{\Omega}(w_{\varepsilon})_{t}\Delta w_{\varepsilon}\,\mathrm{d}x$$

$$\begin{split} &= -\varepsilon \int_{\Omega} |\Delta w_{\varepsilon}|^{2} \,\mathrm{d}x - \int_{\Omega} \mu_{2}(h_{\varepsilon})(1-w_{\varepsilon})u_{\varepsilon}\Delta w_{\varepsilon} \,\mathrm{d}x + \int_{\Omega} \mu_{3}(h_{\varepsilon})F(w_{\varepsilon})\Delta w_{\varepsilon} \,\mathrm{d}x \\ &= -\varepsilon \int_{\Omega} |\Delta w_{\varepsilon}|^{2} \,\mathrm{d}x + \int_{\Omega} \left(\mu_{2}'(h_{\varepsilon})(1-w_{\varepsilon})\nabla h_{\varepsilon}u_{\varepsilon} - \mu_{2}(h_{\varepsilon})\nabla w_{\varepsilon}u_{\varepsilon} + \mu_{2}(h_{\varepsilon})(1-w_{\varepsilon})\nabla u_{\varepsilon}\right) \cdot \nabla w_{\varepsilon} \,\mathrm{d}x \\ &- \int_{\Omega} \left(\mu_{3}'(h_{\varepsilon})\nabla h_{\varepsilon}F(w_{\varepsilon}) + \mu_{3}(h_{\varepsilon})(\tilde{F}(w_{\varepsilon}))^{2}\nabla w_{\varepsilon}\right) \cdot \nabla w_{\varepsilon} \,\mathrm{d}x. \end{split}$$

Hence, using the uniform boundedness of $(u_{\varepsilon}), (h_{\varepsilon}), (w_{\varepsilon})$, continuity of μ'_2, μ'_3 and Young's inequality we obtain

$$\begin{split} &\frac{1}{2}\frac{d}{dt}\int_{\Omega}|\nabla w_{\varepsilon}|^{2}\,\mathrm{d}x+\varepsilon\int_{\Omega}|\Delta w_{\varepsilon}|^{2}\,\mathrm{d}x\\ \leqslant (\|\mu_{2}'\|_{L^{\infty}(0,1)}C_{7}+\|\mu_{3}'\|_{L^{\infty}(0,1)})\int_{\Omega}|\nabla h_{\varepsilon}||\nabla w_{\varepsilon}|\,\mathrm{d}x+\|\mu_{2}\|_{L^{\infty}(0,1)}\int_{\Omega}|\nabla u_{\varepsilon}||\nabla w_{\varepsilon}|\,\mathrm{d}x\\ &+\left(\|\mu_{2}\|_{L^{\infty}(0,1)}C_{7}+\|\mu_{3}\|_{L^{\infty}(0,1)}\right)\int_{\Omega}|\nabla w_{\varepsilon}|^{2}\,\mathrm{d}x\\ \leqslant &C_{24}\left(\int_{\Omega}|\nabla u_{\varepsilon}|^{2}\,\mathrm{d}x+\int_{\Omega}|\nabla w_{\varepsilon}|^{2}\,\mathrm{d}x+\int_{\Omega}|\nabla h_{\varepsilon}|^{2}\,\mathrm{d}x\right) \end{split}$$

Hence, Gronwall's inequality, the fact that $(\nabla w_{0\varepsilon})$ is uniformly bounded in $(L^2(\Omega))^n$ due to the convergence in (6.3.1c), (6.4.5) and (6.4.6) and imply that

$$\int_{\Omega} |\nabla w_{\varepsilon}(t)|^2 \, \mathrm{d}x + \varepsilon \int_0^T \int_{\Omega} |\Delta w_{\varepsilon}|^2 \, \mathrm{d}x \, \mathrm{d}t$$
$$\leqslant C_{25}(T) \left(\|\nabla w_{0\varepsilon}\|_{(L^2(\Omega))^n} + \int_0^T \int_{\Omega} |\nabla u_{\varepsilon}|^2 \, \mathrm{d}x \, \mathrm{d}t + \int_0^T \int_{\Omega} |\nabla h_{\varepsilon}|^2 \, \mathrm{d}x \, \mathrm{d}t \right) \leqslant C_{26}(T)$$

for $t \in (0,T)$ and $\varepsilon \in (0,1)$. Consequently, for all $\varepsilon \in (0,1)$ it holds that

$$\|\nabla w_{\varepsilon}\|_{L^{\infty}(0,T;(L^{2}(\Omega))^{n})} \leqslant C_{26}(T), \qquad (6.4.8)$$

$$\varepsilon \|\Delta w_{\varepsilon}\|_{L^{2}(\Omega \times (0,T))} \leqslant C_{27}(T).$$
(6.4.9)

To obtain a uniform estimate on some norm of the time derivative of w_{ε} , we multiply the equation for w_{ε} from (6.3.2) by $\varphi \in L^2(\Omega)$, integrate over Ω , use the Lipschitz continuity of μ_2 and μ_3 , the uniform boundedness of (u_{ε}) , (w_{ε}) and (h_{ε}) and Hölder's inequality to conclude that

$$\begin{aligned} \left| \int_{\Omega} \partial_t w_{\varepsilon} \varphi \, \mathrm{d}x \right| &\leq \varepsilon \int_{\Omega} \left| \Delta w_{\varepsilon} \right| |\varphi| \, \mathrm{d}x + \int_{\Omega} \left| \mu_2(h_{\varepsilon})(1 - w_{\varepsilon})u_{\varepsilon} \varphi \right| \, \mathrm{d}x + \int_{\Omega} \left| \mu_3(h_{\varepsilon})F(w_{\varepsilon})\varphi \right| \, \mathrm{d}x \\ &\leq \left(\varepsilon \| \Delta w_{\varepsilon} \|_{L^2(\Omega)} + \left(\| \mu_2 \|_{L^{\infty}(0,1)}C_7 + \| \mu_3 \|_{L^{\infty}(0,1)} \right) |\Omega|^{\frac{1}{2}} \right) \| \varphi \|_{L^2(\Omega)}. \end{aligned}$$

Consequently, we conclude from (6.4.9) as $(L^2(\Omega))^* = L^2(\Omega)$ that for all $\varepsilon \in (0,1)$ it holds that

$$\|\partial_t w_{\varepsilon}\|_{L^2(\Omega \times (0,T))} \leqslant C_{28}(T).$$
(6.4.10)

Combining the uniform boundedness of (w_{ε}) with (6.4.8) and (6.4.10) we conclude from Lions-Aubin (Lemma A.3.9) with $H^1(\Omega) \hookrightarrow L^2(\Omega) \hookrightarrow L^2(\Omega)$ and Lemma A.3.1 in the space $L^{\infty}(0,T; (L^2(\Omega))^n) = (L^1(0,T; (L^2(\Omega))^n))^*$ that there is $w \in C([0,T]; L^2(\Omega)) \cap L^{\infty}(0,T; H^1(\Omega))$ and a subsequence s.t. (6.4.4c) - (6.4.4e) hold. From the pointwise a.e. convergence we conclude that $0 \leq w(x,t) = \lim_{k \to \infty} w_{\varepsilon_k}(x,t) \leq 1$ holds a.e. in $\Omega \times (0,T)$.

Due to the pointwise convergence shown in the proof of the last lemma the following corollary is a direct consequence of *Corollary 6.3.3*.

Corollary 6.4.3. For K > 1 and 'small' enough parameters it holds that

$$\|u\|_{L^{\infty}(\Omega\times(0,\infty))} \leqslant K \max\left\{1, \left(\left(4K_{S}|\Omega|^{-\frac{1}{2}}\right)^{\frac{1}{1-\frac{1}{s}}} \frac{2}{\eta}\left(1+\frac{\|\tilde{\mu}_{3}\|_{L^{\infty}(0,1)}}{\mu_{1}}\right)\right)^{\frac{1-\frac{2}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}}\right\}$$

Theorem 6.4.4. There is a bounded nonnegative weak solution (u, w, h) to (6.2.1) in the sense of Definition 6.4.1 satisfying $u \leq C_7$ and $w, h \leq 1$ a.e. in $\Omega \times (0, \infty)$.

Proof. Let T > 0 and $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$. We consider the subsequence (ε_k) from Lemma 6.4.2. Multiplying the equations from (6.3.2) by η , integrating over $\Omega \times (0,T)$ and using partial integration we obtain the weak formulation

$$\begin{split} &-\int_0^T \int_\Omega u_{\varepsilon_k} \partial_t \eta \, \mathrm{d}x \, \mathrm{d}t - \int_\Omega u_{0\varepsilon_k} \eta(\cdot, 0) \, \mathrm{d}x \\ &= -\int_0^T \int_\Omega \psi(w_{\varepsilon_k}, h_{\varepsilon_k}) \nabla u_{\varepsilon_k} \cdot \nabla \eta \, \mathrm{d}x \, \mathrm{d}t \\ &+ \mu_1 \int_0^T \int_\Omega u_{\varepsilon_k}^\alpha \left(1 - J_1(x, h_{\varepsilon_k}) * u_{\varepsilon_k}^\beta - J_2(x, h_{\varepsilon_k}) * w_{\varepsilon_k}^\gamma \right) \eta \, \mathrm{d}x \, \mathrm{d}t \\ &+ \int_0^T \int_\Omega \tilde{\mu}_3(h_{\varepsilon_k}) F(w_{\varepsilon_k}) \eta \, \mathrm{d}x \, \mathrm{d}t, \end{split}$$

$$\begin{split} \int_0^T \int_\Omega \partial_t w_{\varepsilon_k} \eta \, \mathrm{d}x \, \mathrm{d}t &= -\varepsilon_k \int_0^T \int_\Omega \nabla w_{\varepsilon_k} \cdot \nabla \eta \, \mathrm{d}x \, \mathrm{d}t \\ &+ \int_0^T \int_\Omega \left(\mu_2(h_{\varepsilon_k})(1 - w_{\varepsilon_k}) u_{\varepsilon_k} - \mu_3(h_{\varepsilon_k}) F(w_{\varepsilon_k}) \right) \eta \, \mathrm{d}x \, \mathrm{d}t \end{split}$$

and

$$\int_0^T \int_\Omega \partial_t h_{\varepsilon_k} \eta \, \mathrm{d}x \, \mathrm{d}t = D_H \int_0^T \int_\Omega \Delta h_{\varepsilon_k} \eta \, \mathrm{d}x \, \mathrm{d}t + \int_0^T \int_\Omega (g(u_{\varepsilon_k}, w_{\varepsilon_k}) - \lambda h_{\varepsilon_k}) \eta \, \mathrm{d}x \, \mathrm{d}t.$$

From the continuity of ψ and $\tilde{\mu}_3$, (6.4.4c), (6.4.4g) and the dominated convergence theorem we conclude that

$$\psi(w_{\varepsilon_k}, h_{\varepsilon_k}) \nabla \eta \xrightarrow[k \to \infty]{} \psi(w, h) \nabla \eta \text{ in } L^2(0, T; (L^2(\Omega))^n),$$

$$\tilde{\mu}_3(h_{\varepsilon_k}) F(w_{\varepsilon_k}) \xrightarrow[k \to \infty]{} \tilde{\mu}_3(h) F(w) \text{ in } L^2(\Omega \times (0, T)).$$
(6.4.11)

Hence,

$$\int_{0}^{T} \int_{\Omega} \psi(w_{\varepsilon_{k}}, h_{\varepsilon_{k}}) \nabla u_{\varepsilon_{k}} \cdot \nabla \eta \, \mathrm{d}x \, \mathrm{d}t \xrightarrow[k \to \infty]{} \int_{0}^{T} \int_{\Omega} \psi(w, h) \nabla u \cdot \nabla \eta \, \mathrm{d}x \, \mathrm{d}t \tag{6.4.12}$$

follows from (6.4.4a) and compensated compactness (*Lemma A.3.2*). Further, due to (6.2.5a) and the uniform boundedness of (u_{ε_k}) we estimate for $(x,t) \in \Omega \times (0,T)$ that

$$\begin{split} & \left| J_1(x,h_{\varepsilon_k}) * u_{\varepsilon_k}^{\beta}(t) - J_1(x,h) * u^{\beta}(t) \right| \\ \leqslant & \int_{\Omega} \left| J_1(x-y,h_{\varepsilon_k}(y,t)) - J_1(x-y,h(y,t)) | u_{\varepsilon_k}^{\beta}(y,t) \, \mathrm{d}y \right| \\ & + \int_{\Omega} J_1(x-y,h(y,t)) | u_{\varepsilon_k}^{\beta}(y,t) - u^{\beta}(y,t) | \, \mathrm{d}y \end{split}$$

$$\leq C_7^{\beta} \int_{\Omega} L_{J_1}(x-y) |h_{\varepsilon_k}(y,t) - h(y,t)| \,\mathrm{d}y$$

$$+ \int_{\Omega} (L_{J_1}(x-y)h(y,t) + J_1(x-y,0)) |u_{\varepsilon_k}^{\beta}(y,t) - u^{\beta}(y,t)| \,\mathrm{d}y \underset{k \to \infty}{\to} 0$$

due to the dominated convergence theorem combined with (6.4.4b), (6.4.4g), and (6.2.5b) and the uniform boundedness of (u_{ε_k}) and (h_{ε_k}) . Consequently,

$$J_1(\cdot, h_{\varepsilon_k}) * u^{\beta}_{\varepsilon_k} \xrightarrow[k \to \infty]{} J_1(\cdot, h) * u^{\beta} \text{ pointwise a.e. in } \Omega \times (0, T)$$
(6.4.13)

and analogously,

$$J_2(\cdot, h_{\varepsilon_k}) * w^{\gamma}_{\varepsilon_k} \xrightarrow[k \to \infty]{} J_2(\cdot, h) * w^{\gamma} \text{ pointwise a.e. in } \Omega \times (0, T)$$
(6.4.14)

follows. Hence, we conclude combining (6.4.4b), (6.4.13), and (6.4.14) that

$$\begin{aligned} u_{\varepsilon_{k}}^{\alpha}\left(1-J_{1}(\cdot,h_{\varepsilon_{k}})\ast u_{\varepsilon_{k}}^{\beta}-J_{2}(\cdot,h_{\varepsilon_{k}})\ast w_{\varepsilon_{k}}^{\gamma}\right) &\xrightarrow[k\to\infty]{} u^{\alpha}\left(1-J_{1}(\cdot,h)\ast u^{\beta}-J_{2}(\cdot,h)\ast w^{\gamma}\right) \\ \text{pointwise a.e. in } \Omega\times(0,T) \end{aligned}$$

and from the uniform boundedness of (u_{ε_k}) , (w_{ε_k}) and (h_{ε_k}) and Lemma 2.3.2(i) that for all $k \in \mathbb{N}$ it holds that

$$\left|u_{\varepsilon_{k}}^{\alpha}\left(1-J_{1}(x,h_{\varepsilon_{k}})\ast u_{\varepsilon_{k}}^{\beta}(t)-J_{2}(x,h_{\varepsilon_{k}})\ast w_{\varepsilon_{k}}^{\gamma}(t)\right)\right|\leqslant C_{29}.$$

Consequently, the dominated convergence theorem implies that

$$u_{\varepsilon_{k}}^{\alpha} \left(1 - J_{1}(\cdot, h_{\varepsilon_{k}}) * u_{\varepsilon_{k}}^{\beta} - J_{2}(\cdot, h_{\varepsilon_{k}}) * w_{\varepsilon_{k}}^{\gamma} \right) \xrightarrow[k \to \infty]{} u^{\alpha} \left(1 - J_{1}(\cdot, h) * u^{\beta} - J_{2}(\cdot, h) * w^{\gamma} \right)$$

in $L^{2}(\Omega \times (0, T)).$

Combining this with (6.3.1b), (6.4.4b), (6.4.11) and (6.4.12) we conclude that u satisfies (6.4.1), i.e., solves the corresponding equations of (6.2.1) in the sense of *Theorem A.1.1*. Hence, also $u \in C([0,T]; L^2(\Omega))$ holds due to *Theorem A.1.1*.

With the help of Hölder's inequality and (6.4.8) we obtain

$$\varepsilon_{k} \left| \int_{0}^{T} \int_{\Omega} \nabla w_{\varepsilon_{k}} \cdot \nabla \eta \, \mathrm{d}x \, \mathrm{d}t \right| \leq \varepsilon_{k} \| \nabla w_{\varepsilon_{k}} \|_{L^{2}(0,T;(L^{2}(\Omega))^{n})} \| \nabla \eta \|_{L^{2}(0,T;(L^{2}(\Omega))^{n})}$$
$$\leq \varepsilon_{k} C_{26} T^{\frac{1}{2}} \| \nabla \eta \|_{L^{2}(0,T;(L^{2}(\Omega))^{n})} \xrightarrow[k \to \infty]{} 0. \tag{6.4.15}$$

Further, we conclude from the continuity of μ_2 and μ_3 , the pointwise convergences in (6.4.4b), (6.4.4c) and (6.4.4g), the uniform boundedness of (u_{ε_k}) , (w_{ε_k}) and (h_{ε_k}) and the dominated convergence theorem that

$$\mu_2(h_{\varepsilon_k})(1-w_{\varepsilon_k})u_{\varepsilon_k} \xrightarrow[k\to\infty]{} \mu_2(h)(1-w)u \text{ in } L^2(\Omega \times (0,T)), \tag{6.4.16}$$

$$\mu_3(h_{\varepsilon_k})F(w_{\varepsilon_k}) \xrightarrow[k \to \infty]{} \mu_3(h)F(w) \text{ in } L^2(\Omega \times (0,T)).$$
(6.4.17)

Combining (6.3.1c) and (6.4.4c) with (6.4.15) - (6.4.17) we conclude that

$$\int_0^T \int_\Omega \partial_t w \eta \, \mathrm{d}x \, \mathrm{d}t = \int_0^T \left(\mu_2(h)(1-w)u - \mu_3(h)F(w) \right) \eta \, \mathrm{d}x \, \mathrm{d}t.$$

Due to $C_c^{\infty}(\Omega \times (0,T)) \subset W_2^{1,1}(\Omega \times (0,T))$ the fundamental lemma of calculus of variations implies that w satisfies (6.4.2). Moreover, using partial integration we conclude due to (6.3.1c), (6.4.4c), and (6.4.4e) that (especially also for $\eta \in C_c^{\infty}(\overline{\Omega} \times [0,T)) \subset W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$) it holds that

$$-\int_{0}^{T} \int_{\Omega} w\eta_{t} \, \mathrm{d}x \, \mathrm{d}t - \int_{\Omega} w(\cdot, 0)\eta(\cdot, 0) \, \mathrm{d}x \, \mathrm{d}t = \int_{0}^{T} \int_{\Omega} \partial_{t} w\eta \, \mathrm{d}x \, \mathrm{d}t$$
$$= \lim_{k \to \infty} \int_{0}^{T} \int_{\Omega} \partial_{t} w_{\varepsilon_{k}} \eta \, \mathrm{d}x \, \mathrm{d}t$$
$$= \lim_{k \to \infty} \left(-\int_{0}^{T} \int_{\Omega} w_{\varepsilon_{k}} \eta_{t} \, \mathrm{d}x \, \mathrm{d}t - \int_{\Omega} w_{0\varepsilon_{k}}(\cdot)\eta(\cdot, 0) \, \mathrm{d}x \, \mathrm{d}t \right)$$
$$= -\int_{0}^{T} \int_{\Omega} w\eta_{t} \, \mathrm{d}x \, \mathrm{d}t - \int_{\Omega} w_{0}(\cdot)\eta(\cdot, 0) \, \mathrm{d}x \, \mathrm{d}t$$

Hence, we conclude again from the fundamental lemma of calculus of variations that indeed $w(\cdot, 0) = w_0$ a.e. in Ω .

Furthermore, we estimate with the help of Hölder's inequality and the Lipschitz continuity of g that

$$\begin{split} &\int_0^T \int_\Omega |g(u_{\varepsilon_k}, w_{\varepsilon_k}) - g(u, w)| |\eta| \, \mathrm{d}x \, \mathrm{d}t \\ \leqslant & 2L_g \left(\|u_{\varepsilon_k} - u\|_{L^2(\Omega \times (0,T))} + \|w_{\varepsilon_k} - w\|_{L^2(\Omega \times (0,T))} \right) \|\eta\|_{L^2(\Omega \times (0,T))} \xrightarrow[k \to \infty]{} 0 \end{split}$$

due to (6.4.4b) and (6.4.4c). Hence, it follows from (6.4.4f) - (6.4.4h) that

$$\int_0^T \int_\Omega \partial_t h\eta \, \mathrm{d}x \, \mathrm{d}t = D_H \int_0^T \int_\Omega \Delta h\eta \, \mathrm{d}x \, \mathrm{d}t + \int_0^T \int_\Omega (g(u, w) - \lambda h)\eta \, \mathrm{d}x \, \mathrm{d}t$$

for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$. We conclude again from the fundamental lemma of calculus of variations that h satisfies (6.4.3) a.e. in $\Omega \times (0,T)$. Finally, $h(\cdot,0) = h_0$ a.e. in Ω follows as for w. We conclude similary that

$$\int_0^T \int_{\partial\Omega} \nabla h \cdot \nu \eta \, \mathrm{d}\sigma(x) \, \mathrm{d}t = 0$$

which due to $H^1(\Omega) \hookrightarrow L^2(\partial\Omega)$ gives us $\nabla h \cdot \nu = 0$ a.e. on $\partial\Omega \times (0,T)$.

6.5 1D Simulations

In this section we simulate the behavior of solutions to (6.2.1) in one dimension. Thereby, we decompose the domain $\Omega = [-5, 5]$ into an equidistant mesh x_2, \ldots, x_{N-1} with step size dx = 0.05 and the time interval [0, 50] with step size dt = 0.0001. For a simulation of the no-flux boundary condition we add points $x_1 < x_2$ and $x_N > x_{N-1}$ outside of Ω and assume equality of the solutions on the neighboring points. As in *Chapter 5*, we use the method from [111] to discretize the nonlocal integral terms via a composite trapezoidal rule. Moreover, as in [156], we recompute the convolution matrices only every 40 time steps to improve the runtime. Thereby, we assume that changes in the values of the convolution matrices phi_mat_1 and phi_mat_2 (due to changes of h) are negligible within this time interval. Namely, for $i, j \in \{2, \ldots, N-1\}$ the corresponding entry of the kth-convolution matrix is

$$(phi_mat_1)_{ij}^k = J_1(x_i - x_j, h_j^{40k})$$

and with the help of this we compute the n + 1st convolution term at $x_i, i \in \{2, \ldots, N-1\}$ as:

$$(conv_1)_i^{n+1} = dx \sum_{j=3}^{N-2} (phi_mat_1)_{ij}^{[n/40]} (u_j^n)^{\beta} + \frac{dx}{2} \left((phi_mat_1)_{i2}^{[n/40]} (u_2^n)^{\beta} + (phi_mat_1)_{i(N-1)}^{[n/40]} (u_{(N-1)}^n)^{\beta} \right).$$

Analogously we compute phi_mat_2 and $conv_2$. For the discretization of the diffusion term we use finite differences and an upwind scheme. The initial conditions are depicted in *Figure 6.1* and are given by

$$u_{0}(x) = \begin{cases} 0.3e^{-\frac{1}{5}(x+5)^{2}}, & x \in [-5,0], \\ 0.3e^{-5}\left(1-\frac{x}{5}\right), & x \in (0,5], \end{cases}$$
$$w_{0}(x) = \begin{cases} 0.7e^{-(x+5)^{2}}, & x \in [-5,0], \\ 0.7e^{-25}\left(1-\frac{x}{5}\right), & x \in (0,5] \end{cases},$$
$$h_{0}(x) = \begin{cases} \frac{1}{5}(0.3e^{-5}-0.05)x+0.3e^{-5}, & x \in [-5,0] \\ 0.3e^{-5}\left(1-\frac{x}{5}\right), & x \in (0,5]. \end{cases}$$



Figure 6.1: Initial conditions u_0, w_0, h_0 .

We choose the functions $\psi(h, w) = 0.5$ (for simplicity), $\mu_2(h) = h$ (meaning that the net 'deactivation' of *u*-cells is directly proportional to the amount of protons available in the microtumor space), $\mu_3(h) = \tilde{\mu}_3(h) = \frac{h}{1+h}$ (there is no loss of *w*-cells when becoming *u*-cells, the transition primarily to motility- is favored by acidity, but in a limited manner, quickly reaching saturation), $F(w) = \frac{w}{1+w}$, $g(u,w) = \frac{u+w}{1+u+w}$ (both phenotypes are producing acid, also in a limited way), and the constants $D_H = 0.1$ and $\lambda = 1$.

First, we took $\beta = \gamma = \mu_1 = 1$ and explored the influence of the kernels on the minimal value α^* of α for which the solution ceases to exist globally in time (with accuracy to one decimal place). Thereby, we considered as in *Chapter 5* the logistic kernel $J_L(x) = \frac{1}{2+e^x+e^{-x}}$, the uniform kernel $J_U(x) = \chi_{[-1,1]}(x)$ and, moreover, the *h*-dependent kernels

$$J_1(x,h) = \frac{1}{\sqrt{2\pi}} e^{-\frac{x^2}{2}} \left(\frac{h}{1+h} + \frac{1}{10}\right),$$
(6.5.1)



Figure 6.2: Simulation results of model (6.2.1) with $J_1 = J_2 = J_L$, i.e., logistic kernels, $\beta = \gamma = \mu_1 = 1$, $\alpha = 2, 5, 6.1, 6.2$ (columns from left to right, respectively). Component u of the solution starts to become unbounded near $\alpha = 6.2$. In the rightmost column a blow-up occurs in the next time step.

the first of which is a h-dependent shift of a Gaussian, while the latter is a Holling III-type function of h suggesting a slower increase towards saturation, with a certain 'learning effect' as far as the response to more acidity is concerned: as J_2 stands for the interaction of the two cell phenotypes, it accounts for both of them extruding protons, along with the corresponding adaptation of u-cells to interspecific cues.

The first columns of Figures 6.2 and 6.3 show the solution for the critical α from (6.2.2), when J_1 and J_2 are both logistic or uniform, respectively. The solution u aggregates at the position of the initial accumulation of the active cells at the left boundary. In the case of two logistic kernels a stronger aggregation for increasing α values can be observed leading to a blow-up at the left boundary near $\alpha^* = 6.2$. On the other hand, in the case of two uniform kernels u invades the whole domain and aggregates at the right boundary, leading to a blow-up there for approximately $\alpha^* = 14.7$. This invasive behavior can also be observed for all combinations of kernels and parameters $\alpha, \beta, \gamma, \mu$ as long as no blow-up at the left boundary occurs. An overview of the minimal values α^* depending on the kernels can be found in *Table 6.1*.

In further tests we investigated for logistic kernels the influence of β, γ and the growth rate μ_1 on the blow-up behavior. Higher values of β lead to an increase of the minimal value α^* where blow-up occurs. In the case $\beta = 10$ and $\gamma = \mu_1 = 1$ we observed that for $\alpha = 26.9, 27.1, 27.3$ the solution ceases to exist, whereas it exists globally in time for $\alpha = 27, 27.2, 27.4$. Hence, in contrast to *Chapter 5* and [99] we cannot determine a value α^* s.t. for $\alpha < \alpha^*$ the solution is global, whereas it blows-up for $\alpha \ge \alpha^*$. It seems that for $\alpha \ge \alpha^{**} = 27.5$ blow-up occurs but



Figure 6.3: Simulation results of model (6.2.1) with $J_1 = J_2 = J_U$, i.e., uniform kernels, $\beta = \gamma = \mu_1 = 1$, $\alpha = 2, 6.2, 14.6, 14.7$ (columns from left to right, respectively). Component u of the solution starts to become unbounded near $\alpha = 14.7$. In the rightmost column a blow-up occurs in the next time step.

	α^*	Figure
J_1, J_2 logistic	6.2	Figure 6.2
J_1 logistic, J_2 uniform	7.4	
J_1 uniform, J_2 logistic	10.3	
J_1, J_2 uniform	14.7	Figure 6.3
J_1, J_2 from (6.5.1),(6.5.2)	4.1	Figure 6.6

Table 6.1: Minimal value α^* for which the solution ceases to exist for $\beta = \gamma = \mu_1 = 1$ depending on the kernels J_1 and J_2 .

we cannot assure this. In contrast, higher values of γ and/or μ_1 lead to a blow-up for lower α 's, see *Table 6.2* for an overview of the concrete values of α^* along with the respective parameter combinations.

Parameters	α^*
$\beta = 10, \gamma = \mu_1 = 1$	26.9
$\beta = \gamma = 10, \mu_1 = 1$	22.1
$\beta = 10, \gamma = 0.1, \mu_1 = 1$	33.4
$\beta = 1, \gamma = 10, \mu_1 = 1$	4.6
$\beta = 1, \gamma = 0.1, \mu_1 = 1$	25.3
$\beta = \gamma = 1, \mu_1 = 10$	3.6

Table 6.2: Minimal value α^* for which the solution ceases to exist, depending on parameters β, γ, μ_1 . Both convolution kernels are logistic: $J_1 = J_2 = J_L$.



Figure 6.4: Simulations of models (6.2.1) and (6.5.3) with $\alpha = 2$, $\mu_1 = 1$, and (from left to right) $\beta = 1, 10, 100, 1000$ and $\gamma = 1000$ in column 1, $\gamma = 1$ in columns 2-4. First row: $J_1 = J_2 = J_U$; 2nd row: model without w, with $J = J_U$; 3rd row: $J_1 = J_2 = J_L$; 4th row: model without w, with $J = J_L$

Moreover, as in *Chapter 5*, increasing values of β and μ_1 in *Figure 6.4* lead to patterns depending on the kernels J_1 and J_2 . A high value of γ does not seem to lead by itself to patterns, but further experiments suggest that γ influences the height of the peaks, thus leading to less pronounced u-patterns. This is due to the stronger dampening of proliferation, which hinders stronger aggregates. To illustrate the effect of γ we plot in *Figure 6.4* two situations with very different values ($\gamma = 1000$ in the first column and $\gamma = 1$ in the remaining columns).

The performed simulations are very similar to those of the reduced model

$$\begin{cases} \partial_t u = \nabla \cdot (\psi(h)\nabla u) + \mu_1 u^{\alpha} \left(1 - J(x,h) * u^{\beta} \right) & \text{in } \Omega \times (0,\infty), \\ \partial_t h = D_H \Delta h + g(u) - \lambda h & \text{in } \Omega \times (0,\infty) \end{cases}$$
(6.5.3)

without inactive cells w (compare rows 1 and 2 and rows 3 and 4 in Figure 6.4, respectively, for uniform or logistic kernels). System (6.5.3) is a simplification of the model considered in *Chapter 5* without myopic diffusion and taxis, where Turing-like patterns for large values of $\beta \mu_1$ were induced by the nonlocal term. This also seems to be the case here in model (6.2.1). However, the calculation of a strictly positive steady-state (u^*, w^*, h^*) already leads to analytical problems, since this requires even in the corresponding local model without diffusion and for $\mu_2, \mu_3, \tilde{\mu}_3$ independent from h a solution to the nonlinear system

$$0 = \mu_1 (u^*)^{\alpha} \left(1 - (u^*)^{\beta} - (w^*)^{\gamma} \right) + \tilde{\mu}_3 F(w^*),$$

$$0 = \mu_2 (1 - w^*) u^* - \mu_3 F(w^*).$$
(6.5.4)

Comparing the height of the peaks in columns 3 and 4 of *Figure 6.4* for uniform kernels (i.e., first two rows therein) clearly shows the dampening effect of interspecific interactions. Moreover, the solution of (6.5.3) reaches its maximum accumulation at the left boundary faster than the solution of (6.2.1), which is again due to the supplementary interspecific dampening in the latter model. *Figure 6.5* shows that in model (6.5.3) a blow-up already occurs for relatively smaller values of α .



Figure 6.5: Simulations of model (6.5.3) without w with $\beta = \mu_1 = 1$. Columns 1 and 2: J logistic and $\alpha = 4.4, 4.5$. Columns 3 and 4: J uniform and $\alpha = 5.7, 5.8$. In the 2nd and 4th column a blow-up occurs in the next time step..

For the *h*-dependent kernels J_1 and J_2 from (6.5.1) and (6.5.2) the solution *u* in Figure 6.6 rapidly accumulates at the left boundary and then invades the whole domain aggregating much less at the boundaries than in Figures 6.2 and 6.3. As mentioned in Table 6.1 the blow-up already occurs for $\alpha^* = 4.1$. The 4th column in Figure 6.6 shows one example of pattern formation for a certain choice of parameters.



Figure 6.6: Simulations of model (6.2.1) with *h*-dependent kernels J_1 and J_2 from (6.5.1) and (6.5.2) for $\beta = \gamma = \mu_1 = 1$, $\alpha = 2, 4, 4.1$. In the 3rd column a blow-up occurs in the next time step. The 4th column shows patterns for $\alpha = 2$, $\beta = 200$, $\gamma = 10$, $\mu_1 = 100$.

6.6 Discussion

As mentioned in Section 6.2, the model introduced here extends previous settings [99] and, in a certain sense, Chapter 5 and [136]. As in Chapter 5 and [99], the main mathematical challenge comes from the interaction strengths $\alpha, \beta \ge 1$ present in the nonlocal terms; interspecific interactions did not add further difficulties as far as global existence and boundedness are concerned. In contrast to Chapter 5 we do not have here any myopic diffusion, nor taxis terms, which saves us the efforts otherwise needed to estimate first derivatives of the tactic signal. The model with interspecific interactions from [136] involves haptotaxis, but there $\alpha = \beta = \gamma = 1$ and the lack of transitions from one population to another, along with the assumptions made on initial data, convolution kernels, and coefficient functions render the analysis therein more accessible.

The missing diffusion of w-cells required the construction of the approximate problem in Section 6.3. Introducing the term -wu in the dynamics of w-cells helped ensure in that problem the boundedness of w_{ε} , with the aid of a comparison principle. Such term does have a biological motivation as well: it describes competition between active and inactive cells, which in our model is also triggered by the acidity profile, as both tumor cell phenotypes extrude protons in the interstitial space (the active ones more than their quiescent counterparts).

As in *Chapter 5* and [99], the condition (6.2.2) is not sharp: the numerical simulations suggest that the solution also exists globally for certain pairs (α, β) which do not satisfy that requirement. Interestingly, the critical value α^* for which a solution ceases to exist does not seem to be an absolute α -minimum, but can jump to higher or lower values, depending on the particular combination of the other parameters β, γ, μ_1 (even for the same choice of kernels), as seen in *Table 6.2.* Indeed, there seems to be an $\alpha^{**} > \alpha^*$ such that the solution blows up in finite time for $\alpha \ge \alpha^{**}$, but stays global for certain values $\alpha \in (\alpha^*, \alpha^{**})$. A thorough mathematical investigation thereof remains, however, open. As *Table 6.1* shows, α^* also depends on the choice of convolution kernels (for fixed parameter values); this greatly complicates the analysis of blow-up behavior, due to the unlimited degrees of freedom one has for such choices, notwithstanding conditions (6.2.5).

A rigorous mathematical stability and pattern analysis for this kind of PDE-ODE-PDE models seems to be out of reach with the established approaches (see, e.g., [4, 107, 114, 135]), mainly due to the nonlinearities featuring weak Allee and overcrowding/competition effects with the respective interaction strengths α, β, γ , which preclude from identifying nontrivial steady-states even in the absence of diffusion; the phenotypic switch terms only add difficulty to such attempts. The numerical simulations performed in *Section 6.5* give some insight into the long term and patterning behavior of solutions, suggesting that the solution seems to be able to approach in the long term some stable state and to exhibit patterns, depending (as in *Chapter 5*) on the choice of kernels and the parameter combination. The model extension with *w*-cells and interspecific interactions does not change substantially the type and shape of obtained oscillatory patterns, but does have an influence on the peaks of *u*-cell aggregates. The noticed dampening effect also contributes to detering solution blow-up or at least ensuring global boundedness for substantially larger α values which, again, depend on the choice of the convolution kernels and of the interaction strength γ .

Our analysis explicitly required the diffusion coefficient $\psi(w, h)$ to be nondegenerate. Alleviating this assumption leads to further mathematical challenges, when trying to obtain (as usually in such proofs) a bound on ∇u from the ODE for w. Indeed, the problem thereby relies on u being involved in the growth, instead of the decay term. On the other hand, considering such nonlinear diffusion is motivated from a biological viewpoint, in order to account e.g., for chemokinesis [63, 76, 122].

CHAPTER 7

Summary and outlook

To conclude this work, we summarize what was considered, which methods were used, and name possible continuations of this work. We looked at four models involving reaction-diffusion-advection equations with spatial nonlocality, three of them also involving couplings with an ODE and/or a PDE. The considered models describe migration of cells in different biological contexts, in bounded domains. Thereby, we showed the global existence of a weak or very weak solution for each of the considered models. Moreover, we proved the boundedness of the solutions obtained in *Chapters 5* and *6*. In *Chapters 3*, *5*, and *6* numerical simulations were performed. *Chapter 5* also contains an analysis of the long-time behavior of the solution and pattern formation to explain the oscillations in the simulations. *Chapters 3*, *5*, and *6* were already discussed in *Sections 3.6*, *5.7*, and *6.6*, respectively. Therefore, we will only deal with them briefly here and refer to the just mentioned sections for a more detailed discussion of our considerations and related literature. The corresponding assessment of related literature for the model in *Chapter 4* can be found in *Section 4.1*.

In Chapter 3 we analyzed the PDE-PDE-/PDE-ODE-system from (3.4.1) consisting of a reactiondiffusion-advection equation with nonlinear diffusion and nonlocal advection term modeling the development of the cell density and a PDE or an ODE for a diffusible or nondiffusible signal, respectively. Thereby, we combined a prototypical cell-cell and cell-matrix adhesion model with adhesion operator \mathcal{A}_r and a general form of the nonlocal chemotaxis model with nonlocal gradient $\mathring{\nabla}_r$ in the unified framework (3.4.1). This nonlocal model was related to the local unified hapto-/ chemotaxis model (3.4.2) in the sense that our adhesion model was the nonlocal version of the haptotaxis model with nonlinear diffusion and our nonlocal chemotaxis-growth model was the nonlocal version of the local chemotaxis-growth model. We established the connection of these frameworks by demonstrating that the weak solution of the nonlocal model converges in L^2 to the weak solution of the corresponding local model.

The proof relied on the functionwise convergence of the integral operators \mathcal{T}_r and \mathcal{S}_r to the identity operator for diminishing sensing radius r (which did not hold for the original nonlocal operators) and their self-adjoint-like property shown in *Lemmas 3.2.5* and *3.2.7*. These operators were applied to the functions' gradient, and were reformulations of the adhesion and nonlocal gradient operators (on the subdomain Ω_r with distance r to the boundary), respectively. Furthermore, numerical simulations depicted in *Figure 3.1* indicated that in a minimalist model, the difference between the solutions of the models involving \mathcal{A}_r and \mathcal{T}_r is negligible when the cells start at the center of the domain. However, the solutions differed when the cells started near the

boundary, as seen in Figure 3.2.

We showed the existence of a weak-strong solution to the approximate problems (3.4.10) in the case of a diffusible signal involving the functions G_{ε} to assure the nonnegativity of the solution with a Leray-Schauder fixed-point argument and monotone operator theory and concluded the existence of a weak-strong solution to the nonlocal problem (3.4.1) in the sense of *Definition 3.4.6* for a diffusing and nondiffusing signal with the help of several approximations. Further, we established uniform in r estimates on these solutions and proved their convergence to the solution of the local model in the sense of *Definition 3.4.6* for r = 0 using the properties of the reformulated operators indicated above. This convergence could also be seen in *Figures 3.3* and *3.5* for an appropriate choice of parameters that guaranteed well-posedness. Finite-time blow-up occurred in the local model (cf. *Figure 3.4*), whereas the nonlocal model exhibited pattern for large t.

A possible extension of our model would be to include a similar nonlocality in the diffusion term of the adhesion model (see Section 3.6).

In Chapter 4 we considered the reaction-diffusion-advection equation from (4.1.1) that combined degenerate myopic diffusion with self-adhesion and a generalized logistic-type growth term. This extended the model derived in [156] by the term $\mu c(1 - c^{r-1})$ which allowed us to establish the uniform bound in L^r from Lemma 4.5.1. We showed the existence of a global very weak solution in the sense of Definition 4.3.3. The very weak formulation was obtained from (4.1.1) through two partial integrations that shifted all spatial derivatives to the test functions. There, the boundary integral from the first partial integration vanished due to our no-flux boundary condition, but to eliminate the other boundary integral we imposed on our test functions that their derivative in direction $\mathbb{D}\nu$ is zero on the boundary. The density of such functions in H^1 was shown in Lemma 4.6.2, which led to Theorem 4.6.1, where we checked that our very weak formulation is a classical solution to (4.1.1). To show the existence of an 'only' weak solution we lack a uniform bound on ∇c on the whole domain Ω , whose proof seems unlikely, due to the combination of myopic diffusion and degeneracy.

Our equation included the standard adhesion operator also considered in *Chapter 3* from *Definition 4.2.1* into the advection term. In contrast to the approach there, we did not use its reformulation from *Lemma 3.2.1* shifting the application of the nonlocal operator from the function itself to its gradient. Instead, we have rewritten the adhesion operator as in [156] to a convolution with the bounded gradient of an interaction potential, which (thanks to our assumptions on F) illustrated that our operator maps functions from L^1 to bounded functions, a fact that was not used in *Chapter 3*. For this reason, it was sufficient that the approximating sequence of classical solutions (c_{ε}) only converged in L^1 in order to conclude the convergence of the nonlocal term $c_{\varepsilon}Ac_{\varepsilon}$. Moreover, the adhesion operator preserved Hölder-continuity, which was necessary for the existence proof of a classical solution.

The assumptions of a positive distance to the boundary of the domain and some sufficiently low dimension of the degeneracy set of \mathbb{D} and the boundedness of its divergence from (4.3.2c) were necessary for the construction of an approximating sequence of smooth and non-degenerating diffusion tensors \mathbb{D}_{ε} in *Subsection 4.4.1* with the standard approach from Theorem 3 in Section 5.3.3 from [58]. Using this method, the diffusion tensors were uniformly bounded on Ω . Moreover, also their divergences were uniformly bounded on sets compactly contained in $\{\mathbb{D} > 0\}$, which together with the already mentioned uniform boundedness in L^r was used to show the uniform bounds established in Lemma 4.5.2. In contrast to [77] we did not require the convergence of its divergence or additional restrictions on \mathbb{D}_{ε} .

The existence of global classical solutions c_{ε} to the approximate problems involving the diffusion tensors \mathbb{D}_{ε} was shown with a standard fixed-point argument. Thereby, the difficulty in proving their convergence to the desired weak solution lay in the convergence of the term c_{ε}^{r} shown in *Lemma 4.5.4*. This was necessary as we could not apply the de la Vallée-Poussin theorem on the whole domain to obtain L^{1} -convergence, but only on the already mentioned sets compactly contained in the complement of the degeneracy set. Therefore, we introduced the upper box dimension to quantify the required sufficiently low dimensionality of the degeneracy set of \mathbb{D} and constructed a sequence (φ_{δ}) of smooth functions with diminishing support satisfying properties (4.2.5a)-(4.2.5g) and equal one in neighborhoods of { $\mathbb{D} \ge 0$ } that diminish in δ . Using them, we showed the required convergence after splitting c_{ε}^{r} in a function with support in a set with a positive distance to the degeneracy set and a function whose support contained the degeneracy set. Only there the assumptions on r from (4.3.1) and $n \ge 3$ were required to ensure the positivity of the term on the right-hand side of condition (4.2.4).

The next step could be an analysis of the original model from [156]. Our solution approach does not work there, because without the growth term we lack the uniform bound on the approximate solutions in L^r . Moreover, the solution there could be measure-valued rather than a function, as we can only guarantee mass preservation. In [128] local well-posedness was established for an equation of this form coupled with a nonlinear integral equation, where the myopic diffusion was replaced by a quasilinear degenerated diffusion. The model there was also derived in [156] by additionally taking into account the cadherin binding dynamics of a pair of cells. In addition, the numerical simulations are still missing. Thereby, the difficulty lies in the degeneracy of the diffusion.

Chapter 5 dealt with the PDE-PDE-system coupling an reaction-diffusion-advection equation with myopic diffusion, repellent pH-taxis, and a nonlocal intraspecific interaction for the tumor cell density with a reaction-diffusion equation for the acid concentration from (5.3.1). There, the nonlocality was of form $J * u^{\beta}$. Our model extended [99] by replacing the Fickian diffusion with myopic and additionally considers the effects of a soluble signal via repellent pH-taxis. The formal modeling started from a meso-macro-system describing the mesoscopic tumor cell dynamics in response to acidity in the extracellular space. It consisted of a kinetic transport equation in the framework from [116] for the mesoscopic description of cell migration and intraspecific interaction and a macroscopic PDE describing the proton concentration. We deduced the macroscopic equation for the cell population dynamics (5.2.21) by a diffusion-dominated upscaling of the mesoscopic description. Due to the nonlinear source term, we could not apply the method from [25] for a rigorous derivation here. For α and β satisfying (5.3.3), the global existence of a bounded solution was followed with a fixed-point argument and estimates from [99]. Under additional assumptions on the norm of the kernel and the parameters that required especially some 'smallness' of $\beta^2 \|\mu\|_{L^{\infty}}$ as in [99], the tumor cell density approached on a longterm basis either some upper bound or zero, whereas the acid concentration approached some value depending on the concrete form of g from estimate (5.4.1). Thereby, the proof relied on the handling of the myopic diffusion as in [91] and the treatment of nonlocality from [99]. A 1D pattern analysis suggested that the occurrence of Turing-like, Hopf, or wave instabilities is due to the nonlocality and not due to the diffusion and depends on the concrete choice of parameters, especially on the product $\beta\mu$, and the Fourier transform of the kernel. This matched the observations in other non-local models [64, 74, 99, 119, 131, 142, 153] and was confirmed by numerical simulations depicted in *Figures 5.2* and *5.3*.

Further numerical simulations (see Figure 5.1) indicated that the solution exists globally for combinations of larger α compared to β not satisfying (5.3.3). Since the estimates from [99] were restricted to this case, a new approach is required to handle the nonlocal term for such α and β in the proof of global existence. Moreover, the simulations in Figure 5.1 suggested that, as in [99], the maximum value of such α also depends on the kernel. In this context, a further step in the analysis of this model would be the determination of the α about which a blow-up occurs depending on β and the kernel.

The existence proof of a weak solution for a degenerating diffusion tensor \mathbb{D} would require uniform estimates of appropriate norms of the global classical solutions from *Theorem 5.3.6* (and its gradients) that are independent from the lower bounds of the approximating diffusion tensors. This can get very difficult even in 1D (e.g., [149, 151]). Our estimates from the proof of (5.3.4) no longer work here as we lack a uniform lower bound of the diffusion tensors in contrast to *Theorem 5.3.6* and, consequently, the uniform upper bound of the approximate solutions from *Lemma 5.3.4*, which was used in particular for large α for the uniform estimate of the source term, is lost. Similarly, we would need bounds independent from the divergence of the proton concentration, in order to show the existence of a weak solution in the model with haptotaxis instead of chemotaxis. Our estimates cannot be used for this case either, because the uniform bound on the gradients of the acid concentrations from *Lemma 5.3.3* depends on D_H^{-1} .

In contrast to this, in the model presented in *Chapter 6*, we were able to use such estimates to obtain weak solutions in the PDE-ODE-PDE-system (6.2.1) describing the interactions of an active and a quiescent cell population in an acidic environment from classical solutions of the approximating PDE-PDE-PDE-system (6.3.2). Compared to *Chapter 5*, the model took into account a second inactive cell population w whose development was described by an ODE. Moreover, the reaction-diffusion equation for the active cells u did not contain a taxis term or myopic diffusion. Instead, we added a second nonlocality describing interspecific interactions between u and w to the nonlocal reaction term in the equation of u from the last chapter and a growth term of the form $\tilde{\mu}_3(h)F(w)$ that depended on w and the acid concentration h. As the model did not contain a taxis term, we did not require a uniform bound on ∇w_{ε} independent of the diffusion coefficient ε there to conclude the existence of a global uniform bound of the approximate solutions u_{ε} .

We included the term 1 - w into the *u*-dependent growth term in the equation for the quiescent cells *w* in order to ensure its boundedness with the help of a comparison principle in the approximate problem. Otherwise, this boundedness could not even be shown for bounded functions *u*. Further, the boundedness and consequent global existence of the approximate solutions u_{ε} followed similarly to *Chapter 5* using the estimates from [99] to handle the nonlocal intraspecific term. As in *Chapter 5*, the solution was also global for pairs of α and β that did not fulfill condition (6.2.2). In our method, the nonlocal interspecific term did not contribute to the proof of global existence for further pairs of α and β . However, a comparison of the simulations of this model from *Figures 6.2* and 6.3 with those of model (6.5.3) without *w* from *Figure 6.5* suggested that this term damped the blow-up behavior of the solution u. Also, the choice of the kernels, the exponents β and γ and the growth rate μ_1 influenced the minimum value of α^* for which a blow-up occurred (see *Tables 6.1* and *6.2*). Additionally, for certain combinations of parameters, the solution also existed globally for some $\alpha > \alpha^*$. In these cases there seemed to be an $\alpha^{**} > \alpha^*$, s.t. the solution blew-up in finite time for $\alpha \ge \alpha^{**}$, while there were α in (α^*, α^{**}) , where the solution was global. A thorough mathematical investigation of this remains an open problem.

In Figure 6.4 increasing values of β led to oscillations in the case of uniform kernels, which was not the case for increasing γ . Also, for large enough parameters, oscillations occurred for the *h*-dependent kernels from (6.5.1) and (6.5.2) as seen in Figure 6.6. Hence, we assume that as in *Chapter 5*, the formation of Turing-like patterns depends on the kernel and the product $\beta\mu_1$. We lack a pattern analysis here, as the computation of a positive steady state of the corresponding local model without diffusion led to the system of equations (6.5.4), which can only be solved numerically.

One extension of our analysis would be, again, to consider a degenerate diffusion coefficient $\psi(h, w)$. Thereby, the problem is that it is not possible to obtain a bound on ∇u from the equation of w, as is usually the case, since u is contained in the growth and not in the decay term.

Appendix A Additional Theorems

We summarize some results on different types of differential equations, fixed-point and convergence theorems and further results from functional analysis used in this work in Appendices A.1– A.3. In Appendix A.4 we state some results, especially from [99], used in *Part II* and slightly adopted to our needs. Finally, we prove some results on Hölder continuous functions and convolutions from *Chapter 2* in Appendix A.5. If not stated otherwise $\Omega \subset \mathbb{R}^n$, $n \in \mathbb{N}$, denotes a bounded domain with smooth enough boundary throughout this chapter.

A.1 Differential Equations

A.1.1 Linear PDEs

Theorem A.1.1. ([94, Theorem III.5.1]) Let T > 0 and consider the IBVP

$$\begin{cases} u_t - \mathcal{M}u = -f & \text{in } \Omega \times (0, T), \\ \sum_{i=1}^n \left(\sum_{j=1}^n a_{ij} u_{x_j} + a_i u \right) \nu_i = 0 & \text{on } \partial\Omega \times (0, T), \\ u(x, 0) = u_0(x) & \text{in } \Omega, \end{cases}$$
(A.1.1)

where the operator

$$\mathcal{M}u := \sum_{i,j=1}^{n} \left(a_{ij}u_{x_j} + a_i u \right)_{x_i} - au.$$

satisfies the uniform ellipticity and boundedness condition, i.e., there are $\mu_1, \mu_2 > 0$ s.t. for all $\xi \in \mathbb{R}^n, x \in \Omega$ and $t \in (0,T)$ it holds that

$$\mu_1 |\xi|^2 \leq \sum_{i,j=1}^n a_{ij}(x,t) \xi_i \xi_j \leq \mu_2 |\xi|^2$$
(A.1.2)

and $a_i, a, f \in L^{\infty}(\Omega \times (0,T))$ for $i = 1, \ldots, n$.

Then, for any $u_0 \in L^2(\Omega)$ there is a unique weak solution $u \in C([0,T]; L^2(\Omega)) \cap L^2(0,T; H^1(\Omega))$ in the sense that for all $\eta \in W_2^{1,1}(\Omega \times (0,T))$ with $\eta(T) = 0$ it holds that

$$-\int_{0}^{T}\int_{\Omega}u\eta_{t}\,dx\,dt + \int_{0}^{T}\int_{\Omega}\sum_{i=1}^{n}\left(\sum_{j=1}^{n}a_{ij}u_{x_{j}} + a_{i}u\right)\eta_{x_{i}} + au\eta + f\eta\,dx\,dt = \int_{\Omega}u_{0}(x)\eta(x,0)\,dx.$$
(A.1.3)

Moreover, any weak solution $u \in L^{\infty}(0,T;L^{2}(\Omega)) \cap L^{2}(0,T;H^{1}(\Omega))$ is also in $C([0,T];L^{2}(\Omega))$.

Proof. This follows similarly to the proof of Theorem III.5.1 in [94] by adapting it to our boundary condition and is a special case for bounded coefficient functions. Then, the coefficients are in the required spaces for any suitable combination of q, r, q_1, r_1 .

Remark A.1.2. The weak formulation (A.1.3) is equivalent to

$$\int_{\Omega} u(x,t)\eta(x,t) \,\mathrm{d}x - \int_{0}^{t} \int_{\Omega} u\eta_t \,\mathrm{d}x \,\mathrm{d}s + \int_{0}^{t} \int_{\Omega} \sum_{i=1}^{n} \left(\sum_{j=1}^{n} a_{ij} u_{x_j} + a_i u \right) \eta_{x_i} + au\eta + f\eta \,\mathrm{d}x \,\mathrm{d}s$$
$$= \int_{\Omega} u_0(x)\eta(x,0) \,\mathrm{d}x. \tag{A.1.4}$$

for a.e. $t \in (0,T)$ and all $\eta \in W_2^{1,1}(\Omega \times (0,T))$.

Lemma A.1.3. Under the assumptions of Theorem A.1.1 any weak solution $u \in C([0, T]; L^2(\Omega)) \cap L^2(0, T; H^1(\Omega))$ of (A.1.1) in the sense of (A.1.3) satisfies

$$\frac{1}{2} \int_{\Omega} u^2(t) \, dx + \int_0^t \int_{\Omega} \sum_{i=1}^n \left(\sum_{j=1}^n a_{ij} u_{x_j} + a_i u \right) u_{x_i} + a u^2 + f u \, dx \, dt = \frac{1}{2} \int_{\Omega} u_0^2 \, dx$$

for a.e. $t \in (0,T)$,

Proof. This follows similarly to (2.13) in Chapter III in [94] from the corresponding weak formulation (A.1.3) of a solution.

Theorem A.1.4. For $u_0 \in L^{\infty}(\Omega)$ under the assumptions of Theorem A.1.1 there is a constant $K_1 > 0$ depending on $\mu_1, ||a_i||_{L^{\infty}(\Omega \times (0,T))}, ||a||_{L^{\infty}(\Omega \times (0,T))}, ||f||_{L^{\infty}(\Omega \times (0,T))}$ s.t. $||u||_{L^{\infty}(\Omega \times (0,T))} \leq K_1$ holds for any weak solution $u \in C([0,T]; L^2(\Omega)) \cap L^2(0,T; H^1(\Omega))$ of (A.1.1) in the sense of (A.1.3).

Proof. This follows similarly to the proof of Theorem III.7.1 in [94] by adapting it to our boundary condition and is a special case for bounded coefficients. Then, the coefficients are in the required spaces for any suitable combination of q, r.

Theorem A.1.5. ([94, Theorem III.12.1]) Let $\alpha \in (0,1)$ and T > 0. If $a_{ij}, (a_{ij})_{x_i}, a_i, (a_i)_{x_i}, a, f \in C^{\alpha, \frac{\alpha}{2}}(\Omega \times (0,T))$ for i, j = 1, ..., n, then $u \in C^{2+\alpha,1+\frac{\alpha}{2}}(\Omega \times (0,T))$ holds for any weak solution $u \in C([0,T]; L^2(\Omega)) \cap L^2(0,T; H^1(\Omega))$ of (A.1.1) in the sense of (A.1.3)

Theorem A.1.6. Let $\alpha \in (0, 1)$, T > 0 and consider the IBVP

$$\begin{cases} u_t + \mathcal{L}u = f & \text{in } \Omega \times (0, T), \\ \mathcal{B}u = 0 & \text{on } \partial\Omega \times (0, T), \\ u(x, 0) = u_0(x) & \text{in } \Omega, \end{cases}$$
(A.1.5)

where

$$\mathcal{L}u := -\sum_{i,j=1}^{n} a_{ij} u_{x_j x_i} + \sum_{i=1}^{n} a_i u_{x_i} + au$$

satisfies (A.1.2) and

$$\mathcal{B}u = \sum_{i=1}^{n} b_i u_{x_i} + bu$$

satisfies

$$\left|\sum_{i=1}^{n} b_{i} \nu_{i}\right| \ge \delta > 0 \text{ on } \partial\Omega \times (0,T)$$

If $a_{ij}, a_i, a \in C^{\alpha, \frac{\alpha}{2}}(\overline{\Omega} \times [0, T])$ and $b_i, b \in C^{1+\alpha, \frac{1+\alpha}{2}}(\partial \Omega \times [0, T])$ for i, j = 1, ..., n, then for any $f \in C^{\alpha, \frac{\alpha}{2}}(\overline{\Omega} \times [0, T])$ and $u_0 \in C^{2+\alpha}(\overline{\Omega})$ satisfying the compatibility condition $\mathcal{B}u_0 = 0$ on $\partial \Omega \times \{0\}$ there is a unique solution $u \in C^{2+\alpha, 1+\frac{\alpha}{2}}(\overline{\Omega} \times [0, T])$ of (A.1.5) and a constant $K_2 > 0$ depending continuously on the norms of a_{ij}, a_i, a in $C^{\alpha, \frac{\alpha}{2}}(\overline{\Omega} \times [0, T])$ and the norms of b_i, b in $C^{1+\alpha, \frac{1+\alpha}{2}}(\partial \Omega \times [0, T])$ s.t.

$$\left\|u\right\|_{C^{2+\alpha,1+\frac{\alpha}{2}}(\overline{\Omega}\times[0,T])} \leqslant K_2\left(\left\|f\right\|_{C^{\alpha,\frac{\alpha}{2}}(\overline{\Omega}\times[0,T])} + \left\|u_0\right\|_{C^{2+\alpha}(\overline{\Omega})}\right).$$
(A.1.6)

The solution u depends continuously on the coefficients and functions.

Proof. Set $\Phi = 0$ in Theorem IV.5.3 in [94]. The continuous dependence of K_2 on the coefficients and of u on the coefficients and functions follows from the proof of this theorem.

Theorem A.1.7. Let T > 0 and consider the IBVP (A.1.5). Suppose that for i, j = 1, ..., n, $a_{ij}, a_i, a \in C^{\alpha, \frac{\alpha}{2}}(\overline{\Omega} \times [0, T]), b_i, b \in C^{1+\alpha, \frac{1+\alpha}{2}}(\partial\Omega \times [0, T])$ and satisfy $\sum_{i,j=1}^n a_{ij}(x, t)\xi_j\xi_i \ge \mu_1|\xi|^2$ for $\mu_1 > 0$ and $\sum_{i=1}^n b_i\nu_i \ge \delta > 0$. Then, for any $f \in C^{\alpha, \frac{\alpha}{2}}(\overline{\Omega} \times [0, T])$ and $u_0 \in C(\overline{\Omega})$ there is a unique solution $u \in C^{2,1}(\Omega \times (0, T)) \cap C(\overline{\Omega} \times [0, T]) \cap C^{1,0}(\overline{\Omega} \times (0, T))$ of (A.1.5). If additionally $u_0 \in C^{2+\alpha}(\Omega)$ and satisfies the compatibility condition $\mathcal{B}u_0 = 0$ on $\partial\Omega \times \{0\}$, then $u \in C^{2+\alpha, 1+\frac{\alpha}{2}}(\overline{\Omega} \times [0, T])$.

Proof. This is a special case of Theorem 5.18 in [100].

Theorem A.1.8. Let T > 0 and consider for a constant D > 0 the nonhomogeneous heat equation with Neumann boundary condition

$$\begin{cases} u_t = D\Delta u + f & \text{in } \Omega \times (0, T), \\ \partial_\nu u = 0 & \text{on } \partial\Omega \times (0, T), \\ u(x, 0) = u_0(x) & \text{in } \Omega. \end{cases}$$
(A.1.7)

Then, for any $f \in L^2(\Omega \times (0,T))$ and $u_0 \in H^1(\Omega)$ there is a unique solution $u \in W_2^{2,1}(\Omega \times (0,T))$ satisfying

$$\|u\|_{W^{2,1}_{2}(\Omega\times(0,T))} \leq K_{3}(T) \left(\|f\|_{L^{2}(\Omega\times(0,T))} + \|u_{0}\|_{H^{1}(\Omega)}\right).$$

Proof. This is a special case of Theorem IV.9.1 (together with the remark at the end of the chapter for Neumann boundary conditions) in [94] for a nonhomogeneous heat equation with $\Phi = 0, q = 2.$

Theorem A.1.9. ([100, Theorem 2.9]) Let T > 0 and $u, v \in C^{2,1}(\overline{\Omega} \times (0,T)) \cap C(\overline{\Omega} \times [0,T])$. Suppose that \mathcal{L} satisfies (A.1.2), $\sum_{i=1}^{n} a_{ii}, a_i, a \in L^{\infty}(\Omega \times (0,T))$ for $i = 1, \ldots, n, a \ge 0$, $-(b_1, \ldots, b_n)^T$ is a vector that points strongly inside $\Omega \times (0,T)$ on $\partial\Omega \times (0,T)$ and $b \ge 0$ on $\partial\Omega \times (0,T)$. If

$$\begin{cases} u_t + \mathcal{L}u \leqslant v_t + \mathcal{L}v & \text{ in } \Omega \times (0, T), \\ \mathcal{B}u \leqslant \mathcal{B}v & \text{ on } \partial\Omega \times (0, T), \\ u \leqslant v & \text{ in } \overline{\Omega}, \end{cases}$$

then $u \leq v$ holds in $\overline{\Omega} \times [0, T]$.

Proposition A.1.10. ([36, Proposition 13.1]) Let T > 0, $u \in W_p^{2,1}(\Omega \times (0,T))$ for $p \ge n+1$ and consider an operator \mathcal{L} that satisfies (A.1.2) with coefficients that are continuous on $\overline{\Omega} \times [0,T]$. Suppose

$$u_t + \mathcal{L}u \ge 0,$$

holds a.e. in $\Omega \times (0,T)$. If u attains its minimum $m \leq 0$ at $(x_0, t_0) \in \Omega \times (0,T]$, then $u \equiv m$ in $\overline{\Omega} \times [0, t_0]$.

Theorem A.1.11. ([36, Theorem 13.5 with the remark at the end of the chapter]) Let T > 0, $a_{ij}, a_i, a \in C(\overline{\Omega} \times [0,T])$, $b_i \in C^1(\partial\Omega)$ and $b \in C^1(\partial\Omega \times [0,T])$. If $u \in W_p^{2,1}(\Omega \times (0,T))$ for p > n+2 satisfies

$$\begin{cases} u_t + \mathcal{L}u \ge 0 & \text{in } \Omega \times (0, T), \\ \mathcal{B}u \ge 0 & \text{on } \partial\Omega \times (0, T), \\ u(x, 0) \ge 0 & \text{in } \Omega, \end{cases}$$
(A.1.8)

then $u \ge 0$ holds in $\overline{\Omega} \times [0, T]$.

A.1.2 Nonlinear parabolic PDEs

Theorem A.1.12. Let T > 0, $p \ge 2$ and $a : \mathbb{R}^{2n+2} \to \mathbb{R}^n$ and $b : \mathbb{R}^{2n+2} \to \mathbb{R}$ be measurable and satisfy the estimates

$$a(x,t,u,\nabla u) \cdot \nabla u \ge K_4 |\nabla u|^p - \psi_0(x,t), \tag{A.1.9}$$

$$|a(x,t,u,\nabla u)| \leq K_5 |\nabla u|^{p-1} + \psi_1(x,t),$$
(A.1.10)

$$|b(x, t, u, \nabla u)| \leq K_6 |\nabla u|^p + \psi_2(x, t),$$
(A.1.11)

where K_4, K_5, K_6 are positive constants and $\psi_j \in L^{\infty}(\Omega \times (0,T))$, j = 0, 1, 2, are non-negative. Besides, let $u \in C([0,T]; L^2(\Omega)) \cap L^p(0,T; W^1_p(\Omega)) \cap L^{\infty}(\Omega \times (0,T))$ be a weak solution of

$$\begin{cases} \partial_t u - \nabla \cdot a(x, t, u, \nabla u) + b(x, t, u, \nabla u) = 0 & \text{ in } \Omega \times (0, T), \\ a(x, t, u, \nabla u) \cdot \nu(x) = 0 & \text{ on } \partial\Omega \times (0, T), \\ u(x, 0) = u_0(x) & \text{ in } \Omega \end{cases}$$
(A.1.12)

in the sense that

$$\int_{\Omega} u\varphi \, dx \big|_{t_1}^{t_2} + \int_{t_1}^{t_2} \int_{\Omega} -u\varphi_t + a(x,t,u,\nabla u) \cdot \nabla \varphi + b(x,t,u,\nabla u)\varphi \, dx \, dt = 0$$

holds for all $0 \leq t_1 < t_2 \leq T$ and for all $\varphi \in L^p((0,T), W_p^1(\Omega))$ with derivative $\varphi_t \in L^2(\Omega \times (0,T))$. Then, u is Hölder continuous on $\overline{\Omega} \times [\varepsilon, T]$ for any $\varepsilon > 0$, i.e., there are constants $K_7 > 0$ and $\alpha \in (0,1)$ depending only on the constants appearing in (A.1.9) - (A.1.11), the norms of the ψ_j and $||u||_{L^{\infty}(\Omega \times (0,T))}$ and ε s.t. the estimate

$$|u(x_1, t_1) - u(x_2, t_2)| \leq K_7(\varepsilon) \left(|x_1 - x_2|^{\alpha(\varepsilon)} + |t_1 - t_2|^{\frac{\alpha(\varepsilon)}{p}} \right)$$
(A.1.13)

is satisfied for any pair $(x_1, t_1), (x_2, t_2) \in \overline{\Omega} \times [\varepsilon, T]$.

If additionally $u_0 \in C^{\alpha'}(\overline{\Omega})$ for $\alpha' \in (0,1)$, then u satisfies (A.1.13) in $\overline{\Omega} \times [0,T]$ for constants K_7 and α that can be chosen independent from ε . In this case α also depends on α' .

Proof. This is a special case of Theorem 4 in [39] for g = 0 and bounded ψ_i that are in the required spaces for any suitable combination of q, r.

Theorem A.1.13. Let T > 0 and $u \in C([0,T]; L^2(\Omega)) \cap L^2(0,T; H^1(\Omega)) \cap L^{\infty}(\Omega \times (0,T))$ a weak solution to (A.1.12). Suppose that there are $\alpha \in (0,1)$ and $\lambda, \Lambda, \mu_1, \mu_2, \mu_3 > 0$ s.t. for all $(x, t, z, p) \in \Omega \times (0,T) \times \mathbb{R} \times \mathbb{R}^n$ with $|z| \leq M := \|u\|_{L^{\infty}(\Omega \times (0,T))}$ and all $(y, w) \in \Omega \times [-M, M]$ it holds that

$$|a(x,t,z,0)| \leq \mu_1,$$

$$|a(x,t,z,p) - a(y,t,w,p)| \leq \mu_2(1+|p|) \left(|x-y|^{\alpha} + |z-w|^{\alpha}\right),$$

$$|b(x,t,z,p)| \leq \mu_3(1+|p|^2)$$

and that for all $(x, t, z, p) \in \partial \Omega \times (0, T) \times \mathbb{R} \times \mathbb{R}^n$ and all $(s, w) \in (0, T) \times \mathbb{R}$ with $|z|, |w| \leq M$ the estimate

$$|a(x,t,z,p) - a(x,s,w,p)| \leq \mu_2(1+|p|)|t-s|^{\frac{\alpha}{2}}$$

holds. Moreover, suppose that $a_{ij} := \frac{\partial a_i}{\partial p_j}$ satisfies

$$\sum_{i,j=1}^{n} a_{ij}(x,t,z,p)\xi_i\xi_j \ge \lambda |\xi|^2,$$
$$|a_{ij}(x,t,z,p)| \le \Lambda$$

for all $(x, t, z, p) \in \Omega \times (0, T) \times \mathbb{R} \times \mathbb{R}^n$ with $|z| \leq M$ and $\xi \in \mathbb{R}^n$ and that $u_0 \in C^{1+\alpha}(\overline{\Omega})$ satisfies the compatibility condition

$$a(x, 0, u_0, \nabla u_0) \cdot \nu = 0$$
 for $x \in \partial \Omega$.

Then, there are $\delta(\lambda, \Lambda, \alpha) \in (0, 1)$ and $K_8(\lambda, \Lambda, M, \mu_1, \mu_2, \mu_3, \|u_0\|_{C^{1+\alpha}(\overline{\Omega})}, T) > 0$ s.t.

$$\|u\|_{C^{1+\delta,\frac{1+\delta}{2}}(\overline{\Omega}\times[0,T])}\leqslant K_8$$

Proof. This is a special case of Theorem 1.1 in [101] for $\psi = 0$.

Proposition A.1.14. Let T > 0 and consider a function $f(x,t,s) : \overline{\Omega} \times [0,T] \times \mathbb{R} \to \mathbb{R}$ that is continuous in x and t and continuously differentiable in s. Further, let p > n + 2 and $u, w \in W_p^{2,1}(\Omega \times (0,T)) \cap C([0,T]; L^2(\Omega)) \cap L^{\infty}(\Omega \times (0,T))$ s.t. $u(\cdot,0) \leq w(\cdot,0)$ and $u(\cdot,0) \neq w(\cdot,0)$ hold on $\overline{\Omega}$. If the estimates

$$\partial_t u - \Delta u - f(x, t, u) \leq \partial_t w - \Delta w - f(x, t, w)$$
 (A.1.14)

in $\Omega \times (0,T)$ and

$$\frac{\partial u}{\partial \nu} \leqslant \frac{\partial w}{\partial \nu} \tag{A.1.15}$$

on $\partial \Omega \times (0,T)$ are satisfied, then

u < w

holds in $\overline{\Omega} \times (0,T)$.

Proof. This is a special case of Proposition 52.7 in [127] with b = 0 and f independent from ξ .

A.1.3 Monotone Operators

Definition A.1.15. Let V be a reflexive Banach space and $\mathcal{A}: V \to V^*$. The operator \mathcal{A} is called

- monotone if $\langle \mathcal{A}u \mathcal{A}v, u v \rangle_{V^* V} \ge 0$ for all $u, v \in V$,
- maximal monotone if \mathcal{A} is monotone and $\mathcal{A}u = f$ if and only if $\langle f \mathcal{A}v, u v \rangle_{V^*, V} \ge 0$ for all $u \in V$,
- · hemicontinuous if $t \mapsto \langle \mathcal{A}(u+tv), w \rangle_{V^*, V}$ is continuous for all $u, v, w \in V$.

Lemma A.1.16. (Minty's lemma [133, Lemma 2.1 in Chapter II]) Let V a reflexive Banach space and consider an operator $\mathcal{A}: V \to V^*$. If \mathcal{A} is monotone and hemicontinuous, then it is maximal monotone.

Theorem A.1.17. Let T > 0, $p \in (1, \infty)$, p^* s.t. $\frac{1}{p} + \frac{1}{p^*} = 1$, V a separable and reflexive Banach space and H a Hilbert space s.t. $V \hookrightarrow H$ is dense. Consider a family of operators $\mathcal{A}(t, \cdot) : V \to V^*$, $t \in [0, T]$ s.t.

- (i) $\mathcal{A}(\cdot, v) : [0, T] \to V^*$ is measurable for all $v \in V$,
- (ii) $\mathcal{A}(t, \cdot): V \to V^*$ is monotone, hemicontinuous and bounded by

 $\|\mathcal{A}(t,v)\|_{V^*} \leq K_9 \|v\|_{V}^{p-1}$

for all $v \in V$ and a.e. $t \in [0, T]$,

(iii) there are a seminorm [·] on V and $\lambda, \alpha > 0$ s.t.

$$[v] + \|v\|_{H} \ge \alpha \|v\|_{V},$$

$$\langle \mathcal{A}(t, v), v \rangle_{V^{*}, V} \ge \alpha [v]^{p}$$

for a.e. $t \in [0, T]$ and $v \in V$.

Then, for every $f \in L^{p^*}(0,T;V^*)$ and $u_0 \in H$ there is a unique solution $u \in L^p(0,T;V) \cap C([0,T];H)$ with $u_t \in L^{p^*}(0,T;V^*)$ to the Cauchy Problem

$$\begin{cases} u_t(t) + \mathcal{A}(t, u(t)) = f(t) & \text{ in } L^{p^*}(0, T; V^*), \\ u(0) = u_0 & \text{ in } H \end{cases}$$
(A.1.16)

in the sense that for all $\varphi \in V$ and a.e. $t \in (0,T)$

$$\langle u_t, \varphi \rangle_{V^*, V} + \langle \mathcal{A}(t, u(t)), \varphi \rangle_{V^*, V} = \langle f, \varphi \rangle_{V^*, V}$$

and $u(0) = u_0$ in *H*.

Proof. This follows from Propositions 2.1 and 4.1 in Chapter III in [133].

A.1.4 Evolution equations

Lemma A.1.18. ([85, p.56 (iii)] and [148, Lemma 1.3 (ii) and (iii)]) Consider the Neumann heat semigroup $(e^{t\Delta})_{t\geq 0}$ in Ω . Let $\lambda_1 > 0$ the first nonzero eigenvalue of $-\Delta$ under Neumann boundary conditions in Ω . Then, there are constants $K_{10}, K_{11}, K_{12} > 0$ depending only on Ω s.t.

(i) if $1 \leq q then$

$$\|e^{t\Delta}u\|_{L^{p}(\Omega)} \leq K_{10}t^{-\frac{n}{2}\left(\frac{1}{q}-\frac{1}{p}\right)}\|u\|_{L^{q}(\Omega)}$$

for all $u \in L^q(\Omega)$ and $t \in (0, 1)$,

(ii) if $1 \leq q \leq p \leq \infty$ then

$$\|\nabla e^{t\Delta}u\|_{L^{p}(\Omega)} \leq K_{11}\left(1 + t^{-\frac{1}{2} - \frac{n}{2}\left(\frac{1}{q} - \frac{1}{p}\right)}\right)e^{-\lambda_{1}t}\|u\|_{L^{q}(\Omega)}$$

for all $u \in L^q(\Omega)$ and t > 0,

(iii) if $2 \leq p < \infty$ then

$$\|\nabla e^{t\Delta}u\|_{L^p(\Omega)} \leq K_{12}e^{-\lambda_1 t}\|\nabla u\|_{L^p(\Omega)}$$

for all $u \in W_p^1(\Omega)$ and t > 0.

A.1.5 Elliptic PDEs

Lemma A.1.19. Let $\lambda > 0$ and $a_{ij}, b_i \in C^2(\overline{\Omega})$. $i, j = 1, \ldots, n \text{ s.t. } a_{ij} = a_{ji}$. Assume that for some $\alpha > 0$ it holds that $\sum_{i,j=1}^n a_{ij}\xi_i\xi_j \ge \alpha |\xi|^2$ on $\overline{\Omega}$ for all $\xi \in \mathbb{R}^n$ and $\sum_{i=1}^n b_i\nu_i > 0$ on $\partial\Omega$. Then, for all $f \in W^1_{\alpha}(\Omega)$ the elliptic problem

$$\begin{cases} -\sum_{i,j=1}^{n} (a_{ij}u_{x_j})_{x_i} + \lambda u = f & \text{in } \Omega, \\ \sum_{i=1}^{n} b_i u_{x_i} = 0 & \text{on } \partial\Omega \end{cases}$$
(A.1.17)

has a unique solution $u \in C^2(\overline{\Omega})$. Moreover, there are λ_0 and a constant $K_{13} > 0$ s.t. for all $\lambda > \lambda_0$ the estimate

$$\|u\|_{H^{2}(\Omega)} \leq K_{13} \|f\|_{L^{2}(\Omega)}$$
(A.1.18)

holds

Proof. We conclude from Theorems 2.4.2.7 and 2.5.1.1 in [72] that (A.1.17) has a unique solution $u \in \bigcap_{p>1} W_p^3(\Omega) \subset C^2(\overline{\Omega})$. The estimate (A.1.18) follows from Theorem 2.3.3.6 in [72].

A.1.6 ODEs

Lemma A.1.20. Let T > 0. If $f \in C([0,T]; \mathbb{R}^+_0) \cap C^1((0,T); \mathbb{R}^+_0)$ satisfies the inequality

$$f'(t) + K_{14}f(t) \leqslant K_{14}K_{15},$$

on (0,T) for constants $K_{14}, K_{15} > 0$, then for $t \in (0,T)$ it holds that

$$f(t) \leqslant K_{15} + f(0).$$

Proof. Let $t \in (0, T)$. We differentiate

$$\left(f(t)e^{K_{14}t}\right)' = \left(f'(t) + K_{14}f(t)\right)e^{K_{14}t} \leqslant K_{14}K_{15}e^{K_{14}t}.$$

Integrating over [0, t] we obtain

$$f(t)e^{K_{14}t} \leqslant K_{14}K_{15} \int_0^t e^{K_{14}s} \,\mathrm{d}s + f(0) = K_{15} \left(e^{K_{14}t} - 1 \right) + f(0).$$

Consequently,

$$f(t) \leq K_{15} \left(1 - e^{-K_{14}t}\right) + f(0)e^{-K_{14}t} \leq K_{15} + f(0).$$

A.2 Fixed-point theorems

Theorem A.2.1. (Banachs's fixed-point theorem, [58, Section 9.2.1, Theorem 1]) Let X a Banach space and consider a contraction $T: X \to X$. Then, T has a unique fixed-point.

Theorem A.2.2. (Schauder's fixed-point theorem, [154, Section 2.6, Theorem 2.A]) Let X a Banach space and $M \subset X$ nonempty, closed, bounded and convex. Then, a compact operator $T: M \to M$ has a fixed-point.

Theorem A.2.3. (Schaefer's fixed-point theorem, [58, Section 9.2.2, Theorem 4]) Let X a Banach space and consider a compact operator $T: X \to X$. If the set

$$\{u \in X : u = \lambda Tu \text{ for some } \lambda \in [0,1]\}$$

is bounded, then T has a fixed-point.

A.3 Results from functional analysis

In this section we summarize some results from functional analysis.

Lemma A.3.1. ([20, Corollary 3.30]) Let X a separable Banach space. Then, every bounded sequence in its dual space X^* has a weakly-*-convergent subsequence.

Lemma A.3.2. (Compensated compactness, [20, Propositions 3.5 (iv) and 3.13 (iv)]) Let X a Banach space, $x_k \in X$ and $f_k \in X^*$ for $k \in \mathbb{N}$. If $x_k \to x$ in X and $f_k \to f$ in X^* or $x_k \to x$ in X and $f_k \stackrel{*}{\to} f$ in X^* for $k \to \infty$, then

$$\langle f_k, x_k \rangle_{X^*, X} \xrightarrow[k \to \infty]{} \langle f, x \rangle_{X^*, X}.$$

Lemma A.3.3. (A result from [59, p. 6]) Let $p \in (1, \infty)$ and consider a bounded sequence $(f_k)_{k \in \mathbb{N}} \subset L^p(\Omega)$. If

$$\int_E f_k \, dx \xrightarrow[k \to \infty]{} \int_E f \, dx$$

for each bounded, measurable set $E \subset \Omega$, then

$$f_k \xrightarrow[k \to \infty]{} f \text{ in } L^p(\Omega).$$

Lemma A.3.4. ([102, Lemma 1.3]) Let $p \in (1, \infty)$ and $f_k, f \in L^p(\Omega)$, $k \in \mathbb{N}$. If there is a constant $K_{16} > 0$ s.t. $||f_k||_{L^p(\Omega)} \leq K_{16}$ and $f_k \to f$ for $k \to \infty$ a.e. in Ω , then

$$f_k \xrightarrow[k \to \infty]{} f \text{ in } L^p(\Omega).$$

Lemma A.3.5. ([141, Lemma 1.4 in Chapter III]) Let X, Y Banach spaces s.t. $X \hookrightarrow Y$. If $f \in L^{\infty}(0,T;X) \cap C_w([0,T];Y)$, then $f \in C_w([0,T];X)$.

Theorem A.3.6. (*De la Vallée-Poussin theorem*, [19, Theorem 4.5.9]) Let $O \subset \mathbb{R}^n$ open and bounded and $\{f_k\}_{k \in \mathbb{N}} \subset L^1(O)$. Then, the following are equivalent:

- (i) $\{f_k\}_k$ is uniformly integrable,
- (ii) there is an increasing function $G: [0, \infty) \to [0, \infty)$ satisfying $\lim_{t\to\infty} \frac{G(t)}{t} = \infty$ and

$$\sup_{k\in\mathbb{N}}\int_O G(|f_k|)\,dx<\infty.$$

Lemma A.3.7. (*Vitali's lemma*, [37, Theorem 21]) Let $O \subset \mathbb{R}^n$ open and bounded and consider functions $f_k \in L^1(O)$, $k \in \mathbb{N}$, s.t. $f_k \to f$ for $k \to \infty$ a.e. in O. Then, the following are equivalent:

- (i) $f \in L^1(O)$ and $f_k \to f$ in $L^1(O)$ for $k \to \infty$,
- (ii) $\{f_k\}_k$ is uniformly integrable.

Lemma A.3.8. ([133, Chapter III Proposition 1.2]) Let T > 0, $p, q \in (0, 1)$ s.t. $\frac{1}{p} + \frac{1}{q} = 1$, Va Banach space and H a Hilbert spaces s.t. $V \hookrightarrow H \hookrightarrow V^*$ are dense. If $f \in L^p(0, T; V)$ and $\partial_t f \in L^q(0, T; V^*)$, then $f \in C([0, T]; H)$ with

$$\|f\|_{C([0,T];H)} \leq K_{17} \left(\|f\|_{L^p(0,T;V)} + \|\partial_t f\|_{L^q(0,T;V^*)} \right)$$

and

$$\frac{d}{dt} \|f\|_{H}^{2} = 2\langle \partial_{t}f, f \rangle_{V^{*}, V}.$$

Lemma A.3.9. (*Lions-Aubin lemma*, [134, Corollary 4]) Let X, B, Y Banach spaces s.t. $X \hookrightarrow \hookrightarrow B \hookrightarrow Y$ and $p \in [1, \infty)$, $r \in (1, \infty)$.

- (i) If $\{f_k\}_{k\in\mathbb{N}}$ is bounded in $L^p(0,T;X)$ and $\{\partial_t f_k\}_{k\in\mathbb{N}}$ is bounded in $L^1(0,T;Y)$, then $\{f_k\}_{k\in\mathbb{N}}$ is relatively compact in $L^p(0,T;B)$.
- (ii) If $\{f_k\}_{k\in\mathbb{N}}$ is bounded in $L^{\infty}(0,T;X)$ and $\{\partial_t f_k\}_{k\in\mathbb{N}}$ is bounded in $L^r(0,T;Y)$, then $\{f_k\}_{k\in\mathbb{N}}$ is relatively compact in C([0,T];B).

Lemma A.3.10. (Partition of unity, [2, Theorem 3.15]) Let $K \subset \Omega$ compact and consider an open covering $\{O_k\}_{k=1,...,N}$ of K. Then, there is a partition of unity $\{\psi_k\}_{k=1,...,N}$ s.t. $0 \leq \psi_k \leq 1$, $\psi_k \in C_c^{\infty}(O_k)$ for k = 1, ..., N and $\sum_{k=1}^N \psi_k = 1$ on K.

Theorem A.3.11. ([88, Chapter 6, Theorems 6.27]) Let $S \subset \mathbb{R}^n$ open and bounded, $x_0 \in \Omega$ and $f: S \times \Omega \to \mathbb{R}$ a map s.t.

• for any $x \in \Omega$ the map $y \mapsto f(y, x)$ is in $L^1(S)$,

- for a.e. $y \in S$, the map $x \mapsto f(y, x)$ is continuous at x_0 ,
- there is $h \in L^1(S)$, $h \ge 0$ s.t. $|f(\cdot, x)| \le h$ a.e. in S for all $x \in \Omega$.

Then, the map $F: \Omega \to \mathbb{R}, x \mapsto \int_S f(y, x) \, dy$ is continuous in x_0 .

Theorem A.3.12. ([88, Chapter 6, Theorems 6.28]) Let $S \subset \mathbb{R}^n$ open and bounded, $I \subset \mathbb{R}$ a nontrivial open interval and $f: S \times I \to \mathbb{R}$ a map with the following properties:

- for any $x \in I$ the map $y \mapsto f(y, x)$ is in $L^1(I)$,
- for a.e. $y \in S$, the map $x \mapsto f(y, x)$ is differentiable with derivative f',
- there is $h \in L^1(S)$, $h \ge 0$ s.t. $|f'(\cdot, x)| \le h$ a.e. in S for all $x \in \Omega$.

Then, for any $x \in I$ it holds that $f'(\cdot, x) \in L^1(S)$ and $F(x) := \int_S f(y, x) dy$ is differentiable with derivative $F'(x) = \int_S f'(y, x) dy$.

A.4 Results used in part II

Lemma A.4.1. Let $u \in C^1(\overline{\Omega})$ and consider $\alpha, \beta \ge 1$ satisfying

$$\alpha < \begin{cases} 1 + \beta, & n = 1, 2, \\ 1 + \frac{2\beta}{n}, & n > 2. \end{cases}$$
(A.4.1)

Moreover, let $q > \max\{1, \beta + \alpha - 1\}$,

$$\max\left\{\frac{n(\alpha-1)}{q},\frac{2(\alpha-1)}{q},1\right\} < r \leqslant \frac{2(q+\alpha-1)}{q}$$

and

$$s \begin{cases} = \infty, & n = 1, \\ \in \left(\frac{2qr}{qr - 2(\alpha - 1)}, \infty\right), & n = 2, \\ = \frac{2n}{n - 2}, & n > 2. \end{cases}$$
(A.4.2)

Then, for any $K_{18} > 0$ it holds that

$$\int_{\Omega} u^{q+\alpha-1} dx \leq \frac{2(q-1)}{q^2 K_{18}} \int_{\Omega} |\nabla u^{\frac{q}{2}}|^2 dx + K_{19}(K_{18},q,r) \| u^{\frac{q}{2}} \|_{L^r(\Omega)}^{2r \frac{2(q+\alpha-1)}{s}-q} + K_{20}(r)^{\frac{qr-2(q+\alpha-1)}{\frac{qr}{s}-q}},$$
(A.4.3)

where

$$\begin{split} K_{19}(K_{18},q,r) &:= 2\left(\frac{K_{21}^2 q^2 K_{18}}{q-1}\right)^{\frac{qr-2(q+\alpha-1)}{qr(\frac{2}{s}-1)+2(\alpha-1)}} + K_{20}(r)^{\frac{qr-2(q+\alpha-1)}{\frac{qr}{s}-q}},\\ K_{20}(r) &:= 4K_S(s)|\Omega|^{\frac{r-2}{2r}},\\ K_{21} &:= 2K_S(s)\left(1+2K_P\right). \end{split}$$

Here, $K_S(s)$ denotes the Sobolev embedding constant from $H^1(\Omega)$ into $L^s(\Omega)$ from Lemma 2.2.8 and $K_P > 0$ denotes the constant from the Poincaré inequality. Moreover, for any $K_{18}, K_{22} > 0$ (after setting $r = \frac{q+\alpha-1+\beta}{q}$) the estimate

$$\int_{\Omega} u^{q+\alpha-1} dx \leq \frac{2(q-1)}{q^2 K_{18}} \int_{\Omega} |\nabla u^{\frac{q}{2}}|^2 dx + \frac{1}{K_{22}} \int_{\Omega} u^{\beta} dx \int_{\Omega} u^{q+\alpha-1} dx + K_{23} \left(K_{18}, K_{22}, q\right)$$
(A.4.4)

holds, where

$$K_{23}\left(K_{18}, K_{22}, q\right) := \left(2\left(\frac{K_{21}^2 q^2 K_{18}}{q-1}\right)^{\frac{q+\alpha-1-\beta}{q-\alpha+1+\beta-2\frac{q+\alpha-1+\beta}{s}}} + K_{24}(q)^{\frac{q+\alpha-\beta-1}{q-\frac{q+\alpha-1+\beta}{s}}}\right)^{\frac{q-\alpha+1+\beta-\frac{2(q+\alpha-1+\beta)}{s}}{\beta+1-\alpha-\frac{2\beta}{s}}} \cdot K_{22}^{\frac{q-\alpha+1+\beta-\frac{2(q+\alpha-1+\beta)}{s}}{s}} + K_{24}(q)^{\frac{q+\alpha-\beta-1}{q-\frac{q+\alpha-1+\beta}{s}}},$$

and

$$K_{24}(q) := 4K_S(s)|\Omega|^{\frac{1}{2} - \frac{q}{q + \alpha - 1 + \beta}}.$$

Proof. Let

$$\lambda = \frac{\frac{q}{2(q+\alpha-1)} - \frac{1}{r}}{\frac{1}{s} - \frac{1}{r}}.$$

Then, $\lambda \in [0,1)$ and $\frac{\lambda(q+\alpha-1)}{q} \in [0,1)$ hold due to our choice of parameters. We state inequality (2.11) from the proof of Theorem 1.1 in [99] (with $B(x,\delta)$ replaced by Ω)

$$\int_{\Omega} u^{q+\alpha-1} \, \mathrm{d}x \leq 2 \left(K_{21} \| \nabla u^{\frac{q}{2}} \|_{(L^{2}(\Omega))^{n}} \right)^{\frac{2\lambda(q+\alpha-1)}{q}} \| u^{\frac{q}{2}} \|_{L^{r}(\Omega)}^{\frac{2(1-\lambda)(q+\alpha-1)}{q}} + \left(K_{20}(r)^{\lambda} \| u^{\frac{q}{2}} \|_{L^{r}(\Omega)} \right)^{\frac{2(q+\alpha-1)}{q}},$$

where

$$K_{20}(r) := 4K_S(s)|\Omega|^{\frac{r-2}{2r}}, K_{21} := 2K_S(s)\left(1 + 2K_P\right).$$

We proceed as in the proof of (2.14) in [99]. Applying Young's inequality twice and regrouping the terms we conclude that for $K_{18} > 0$ it holds that

$$\begin{split} \int_{\Omega} u^{q+\alpha-1} \, \mathrm{d}x \leqslant & \frac{2(q-1)}{q^2 K_{18}} \| \nabla u^{\frac{q}{2}} \|_{(L^2(\Omega))^n}^2 \\ &+ 2 \left(K_{21}^{\frac{2\lambda(q+\alpha-1)}{q}} \| u^{\frac{q}{2}} \|_{L^r(\Omega)}^{\frac{2(1-\lambda)(q+\alpha-1)}{q}} \left(\frac{q^2 K_{18}}{q-1} \right)^{\frac{\lambda(q+\alpha-1)}{q}} \right)^{\frac{q}{q-\lambda(q+\alpha-1)}} \\ &+ K_{20}(r)^{\frac{2\lambda(q+\alpha-1)}{q}} \left(\| u^{\frac{q}{2}} \|_{L^r(\Omega)}^{\frac{2(1-\lambda)(q+\alpha-1)}{q-\lambda(q+\alpha-1)}} + 1 \right) \\ &\leqslant & \frac{2(q-1)}{q^2 K_{18}} \| \nabla u^{\frac{q}{2}} \|_{(L^2(\Omega))^n}^2 + K_{19}(K_{18},q,r) \| u^{\frac{q}{2}} \|_{L^r(\Omega)}^{\frac{2(1-\lambda)(q+\alpha-1)}{q-\lambda(q+\alpha-1)}} + K_{20}(r)^{\frac{2\lambda(q+\alpha-1)}{q}}, \end{split}$$
(A.4.5)

where

$$K_{19}(K_{18},q,r) := 2\left(\frac{K_{21}^2q^2K_{18}}{q-1}\right)^{\frac{\lambda(q+\alpha-1)}{q-\lambda(q+\alpha-1)}} + K_{20}(r)^{\frac{2\lambda(q+\alpha-1)}{q}}.$$

Inserting the definition of λ we obtain inequality (A.4.3). Moreover, we state inequality (2.17) from the proof of Theorem 1.1 in [99], i.e.,

$$\|u^{\frac{q}{2}}\|_{L^{\frac{q+\alpha-1+\beta}{q-\lambda(q+\alpha-1)}}(\Omega)}^{\frac{2(1-\lambda)(q+\alpha-1)}{q-\lambda(q+\alpha-1)}} \leqslant \left(\|u^{\frac{q}{2}}\|_{L^{\frac{2\beta}{q}}(\Omega)}^{\frac{2\beta}{q}}\|u^{\frac{q}{2}}\|_{L^{\frac{2(q+\alpha-1)}{q}}(\Omega)}^{\frac{2(q+\alpha-1)}{q}}\right)^{\frac{q-2(q+\alpha-1)}{s}},$$
(A.4.6)

where due to our choice of α, β and s in (A.4.1) and (A.4.2) it holds that

$$\frac{q - \frac{2(q+\alpha-1)}{s}}{q - \alpha + 1 + \beta - \frac{2(q+\alpha-1+\beta)}{s}} < 1$$

Now, we estimate the term from (A.4.5) for $r = \frac{q+\alpha-1+\beta}{q}$ using (A.4.6) as in (2.19) from [99]. Young's inequality leads for any $K_{22} > 0$ to the estimate

$$K_{19}(K_{18},q,r) \| u^{\frac{q}{2}} \|_{L^{\frac{q+\alpha-1}{q-\lambda(q+\alpha-1)}}}^{\frac{2(1-\lambda)(q+\alpha-1)}{q-\lambda(q+\alpha-1)}} (\Omega)$$

$$\leq \frac{1}{K_{22}} \int_{\Omega} u^{\beta} \, \mathrm{d}x \int_{\Omega} u^{q+\alpha-1} \, \mathrm{d}x + K_{19}(K_{18},q,r)^{\frac{q-\alpha+1+\beta-2(q+\alpha-1+\beta)}{\beta+1-\alpha-\frac{2\beta}{s}}} K_{22}^{\frac{q-2(q+\alpha-1)}{s}}.$$

Inserting this estimate and our choice of r into (A.4.5) we arrive at (A.4.4).

Lemma A.4.2. Consider s as in Lemma A.4.1. Set $q_k := 2^k + h$ for $h := \frac{2(s-1)(\alpha-1)}{s-2}$, $k \in \mathbb{N}_0$. Then, for all k it holds that

$$\frac{\frac{2(q_k+\alpha-1)}{s} - q_k}{2q_{k-1}\left(\frac{2}{s}-1\right) + 2(\alpha-1)} = 1,$$
(A.4.7)

$$\frac{2q_{k-1} - 2(q_k + \alpha - 1)}{2q_{k-1}\left(\frac{2}{s} - 1\right) + 2(\alpha - 1)} = \frac{s}{s-2}$$
(A.4.8)

and

$$\frac{2q_{k-1} - 2(q_k + \alpha - 1)}{\frac{2q_{k-1}}{s} - q_k} \leqslant \alpha + 1.$$
(A.4.9)

Proof. See part 2 of the proof of Theorem 1.1 in [99], where they show (in their notation) that $\frac{Q_k}{r_k} = 2$ in (2.28). Comparing this to the term in (A.4.7) we obtain the desired equality. Moreover, they show (in their notation) in (2.32) that $\frac{\lambda_k(q_k+\alpha-1)}{q_k-\lambda_k(q_k+\alpha-1)} = \frac{s}{s-2}$ where the left-hand side equals $\frac{2q_{k-1}-2(q_k+\alpha-1)}{2q_{k-1}(\frac{2}{s}-1)+2(\alpha-1)}$. This gives us (A.4.8). We obtain (A.4.9) from (2.33) in [99], where $\frac{2q_{k-1}-2(q_k+\alpha-1)}{\frac{2q_{k-1}}{2q_k}-q_k} = \frac{2\lambda_k(q_k+\alpha-1)}{q_k} \leq \alpha+1$ was shown

Lemma A.4.3. Let $y_k \in C([0,\infty)) \cap C^1(0,\infty)$ nonnegative for $k \in \mathbb{N}_0$ and satisfying

$$y'_{k}(t) + c_{k}y_{k}(t) \leq c_{k}A_{k} \max\left\{1, \sup_{t \geq 0} y^{2}_{k-1}(t)\right\},$$

where $A_k = \bar{a}2^{Dk} \ge 1$ and $c_k, \bar{a}, D > 0$. We assume that $y_k(0) \le bK^{2^k}$ holds for some constants $b \ge 1$ and K > 0. Then, for all $m \ge 1$ it holds that

$$y_k(t) \leqslant (2\bar{a})^{2^{k-m+1}-1} 2^{D\left(2(2^{k-m}-1)+m2^{k-m+1}-k\right)} \max\left\{\sup_{t\geq 0} y_{m-1}^{2^{k-m+1}}(t), b^{2^{k-m}} K^{2^k}, 1\right\}.$$

Proof. See Lemma 2.1 in [99] and adapt the proof for $y_k(0) \leq bK^{2^k}$ instead of $y_k(0) \leq K^{2^k}$. \Box

Lemma A.4.4. Let $\Omega \subset \mathbb{R}^n$ a bounded convex domain. Let $p = \infty$ if n = 1, $p \in (2, \infty)$ if n = 2and 2 if <math>n > 2. Then, it holds that

$$\|u\|_{L^p(\Omega)} \leq K_S(p) \|u\|_{H^1(\Omega)}$$

where

$$K_{S}(p) = \begin{cases} \max\left\{1, \frac{\operatorname{diam}(\Omega)|V|^{\frac{1}{2}}}{|\Omega|}\right\} & n = 1, \\ \sqrt{2}\max\left\{|\Omega|^{\frac{1}{p} - \frac{1}{2}}, \frac{\operatorname{diam}(\Omega)^{1 + \frac{p+2}{2p}n}\pi^{\frac{p+2}{4p}n}}{n|\Omega|} \frac{\Gamma(\frac{p-2}{4p}n)}{\Gamma(\frac{p+2}{4p}n)}\right\} \sqrt{\frac{\Gamma(\frac{n}{p})}{\Gamma(\frac{p-1}{p}n)}} \left(\frac{\Gamma(n)}{\Gamma(\frac{n}{2})}\right)^{\frac{p-2}{2p}} & n \ge 2. \end{cases}$$

Here, $V := \bigcup_{x \in \Omega} \Omega_x$, where $\Omega_x := \{x - y : y \in \Omega\}$ for $x \in \Omega$, and Γ denotes the Gamma function given by $\Gamma(x) = \int_0^\infty t^{x-1} e^{-t} dt$ for x > 0.

Proof. See Theorems 2.1, 3.2 and 3.4 in [108].

Lemma A.4.5. Let $\Omega \subset \mathbb{R}^n$ a bounded convex domain. Consider $J \in L^1(B_{\operatorname{diam}(\Omega)}(0))$ with $\|J\|_{L^1(B_{\operatorname{diam}(\Omega)}(0))} = 1$ and $u \in C^1(\overline{\Omega})$. Then,

$$\int_{\Omega} \int_{\Omega} J(x-y)(u(x)-u(y))^2 \, dy \, dx \leq |\operatorname{diam}(\Omega)|^2 \int_{\Omega} |\nabla u(x)|^2 \, dx.$$

Proof. This follows due to the convexity of Ω as in the proof of (3.6) in the proof of Proposition 3.1 in [99] with $B(x, \delta)$ replaced by Ω .

Lemma A.4.6. Let $t_0 \ge 0$, U > 0 and $u : \Omega \times (t_0, \infty) \rightarrow [0, U]$, $\varphi : [0, U] \rightarrow [0, \infty)$ uniformly continuous satisfying

$$\int_{t_0}^{\infty} \int_{\Omega} (\varphi(u(x,t)))^2 \, dx \, dt < \infty.$$

Then, also

$$\|\varphi(u(t))\|_{L^{\infty}(\Omega)} \xrightarrow[t \to \infty]{} 0.$$

Proof. See proof of Lemma 3.10 in [139] for φ as above.

A.5 Proofs of some lemmas from Chapter 2

Proof of Lemma 2.2.3. (i) See (4.7) in [70].

(ii) For $r \in (0,1)$ and $v, w \in \mathbb{R}$ the estimate $|v+w|^r \leq |v|^r + |w|^r$ holds. Hence, we estimate for $x, y \in \overline{\Omega}$ with v := u(x) - u(y) and w := u(y) or v := u(y) - u(x) and w := u(x) that

$$|u(x)^r - u(y)^r| \leq |u(x) - u(y)|^r \leq K_{25}|x - y|^{\vartheta r}.$$

(iii) We estimate for $x, y \in \overline{\Omega}$ with the mean value theorem that

$$|u^{r}(x) - u^{r}(y)| \leq r ||u||_{L^{\infty}(\Omega)}^{r-1} |u(x) - u(y)| \leq K_{26} |x - y|^{\vartheta}.$$

(iv) The function u is continuous. Hence, there is $m := \min_{x \in \overline{\Omega}} |u(x)| > 0$ and we can estimate for $x, y \in \overline{\Omega}$ that

$$\left|\frac{1}{u(x)} - \frac{1}{u(y)}\right| = \frac{|u(y) - u(x)|}{|u(y)u(x)|} \le \frac{|u(y) - u(x)|}{m^2} \le K_{27}|x - y|^{\vartheta}.$$

Proof of Lemma 2.2.9. (i) Let $u \in C^{1+\vartheta,\frac{1+\vartheta}{2}}(\overline{\Omega} \times [0,T])$. We estimate the norm of $u - u(\cdot,0)$ term by term. Let $t, t' \in [0,T], t \neq t'$ and $x \in \overline{\Omega}$. First, we estimate

$$\frac{|u(x,t) - u(x,0) - (u(x,t') - u(x,0))|}{|t - t'|^{\frac{\vartheta}{2}}} = \frac{|u(x,t) - u(x,t')|}{|t - t'|^{\frac{\vartheta}{2}}} = |t - t'|^{\frac{1}{2}} \frac{|u(x,t) - u(x,t')|}{|t - t'|^{\frac{1+\vartheta}{2}}} \leqslant T^{\frac{1}{2}} \langle u \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{1+\vartheta}{2}}.$$
(A.5.1)

Moreover, we use the continuous embedding of $W^1_{\infty}(\Omega)$ into $C^{\vartheta}(\overline{\Omega})$ with constant $K_I(\vartheta)$ from Lemma 2.2.8(ii) to estimate that

$$\|u - u(\cdot, 0)\|_{C(\overline{\Omega} \times [0,T])} + \langle u - u(\cdot, 0) \rangle_{x,\overline{\Omega} \times [0,T]}^{\vartheta}$$

$$\leq K_{I}(\vartheta) \left(\|u - u(\cdot, 0)\|_{C(\overline{\Omega} \times [0,T])} + \sum_{i=1}^{n} \|u_{x_{i}} - u_{x_{i}}(\cdot, 0)\|_{C(\overline{\Omega} \times [0,T])} \right)$$

$$\leq K_{I}(\vartheta) \left(T^{\frac{1+\vartheta}{2}} \langle u \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{1+\vartheta}{2}} + T^{\frac{\vartheta}{2}} \sum_{i=1}^{n} \langle u_{x_{i}} \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{\vartheta}{2}} \right)$$
(A.5.2)

Putting this together with the supremum of (A.5.1) we obtain (2.2.2) due to T < 1.

(ii) Let $u \in C^{2+\vartheta,1+\frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])$, $t, t' \in [0,T]$, $t \neq t'$ and $x \in \overline{\Omega}$. First, we conclude from the mean value theorem that

$$\|u(x,t) - u(x,0)\|_{C(\overline{\Omega} \times [0,T])} + \langle u - u(x,0) \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{1+\vartheta}{2}} \leqslant (T + T^{\frac{1-\vartheta}{2}}) \|u_t\|_{C(\overline{\Omega} \times [0,T])}.$$

Moreover, (i) implies for i = 1, ..., n that

$$\|u_{x_i} - u_{x_i}(\cdot, 0)\|_{C^{\vartheta, \frac{\vartheta}{2}}(\overline{\Omega} \times [0,T])} \leq \max\{1, K_I(\vartheta)\} T^{\frac{\vartheta}{2}} \|u_{x_i}\|_{C^{1+\vartheta, \frac{1+\vartheta}{2}}(\overline{\Omega} \times [0,T])}.$$

Together with the above estimates we obtain (2.2.3).

(iii) We estimate as before with the help of the mean value theorem and the Sobolev embedding that

$$\langle u - u(\cdot, 0) \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{\vartheta}{2}} \leqslant T^{\frac{2-\vartheta}{2}} \| u_t \|_{C(\overline{\Omega} \times [0,T])}, \\ \| u - u(\cdot, 0) \|_{C(\overline{\Omega} \times [0,T])} + \langle u - u(\cdot, 0) \rangle_{x,\overline{\Omega} \times [0,T]}^{\vartheta} \\ \leqslant K_I(\vartheta) \left(T \| u_t \|_{C(\overline{\Omega} \times [0,T])} + T^{\frac{1+\kappa}{2}} \sum_{i=1}^n \langle u_{x_i} \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{1+\kappa}{2}} \right).$$
Proof of Lemma 2.3.2. (i) Let $u, h \in L^{\infty}(\Omega \times (0,T)), x \in \overline{\Omega}$ and $t \in [0,T]$. Using our assumptions on J we can estimate

$$\begin{split} \left| J(x,h) * u^{\beta}(t) \right| &\leq \left| \int_{\Omega} J(x-y,h(y,t)) u^{\beta}(y,t) \, \mathrm{d}y \right| \\ &\leq \left\| u \right\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \left| \int_{\Omega} L(x-y) |h(y,t)| + \left| J(x-y,0) \right| \, \mathrm{d}y \right| \\ &\leq \left\| u \right\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \left(\left\| L \right\|_{L^{1}(S)} \left\| h \right\|_{L^{\infty}(\Omega \times (0,T))} + \left\| J(x,0) \right\|_{L^{1}(S)} \right). \end{split}$$

Taking the supremum on the left-hand side, we conclude that $J(\cdot, h) * u^{\beta} \in L^{\infty}(\Omega \times (0, T))$.

(ii) Let $u, h \in C^{\vartheta, \frac{\vartheta}{2}}(\overline{\Omega} \times [0, T]), u \ge 0, x \in \overline{\Omega}$ and $t_1, t_2 \in [0, T], t_1 \ne t_2$. Then, we can estimate using our assumptions on J, the Hölder continuity of u and h and the mean value theorem that

$$\begin{aligned} |J(x,h) * u^{\beta}(t_{1}) - J(x,h) * u^{\beta}(t_{2})| \\ &= \left| \int_{\Omega} J(x-y,h(y,t_{1}))u^{\beta}(y,t_{1}) - J(x-y,h(y,t_{2}))u^{\beta}(y,t_{2}) \, \mathrm{d}y \right| \\ &\leq \left| \int_{\Omega} \left(J(x-y,h(y,t_{1})) - J(x-y,h(y,t_{2}))u^{\beta}(y,t_{1}) \right) \, \mathrm{d}y \right| \\ &+ \left| \int_{\Omega} J(x-y,h(y,t_{2})) \left(u^{\beta}(y,t_{1}) - u^{\beta}(y,t_{2}) \right) \, \mathrm{d}y \right| \\ &\leq \| u \|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \int_{\Omega} L(x-y) |h(y,t_{1}) - h(y,t_{2})| \, \mathrm{d}y \\ &+ \beta \| u \|_{L^{\infty}(\Omega \times (0,T))}^{\beta-1} \int_{\Omega} (L(x-y) |h(y,t_{2})| + |J(x-y,0)|) |u(y,t_{1}) - u(y,t_{2})| \, \mathrm{d}y \\ &\leq K_{28} \left(\| u \|_{L^{\infty}}, \| h \|_{L^{\infty}}, \| L \|_{L^{1}}, \| J(\cdot,0) \|_{L^{1}}, \beta \right) T^{\frac{\vartheta}{2}} \left(\langle h \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{\vartheta}{2}} + \langle u \rangle_{t,\overline{\Omega} \times [0,T]}^{\frac{\vartheta}{2}} \right). \end{aligned}$$
(A.5.3)

Now, let $x_1, x_2 \in \overline{\Omega}, x_1 \neq x_2$ and $t \in [0, T]$. Then, we can estimate

$$\begin{aligned} |J(x_{1},h) * u^{\beta}(t) - J(x_{2},h) * u^{\beta}(t)| \\ &= \left| \int_{\mathbb{R}^{n}} J(x_{1} - y,h(y,t)) u^{\beta}(y,t) \chi_{\Omega}(y) - J(x_{2} - y,h(y,t)) u^{\beta}(y,t) \chi_{\Omega}(y) \, \mathrm{d}y \right| \\ &= \left| \int_{\mathbb{R}^{n}} J(z,h(x_{1} - z,t)) u^{\beta}(x_{1} - z,t) \chi_{\Omega}(x_{1} - z) \right| \\ &- J(z,h(x_{2} - z,t)) u^{\beta}(x_{2} - z,t) \chi_{\Omega}(x_{2} - z) \, \mathrm{d}z \end{aligned}$$

$$\leq \left| \int_{\mathbb{R}^{n}} \left(J(z,h(x_{1} - z,t)) u^{\beta}(x_{1} - z,t) - J(z,h(x_{2} - z,t)) u^{\beta}(x_{2} - z,t) \right) \chi_{\Omega}(x_{1} - z) \chi_{\Omega}(x_{2} - z) \, \mathrm{d}z \end{aligned}$$
(A.5.4)

+
$$\int_{\mathbb{R}^n} \left| J(z, h(x_1 - z, t)) u^\beta(x_1 - z, t) \chi_\Omega(x_1 - z) (1 - \chi_\Omega(x_2 - z)) \right| dy$$
 (A.5.5)

+
$$\int_{\mathbb{R}^n} \left| J(z, h(x_2 - z, t)) u^\beta(x_2 - z, t) (1 - \chi_\Omega(x_1 - z)) \chi_\Omega(x_2 - z) \right| dz.$$
 (A.5.6)

We can estimate (A.5.4) analogously to (A.5.3). To estimate (A.5.5) we define for $x_1, x_2 \in \overline{\Omega}$ the sets

$$S_{x_1x_2} := \left\{ z \in \mathbb{R}^n : x_1 - z \in \overline{\Omega}, \, x_2 - z \notin \overline{\Omega} \right\},\$$

$$G_{x_1x_2} := \left\{ z \in \mathbb{R}^n : x_1 - z \in \overline{\Omega}, \, \text{dist} \, (x_1 - z, \, \partial\Omega) \leq |x_1 - x_2| \right\}.$$

Let $z \in S_{x_1x_2}$. Then, $a := x_1 - z \in \overline{\Omega}$ and $x_2 - z = x_2 - x_1 + a \notin \overline{\Omega}$. Assume dist $(x_1 - z, \partial\Omega) > |x_1 - x_2|$. Consequently, $\overline{B_{|x_1 - x_2|}(a)} \subset \overline{\Omega}$ and $a + x_2 - x_1 \in \overline{\Omega}$ which leads to a contradiction. Hence, $S_{x_1x_2} \subset G_{x_1x_2}$ and $|S_{x_1x_2}| \leq |G_{x_1x_2}| \leq K_{29}(\Omega)|x_1 - x_2|$ holds for sufficiently smooth $\partial\Omega$. Then, we can estimate

$$\begin{split} |(\mathbf{A}.5.5)| &= \left| \int_{S_{x_1x_2}} J(z, h(x_1 - z, t)) u^{\beta}(x_1 - z, t) \, \mathrm{d}z \right| \\ &\leqslant \|u\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \int_{G_{x_1x_2}} L(z) |h(x_1 - z, t)| + |J(z, 0)| \, \mathrm{d}z \\ &\leqslant \|u\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \max\{1, \|h\|_{L^{\infty}(\Omega \times (0,T))}^{\beta} \left(\|L\|_{L^p(S)} + \|J(\cdot, 0)\|_{L^p(S)} \right) |G_{x_1x_2}|^{\frac{p-1}{p}} \\ &\leqslant K_{30} |x_1 - x_2|^{\frac{p-1}{p}}. \end{split}$$

We can estimate (A.5.6) analogously. Putting this together with the estimates of the other terms, we conclude $J(\cdot, h) * u^{\beta} \in C^{\kappa, \frac{\kappa}{2}}(\overline{\Omega} \times [0, T])$ for $\kappa := \min\{\vartheta, \frac{p-1}{p}\}$.

Bibliography

- J. Adams. "Fascin Protrusions in Cell Interactions". In: Trends in cardiovascular medicine 14.6 (2004), pp. 221–226.
- [2] R. Adams and J. Fournier. Sobolev Spaces. 2. Edition. Vol. 140. Pure and Applied Mathematics. Elsevier Ltd., Oxford, England, 2003.
- [3] A. R. A. Anderson, M. A. J. Chaplain, E. L. Newman, R. J. C. Steele, and M. Thompson A. "Mathematical Modelling of Tumour Invasion and Metastasis". In: *Computational and Mathematical Methods in Medicine* 2.2 (2000), pp. 129–154.
- [4] G. Andreguetto Maciel and R. Martinez-Garcia. "Enhanced species coexistence in Lotka-Volterra competition models due to nonlocal interactions". In: *Journal of Theoretical Biology* 530 (2021), p. 110872.
- [5] N. J. Armstrong, K. J. Painter, and J. A. Sherratt. "A Continuum Approach to Modelling Cell-Cell Adhesion". In: *Journal of Theoretical Biolog* 243.1 (2006), pp. 98–113.
- [6] G. I. Bell. "Models for the Specific Adhesion of Cells to Cells". In: Science 200.4342 (1978), pp. 618–627.
- [7] G. I. Bell, M. Dembo, and P. Bongrand. "Cell adhesion. Competition between nonspecific repulsion and specific bonding". In: *Biophysical Journal* 45.6 (1984), pp. 1051–1064.
- [8] N. Bellomo. Modeling Complex Living Systems: A Kinetic Theory and Stochastic Game Approach. Modeling and Simulation in Science, Engineering and Technology. Birkhäuser Boston, c/o Springer Science & Business Media LLC, New York, NY, USA, 2008.
- [9] N. Bellomo, A. Bellouquid, Y. Tao, and M. Winkler. "Toward a mathematical theory of Keller-Segel models of pattern formation in biological tissues". In: *Mathematical Models* and Methods in Applied Sciences 25.9 (2015), pp. 1663–1763.
- [10] N. Bellomo and A. Bellouquid. "On the derivation of angiogenesis tissue models: From the micro-scale to the macro-scale". In: *Mathematics and Mechanics of Solids* 20.3 (2015), pp. 268–279.
- [11] N. Bellomo, A. Bellouquid, L. Gibelli, and N. Outada. A Quest Towards a Mathematical Theory of Living Systems. Modeling and Simulation in Science, Engineering and Technology. Springer International Publishing AG, Cham, Switzerland, 2017.
- [12] S. Bian and L. Chen. "A nonlocal reaction diffusion equation and its relation with Fujita exponent". In: *Journal of Mathematical Analysis and Applications* 444.2 (2016), pp. 1479– 1489.
- [13] S. Bian, L. Chen, and E. A. Latos. "Chemotaxis model with nonlocal nonlinear reaction in the whole space". In: *Discrete and Continuous Dynamical Systems - Series A* 38.10 (2018), pp. 5067–5083.

- [14] S. Bian, L. Chen, and E. A. Latos. "Global existence and asymptotic behavior of solutions to a nonlocal Fisher-KPP type problem". In: *Nonlinear Analysis: Theory, Methods and Applications* 149 (2017), pp. 165–176.
- [15] S. Bian, L. Chen, and E. A. Latos. "Nonlocal nonlinear reaction preventing blow-up in supercritical case of chemotaxis system". In: *Nonlinear Analysis* 176 (2018), pp. 178–191.
- [16] V. Bitsouni, D. Trucu, M. A. J. Chaplain, and R. Eftimie. "Aggregation and travelling wave dynamics in a two-population model of cancer cell growth and invasion". In: *Mathematical Medicine and Biology: A Journal of the IMA* 35.4 (2018), pp. 541–577.
- [17] L. Blanchoin, R. Boujemaa-Paterski, C. Sykes, and J. Plastino. "Actin dynamics, architecture, and mechanics in cell motility." In: *Physiological reviews* 94.1 (2014), pp. 235– 263.
- [18] E. Boedtkjer and S. F. Pedersen. "The Acidic Tumor Microenvironment as a Driver of Cancer". In: Annual Review of Physiology 82 (2020), pp. 103–126.
- [19] V. Bogachev. *Measure Theory*. Vol. 1. Measure Theory. Springer-Verlag Berlin Heidelberg, Berlin, Heidelberg, Germany, 2007.
- [20] H. Brezis. Functional Analysis, Sobolev Spaces and Partial Differential Equations. Universitext. Springer Science & Business Media, New York, NY, USA, 2010.
- [21] A. Buttenschön, T. Hillen, A. Gerisch, and K. J. Painter. "A space-jump derivation for nonlocal models of cell-cell adhesion and non-local chemotaxis". In: Journal of Mathematical Biology 76.1-2 (2018), pp. 429–456.
- [22] E. Butturini, A. Carcereri de Prati, D. Boriero, and S. Mariotto. "Tumor Dormancy and Interplay with Hypoxic Tumor Microenvironment". In: *International Journal of Molecular Sciences* 20.17 (2019), p. 4305.
- [23] J. Carrillo, H. Murakawa, M. Sato, H. Togashi, and O. Trush. "A population dynamics model of cell-cell adhesion incorporating population pressure and density saturation". In: *Journal of Theoretical Biology* 474 (2019), pp. 14–24.
- [24] S. Casas-Tinto and M. Portela. "Cytonemes, Their Formation, Regulation, and Roles in Signaling and Communication in Tumorigenesis". In: *International Journal of Molecular Sciences* 20.22 (2019), p. 5641.
- [25] F. A. C. C. Chalub, P. A. Markowich, B. Perthame, and C. Schmeiser. "Kinetic Models for Chemotaxis and their Drift-Diffusion Limits". In: *Monatshefte für Mathematik* 142.1-2 (2004), pp. 123–141.
- [26] M. A. J. Chaplain, M. Lachowicz, Z. Szymańska, and D. Wrzosek. "Mathematical modelling of cancer invasion: the Importance of cell-cell adhesion and cell-matrix adhesion". In: *Mathematical Models and Methods in Applied Sciences* 21.04 (2011), pp. 719–743.
- [27] A. Chauvière, T. Hillen, and L. Preziosi. "Modeling cell movement in anisotropic and heterogeneous network tissues". In: *Networks and Heterogeneous Media* 2.2 (2007), pp. 333– 357.
- [28] L. Chen, K. J. Painter, C. Surulescu, and A. Zhigun. "Mathematical models for cell migration: a non-local perspective". In: *Philosophical Transactions of the Royal Society* B 375.1807 (2020), p. 20190379.

- [29] M. Conte, Y. Dzierma, S. Knobe, and C. Surulescu. "Mathematical modeling of glioma invasion and therapy approaches via kinetic theory of active particles". In: *Mathematical Models and Methods in Applied Sciences* 33.5 (2023), pp. 1009–1051.
- [30] M. Conte and N. Loy. "A Non-Local Kinetic Model for Cell Migration: A Study of the Interplay Between Contact Guidance and Steric Hindrance". In: SIAM Journal on Applied Mathematics 84.3 (2024), S429–S451.
- [31] M. Conte and C. Surulescu. "Mathematical modeling of glioma invasion: acid- and vasculature mediated go-or-grow dichotomy and the influence of tissue anisotropy". In: Applied Mathematics and Computation 407 (2021), p. 126305.
- [32] G. Corbin, A. Hunt, A. Klar, F. Schneider, and C. Surulescu. "Higher-order models for glioma invasion: From a two-scale description to effective equations for mass density and momentum". In: *Mathematical Models and Methods in Applied Sciences* 28.09 (2018), pp. 1771–1800.
- [33] G. Corbin, A. Klar, C. Surulescu, C. Engwer, M. Wenske, J. Nieto, and J. Soler. "Modeling glioma invasion with anisotropy- and hypoxia-triggered motility enhancement: From subcellular dynamics to macroscopic PDEs with multiple taxis". In: *Mathematical Models* and Methods in Applied Sciences 31.01 (2021), pp. 177–222.
- [34] P. Costa and M. Parsons. "Chapter Two New Insights into the Dynamics of Cell Adhesions". In: International Review of Cell and Molecular Biology 283 (2010), pp. 57–91.
- [35] E. Crétel, A. Pierres, A.-M. Benoliel, and P. Bongrand. "How Cells Feel Their Environment: A Focus on Early Dynamic Events". In: *Cellular and molecular bioengineering* 1 (2008), pp. 5–14.
- [36] D. Daners and P. Koch-Medina. Abstract Evolution Equations, Periodic Problems, and Applications. Vol. 279. Pitman Research Notes in Mathematics Series. Longman Scientific & Technical, Harlow, England, 1992.
- [37] C. Dellacherie and P.-A. Meyer. *Probabilities and Potential*. Vol. 29. North Holland Mathematics Studies. Hermann, Paris, France; Northolland Publishing Company, Amsterdam, Netherlands, 1978.
- [38] J. S. Desgrosellier and D. A. Cheresh. "Integrins in cancer: biological implications and therapeutic opportunities." In: *Nature reviews. Cancer* 10 (2010), pp. 9–22.
- [39] E. Di Benedetto. "On the local behaviour of solutions of degenerate parabolic equations with measurable coefficients". In: Annali della Scuola Normale Superiore di Pisa - Classe di Scienze. 4th ser. 13.3 (1986), pp. 487–535.
- [40] P. DiMilla, K. Barbee, and D. Lauffenburger. "Mathematical model for the effects of adhesion and mechanics on cell migration speed". In: *Biophysical Journal* 60.1 (1991), pp. 15–37.
- [41] R. B. Dickinson and R. T. Tranquillo. "A stochastic model for adhesion-mediated cell random motility and haptotaxis". In: *Journal of Mathematical Biology* 31.6 (1993), pp. 563– 600.
- [42] A. Dietrich, N. Kolbe, N. Sfakianakis, and C. Surulescu. "Multiscale modeling of glioma invasion: from receptor binding to flux-limited macroscopic PDEs". In: *Multiscale Model*ing and Simulation 20.2 (2022), pp. 685–713.

- [43] P. Domschke, D. Trucu, A. Gerisch, and M. A. J. Chaplain. "Mathematical modelling of cancer invasion: implications of cell adhesion variability for tumour infiltrative growth patterns". In: *Journal of Theoretical Biology* 361 (2014), pp. 41–60.
- [44] P. Domschke, D. Trucu, A. Gerisch, and M. A. J. Chaplain. "Structured models of cell migration incorporating molecular binding processes". In: *Journal of Mathematical Biology* 75.6-7 (2017), pp. 1517–1561.
- [45] J. Dyson, S. A. Gourley, and G. F. Webb. "A non-local evolution equation model of cellcell adhesion in higher dimensional space". In: *Journal of biological dynamics* 7 (2013), pp. 68–87.
- [46] J. Dyson, S. A. Gourley, R. Villella-Bressan, and G. F. Webb. "Existence and Asymptotic Properties of Solutions of a Nonlocal Evolution Equation Modeling Cell-Cell Adhesion". In: SIAM Journal on Mathematical Analysis 42.4 (2010), pp. 1784–1804.
- [47] M. Eckardt, K. J. Painter, C. Surulescu, and A. Zhigun. "Nonlocal and local models for taxis in cell migration: a rigorous limit procedure". In: *Journal of Mathematical Biology* 81.6-7 (2020), pp. 1251–1298.
- [48] M. Eckardt and C. Surulescu. On a PDE-ODE-PDE model for two interacting cell populations under the influence of an acidic environment and with nonlocal intra- and interspecific growth limitation. 2024. arXiv:2409.12657.
- [49] M. Eckardt and C. Surulescu. "On a mathematical model for cancer invasion with repellent pH-taxis and nonlocal intraspecific interaction". In: Zeitschrift für angewandte Mathematik und Physik 75.2 (2024), p. 41.
- [50] M. Eckardt and A. Zhigun. "Global existence of solutions to a nonlocal equation with degenerate anisotropic diffusion". In: *Journal of Mathematical Analysis and Applications* 543.2 (2025), p. 128971.
- [51] R. Eftimie. Hyperbolic and Kinetic Models for Self-organised Biological Aggregations: A Modelling and Pattern Formation Approach. Vol. 2232. Lecture Notes in Mathematics, Mathematical Biosciences Subseries. Springer Nature Switzerland AG, Cham, Switzerland, 2018.
- [52] S.-I. Ei, H. Ishii, S. Kondo, T. Miura, and Y. Tanaka. "Effective nonlocal kernels on reaction-diffusion networks". In: *Journal of Theoretical Biology* 509 (2021), p. 110496.
- [53] C. Engwer, T. Hillen, M. Knappitsch, and C. Surulescu. "Glioma follow white matter tracts: a multiscale DTI-based model". In: *Journal of Mathematical Biology* 71.3 (2015), pp. 551–582.
- [54] C. Engwer, A. Hunt, and C. Surulescu. "Effective equations for anisotropic glioma spread with proliferation: a multiscale approach and comparisons with previous settings". In: *Mathematical Medicine and Biology. A Journal of the IMA* 33.4 (2016), pp. 435–459.
- [55] C. Engwer, M. Knappitsch, and C. Surulescu. "A multiscale model for glioma spread including cell-tissue interactions and proliferation". In: *Mathematical Biosciences and En*gineering 13.2 (2016), pp. 443–460.
- [56] C. Engwer, C. Stinner, and C. Surulescu. "On a structured multiscale model for acidmediated tumor invasion: the effects of adhesion and proliferation". In: *Mathematical Models and Methods in Applied Sciences* 27.7 (2017), pp. 1355–1390.

- [57] D. S. Eom and D. M. Parichy. "A macrophage relay for long-distance signaling during postembryonic tissue remodeling". In: *Science* 355.6331 (2017), pp. 1317–1320.
- [58] L. Evans. Partial Differential Equations. Vol. 19. Graduate studies in mathematics. American Mathematical Society, Providence, RI, USA, 1998.
- [59] L. C. Evans. Weak convergence methods for nonlinear partial differential equations. Vol. 74. CBMS Regional Conference Series in Mathematics. American Mathematical Society, Providence, RI, USA, 1990.
- [60] K. J. Falconer. Fractal Geometry: Mathematical Foundations and Applications. John Wiley & Sons, Ltd., Chichester, England, 1990.
- [61] M. Flinck, S. H. Kramer, and S. F. Pedersen. "Roles of pH in control of cell proliferation". In: Acta Physiologica 223.3 (2018), e13068.
- [62] M. Flinck, S. H. Kramer, J. Schnipper, A. P. Andersen, and S. F. Pedersen. "The acidbase transport proteins NHE1 and NBCn1 regulate cell cycle progression in human breast cancer cells". In: *Cell Cycle* 17.9 (2018), pp. 1056–1067.
- [63] S. Y. Y. Fok, J. S. Rubin, F. Pixley, J. Condeelis, F. Braet, and L. L. Soon. "Rapid chemokinetic movement and the invasive potential of lung cancer cells; a functional molecular study". In: *BMC Cancer* 6 (2006), p. 151.
- [64] M. A. Fuentes, M. N. Kuperman, and V. M. Kenkre. "Analytical Considerations in the Study of Spatial Patterns Arising from Nonlocal Interaction Effects". In: *The Journal of Physical Chemistry B* 108.29 (2004), pp. 10505–10508.
- [65] G. Garcia and C. Parent. "Signal relay during chemotaxis". In: Journal of Microscopy 231.3 (2008), pp. 529–534.
- [66] B. Geiger, R. Zaidel-Bar, and M. Rao. "The Molecular Architecture of Cell-Cell Adhesion". In: *Encyclopedia of Cell Biology* 3 (2016), pp. 181–191.
- [67] A. Gerisch. "On the approximation and efficient evaluation of integral terms in PDE models of cell adhesion". In: *IMA Journal of Numerical Analysis* 30.1 (2010), pp. 173– 194.
- [68] A. Gerisch and M. A. J. Chaplain. "Mathematical modelling of cancer cell invasion of tissue: local and non-local models and the effect of adhesion". In: *Journal of Theoretical Biology* 250.4 (2008), pp. 684–704.
- [69] A. Gerisch and K. J. Painter. "Mathematical Modeling of Cell Adhesion and Its Applications Mathematical Modeling of Cell Adhesion and Its Applications". In: *Cell Mechanics: From Single Scale-Based Models to Multiscale Modeling*. Chapman & Hall/CRC, Boca Raton, FL, USA, 2010, pp. 319–349.
- [70] D. Gilbarg and N. Trudinger. Elliptic Partial Differential Equations of Second Order. Vol. 224. Classics in Mathematics. Springer-Verlag Berlin Heidelberg, Berlin, Heidelberg, Germany, 2001.
- [71] L. González-Méndez, I. Seijo-Barandiarán, and I. Guerrero. "Cytoneme-mediated cell-cell contacts for Hedgehog reception". In: *eLife* 6 (2017), e24045.
- [72] P. Grisvard. *Elliptic Problems in Nonsmooth Domains*. Vol. 69. Classics in Applied Mathematics. Society for Industrial and Applied Mathematics, Philadelphia, PA, USA, 2011.

- [73] S. C. Gupta and Y.-Y. Mo. "Abstract 168: MALAT1 is crucial for epithelial-mesenchymal transition of breast cancer cells in acidic microenvironment". In: *Cancer Research* 75.15 Supplement (2015), pp. 168–168.
- [74] R. Han, B. Dai, and Y. Chen. "Pattern formation in a diffusive intraguild predation model with nonlocal interaction effects". In: AIP Advances 9.3 (2019), p. 035046.
- [75] D. Hanahan and R. A. Weinberg. "Hallmarks of Cancer: The Next Generation". In: Cell 144.5 (2011), pp. 646–674.
- [76] H. Hayashi, O. Aharonovitz, R. T. Alexander, N. Touret, W. Furuya, J. Orlowski, and S. Grinstein. "Na+/H+ exchange and pH regulation in the control of neutrophil chemokinesis and chemotaxis". In: *American Journal of Physiology-Cell Physiology* 294.2 (2008), pp. C526–C534.
- [77] F. Heihoff. "Global solutions to a haptotaxis system with a potentially degenerate diffusion tensor in two and three dimensions". In: *Nonlinearity* 36.2 (2023), pp. 1245–1278.
- [78] T. Hillen. "M⁵ mesoscopic and macroscopic models for mesenchymal motion". In: Journal of Mathematical Biology 53.4 (2006), pp. 585–616.
- [79] T. Hillen, K. J. Painter, and C. Schmeiser. "Global existence for chemotaxis with finite sampling radius". In: *Discrete and Continuous Dynamical Systems - Series B* 7.1 (2007), pp. 125–144.
- [80] T. Hillen, K. J. Painter, and M. Winkler. "Global solvability and explicit bounds for non-local adhesion models". In: *European Journal of Applied Mathematics* 29.04 (2018), pp. 646–684.
- [81] T. Hillen. "A Classification of Spikes and Plateaus". In: SIAM Review 49.1 (2007), pp. 35– 51.
- [82] T. Hillen and A. Buttenschön. "Nonlocal Adhesion Models for Microorganisms on Bounded Domains". In: SIAM Journal on Applied Mathematics 80.1 (2020), pp. 382–401.
- [83] T. Hillen and K. J. Painter. "Transport and Anisotropic Diffusion Models for Movement in Oriented Habitats". In: *Dispersal, Individual Movement and Spatial Ecology: A Mathematical Perspective.* Vol. 2071. Lecture Notes in Mathematics, Mathematical Biosciences Subseries. Springer-Verlag Berlin Heidelberg, Berlin, Heidelberg, Germany, 2013, pp. 177– 222.
- [84] R. A. Horn and C. R. Johnson. *Matrix Analysis.* 2. Edition. Cambridge University Press, New York, 2013.
- [85] D. Horstmann and M. Winkler. "Boundedness vs. blow-up in a chemotaxis system". In: Journal of Differential Equations 215 (2005), pp. 52–107.
- [86] R. Horwitz and D. Webb. "Cell migration". In: Current Biology 13.19 (2003), R756–R759.
- [87] N. I. Kavallaris and T. Suzuki. Non-Local Partial Differential Equations for Engineering and Biology: Mathematical Modeling and Analysis. Springer International Publishing AG, Cham, Switzerland, 2018.
- [88] A. Klenke. Probability Theory: A Comprehensive Course. 3. Edition. Universitext. Springer Nature Switzerland AG, Cham, Switzerland, 2020.

- [89] N. Kolbe, N. Sfakianakis, C. Stinner, C. Surulescu, and J. Lenz. "Modeling multiple taxis: Tumor invasion with phenotypic heterogeneity, haptotaxis, and unilateral interspecies repellence". In: *Discrete and Continuous Dynamical Systems - Series B* 26.1 (2021), pp. 443–481.
- [90] T. B. Kornberg and S. Roy. "Cytonemes as specialized signaling filopodia". In: Development 141.4 (2014), pp. 729–736.
- [91] P. Kumar, J. Li, and C. Surulescu. "Multiscale modeling of glioma pseudopalisades: contributions from the tumor microenvironment". In: *Journal of Mathematical Biology* 82.6 (2021), p. 49.
- [92] P. Kumar and C. Surulescu. "A Flux-Limited Model for Glioma Patterning with Hypoxia-Induced Angiogenesis". In: Symmetry 12.11 (2020), p. 1870.
- [93] E. Kuusela and W. Alt. "Continuum model of cell adhesion and migration". In: Journal of Mathematical Biology 58.1-2 (2009), pp. 135–161.
- [94] O. Ladyzhenskaya, V. Solonnikov, and N. Ural'tseva. Linear and quasi-linear equations of parabolic type. Vol. 23. Translations of Mathematical Monographs. American Mathematical Society, Providence, RI, USA, 1968.
- [95] L. Lamalice, F. Le Boeuf, and J. Huot. "Endothelial Cell Migration During Angiogenesis". In: Circulation Research 100.6 (2007), pp. 782 –794.
- [96] D. E. Leckband. "Design Rules for Biomolecular Adhesion: Lessons from Force Measurements". In: Annual review of chemical and biomolecular engineering 1 (2010), pp. 365– 89.
- [97] C. T. Lee, M. F. Hoopes, J. Diehl, W. Gilliland, G. Huxel, E. V. Leaver, K. McCann, J. Umbanhowar, and A. Mogilner. "Non-local Concepts and Models in Biology". In: *Journal of Theoretical Biology* 210.2 (2001), pp. 201–219.
- [98] Y.-H. Lee and A. A. von Davier. "Equating Through Alternative Kernels". In: Statistical Models for Test Equating, Scaling, and Linking. Statistics for Social and Behavioral Sciences. Springer Science & Business Media LLC, New York, NY, USA, 2011, pp. 159– 173.
- [99] J. Li, L. Chen, and C. Surulescu. "Global boundedness, hair trigger effect, and pattern formation driven by the parametrization of a nonlocal Fisher-KPP problem". In: *Journal* of Differential Equations 269.11 (2020), pp. 9090–9122.
- [100] G. M. Lieberman. Second Order Parabolic Differential Equations. World Scientific Publishing Co. Pe. Ltd., Singapore, 1996.
- [101] G. M. Lieberman. "Hölder continuity of the gradient of solutions of uniformly parabolic equations with conormal boundary conditions". In: Annali di Matematica Pura ed Applicata 148.1 (1987), pp. 77–99.
- [102] J.-L. Lions. Quelques méthodes de résolution des problèmes aux limites non linéaires. Dunod, Gauthier-Villars, Paris, France, 1969.
- [103] A. Lorz, T. Lorenzi, M. E. Hochberg, J. Clairambault, and B. Perthame. "Populational adaptive evolution, chemotherapeutic resistance and multiple anti-cancer therapies". In: *ESAIM: Mathematical Modeling and Numerical Analysis* 47.2 (2013), pp. 377–399.

- [104] A. Lorz, S. Mirrahimi, and B. Perthame. "Dirac Mass Dynamics in Multidimensional Nonlocal Parabolic Equations". In: *Communications in Partial Differential Equations* 36.6 (2011), pp. 1071–1098.
- [105] N. Loy and L. Preziosi. "Kinetic models with non-local sensing determining cell polarization and speed according to independent cues". In: *Journal of Mathematical Biology* 80.1-2 (2019), pp. 373–421.
- [106] P. Lu, V. Weaver, and Z. Werb. "The extracellular matrix: A dynamic niche in cancer progression". In: *Journal of Cell Biology* 196.4 (2012), pp. 395–406.
- [107] S. M. Merchant and W. Nagata. "Instabilities and spatiotemporal patterns behind predator invasions with nonlocal prey competition". In: *Theoretical Population Biology* 80.4 (2011), pp. 289–297.
- [108] M. Mizuguchi, K. Tanaka, K. Sekine, and S. Oishi. "Estimation of Sobolev embedding constant on a domain dividable into bounded convex domains". In: *Journal of Inequalities* and Applications 2017 (2017), p. 299.
- [109] H. Murakawa and H. Togashi. "Continuous models for cell-cell adhesion". In: Journal of Theoretical Biology 374 (2015), pp. 1–12.
- [110] J. D. Murray. Mathematical Biology II: Spatial Models and Biomedical Applications. 3. Edition. Vol. 18. Interdisciplinary Applied Mathematics. Springer-Verlag New York, New York, NY, USA, 2003.
- [111] G. Nadin, B. Perthame, and M. Tang. "Can a traveling wave connect two unstable states? The case of the nonlocal Fisher equation". In: *Comptes Rendus Mathematique* 349.9-10 (2011), pp. 553–557.
- [112] S. S. Nazari, A. D. Doyle, C. K. E. Bleck, and K. M. Yamada. "Long Prehensile Protrusions Can Facilitate Cancer Cell Invasion through the Basement Membrane". In: *Cells* 12.20 (2023), p. 2474.
- [113] M. Negreanu and J. I. Tello. "On a competitive system under chemotactic effects with non-local terms". In: *Nonlinearity* 26.4 (2013), pp. 1083–1103.
- [114] W. Ni, J. Shi, and M. Wang. "Global stability and pattern formation in a nonlocal diffusive Lotka-Volterra competition model". In: *Journal of Differential Equations* 264.11 (2018), pp. 6891–6932.
- [115] E. Nicola. "Interfaces between Competing Patterns in Reaction-diffusion Systems with Nonlocal Coupling". PhD thesis. Max-Planck-Institut f
 ür Physik komplexer Systeme Dresden, 2001.
- [116] H. G. Othmer, S. R. Dunbar, and W. Alt. "Models of dispersal in biological systems". In: *Journal of Mathematical Biology* 26.3 (1988), pp. 263–298.
- [117] H. G. Othmer and T. Hillen. "The Diffusion Limit of Transport Equations II: Chemotaxis Equations". In: SIAM Journal on Applied Mathematics 62.4 (2002), pp. 1222–1250.
- [118] K. J. Painter, N. J. Armstrong, and J. A. Sherratt. "The impact of adhesion on cellular invasion processes in cancer and development". In: *Journal of Theoretical Biology* 264.3 (2010), pp. 1057 –1067.

- [119] S. Pal, S. Ghorai, and M. Banerjee. "Analysis of a Prey-Predator Model with Non-local Interaction in the Prey Population". In: *Bulletin of Mathematical Biology* 80 (2018), pp. 906–925.
- [120] S. Peppicelli, F. Bianchini, E. Torre, and L. Calorini. "Contribution of acidic melanoma cells undergoing epithelial-to-mesenchymal transition to aggressiveness of non-acidic melanoma cells". In: *Clinical & Experimental Metastasis* 31.4 (2014), pp. 423–433.
- [121] A. J. Perumpanani, J. A. Sherratt, J. Norbury, and H. M. Byrne. "Biological inferences from a mathematical model for malignant invasion". In: *Invasion Metastasis* 16.4-5 (1996), pp. 209–221.
- [122] R. Petrie, A. Doyle, and K. Yamada. "Random versus directionally persistent cell migration". In: Nature Reviews Molecular Cell Biology 10.8 (2009), pp. 538–49.
- [123] M. W. Pickup, J. K. Mouw, and V. M. Weaver. "The extracellular matrix modulates the hallmarks of cancer". In: *EMBO reports* 15.12 (2014), pp. 1243–1253.
- [124] R. G. Plaza. "Derivation of a bacterial nutrient-taxis system with doubly degenerate crossdiffusion as the parabolic limit of a velocity-jump process". In: *Journal of Mathematical Biology* 78.6 (2019), pp. 1681–1711.
- [125] E. Prieto-García, C. V. Díaz-García, I. García-Ruiz, and M. T. Agulló-Ortuño. "Epithelialto-mesenchymal transition in tumor progression". In: *Medical Oncology* 34.7 (2017).
- [126] L. K. Putney and D. L. Barber. "Na-H exchange-dependent increase in intracellular pH times G2/M entry and transition". In: *Journal of Biological Chemistry* 278.45 (2003), pp. 44645–44649.
- [127] P. Quittner and P. Souplet. Superlinear Parabolic Problems: Blow-up, Global Existence and Steady States. Birkhäuser Advanced Texts. Birkhäuser Verlag AG, Basel, Switzerland, 2007.
- [128] M. L. Rajendran and A. Zhigun. Local well-posedness for a novel nonlocal model for cellcell adhesion via receptor binding. 2024. arXiv:2404.15222.
- [129] G. Ren. "Global boundedness and asymptotic behavior in an attraction-repulsion chemotaxis system with nonlocal terms". In: Zeitschrift für angewandte Mathematik und Physik 73.5 (2022), p. 200.
- [130] I. Sáenz-de Santa-María, C. Bernardo-Castiñeira, E. Enciso, I. García-Moreno, J. Chiara, C. Suarez, and M.-D. Chiara. "Control of long-distance cell-to-cell communication and autophagosome transfer in squamous cell carcinoma via tunneling nanotubes". In: Oncotarget 8.13 (2017), pp. 20939–20960.
- [131] B. L. Segal, V. A. Volpert, and A. Bayliss. "Pattern formation in a model of competing populations with nonlocal interactions". In: *Physica D: Nonlinear Phenomena* 253 (2013), pp. 12–22.
- [132] J. A. Sherratt, S. A. Gourley, N. J. Armstrong, and K. J. Painter. "Boundedness of solutions of a non-local reaction-diffusion model for adhesion in cell aggregation and cancer invasion". In: *European Journal of Applied Mathematics* 20.1 (2009), pp. 123–144.
- [133] R. E. Showalter. Monotone Operators in Banach Space and Nonlinear Partial Differential Equations. Vol. 49. Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, USA, 1997.

- [134] J. Simon. "Compact sets in the space Lp(O,T; B)". In: Annali di Matematica Pura ed Applicata 146 (1987), pp. 65–96.
- [135] M. I. Simoy and M. N. Kuperman. "Non-local interaction effects in models of interacting populations". In: *Chaos, Solitons & Fractals* 167 (2023), p. 112993.
- [136] Z. Szymańska, C. Rodrigo Morales, M. Lachowicz, and M. A. J. Chaplain. "Mathematical modeling of cancer invasion of tissue: the role and effect of nonlocal interactions". In: *Mathematical Models and Methods in Applied Sciences* 19.2 (2009), pp. 257–281.
- [137] Z. Szymańska, J. Skrzeczkowski, B. Miasojedow, and P. Gwiazda. "Bayesian inference of a non-local proliferation model". In: *Royal Society Open Science* 8.11 (2021), p. 211279.
- [138] Y. Tao and M. Winkler. "A Chemotaxis-Haptotaxis Model: The Roles of Nonlinear Diffusion and Logistic Source". In: SIAM Journal on Mathematical Analysis 43.2 (2011), pp. 685–704.
- [139] Y. Tao and M. Winkler. "Large Time Behavior in a Multidimensional Chemotaxis-Haptotaxis Model with Slow Signal Diffusion". In: SIAM Journal on Mathematical Analysis 47.6 (2015), pp. 4229–4250.
- [140] I. W. Taylor and P. J. Hodson. "Cell cycle regulation by environmental pH". In: Journal of Cellular Physiology 121.3 (1984), pp. 517–525.
- [141] R. Temam. Navier-Stokes Equations: Theory and Numerical Analysis. North-Holland Publishing Company, Amsterdam, Netherlands, 1977.
- [142] C. Tian, Z. Ling, and L. Zhang. "Nonlocal interaction driven pattern formation in a prey-predator model". In: Applied Mathematics and Computation 308 (2017), pp. 73–83.
- [143] A. Uatay. "Multiscale Mathematical Modeling of Cell Migration: From Single Cells to Populations". PhD thesis. TU Kaiserslautern, 2019.
- [144] S. Valdebenito, S. Malik, R. Luu, O. Loudig, M. Mitchell, G. Okafo, K. Bhat, B. Prideaux, and E. A. Eugenin. "Tunneling nanotubes, TNT, communicate glioblastoma with surrounding non-tumor astrocytes to adapt them to hypoxic and metabolic tumor conditions". In: *Scientific Reports* 11.1 (2021), p. 14556.
- [145] V. Volpert. Elliptic partial differential equations. Volume 2: Reaction-Diffusion Equations. Vol. 104. Monographs in Mathematics. Reaction-diffusion equations. Springer Basel, Basel, Switzerland, 2014.
- [146] M. Ward and D. Hammer. "A theoretical analysis for the effect of focal contact formation on cell-substrate attachment strength". In: *Biophysical Journal* 64.3 (1993), pp. 936–959.
- [147] J. H. Wen, O. Choi, H. Taylor-Weiner, A. Fuhrmann, J. V. Karpiak, A. Almutairi, and A. J. Engler. "Haptotaxis is cell type specific and limited by substrate adhesiveness". In: *Cellular and Molecular Bioengineering* 8.4 (2015), pp. 530–542.
- [148] M. Winkler. "Aggregation vs. global diffusive behavior in the higher-dimensional Keller-Segel model". In: Journal of Differential Equations 248.12 (2010), pp. 2889–2905.
- [149] M. Winkler. "Singular structure formation in a degenerate haptotaxis model involving myopic diffusion". In: Journal de Mathématiques Pures et Appliquées 112 (2018), pp. 118– 169.

- [150] M. Winkler and C. Stinner. "Refined regularity and stabilization properties in a degenerate haptotaxis system". In: Discrete and Continuous Dynamical Systems - Series A 40.6 (2020), pp. 4039–4058.
- [151] M. Winkler and C. Surulescu. "Global weak solutions to a strongly degenerate haptotaxis model". In: *Communications in Mathematical Sciences* 15.6 (2017), pp. 1581–1616.
- [152] D. C. Worth and M. Parsons. "Adhesion dynamics: Mechanics and measurements". In: The International Journal of Biochemistry and Cell Biology 40.11 (2008), pp. 2397–2409.
- [153] S. Zaytseva, J. Shi, and L. B. Shaw. "Model of pattern formation in marsh ecosystems with nonlocal interactions". In: *Journal of Mathematical Biology* 80.3 (2020), pp. 655–686.
- [154] E. Zeidler. Nonlinear functional analysis and its applications I: Fixed-Point Theorems. Springer-Verlag New York Inc., New York, NY, USA, 1986.
- [155] A. Zhigun and C. Surulescu. "A Novel Derivation of Rigorous Macroscopic Limits from a Micro-Meso Description of Signal-Triggered Cell Migration in Fibrous Environments". In: SIAM Journal on Applied Mathematics 82.1 (2022), pp. 142–167.
- [156] A. Zhigun and M. L. Rajendran. "Modelling non-local cell-cell adhesion: a multiscale approach". In: Journal of Mathematical Biology 88.55 (2024), p. 55.

Academic curriculum vitae

since 04/2019	Doctoral studies in Mathematics, Technische Universität Kaiserslautern (since 01/2023 Rheinland-Pfälzische Technische Universität Kaiserslautern-Landau)
04/2017 - 10/2018	M. Sc. Mathematics with subsidiary subject Computer Science, Technische Universität Darmstadt
09/2015 - 05/2016	Studied abroad under the ERASMUS program, Université de Bordeaux I, France
10/2023 - 03/2017	B. Sc. Mathematics bilingual with subsidiary subject Computer Science, Technische Universität Darmstadt
06/2013	Abitur, Edith-Stein-Schule Darmstadt

Akademischer Werdegang

seit 04/2019	Promotion Mathematik, Technische Universität Kaiserslautern (seit 01/2023 Rheinland-Pfälzische Technische Universität Kaiserslautern-Landau)
04/2017 - 10/2018	M. Sc. Mathematik mit Nebenfach Informatik, Technische Universität Darmstadt
09/2015 - 05/2016	Auslandsaufenthalt im Rahmen des ERASMUS-Programms, Université de Bordeaux I, Frankreich
10/2023 - 03/2017	B. Sc. Mathematik bilingual mit Nebenfach Informatik, Technische Universität Darmstadt
06/2013	Abitur, Edith-Stein-Schule Darmstadt